SPECTRAL NETWORKS AND BETTI LAGRANGIANS

ROGER CASALS AND YOON JAE NHO

ABSTRACT. We develop the theory of spectral networks in real contact and symplectic topology. First, we establish the existence and pseudoholomorphic characterization of spectral networks for Lagrangian fillings in the cotangent bundle of a smooth surface. These are proven via analytic results on the adiabatic degeneration of Floer trajectories and the explicit computation of continuation strips. Second, we construct a Family Floer functor for Lagrangian fillings endowed with a spectral network and prove its equivalence to the non-abelianization functor. In particular, this implies that both the framed 2d-4d BPS states and the Gaiotto-Moore-Neitzke non-abelianized parallel transport are realized as part of the A_{∞} -operations of the associated 4d partially wrapped Fukaya categories. To conclude, we present a new construction relating spectral networks and Lagrangian fillings using Demazure weaves, and show the precise relation between spectral networks and augmentations of the Legendrian contact dg-algebra.

Contents

1. Introduction		2	
1.1.	Scientific context	3	
1.2.	Main results	4	
2. Betti Lagrangians and Flowline Trajectories		15	
2.1.	Real and holomorphic cotangent bundles	15	
2.2.	Betti surfaces	16	
2.3.	Betti Lagrangians	18	
2.4.	Real Morse flowlines	24	
2.5.	Asymptotics for conical ends (trapping lemma)	26	
2.6.	Holomorphic Morse flowlines for meromorphic spectral curves	31	
2.7.	Asymptotics for WKB trajectories	32	
3. Mo	3. Morse spectral networks		
3.1.	Morse flowtrees and soliton classes	35	
3.2.	Definitions on spectral networks	39	
3.3.	Proof of Theorem 1.(i)	44	
3.4.	Proof of Theorem 1.(ii)	49	
3.5.	Spectral networks and 2d-4d BPS indices	51	
3.6.	Spectral networks and rigid flowtrees	53	
4. Adiabatic convergence for D_4^- -trees			

ROGER CASALS AND YOON JAE NHO

4.1.	ε -strips and their limits	
4.2.	Proof of Proposition 4.3	64
4.3.	Proof of Theorem 4.2	68
5. Family Floer and Non-abelianized Local Systems		
5.1.	The functor $\Phi_{\mathcal{W}}$: non-abelianization of local systems	
5.2.	The functor \mathbb{F} : Family Floer and local systems	
5.3.	Equivalence between \mathbb{F} and $\Phi_{\mathcal{W}}$	85
5.4.	Proof of Theorem 2	
6. Spe	ectral networks and 4d Fukaya Categories	
6.1.	The Yoneda module of an exact Betti Lagrangian	
6.2.	Proof of Theorem 6.3 (Floer $\mu_{1 1}$ is spectral transport)	
6.3.	Cylindrized Betti Lagrangians and spectral networks	
6.4.	Wrapping up the description of $\mathcal{Y}_{(L_o,V)}$	
7. Spectral networks and weaves		
7.1.	Weaves on Betti surfaces	
7.2.	Spectral networks for Demazure weaves	
7.3.	Explicit computations of spectral networks from Demazure weaves	
Appendix A. Aspects of the geometry of pseudo-holomorphic curves		
A.1.	Geometric boundedness of meromorphic spectral curves	
A.2.	Truncated reverse isoperimetric inequality	
A.3.	Local systems and twisted local systems	
References		

1. INTRODUCTION

The object of this article is to develop the theory of spectral networks within the context of real contact and symplectic topology. First, we provide foundational results on the existence and characterization of spectral networks for Lagrangian fillings in the cotangent bundle of a smooth surface. In particular, we characterize the walls of spectral networks in terms of pseudoholomorphic strips. These are proven via new results on the adiabatic degeneration of Floer trajectories, the behavior of the associated flowlines, and explicit computations of continuation strips. Second, we construct a Family Floer functor for Betti Lagrangians endowed with a spectral network and show that the non-abelian parallel transport in Gaiotto-Moore-Neitzke's supersymmetric N = 4 context is both generalized and made rigorous by this Floer-theoretic parallel transport. Third, the article brings forth new constructions and applications relating spectral networks and Lagrangian fillings, including the study of spectral networks from Demazure weaves, their relation to the Legendrian contact dg-algebra, and an interpretation of framed 2d-4d BPS states and the non-abelianization maps as part of the A_{∞} -operations and module categories of partially wrapped Fukaya categories.

SPECTRAL NETWORKS AND BETTI LAGRANGIANS

A central theme of this work is that spectral networks can be studied and characterized in terms of the Floer theory of real Lagrangians in the real cotangent bundle of a smooth surface. This allows for a key generalization of the standard physical theory of spectral networks, which itself focuses on meromorphic spectral curves in the holomorphic cotangent bundle of a Riemann surface. This generalization is needed to define and study the Betti moduli space in the wild case for smooth asymptotic data, not necessarily meromorphic Stokes data. In particular, it allows us to tackle asymptotics associated to any positive braid, including and generalizing all braid varieties, and perform computations on their associated cluster structures, BPS monodromy and Donaldson-Thomas theory. Indeed, the Floer-theoretic framework and results we establish provide a precise bridge to apply the new developments on cluster algebras in 4-dimensional symplectic topology to the study of spectral networks.

The introduction of spectral networks in mathematical physics has lead to an outstanding number of breakthroughs in string theory. The present work also serves the purpose of establishing a rigorous mathematical foundation for many beautiful insights and claims from the physics community, in our case within the context of contact and symplectic topology. A number of such conjectured properties and examples are discussed and proven throughout this manuscript, including the characterization of spectral networks in terms of pseudo-holomorphic disks and the study of higher-order Stokes phenomena, including the Berk-Nevins-Roberts example and beyond.

1.1. Scientific context. In 2012, within the context of supersymmetric gauge theories in mathematical physics, D. Gaiotto, G. Moore and A. Neitzke introduced and developed the study of spectral networks in [GMN13a; GMN13b; GMN14]. These contributions to physical theories of class Sallowed for the computation of different BPS degeneracies in the associated 4-dimensional N = 2theories (e.g. coupling to surface defects) and a better understanding of how the wall-crossing phenomenon in coupled 2d-4d systems [GMN12; GMN13c] related to the BPS spectrum. Since then, an abundance of exciting work illustrates that spectral networks are part of a fertile and productive ground for mathematical physics. To wit, further explorations of different types of BPS states were achieved in [Gab+17; GLM15; LP16; LP17; Lon18] and [DMN20; Erg+21; HN17; HHN20], non-abelianization and quantum holonomies were studied in [Gab17; HN16; HN20; KLS23; NY20] and see also the surveys [Nei14; Nei21].

In pure mathematics, spectral networks have started to be used in a variety of areas, including exact WKB analysis [Ale+24; GHN23; IK22; IK23] (see also the preceding works [AKT01; HLOD04]), the study of moduli of Higgs bundles and their Hitchin fibrations [Dum+21; FN21; Kat+15; Wil16], and that of spaces of stability conditions [BS15; HKS24; Smi15]; see also [FN24; Kuw24; Nho24; Sue24]. Throughout these contributions, there is a focus towards the case $G = SL_2(\mathbb{C})$, where the spectral curve is a 2-fold branch cover of the base. This case is rather well-understood by now, in part due to its direct relation to the theory of quadratic differentials on Riemann surfaces and the combinatorics of their associated foliations. The higher rank case, for $G = SL_n(\mathbb{C})$, remains an interesting subject of exploration, cf. [Kat+15; Smi21], and a main focus of this article.

In the $SL_2(\mathbb{C})$ case, a first connection between the Floer theory of real Lagrangian surfaces and framed 2d-4d BPS states was hinted by the first author in [CM18], being further developed in [Nho24] by the second author. Subsequently, the introduction of Legendrian weaves [CZ22]has provided key insights to establish the general connection between Floer theory and spectral networks in all higher ranks, significantly beyond the theory of quadratic differentials. Equipped with new tools in symplectic topology [CL22; CZ22; GPS20; GPS24b], the present manuscript now greatly generalizes the results in [CM18; Nho24] to arbitrary rank, and crystallizes a number of further insights relating spectral networks to both partially wrapped Fukaya categories [GPS20; GPS24b; Syl19 and the construction of cluster structures on braid varieties [Cas+24; Cas+25;CW24]. From the Betti perspective on wild character varieties, a key difference between rank n = 2 and higher rank spectral networks is that, in the latter case, asymptotics are not constrained to be meromorphic. Namely, the putative spectral curves can have asymptotics dictated by non-algebraic links, and thus they are not necessarily associated to meromorphic curves. Thus far, all higher rank explorations in the literature focus on the algebraic case, on the so-called WKB spectral networks, where the spectral curve is a holomorphic subvariety of the holomorphic cotangent bundle of a Riemann surface. The results we present go much beyond such algebraic constraints.

A key conceptual advance of this article is that the technique of spectral networks also works for real Lagrangian fillings of Legendrian links, regardless of whether the Lagrangian surface and their asymptotic Legendrian links have algebraic origin or not. For this, we introduce a more general definition of spectral network than [GMN13b], where only an almost complex structure is used and the necessary analytical properties of pseudo-holomorphic strips and their adiabatic limits can be proven. An advantage of intertwining spectral networks with some of the current tools from real contact and symplectic topology is that structural results known within the latter translate into theorems for spectral networks. An instance of this is the invariance of BPS soliton classes under real Hamiltonian isotopies, and not just holomorphic deformations. Another example is the functorial behavior of Floer-theoretic invariants for Legendrian links under Lagrangian cobordisms, which readily explains how to glue and compute spectral networks associated to Demazure weaves. These new classes of spectral networks, built from Demazure weaves, include and generalize many guiding examples in the literature and allow for direct computations of the BPS monodromy generator by using explicit reddening sequences from their associated cluster algebras.

1.2. Main results. Let S be a smooth closed oriented surface and $m \subset S$ a finite collection of marked points. A triple $(S, m; \Lambda)$ is said to be a Betti surface of rank n if Λ is a collection of Legendrian links in the ideal contact boundary $(T^{\infty}S, \xi_{st})$, each of which is associated to an n-stranded positive braid over the Legendrian fiber of a marked point in m. Let us denote by $S := \mathbf{S} \setminus \mathbf{m}$ the complement of the marked points. By depicting the wavefronts of the links in Λ onto $S \subset \mathbf{S}$, a Betti surface is pictorially drawn as in Figure 1.



FIGURE 1. A Betti surface **S** of genus 3 and rank 3, with 9 marked points $m = \{m_1, \ldots, m_9\}$, in orange, and their associated Legendrian links in Λ , in blue. The Legendrian links associated to marked points m_1, m_5, m_8 have concentric circles as their fronts, which corresponds to a regular singularity. A non-concentric Legendrian isotopy class corresponds to an irregular singularity.

The Legendrian links in Λ are the contact topological incarnation of the Stokes data from the theory of irregular singularities of meromorphic connections, cf. [KKP08, Remark 2.17]. In particular, any meromorphic spectral curve in the sense of [GMN13b] yields a well-defined Betti surface of the corresponding rank, where the fronts of the Legendrians in Λ are the associated Stokes diagrams. Details are provided in Section 2, cf. also [Boa21; She+19] and [Su25, Appendix B]. At core, a Betti Lagrangian $L \subset (T^*S, \omega_{st})$ is a Lagrangian submanifold whose projection onto S is an *n*-fold branched cover and whose asymptotics are dictated by the Legendrian links in Λ ; this is detailed in Definition 2.6. There are two important classes of such Betti Lagrangians, both discussed in Section 2.3:

- (i) Exact Betti Lagrangians, where the restriction of the Liouville form λ_{st} to L is an exact 1-form, i.e. all real periods of λ_{st} on L vanish. This class is defined with S a smooth punctured surface and it yields Lagrangian fillings $L \subset (T^*S, \lambda_{st})$ of the Legendrian link given by the union of all Legendrian links Λ , after cylindrization near the contact boundary; cf. Section 6.3 and [CG22; CG24; CZ22; EHK16].
- (ii) If S is endowed with a Riemann surface structure \mathbb{S} , we can consider meromorphic Betti Lagrangians. These are the Betti Lagrangians that appear as the real part of a holomorphic Lagrangian curve $\Sigma \subset (\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$ in the holomorphic cotangent bundle $(\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$ with asymptotics dictated by the irregular classes at \boldsymbol{m} . These are the spectral curves studied in the literature, including those in [GMN13a; GMN13b; GMN14] and those featuring in

6

ROGER CASALS AND YOON JAE NHO

the irregular Riemann-Hilbert correspondence, see e.g. [BB04; Boa21].

The theory of spectral networks in [GMN13a; GMN13b; GMN14] is devoted to Betti Lagrangians of type (ii), in Riemann surfaces S. The resulting spectral networks satisfy a number of additional properties, implied by the underlying holomorphicity: we refer to those spectral networks in [GMN13a; GMN13b; GMN14] as WKB spectral networks. In order to develop the theory of spectral networks in the real setting of (T^*S, λ_{st}) , so as to include the important class (i) above, Section 3 introduces a generalization of those WKB spectral networks, which we refer to as a Morse spectral networks. In a nutshell, a Morse spectral network $\mathcal{W} \subset S$ compatible with L is a set of gradient flow trajectories in S given by the difference functions of the sheets of $L \subset T^*S$ above S satisfying a set of non-trivial interaction and asymptotic constraints. These constraints exclude a number of pathological behaviors. For instance, it is proven in Section 3 that Morse spectral networks do not allow for ill-behaved asymptotics at infinity, such as trajectories reaching m in finite time, nor trajectories getting arbitrarily close to m and then steering away from m. In fact, trajectories should converge to Reeb chords of Λ , which generalize the points of maximal decay in Stokes diagrams. In the interior, Morse spectral networks also disallow for high-valence vertex interactions and tangencies between trajectories and sink points inside of S, among others. In general, these are all possible behaviors for such difference-function gradient trajectories that occur without a priori constraints on the geometry of L. In addition, a Morse spectral network must have a compatible counting function μ on such trajectories, generalizing the count of framed 2d-4d BPS states.¹ These interaction constraints and the necessary properties of μ , both crucial to the concept of a Morse spectral network, are specified in Section 3.2: a depiction of the key allowed local models for a Morse spectral network \mathcal{W} is in Figure 2.

A. The first natural question that arises is the existence of such Morse spectral networks compatible with an exact Betti Lagrangian $L \subset (T^*S, \omega_{st})$. A main difficulty is that the gradient flow trajectories given by the sheets of L over S might not satisfy the necessary asymptotic and interaction constraints. Also, it might not be clear how to generally assign the counting function on trajectories given an arbitrary such Lagrangian $L \subset (T^*S, \omega_{st})$. Similarly, in the holomorphic setting, it has not been established that a finite-energy WKB spectral network compatible with a given a meromorphic Betti Lagrangian $\Sigma \subset (\mathbb{T}^*S, \omega_{\mathbb{C}})$ actually does exist. In practice, it is also important to guarantee that there are finitely many trajectories in a spectral network, given an energy cut-off (or equivalently, a mass cut-off in physics literature). In fact, due to its conceptual importance, we include this condition into the definition of the WKB spectral network. The first result in this manuscript resolves the matter of existence, including the (energy)-finiteness constraint:

¹This is in line with the 4d BPS invariants of [Bri19, Definition 2.1]. Nevertheless, we do not consider a central charge in this general real setting; that might be the focus of future work.



FIGURE 2. The local models for vertices in spectral networks: initial, interaction and non-interaction. There are two types of interaction vertices: creation and 6valent type. The inconsistent vertex type is allowed for pre-spectral networks, but is *not* allowed for spectral networks.

Theorem 1 (Existence of Spectral Networks). Let S be a Betti surface.

- (i) Let $L \subset (T^*S, \lambda_{st})$ be an exact Betti Lagrangian and (S, g) a metric adapted to L. Then there exists a finite Morse spectral network compatible with L and (S, g).
- (ii) Let S a Riemann surface structure for S, $(\mathbb{T}^*S, \omega_{\mathbb{C}})$ its holomorphic cotangent bundle, and $\Sigma \subset (\mathbb{T}^*S, \omega_{\mathbb{C}})$ a meromorphic spectral curve with O(-1) ends. Then, for dense subset of phases $\theta \in S^1$, there exists a generic WKB spectral network compatible with the real Lagrangian $e^{i\theta}\Sigma \subset (T^*S, \lambda_{st})$.

In the statement of Theorem 1, the dense set of phases in S^1 is a countable intersection of open dense sets. The notation $e^{i\theta}\Sigma$ stands for the real part of the holomorphic subvariety Σ after being acted by a rotation of phase θ , see Section 2 for details. In the real case (*i*), our proof of Theorem 1 actually yields a compatible Morse spectral network whose only interaction vertices are creation vertices, which is in itself an interesting difference with the WKB case. For the proof of Theorem 1, presented in Section 3, both the trapping lemma in Section 2.5 and the preliminary transversality established in Section 3.3.1 are important technical aspects, controlling the behavior of flow trajectories and ensuring necessary asymptotics and interaction constraints. The proof of Theorem 1 in the WKB case (*ii*) nicely intertwines with the heuristic construction proposed in [GMN13b]: indeed, we prove that their proposed procedure gives an algorithm that both terminates and outputs an object which is indeed a WKB spectral network, satisfying the energy-finiteness condition, or rather, a stronger condition that we call gapped (c.f Definition 3.12). In this latter WKB case, see also [Kuw24, Theorem 1.2], which discusses the construction of certain WKB spectral networks via sheaf quantization. Note that in the WKB case (*ii*), the real Lagrangian $e^{i\theta}\Sigma \subset (T^*S, \lambda_{st})$ is typically not an exact Lagrangian, as the periods of Σ might not all have the same phase $\theta + \pi/2$.

B. Floer theory, broadly understood as the study of symplectic topology through pseudoholomorphic methods, infinite-dimensional variational analysis on path spaces and their homological incarnations, is a pillar of contact and symplectic topology. See for instance [AS19; EGH00; Flo88; Flo89a; Gro85; PSS96; Vit99] and the monographs [Fuk+09a; Fuk+09b; MS04; Sei08b], to name some key references. The context for Theorem 1 certainly belongs to real symplectic topology: a real Lagrangian submanifold L inside the cotangent bundle (T^*S, ω_{st}) . Yet, spectral networks $W \subset S$ are defined in terms of gradient trajectories on S: even if they are trajectories of functions describing L, there are no immediate Floer-theoretic aspects to the notion of a spectral network. Based on examples, e.g. when certain 4d BPS states correspond to holomorphic disks [GMN13b, Section 6.2], it has been hoped by the mathematical physics community that spectral networks $W \subset S$ can in fact be given a characterization entirely in Floer-theoretic terms. Given the effectiveness of Floer theory and all its known functorial properties, establishing a rigorous connection between Floer theory and spectral networks seems particularly desirable from the viewpoint of pure mathematics as well. Our second result proves such characterization:

Theorem 2 (Floer-theoretic characterization of Spectral Networks). Let S be a Betti surface, $\Sigma \subset (T^*S, \lambda_{st})$ a Betti Lagrangian and $W \subset S$ a spectral network compatible with Σ . Then the following characterization holds:

A point $p \in S$ belongs to an active wall of $\mathcal{W} \iff \exists$ broken pseudo-holomorphic strip in (T^*S, λ_{st}) between the fiber T_p^*S and Σ in the adiabatic limit.

Furthermore, the following two properties are satisfied:

- (i) If p ∈ W, then the relative homology class in Σ associated any such pseudo-holomorphic strip is uniquely determined by W, and such strip must adiabatically degenerate to a subset of W.
- (ii) For Σ an exact Betti Lagrangian, the characterization holds for (genuine) pseudoholomorphic strips, without being potentially broken.

When Σ is non-exact, we demand Σ to be meromorphic outside some compact subset of T^*S , for some technical reasons. All known interesting examples of non-exact Betti Lagrangians fall into such a category. A first important aspect of Theorem 2 is the presence of an adiabatic limit: it refers to the limit $\varepsilon \to 0$ when the Betti Lagrangian $\Sigma \subset (T^*S, \lambda_{st})$ is rescaled by a factor of $\varepsilon \in \mathbb{R}_+$ under the fiberwise radial scaling action of the cotangent bundle.² Since even in the simplest cases pseudo-holomorphic strips are in bijective correspondence with flow-trees only in the adiabatic limit, the adiabatic hypothesis in Theorem 2 is needed in any such characterization. A second relevant fact is that, for exact Betti Lagrangians, all the walls of the spectral networks produced in Theorem 1 are active. Therefore Theorem 2 gives a complete characterization of points in a spectral network, without even considering framed 2d-4d BPS indices.

Appropriately understood, Theorem 2 generalizes the analysis relating flow-trees and pseudoholomorphic strips from [Ekh07]. The result in ibid. are nevertheless far from concluding Theorem 2: a key new ingredient is establishing the convergence from pseudo-holomorphic strips to trees while satisfying the initial edge constraints, which is crucial to control the behavior of D_4^- trees near branch points. This convergence result is proven in Section 4. In addition, the flowtree results in the literature do not apply to non-exact Lagrangians (and so would not be able to deal with any of [GMN13b; GMN13a; GMN14]) and also require generic swallowtails front singularities. Since D_4^- -singularities are necessarily present in a branch cover and the study of spectral curves, we also further develop the machinery of Morse flowtrees for this context.

Theorem 2 is proven at the end of Section 5, after developing the necessary analytical results in the bulk of Sections 4 and 5. At core, it is the combination of two different of results. First, we establish no-go theorems in Section 4, cf. Theorem 4.2, using adiabatic degeneration. We are able to use the adiabatic limit to show that certain pseudoholomorphic strips do *not* exist, as stated in Theorem 2: it allows us to prove that points in the complement of the spectral networks cannot support such a pseudoholomorphic strip and that, for any point in the spectral network \mathcal{W} , the only possible homology class for a potential pseudoholomorphic strip through that point must be the soliton class determined by \mathcal{W} and the point. Second, we prove that there exist such pseudoholomorphic strips for points in \mathcal{W} , as stated in Theorem 2, through the explicit computations established in Section 5.

C. In Theorems 1 and 2, the more foundational aspects of existence and characterizations are addressed. A central aspect of spectral networks in the supersymmetric theories of [GMN13b] is the non-abelianization functor, cf. [GMN12, Section 10] and [GMN14, Section 5]. In the supersymmetric N = 4 context, it is a construction from the infrared to the ultraviolet that carries $GL_1(\mathbb{C})$ -local systems on the Seiberg-Witten curve to $GL_n(\mathbb{C})$ -local system on the space of ultraviolet parameters. In topological terms, each spectral network $\mathcal{W} \subset S$ partially defines a functor

²In the literature in geometric analysis, adiabatic limit refers to the process of degenerating a Riemannian metric in certain directions. In symplectic topology, one typically starts with the symplectic 2-form and a compatible almost complex structure, from which a Riemannian metric is produced. The fiberwise rescaling action can be seen as a rescaling of the almost complex structure, which itself is then seen as rescaling the metric. In this case, the limit metrics when $\varepsilon \to 0$ degenerate to be supported in the zero section of T^*S .

 $\Phi_{\mathcal{W}}$: Loc[†](L) --> Loc[†](S) from rank-*n* twisted local systems on L to rank-1 twisted local systems in S, i.e. from a Betti Lagrangian $L \subset (T^*S, \lambda_{st})$ to the zero section S. The functor $\Phi_{\mathcal{W}}$ is constructed combinatorially from \mathcal{W} , introducing detour paths that act as instanton corrections and abide by formulas similar to the Hori-Vafa wall-crossing formulas. Technically, given $V \in \operatorname{Loc}^{\dagger}(L)$, the physics framework is such that $\Phi_{\mathcal{W}}(V)$ has parallel transport only defined for paths transverse to \mathcal{W} and endpoints in the complement \mathcal{W}^c , thus our use of the dashed arrows in $\Phi_{\mathcal{W}}$: Loc[†](L) --> Loc[†](S), cf. Section 5.1.

In this paper we develop Floer theory for exact Betti Lagrangians in the presence of a spectral network \mathcal{W} , in Section 5, constructing Floer-type cochain complexes with differentials recording certain counts of (actual non-adiabatic) pseudo-holomorphic disks. As a consequence, we are able to construct an actual functor $\mathbb{F} : \operatorname{Loc}^{\dagger}(L) \longrightarrow \operatorname{Loc}^{\dagger}(S)$ using Family Floer techniques. The functor \mathbb{F} is defined for all paths and built as a direct limit of functors \mathbb{F}_{ε} depending on the adiabatic parameter. This is presented in Section 5.2; here \mathbb{F} stands both for Family and for Floer. Then we establish the comparison between these two functors in Section 5.3. The results can be summarized as follows:

Theorem 3 (Family Floer and Non-abelianization). Let S be a Betti surface, $L \subset (T^*S, \lambda_{st})$ an exact Betti Lagrangian and $W \subset S$ a spectral network compatible with Σ . Then:

(1) Family Floer with respect to L can be defined and yields a functor

 $\mathbb{F}: \operatorname{Loc}^{\dagger}(L) \longrightarrow \operatorname{Loc}^{\dagger}(S),$

invariant under compactly supported Hamiltonian isotopies of L. Specifically, it is built from a type of Floer complexes $CF^*(T_z^*S, L)$ between cotangent fibers and L, which can be constructed and shown to be well-defined.

(2) In the adiabatic limit $\varepsilon \to 0$, the two functors $\mathbb{F}, \Phi_{\mathcal{W}} : \operatorname{Loc}^{\dagger}(\varepsilon L) \longrightarrow \operatorname{Loc}^{\dagger}(S),$ are equivalent.

As said above, the Floer-theoretic functor \mathbb{F} in Theorem 3.(1) is defined as a direct limit of Family Floer functors \mathbb{F}_{ε} that depends on the adiabatic parameter ε : at core, \mathbb{F} uses the spectral network \mathcal{W} in order to correct the technical issues that often obstruct when applying Family Floer methods, including caustics. The comparison in Theorem 3.(2) does require scaling the Betti Lagrangian and thus the comparison $\mathbb{F} = \Phi_{\mathcal{W}}$ only makes sense in the adiabatic limit. Theorem 3 is proven in Section 5.2 for Part (1) and Section 5.3 for Part (2). Note that the case of quadratic differentials, with $\mathrm{GL}_2(\mathbb{C})$, can be argued from [Nho24], whereas Theorem 3 works for any higher rank exact Betti Lagrangians. We emphasize that the functor \mathbb{F} is well-defined whereas, technically, $\Phi_{\mathcal{W}}$ cannot parallel transport starting at \mathcal{W} : Theorem 3.(2) is thus an equality where $\Phi_{\mathcal{W}}$ is defined. More conceptually, it shows how the framework of Floer theory not only captures but generalizes that of spectral networks: the former being able to be used to define and extend the latter. An interesting question is whether Theorem 3 could be extended to general Betti Lagrangians. For such Lagrangians, we are currently only able to explicitly compute parallel transport maps for very short line segments. Such a computation is the key component behind the proof of Theorem 2. We leave this question for the future research.

A corollary of Theorem 3 is that we can extract a symplectic invariant of the exact Betti Lagrangians $L \subset T^*S$ from W, up to compactly supported Hamiltonian isotopies. Indeed, Theorem 3 implies that the set of soliton classes in W associated to Reeb chords of $\partial_{\infty}L$ is an invariant of L. This is proven in Theorem 5.8.³ Note that the cluster algebra structures built in [Cas+25], corresponding to a Betti surface $S = S^2$ with one irregular singularity, have cluster variables defined by Floer parallel transport along relative cycles. In combination with that work, Theorem 3 then formalizes the cluster-like coordinates sketched in the supersymmetric context in [GMN13b, Section 10] and [Nei14].

D. Fukaya A_{∞} -categories, in their many forms, are a central object in Floer theory and the study of symplectic topology via pseudo-holomorphic methods, see e.g. [Fuk+09a; GPS20; Sei08b]. The Floer complexes we constructed for Theorem 3 and the results developed in Sections 4 and 5 can be enhanced to an A_{∞} -categorical level. This is achieved in Section 6, as follows:

- (i) First, due to its asymptotic ends, an exact Betti Lagrangian $L \subset T^*S$ does not define an object of $\mathfrak{W}(T^*S)$ or any partially wrapped version thereof. That said, we construct in Section 6.1 an A_{∞} -module $\mathcal{Y}_{(L,V)}$ of $\mathfrak{W}(T^*S)$ that behaves as if it was the Yoneda module for such an object. More crucially, we then prove in Section 6.2 that its $\mu_{1|1} A_{\infty}$ -operation precisely captures the non-abelian parallel transport.
- (ii) Second, in Section 6.3 we geometrically modify L via cylindrization so as to obtain a Lagrangian L_{\circ} which defines an object in the partially wrapped Fukaya category $\mathfrak{W}(T^*S, \Lambda)$. We also establish that its associated $\mu_2 A_{\infty}$ -map recovers non-abelian parallel transport, now in the partially wrapped case.
- (iii) Third, we compare the A_{∞} -modules presented in (i) and (ii) above under the inclusion functor $i_* : \mathfrak{W}(T^*S) \to \mathfrak{W}(T^*S, \Lambda)$ and completely describe the Yoneda A_{∞} -module of

³It is an effective invariant, as it at least matches the information contained in the augmentation induced by L for the Legendrian contact dg-algebra of $\partial_{\infty}L$, when the latter is defined. Thus it already distinguishes Lagrangians before and after a surgery.

 L_{\circ} . This description of this A_{∞} -module of $\mathfrak{W}(T^*S, \Lambda)$ associated to the cylindrized Betti Lagrangian L_{\circ} is achieved in Section 6.4 by proving a generation result for $\mathfrak{W}(T^*S, \Lambda)$, cf. Proposition 6.19, showing that its generated by internal and infinity fibers, without linking disks being in the generating set.

These are all established in detail in Section 6. See Proposition 6.1, Theorem 6.3 and Corollary 6.4 for (i), Section 6.3.1 and Theorem 6.12 for (ii), and Corollary 6.13 and Proposition 6.19 for (iii). We summarize some of the key outcomes in the following result:

Theorem 4 (Spectral curves and 4d Partially Wrapped Fukaya Categories). Let $L \subset (T^*S, \lambda_{st})$ be an exact Betti Lagrangian endowed with a local system $V \in \text{Loc}(L)$ and $\mathcal{W} \subset S$ a compatible Morse spectral network. Then:

(i) There exists an A_{∞} -module $\mathcal{Y}_{(L,V)}: \mathfrak{W}(T^*S) \longrightarrow \mathrm{mod}_k$ with cohomologies

$$H^*(\mathcal{Y}_{(L,V)}(P)) \cong HF^*(P,(L,V)),$$

where $P \subset T^*S$ is any exact cylindrical Lagrangian with compact horizontal support. Furthermore, the minimal geodesic generator $[\alpha_{zw}] \in \operatorname{Hom}_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S)$ of an \mathcal{W} -adapted pair $z, w \in S$ satisfies

$$\mu_{1|1}([\alpha_{zw}], \cdot) = \Phi_{\mathcal{W}}(V)(\alpha_{zw}).$$

In particular, there exists an A_{∞} -quasi-equivalence $\mathfrak{W}(T^*S)$ -mod $\cong C_{-*}(\Omega_z S)$ -mod,

$$\mathcal{Y}_{(L,V)} \simeq \Phi_{\mathcal{W}}(V),$$

as A_{∞} -modules over $\mathfrak{W}(T^*S)$.

(ii) Consider the Yoneda A_{∞} -module $\mathcal{Y}_{(L_{\circ},V)} : \mathfrak{W}(T^*S, \Lambda) \longrightarrow \text{mod}_k$ of the cylindrized Betti Lagrangian $L_{\circ} \subset T^*S$ and its $\mu_2 A_{\infty}$ -operation. Then the minimal geodesic $\alpha_{zw} \in \Omega_{z,w}$ satisfies

$$\mu_2(i_*[\alpha_{zw}], \cdot) = \Phi_{\mathcal{W}}(V)(\alpha_{zw})$$

and there is a homotopy

 $\mathcal{Y}_{(L_\circ,V)} \circ i_* \simeq \mathcal{Y}_{(L,V)}$

of A_{∞} -modules over $\mathfrak{W}(T^*S)$.

At core, Theorem 4 establishes that the non-abelian parallel transport of Gaiotto-Moore-Neitzke studied in [GMN12; GMN13b; GMN14], and the associated counts of framed 2d-4d BPS states, can be realized as parts of the A_{∞} -operations in either wrapped or partially wrapped 4-dimensional Fukaya categories. Theorem 4.(i) does so in the wrapped category $\mathfrak{W}(T^*S)$ and Theorem 4.(ii) in the partially wrapped category $\mathfrak{W}(T^*S, \Lambda)$, also establishing the comparison. At a technical level, we use in Section 6 the modern construction of partially wrapped categories in [GPS20] and establish our results by indeed using their definitions in terms of homotopy colimits, carefully defining the class associated to a minimal geodesic α_{zw} by studying the necessary continuation maps, cf. Sections 6.2 and 6.3.2.

E. Theorems 1 and 2 establish existence and characterizations of spectral networks, and Theorems 3 and 4 develop the Floer-theoretic results needed to rigorously define and establish non-abelianization in spectral networks, first analytically and then categorically, respectively. The last result we present is a new construction of spectral networks, providing both novel examples and crystallizing connections to cluster algebras.

The input of the construction is a Legendrian weave $\mathfrak{w} \subset S$, as introduced in [CZ22], specifically a Demazure weave, cf. Section 7.1, [Cas+24, Section 4.1] or [Cas+25, Section 4.1]. Such a weave $\mathfrak{w} \subset S$ defines an exact Betti Lagrangian $L_{\mathfrak{w}} \subset (T^*S, \lambda_{st})$ with asymptotics determined by a Legendrian $\Lambda_{\partial\mathfrak{w}} \subset (T^{\infty}S, \xi_{st})$, depending only on the boundary $\partial\mathfrak{w}$ of the weave. In Section 7.2 we define a certain object $\mathcal{W}_{\mathfrak{w}} \subset S$, associated to a weave \mathfrak{w} , entirely in combinatorial terms: we refer to it as the *augmentation forest* of \mathfrak{w} . These $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ are built combinatorially from \mathfrak{w} , in contrast to the intrinsically analytic nature of any spectral network associated to $L_{\mathfrak{w}}$ and its associated augmentation $\varepsilon_{\mathfrak{w}} : \mathcal{A}(\Lambda_{\partial\mathfrak{w}}) \longrightarrow k[H_1(L_{\mathfrak{w}})]$ of the Legendrian contact dg-algebra $\mathcal{A}(\Lambda_{\partial\mathfrak{w}})$. Figure 3 illustrates two examples of weaves and their associated spectral networks.

For simplicity, we focus on $S = \mathbb{R}^2$ and their $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$, as this already yields many interesting new examples; see [CN22] for specifics on the Legendrian dg-algebra $\mathcal{A}(\Lambda_{\partial \mathfrak{w}})$. In Section 7, after providing the construction of the augmentation forest $\Theta_{\mathfrak{w}}$, we conclude the following result and use it to explicitly compute several examples, cf. Section 7.3:

- **Theorem 5** (Augmentations, weaves and Spectral Networks). Let $\mathfrak{w} \subset \mathbb{R}^2$ be a Demazure weave and $L_{\mathfrak{w}} \subset (T^*\mathbb{R}^2, \lambda_{st})$ its associated exact Betti Lagrangian. Then:
 - (1) The augmentation forest $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ is a Morse spectral network compatible with $L_{\mathfrak{w}}$. In addition, it is a finite and creative spectral network.
 - (2) The flowlines of the spectral network $\mathcal{W}_{\mathfrak{w}}$ are exactly given by the union of D_4^- -trees obtained by adiabatically degenerating the rigid pseudoholomorphic strips contributing to the augmentation $\varepsilon_{\mathfrak{w}}$ induced by $L_{\mathfrak{w}}$.

Theorem 5 has two advantages. First, there are many weaves with the same boundary $\partial \mathbf{w}$, e.g. [CG22, Section 2] and [CZ22, Section 7.1], and thus Theorem 5 provides many examples of spectral networks with the *same* asymptotic conditions: the spectral curves $L_{\mathbf{w}}$ for different such weaves \mathbf{w} are typically *not* Hamiltonian isotopic, but they are all exact Lagrangian fillings of the same $\Lambda_{\partial \mathbf{w}}$. The associated spectral networks $\Theta_{\mathbf{w}}$ are themselves also different: in fact, the notion of (Lagrangian disk) mutation for Betti Lagrangians aligns with the notion of mutation of spectral networks, corresponding to the appearance of a 4d BPS state in a degenerate spectral curve, cf. Section 7.3.2. Second, since the construction of $\Theta_{\mathbf{w}}$ is combinatorial, not requiring to solve



FIGURE 3. (1) A weave \mathfrak{w} and the spectral network $\mathcal{W}_{\mathfrak{w}}$ constructed in Section 7.2. (2) The spectral network $\mathcal{W}_{\mathfrak{w}}$ in (1), drawn on its own, without the weave \mathfrak{w} . (3) Another weave with its associated spectral network. In this case, the spectral network recovers precisely the Berk-Nevins-Roberts network from [BNR82] by adiabatically degenerating the rigid pseudoholomorphic strips contributing to the augmentation associated to the Lagrangian filling of the weave.

any differential equation, Theorem 5 is an effective tool for understanding which spectral networks are associated to irregular data. In particular, given a higher-order linear differential equation – e.g. with new Stokes curves as in [BNR82] – one can qualitatively draw spectral networks associated to it once the Legendrian associated to the Stokes data is computed, which only requires the asymptotic range. Section 7.3 gives several such examples and computations.

Acknowledgements: We thank Merlin Christ, Ian Le, Andrew Neitzke and Daping Weng for useful discussions. Specifically, the construction in Section 7.2 was first discussed by R.C. with

I. Le and D. Weng in relation to another project and we are thankful for their input and generosity. R.C. is also particularly grateful to M. Aganagić and P. Zhou for their interest and encouragement on this work during his visit to their String Theory-Math Seminar at UC Berkeley. Y.J.N. is grateful to Yong-Geun Oh for his interest and continued support. R.C. is supported by the NSF CAREER DMS-1942363, a Sloan Research Fellowship of the Alfred P. Sloan Foundation and a UC Davis College of L&S Dean's Fellowship. □

2. Betti Lagrangians and Flowline Trajectories

This section introduces Betti surfaces in Section 2.2 and Betti Lagrangians in 2.3, where exact Betti Lagrangians and meromorphic spectral curves are also discussed. Real and holomorphic flowlines are discussed in Sections 2.4 and 2.6 respectively. Accordingly, the trapping lemma is proved in Section 2.5 and the asymptotics of WKB trajectories are studied in Section 2.7. We start with a brief discussion on real and holomorphic cotangent bundles in Section 2.1.

2.1. Real and holomorphic cotangent bundles. Let S be a smooth real surface, we denote by (T^*S, λ_{st}) its real cotangent bundle endowed with the canonical Liouville 1-form $\lambda_{st} \in \Omega^1(T^*S)$. It is the unique 1-form such that $\eta^*\lambda_{st} = \eta$ for any $\eta \in \Omega^1(S)$, and its differential is denoted by $\omega_{st} = d\lambda_{st}$. The ideal contact boundary [Gir20, Prop. 2] of this Liouville domain is denoted $(T^{\infty}S, \xi_{st})$. In this article, we focus on real Lagrangian surfaces $L \subset (T^*S, \lambda_{st})$ with boundary on a specified Legendrian $\Lambda \subset (T^{\infty}S, \xi_{st})$. The Legendrian boundary Λ will always be smooth, and L will typically be embedded, though we also consider immersed, and more singular, Lagrangian subsets in certain parts. We refer to [AG01; Arn90] or [Gei08; MS98] for initiating matters on contact and symplectic topology.

Let S be endowed with a Riemann surface structure \mathbb{S} , so that \mathbb{S} is a complex 1-dimensional manifold and $J_{\mathbb{S}}: TS \longrightarrow TS$ is its defining (almost) complex structure. The holomorphic cotangent bundle will be denoted by $(\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$, and its canonical (complex-valued) holomorphic symplectic form by $\omega_{\mathbb{C}} \in \Omega^2_{\mathbb{C}}(\mathbb{T}^*\mathbb{S}) = \Omega^{2,0}(T^*S, J_{\mathbb{S}})$. Here the splitting of $\Omega^{\bullet}(T^*S)$ into its $\Omega^{i,j}(T^*S, J_{\mathbb{S}})$ parts is determined by the eigenspaces of $J_{\mathbb{S}}$. As in the real case, this 2-form can be written as $d\omega_{\mathbb{C}} = \lambda_{\mathbb{C}}$, where the primitive $\lambda_{\mathbb{C}} \in \Omega^1_{\mathbb{C}}(\mathbb{T}^*\mathbb{S}) = \Omega^{1,0}(T^*S, J_{\mathbb{S}})$ is the canonical holomorphic Liouville form. In this holomorphic context, holomorphic Lagrangian submanifolds will be denoted by $\Sigma \subset (\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$. Given $(\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$, we can consider its underlying real rank-2 bundle T^*S and the real-valued smooth symplectic 2-form $\operatorname{Re}(e^{i\theta}\omega_{\mathbb{C}})$, for a choice of $\theta \in S^1$. We refer to this pair $(T^*S, \operatorname{Re}(e^{i\theta}\omega_{\mathbb{C}}))$ over Sas the θ -part $(\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$; when $\theta = 0$, this is also said to be the real part of $(\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$.

Remark 2.1. (i). Let $\theta \in S^1$ and $\Sigma \subset (\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$ be a holomorphic Lagrangian submanifold. Then, then underlying real submanifold $L \subset T^*S$ of Σ is a real Lagrangian submanifold of $(T^*S, Re(e^{i\theta}\omega_{\mathbb{C}}))$, where we can identify TL and T Σ as real rank-2 bundles. Indeed, $\omega_{\mathbb{C}}|_{T\Sigma} \equiv 0$ if and only if $e^{i\theta}\omega_{\mathbb{C}}|_{T\Sigma} \equiv 0$, which implies that $Re(e^{i\theta}\omega_{\mathbb{C}}|_{T\Sigma}) \equiv 0$. (ii) Studying the real Lagrangian L in (i) in the θ -part $(T^*S, Re(e^{i\theta}\omega_{\mathbb{C}}))$ is equivalent to studying the real Lagrangian underlying $e^{i\theta}\Sigma$ for (T^*S, ω_{st}) , where $e^{i\theta}\Sigma$ is the result of applying fiberwise multiplication to Σ by the scalar $e^{i\theta}$. We thus denote such real Lagrangians as $L = e^{i\theta}\Sigma$.

(iii) For a generic $\theta \in S^1$, the real Lagrangian $e^{i\theta}\Sigma$ submanifold of (T^*S, ω_{st}) is not exact, i.e. λ_{st} defines a closed but non-exact real 1-form when restricted to $e^{i\theta}\Sigma$. For instance, the real part of Σ is exact if and only if all the complex periods of Σ are purely imaginary. Nevertheless, note also that for generic $\theta \in S^1$, $e^{i\theta}\Sigma \subset (T^*S, \omega_{st})$ does not bound any holomorphic disks either (in an appropriate adiabatic limit) and, in many aspects, behaves similarly to an exact Lagrangian. \Box



FIGURE 4. A smooth surface **S** of genus 3 with 9 marked points $\boldsymbol{m} = \{m_1, \ldots, m_9\}$, in orange, and fronts for their associated Legendrian links in $\boldsymbol{\Lambda} = \{\Lambda_1, \ldots, \Lambda_9\}$, in blue. The annular neighborhoods $A_i \subset S$ for each marked point are shaded in pink. The positive braid words $\beta_1, \beta_5, \beta_8$ are the empty braid words in 3-strands and, for instance, we can choose $\beta_6 = \sigma_2^2 \sigma_1 \sigma_2 \sigma_1^2$ and $\beta_9 = \sigma_1 \sigma_2$.

2.2. Betti surfaces. The asymptotics of Betti Lagrangians will be constrained by Legendrian links. Given a marked oriented closed surface (\mathbf{S}, \mathbf{m}) , denote $S = \mathbf{S} \setminus \mathbf{m}$ and consider a set $\{A_1, \ldots, A_{|\mathbf{m}|}\}$ of small annular neighborhoods $A_i \subset S$, each obtained by considering a small disk neighborhood of the *i*th marked point in S and removing such marked point. The required asymptotics of $L \subset (T^*S, \lambda_{st})$ will be prescribed by a set $\mathbf{\Lambda} = \{\Lambda_1, \ldots, \Lambda_{|\mathbf{m}|}\}$ of Legendrians, each in the ideal contact boundary $\Lambda_i \subset (T^{\infty}A_i, \xi_{st})$. Since projection of the ideal contact boundary onto the zero section is a Legendrian fibration, the front for Λ_i can (and will) be drawn in $A_i \subset S$. See Figure 4 for a depiction of this setup. Given a collection $\beta = \{\beta_1, \ldots, \beta_{|\mathbf{m}|}\}$ of positive braid words $\beta_i \in \operatorname{Br}_n^+$ in *n*-strands, $i \in [1, |\mathbf{m}|]$, consider the Legendrian link $\Lambda_\beta \subset (T^{\infty}S, \xi_{st})$ whose front is given by drawing the braid word β_i circularly around A_i , according to the orientation of S; cf. [CN22, Section 2.2] and see again Figure 4. The Legendrian isotopy class of Λ_β is independent of the braid word representative of its associated braid element in the braid group. **Remark 2.2.** The braids in β need not be algebraic braids, i.e. they are not constrained to be certain iterated cables of the unknot. See the proof of [CG22, Corollary 1.6] for infinitely many examples of non-cable satellite knots and hyperbolic knots that are associated to positive braids (and necessarily not algebraic). In the meromorphic setting, as in the literature on spectral curves [GMN13a; GMN13b; GMN14], the braids are necessarily algebraic.

Formalizing this, a marked oriented closed surface (\mathbf{S}, \mathbf{m}) with such Legendrians links leads to the following:

Definition 2.3 (Betti surfaces). A Betti surface of rank n is a triple $(\mathbf{S}, \boldsymbol{m}; \boldsymbol{\Lambda})$ such that $\boldsymbol{\Lambda}$ is a collection of Legendrian links in the ideal contact boundary $(T^{\infty}S, \xi_{st})$, each of which is associated to an n-stranded positive braid circularly drawn around a marked point in \boldsymbol{m} .

See Figure 4 for an example of Definition 2.3. For each $\Lambda_{\beta_i} \subset (T^{\infty}A_i, \xi_{st})$, a Reeb chord is said to be positive if, when seen as a crossing in the Lagrangian projection, the upper strand travels from northwest to southeast with respect to the given orientations on the strands; else, we say it is *negative*. For notation purposes, Λ_{β} is said to be Reeb-positive if all the Reeb chords of all Λ_i are positive. After a C^{∞} -small Legendrian isotopy, we can and do assume that each of the Reeb chords and transverse double points (of the fronts in A_i) are at different angles. The angles in the core $S^1 \subset A_i$ where there is a Reeb chord for Λ_{β_i} are said to be *crossing points* for that core circle.

Remark 2.4. By [CN22, Section 2.2], if β_i contains a half-twist w_0 as a subword, then the braid for its front can be drawn such that Reeb chords are in bijection with the crossings of β . Specifically, the front can be drawn so that there is exactly one Reeb chord (arbitrarily close and) preceding each crossing, and these are all the Reeb chords.

For Floer-theoretic purposes, it is useful to regard the marked points \boldsymbol{m} as giving the *horizontal* infinity of S, e.g. as opposed to a 1-point compactification of S. In detail, for each $m_i \in \boldsymbol{m}$, we identify the associated annular neighborhood $A_i \subset S$ with $A_i \cong S^1 \times (1, \infty)$ with coordinates $(e^{i\theta}, r) \in S^1 \times (1, \infty)$, so that any point converges to the point m_i in \mathbf{S} as $r \to \infty$. For a Reeb chord with a given crossing point θ_0 , we call the ray $(e^{i\theta_0}, r) \subset A_i$ its Reeb chord ray.

Example 2.5 (Betti surfaces from Stokes data). As hinted in [KKP08, Remark 2.17], the data of the irregular singularities of a meromorphic connection on a Riemann surface S leads to a Betti surface. This is beautifully crystallized in [Boa21], specifically in the notion of an irregular curve $(\mathbb{S}, \boldsymbol{m}, \Theta)$, as defined in [Boa14, Definition 8.1] or [Boa21, Definition 5.5]; cf. also [Sab13, Part I], [She+19, Section 3.3], [Su25, Appendix B]. In a nutshell, an irregular curve is a marked Riemann surface $(\mathbb{S}, \boldsymbol{m})$ with the data Θ of an irregular class at each marked point: the irregular class codifies the formal type of the pole of a meromorphic connection, up to formal gauge transformation. In practice, this can be written as a finite Puiseux series Q, representing an orbit of the Galois action of the profinite group $\hat{\mathbb{Z}}$ on $\mathbb{C}((z^{1/\infty}))/z^{-1}\mathbb{C}[[z^{1/\infty}]] := \bigcup_{N \in \mathbb{N}} \mathbb{C}((z^{1/N}))/z^{-1}\mathbb{C}[[z^{1/N}]]$. Here we will

work with representatives Q of the form

$$Q = \sum_{k_j \in \mathbb{Q}_{>0}} c_{k_j} z^{-k_j}, \quad c_{k_j} \in \mathfrak{t}_{reg} \subset \mathfrak{gl}_n,$$

modulo the action of $c_{k_j} \to c_{k_j} e^{2\pi i n/\operatorname{ram}(Q)}$, $n \in [1, \operatorname{ram}(Q)]$, where $\operatorname{ram}(Q)$ is the smallest number such that $Q \in \mathbb{C}((z^{1/\operatorname{ram}(Q)}))$. The irregular part of the formal type of the meromorphic connection is then given by d - dQ. Since Q is a finite Puiseux series with pole orders strictly greater than 0, it can be seen as a multivalued function on \mathbb{C}^* . The Stokes diagram associated to such irregular data is given by the graphs of the multivalued functions $\operatorname{Re}(Q(re^{i\theta}))$ on S^1_{θ} , for fixed r large enough. For each marked point m_i , the Stokes diagram in A_i is precisely the circular closure of a positive braid, and thus can be interpreted as a front for a Legendrian of the type Λ_{β_i} above.

In summary, the Betti surface associated to an irregular curve $(\mathbb{S}, \boldsymbol{m}, \Theta)$ is precisely the underlying smooth surface $(\mathbf{S}, \boldsymbol{m}, \Lambda_{\Theta})$, where Λ_{Θ} is the Legendrian link in $(T^{\infty}S, \xi_{st})$ defined by the Stokes diagrams of the irregular data Θ , understood as fronts in S. In these examples, the points of maximal decay for the Stokes diagram of an irregular type give the Reeb rays for the associated Legendrian link; these are also referred to as singular directions or anti-Stokes directions of the Stokes diagram. By construction, the Stokes directions (a.k.a. oscillatory directions) of the Stokes diagram are precisely the directions where there is a crossing in the front of the associated Legendrian link, i.e. the angles at which we draw the crossings of the associated braid word. \Box

2.3. Betti Lagrangians. Given a Betti surface $(\mathbf{S}, \boldsymbol{m}, \boldsymbol{\Lambda})$, we introduce Betti Lagrangians: the Lagrangian surfaces in (T^*S, λ_{st}) for which we construct and study spectral networks. These Lagrangians are our main object of study. There are two important classes, *exact Betti Lagrangians* and *meromorphic spectral curves*, discussed respectively in Subsections 2.3.1 and 2.3.2. Many of the Lagrangian fillings of Legendrian links studied in the literature, e.g. [CG22; CW24; CZ22; EHK16], provide plenty of examples of this first class of exact Betti Lagrangians. The irregular curves studied in the context of wild character varieties and the irregular Riemann-Hilbert or non-Abelian Hodge correspondences give many examples of the second class, e.g. cf. [BB04; Boa14; Boa14; FN21; GMN12; GMN13b; GMN14; Kat+15; Sab13].

To define Betti Lagrangians, we first discuss the notion of Lagrangian multigraphs on a smooth manifold M, as follows. By definition, a properly embedded Lagrangian submanifold $L \subset T^*M$ is said to be a Lagrangian multi-graph if outside a codimension-1 properly closed subset $K_L \subset L$, the projection $\pi : L \longrightarrow M$ is a smooth immersion outside K_L and it is a covering over $M - \pi(K_L)$. Given a point $m \in M - \pi(K_L)$, a neighborhood of the preimage $\pi^{-1}(m) \cap (L - K_L)$ in M consists of n disjoint Lagrangian disks, the smooth sheets of L over m. Each such smooth sheet can be realized as the Lagrangian graph of the differential of a smooth function $f_i :\longrightarrow \mathbb{R}, i \in [1, n]$, with each f_i well-defined up to constant.

Definition 2.6 (Betti Lagrangians). Let $(\mathbf{S}, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface. A Betti Lagrangian $L \subset T^*S$ is a weakly bounded Lagrangian multigraph $L \subset T^*S$ such that the restriction of the projection

 $\pi: L \to S$ onto the zero section S is a degree-n simple branched cover with finitely many branched points.

In Definition 2.6 we have used the notation *weakly bounded*, which we use to encode a number of Riemannian properties of the Lagrangian multigraph L. Its precise meaning is as follows:

Definition 2.7. A Lagrangian multigraph $L \subset T^*M$ is said to be *weakly bounded* if there exists a Riemannian metric (M, g), constants $inj(g), \rho \in \mathbb{R}_+$ and a precompact subset $K \subset M$ containing $\pi(K_L)$ such that:

- (i) The Riemannian metric g is complete, the C^{∞} -norm of all the derivatives of the curvature tensor are uniformly bounded (a.k.a. geometrically bounded), and the minimal injectivity radius of g is bounded below by inj(g) > 0.
- (ii) Given a point $m \in M N_{2inj(g)}(K)$, at distance at least 2r from K, and the functions $f_i: M \longrightarrow \mathbb{R}$ describing the sheets of L over $B_{inj(g)}(m)$ as $gr(df_i)$, then we must have

$$\inf_{B_{\mathrm{inj}(g)}(m)} \left| df_i - df_j \right| > \rho, \quad \forall q \in B_{\mathrm{inj}(g)}(m).$$

By definition, a weakly bounded Lagrangian multigraph L is said to be *uniformly bounded* if L lies in the R-disk bundle $D_R T^*S \subset T^*S$ for some $R \in \mathbb{R}_+$ and for the smooth sheets of its fiberwise ε -scaled image $\varepsilon L \subset T^*S$ over $B_{inj(g)}(m)$, $m \in M - N_{2inj(g)}(K)$, there exist uniform bounds C_k such that $\|\nabla^k f_i\|_g \leq C_k$ for all $i \in [1, n]$, where ∇ is Levi-Civita connection of (M, g) and there is no dependence on ε .

Intuitively, the second condition for a uniformly bounded Lagrangian multigraph in Definition 2.7 is that away from the preimage of the locus K, the sheets of the rescaled Lagrangian $\varepsilon L \subset T^*S$ uniformly C^{∞} -converge with respect to g to the zero section as $\varepsilon \to 0$.

It is worth providing a local description of the Betti Lagrangian near its branch point. The germ $\{w^2 - z = 0\}$ in $\mathbb{C}^2 = T^*\mathbb{C}$ is called the holomorphic cusp singularity, and we call its real part the D_4^- -singularity. Near its branch points, L is always locally equivalent to the D_4^- -singularity germ. For the rest of the paper, We will assume that given a branch point b, there exists some local conformal coordinate z near b, and some constant holomorphic covector c such that near the ramification locus $\pi^{-1}(b)$, L is equal to the real part of $\{(\lambda_{\mathbb{C}} - c)^2 = zdz^2\} \subset T^*_{\mathbb{C}}\mathbb{C}$ near z = 0. We will furthermore assume that the smooth sheets of L are all holomorphic.

2.3.1. Exact Betti Lagrangians. For exact Betti Lagrangians, the analysis uses the conical ends, as with the framework of Lagrangian fillings from [EHK16, Section 2.2]. Consider the Legendrian fiber in $(T^{\infty}\mathbf{S}, \xi_{st})$ above a marked point $m_i \in \mathbf{m}$ and identify a neighborhood of it with (J^1S^1, ξ_{st}) , where the fiber is mapped to the zero section. Suppose that the Legendrian link $\Lambda_i \subset (T^{\infty}S, \xi_{st})$ is parametrically given by $(x(\theta), y(\theta), z(\theta))$ where $x(\theta) = \theta$ and the contact 1-form is dz - ydx, $(x, y) \in T^*S^1$ and $z \in \mathbb{R}$. Then a conical end is defined as follows: **Definition 2.8** (Conical ends). Let $L \subset (T^*D^2, \lambda_{st})$ be the germ of a Lagrangian surface and $\Lambda \subset (J^1S^1_{\theta}, \xi_{st})$ be parametrized by $(x(\theta), y(\theta), z(\theta))$, where S^1 is identified with the Legendrian fiber $T_0^{\infty}D^2$ of $(T^{\infty}D^2, \xi_{st})$ above 0. By definition, L has a conical end of type Λ if L admits a parametrization of the form

(1)
$$(r,\theta) \longmapsto (x(\theta), f(r)y(\theta), r, f'(r)z(\theta)) \in T^*D^2,$$

for some strictly increasing positive real function $f(r) : [1, \infty) \longrightarrow \mathbb{R}$ which is linear at infinity. In other words, L is the multigraph associated to the multi-valued function $z(\theta)f(r) : S^1 \times [1, \infty) \longrightarrow \mathbb{R}$.

The conical ends in Definition 2.8 allow us to introduce the following:

Definition 2.9 (Exact Betti Lagrangians). Let $(\mathbf{S}, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface. An *exact Betti* Lagrangian $L \subset (T^*S, \lambda_{st})$ of rank n is an exact Lagrangian submanifold $L \subset (T^*S, \lambda_{st})$ such that

(i) The restriction $\pi: L \to S$ of the projection $T^*S \longrightarrow S$ onto the zero section is a degree-*n* simple branched cover, with finitely many branch points.

(ii) L has conical ends near each of the marked points in \boldsymbol{m} .

Let (S, g) be a Riemannian metric such that g can be written as $g = dr^2 + r^2 d\theta^2$ near each of the horizontal infinities of the Betti surface S. Then the conical condition in Definition 2.9.(ii) implies that exact Betti Lagrangians are uniformly bounded with respect to such Riemannian metric g. Therefore, exact Betti Lagrangians are Betti Lagrangians, according to Definition 2.6.

2.3.2. Meromorphic spectral curves. Let us introduce a second important class of Betti Lagrangians, meromorphic spectral curves, associated to an irregular curve $(\mathbb{S}, \boldsymbol{m}, \Theta)$; see [Boa21, Definition 5.5] and Example 2.5 above for irregular curves. Intuitively, these spectral curves are given by meromorphic multi-graphs on \mathbb{S} with their behavior at a pole in \boldsymbol{m} specified by the irregular data Θ . Locally, the behavior at poles can be described as follows.

Consider a germ of a holomorphic Lagrangian multi-graph $\Sigma \subset T^*\mathbb{C}^*$ over \mathbb{C}^* without singularities. Since the projection of Σ to the base \mathbb{C}^* defines an *n*-sheeted covering, we can consider the elementary symmetric polynomial \mathbf{c}_i , $i \in [1, n]$, over the distinct sheets $\lambda_1, \ldots, \lambda_n$ of Σ and obtain holomorphic functions $\mathbf{c}_i(\lambda_1, \ldots, \lambda_n)$ on \mathbb{C}^* , $i \in [1, n]$. By definition, Σ is said to extend to a *meromorphic multi-graph* over \mathbb{C} if the elementary symmetric polynomials extend over \mathbb{C} as meromorphic functions with poles over the origin. After possibly passing to a ramified cover, under $z \to z^r$ for some $r \in \mathbb{N}$, we regard the sheets $\lambda_1, \ldots, \lambda_n$ of Σ as giving holomorphic functions on \mathbb{C}^* . By Lagrange's upper bound $\max\{1, \sum_{i=1}^n \left| \frac{\mathbf{c}_i}{\mathbf{c}_n} \right|\}$ on the norm of complex roots, cf. [Lag73, Chapter IV], there exists $N \in \mathbb{N}$ large enough such that the norm of $z^N \lambda_i(z^r)$ is uniformly bounded for all $i \in [1, n]$. Therefore, the functions $\lambda_i(z^r) : \mathbb{C}^* \longrightarrow \mathbb{C}$ all extend to germs of meromorphic functions on \mathbb{C} near the origin and we can regard the sheets of Σ as the Lagrangian graphs of Puiseux series with finite polar parts.

Given a meromorphic multi-graph over \mathbb{C} , as above, we consider the differential operator

$$\partial_z^n - \sum_{i=1}^n \mathfrak{c}_i(\lambda_1, \dots, \lambda_n) \partial_z^{n-i}, \quad z \in \mathbb{C}.$$

By reducing this operator to a system of order 1 differential operators, it defines a meromorphic connection on the rank-*n* trivial bundle over \mathbb{C} with a single pole at the origin. The irregular data is then given by considering the irregular class of such meromorphic connection. Intuitively, these Galois orbits are given by a local primitive of the holomorphic Liouville form λ_{st} over the distinct components of the multi-graph Σ , after dropping the $o(\log(z))$ terms.

Definition 2.10 (Meromorphic Spectral Curves). Let $(\mathbb{S}, \boldsymbol{m}, \Theta)$ be an irregular curve. A holomorphic submanifold $\Sigma \subset T^*S$ is said to be a *meromorphic spectral curve* if

- (i) the germ of Σ is a meromorphic multi-graph with poles on \boldsymbol{m} ,
- (ii) the corresponding formal Puiseux series has the same orbits as Θ, with the same multiplicities.

In addition, a meromorphic spectral curve is said to have O(-1)-ends if the sheets $\lambda_1, \ldots, \lambda_n$ and their difference functions have growth bounded below by z^{-1} , locally in $z \in \mathbb{C}^*$ with a pole at z = 0.

For the purposes of our results, including Theorems 1, 2, 5 and 4, we always consider meromorphic spectral curve with O(-1)-ends. First, O(-1)-ends are necessary because, without such hypothesis, one could add a holomorphic perturbation to force the difference between the sheets to blow-up slower than z^{-1} as $z \to 0$. Second, having O(-1)-ends implies that the θ -real part of a meromorphic spectral curve is a Betti Lagrangian as in Definition 2.6, as it implies that the underlying multi-graph is weakly bounded. In practice, here and in all situations we shall choose a Kähler metric on S which coincides with $|z|^{-2}dz \otimes dz$ near the marked points.

Remark 2.11. (i) If an irregular class in the irregular data Θ has multiplicity, the associated components of this same type coincide in $\mathbb{C}((z^{1/\infty}))/z^{-1}\mathbb{C}[[z^{1/\infty}]]$ and thus differ by a finite Puiseux series in $z^{-1}\mathbb{C}[[z^{1/\infty}]]$. In contact topological terms, the Legendrian link associated to an irregular class with multiplicity $m \in \mathbb{N}$ corresponds to the m-copy Reeb push-off of the Legendrian link associated to the irregular class (without multiplicity). This choice guarantees that the O(-1)-condition can be satisfied.

(ii) It follows from the construction of Stokes diagrams, see Example 2.5, that the condition that the difference of the sheets grows faster or equal than z^{-1} is equivalent to requiring that the order of the poles of the multi-valued 1-form $dQ_i - dQ_j$, $i, j \in [1, n]$, are all greater or equal than 1. That is, the (difference of the) polar terms of Q themselves are of degree greater than 0, e.g. no holomorphic parts.

(iii) An alternative viewpoint of meromorphic multi-graph is as follows, cf. [IM21; Kuw24]. Given a rank n meromorphic multi-graph, its cameral cover is given by taking all the possible permutations of the sheets. Then, each pair $i, j \in [1, n], i < j$, induces a covering of \mathbb{C}^* over which the germ of the cameral cover is realized as the germ of a rank 2 quadratic differential near its pole. Then the condition that the difference of the sheets grow faster or equal than z^{-1} is equivalent to requiring that the order of the poles of this quadratic differential are all greater or equal than $2.\square$

Example 2.12. (1) Airy Equation. As an explicit instance of Example 2.5, consider the trivial \mathbb{C}^2 bundle over the Riemann surface $\mathbb{S} = \mathbb{C}$ associated to the Airy differential equation f''(x) = xf(x), $x \in \mathbb{C}$. It has an irregular singularity at $x = \infty$, which we take to be the unique marked point of the associated irregular curve $\mathbb{CP}^1 = \mathbb{S} \cup \{\infty\}$; this is the equation originally studied by G. Stokes [Sto09]. In the x-coordinate, the associated meromorphic connection can be written as

$$\nabla_m := d - \begin{bmatrix} 0 & 1 \\ x & 0 \end{bmatrix} dx = d - dQ, \quad Q = \begin{bmatrix} 0 & x \\ x^2/2 & 0 \end{bmatrix}, \quad x \in \mathbb{C}.$$

The formal irregular type of ∇_m at $x = \infty$ is given by $\langle x^{3/2} \rangle$, as the eigenvalues of Q are $e^{\pm x^{3/2}/\sqrt{2}}$. Thus we have the irregular curve $(\mathbb{S}, \boldsymbol{m}, \Theta) = (\mathbb{CP}^1, \infty, \langle x^{3/2} \rangle)$. Its associated Stokes diagram at $x = \infty$ is described by the braid word $\beta = \sigma_1^3$, so the Betti surface is $(S, \boldsymbol{m}, \Lambda) = (S^2, \infty, \Lambda_{\sigma_1^3})$. After satelliting the Legendrian zero section of (J^1S^1, ξ_{st}) to the max-tb Legendrian unknot in (\mathbb{R}^3, ξ_{st}) , where S^1 is the Legendrian fiber of $T^{\infty}S^2 \longrightarrow S^2$ at infinity, the Legendrian knot $\Lambda_{\sigma_1^3}$ satellites to the max-tb unknot itself, presented in the front as the (-1)-closure of $\beta = \sigma_1^3$. Any θ -part of the associated meromorphic spectral curve $\Sigma = \{(x, \eta) \in T^*\mathbb{C} : \eta^2 = x\} \subset T^*\mathbb{C}, \Sigma \cong \mathbb{C}$, thus gives a Betti Lagrangian disk filling of the max-tb unknot. In terms of weaves, giving the front of the Legendrian lift of the θ -part of Σ , this particular meromorphic spectral curve gives a 2-weave with a unique trivalent vertex.

For coherence with previous notation, we can rewrite the above in terms of the coordinate $z = x^{-1}$, locally around the pole, as the coordinate z is the one used in Example 2.5 and previously in this subsection. Then the Airy equation transforms to $z^4 f''(z) + 2z^3 f'(z) = f(z)z^{-1}$: at $z \neq 0$, this is equivalent to $f''(z) + 2z^{-1}f'(z) = f(z)z^{-5}$, which has an irregular singularity at z = 0 because the coefficient z^{-5} of f(z) has a pole greater than 2 at z = 0. Its analysis of the irregular type at z = 0 coincides with that of the Airy equation at $z^{-1} = x = \infty$.

(2) Irreducible isolated plane curve singularities. The discussion in (1) readily generalizes for differential operators $P(x, \partial_x)$ on $\mathbb{CP}^1 \setminus \{\infty\}$, $x \in \mathbb{C}$ whose spectral curve $\Sigma = \{(x, \eta) \in T^*\mathbb{C} : P(x, \eta) = 0\}$ has an irreducible isolated singularity at the origin $(x, \eta) = 0$. In these cases, the irregular curve $(\mathbb{S}, \boldsymbol{m}, \Theta) = (\mathbb{CP}^1, \infty, \Theta_P)$ has as irregular data Θ_P a Stokes diagram given by a braid $\beta(P)$ that represents – upon satelliting along the unknot – the link of the singularity $P : \mathbb{C}^2 \longrightarrow \mathbb{C}$. The Betti surface is thus of the form $(S, \boldsymbol{m}, \Lambda) = (S^2, \infty, \Lambda_{\beta(P)})$, where the smooth type of $\Lambda_{\beta(P)}$, as a link in $S^3 = \partial(T^*\mathbb{C})$, is that of the link of the singularity. The theory of algebraic links allows to readily compute such smooth type using a Puiseux series for P (via Newton exponents), expressing this link as an iterated cable link with positive enough (p_i, q_i) coefficients; see e.g. [EN85, Chapter I & II] or [Ghy17, Chapters 17 & 18]. By [Cas22, Prop. 2.2], algebraic links have a unique max-tb Legendrian representative, and thus the Legendrian isotopy type of $\Lambda_{\beta(P)}$ is uniquely determined by the irregular data Θ_P .

This construction leads to a large class of Betti Lagrangians in $(\mathbb{R}^4, \omega_{st})$ which are Lagrangian fillings of max-tb Legendrian representatives of algebraic links.⁴ At the same time, their associated meromorphic connections are also important objects of study: these include the differential equations of Airy, Bessel, Clifford, Weber, Whittaker and general differential operators of Airy type (cf. [HJ22; Kat87]), among others.⁵ By construction, these spectral curves are singular. In order to obtain smooth (embedded) Lagrangian fillings for such links $\Lambda_{\beta(P)}$ one must morsify the singularity P: as explained in [Cas22], any real morsification will give rise to an exact embedded Lagrangian filling. Furthermore, since the front singularities of holomorphic Legendrians are real codimension-2, these Lagrangian fillings can all be described by weaves [CZ22]. Building on (1) above, a simple instance is that, for generic θ , the θ -part of the meromorphic spectral curve associated to a morsification of $f''(x) = x^3 f(x)$ gives Betti Lagrangians for the Betti surface $(S, \boldsymbol{m}, \Lambda) = (S^2, \infty, \Lambda_{\sigma_1^5})$: as we vary $\theta \in S^1$, this actually recovers the 5 known Lagrangian fillings of the max-tb trefoil. In the general case, there is a moduli for the space of meromorphic connections on $\mathbb C$ with such fixed irregular data at $x = \infty$, including such morsifications, and it is isomorphic to the moduli of sheaves with singular support on these Legendrian links. See e.g. [CW24; Su25]. The θ -parts of different such meromorphic spectral curves give rise to different Lagrangian fillings: both varying θ and varying the meromorphic connection typically changes the Hamiltonian isotopy type of the Betti Lagrangian, see e.g. [Hug23] for how varying $\theta \in S^1$ for a fixed morsification of $f''(x) = x^m f(x)$ leads to different orbits of Lagrangian fillings for (2, m)-torus links.

(3) BNR Equation. Consider the Berk-Nevins-Roberts differential equation, cf. [BNR82]. This is the order 3 differential equation

$$\partial_x^3 f(x) - 3\partial_x f(x) + x f(x) = 0, \quad x \in \mathbb{C},$$

known for being one of the first studied examples with higher rank Stokes phenomenon. In the coordinate $z = x^{-1}$, the equation is transformed to

$$z^{6}\partial_{z}^{3}f(z) + 6z^{5}\partial_{z}^{2}f(z) + (6z^{4} - 3z^{2})\partial_{z}f(z) - z^{-1}f(z) = 0, \quad z \in \mathbb{C}$$

⁴Technically, as given, they are asymptotically Lagrangian fillings, but this can be corrected as in [Cha10, Section 5.1].

⁵See 48min in P. Boalch's talk "First Steps in Global Lie Theory" at the Simons Center during the program "The Stokes Phenomenon and its applications in Mathematics and Physics" as well as his *spirograph Stokes diagrams* applet in his website.

which has an irregular singularity at z = 0, e.g. because the coefficient of f(z) has a pole of order 7. A direct computation of the irregular data gives that the irregular curve is

$$(\mathbb{S}, \boldsymbol{m}, \Theta) = (\mathbb{CP}^1, \boldsymbol{m}, \langle x^{4/3} \rangle)$$

The positive 3-stranded braid word associated to this irregular data is $\beta = (\sigma_1 \sigma_2)^2$ and thus the Betti surface is $(S, \boldsymbol{m}, \Lambda) = (S^2, \infty, \Lambda_{(\sigma_1 \sigma_2)^2})$. The Betti Lagrangian $L \subset (T^* \mathbb{R}^2, \lambda_{st})$ associated to this meromorphic spectral curve $\Sigma \subset T^* \mathbb{C}$ consists of a unique embedded Lagrangian disk, as it is graphical in $x \in \mathbb{C}$. Indeed, it is a 3-weave that can be parametrized by taking the real part of the holomorphic Legendrian lift of

$$\Sigma = \{ (x,\eta) \in T^* \mathbb{C} : \eta^3 - 3\eta + x = 0 \} \subset T^* \mathbb{C},$$

understood as a holomorphic Lagrangian. Such Σ is a morsification of the function $x(\eta) = \eta^3$, so it fits within the description of (2) above, for a smooth germ in this case. In real coordinates $x = u + iv, u, v \in \mathbb{R}^2$, the front for this real Legendrian lift is parametrized by

$$\sigma: \mathbb{R}^2 \longrightarrow \mathbb{R}^3, \quad \sigma(u, v) = \left(u^3 - 3uv^2, 3u^2v - v^3, 3(u^3v - uv^3 - uv)\right),$$

which readily exhibits the boundary braid $\beta = (\sigma_1 \sigma_2)^2$ and the embedded Lagrangian disk, as there are no Reeb chords in this front. The contact topological proof that the Berk-Nevins-Roberts equation has no moduli is then simply the observation that $\Lambda_{(\sigma_1 \sigma_2)^2}$, understood as a Legendrian link in (S^3, ξ_{st}) after satelliting the Legendrian (unit) cotangent fiber along the max-tb Legendrian unknot, is actually Legendrian isotopic to the max-tb Legendrian unknot. Note that, in accordance to this, taking different θ -parts leads to a rotation of the 3-weave and the Hamiltonian isotopy type of the Betti Lagrangian remains the same.

2.4. Real Morse flowlines. Let (M, g) be a Riemannian manifold and $L \subset T^*M$ a Betti Lagrangian. Consider $m \in M - \pi(K_L)$ and let $\{f_1, \ldots, f_n\}$ define the smooth sheets of L over m, via the graphs of df_i .

Definition 2.13. A smooth map $\gamma : (-\epsilon, \epsilon) \longrightarrow M - K_L$ is said to be an *ij*-gradient flowline for L if it satisfies

(2)
$$\gamma'(s) + \nabla_g(f_i - f_j)(\gamma(s)) = 0, \quad \gamma(0) = m, \quad s \in (-\epsilon, \epsilon),$$

for the two local sheets f_i, f_j of L along $\gamma, i, j \in [1, n]$. The domain of definition of a flowline can be extended as long as it does not converge to a point in $\pi(K_L)$: a flowlines obtained by extending the domain of definition are called *continuations*. Maximal continuations of solutions of the gradient flowline equation (2) are said to be *trajectories*.

Given an *ij*-gradient flowline $\gamma: (-\epsilon, \epsilon) \longrightarrow M - K_L$, consider the lift

$$\bar{\gamma}^i : (-\epsilon, \epsilon) \longrightarrow L, \quad \bar{\gamma}^i(s) = (\gamma(s), d(f_i \circ \gamma)(s))$$

of γ to the *i*th sheet of L, and note the following energy inequality:

Lemma 2.14. Let (M, g) be a Riemannian manifold, $L \subset (T^*M, d\lambda)$ a Betti Lagrangian with λ the Liouville form, and $\gamma : (a, b) \longrightarrow M$ an ij-flowline for L. Then

$$\int_{\bar{\gamma}^i - \bar{\gamma}^j} \lambda = -\int_a^b \left| \nabla (f_i - f_j) \right|^2 = (f_i - f_j)(\gamma(b)) - (f_i - f_j)(\gamma(a))$$

Proof. This follows from $(\bar{\gamma}^i - \bar{\gamma}^j)^* \lambda = d(f_i - f_j)(\gamma'(s)) = \partial_s f$ and Equation (2).

Lemma 2.14 implies that the relative homology class $[\bar{\gamma}^j - \bar{\gamma}^i] \in H_1(L, \mathfrak{t})$ has positive λ -length, where \mathfrak{t} is the appropriate set of points in L giving the boundaries of the lifts $\bar{\gamma}^i$. This relative homology class $\bar{\gamma}^i - \bar{\gamma}^j$ is said to be the *cotangent ij-lift* of γ , and its λ -length above is said to be its *flow-energy*, a.k.a. its symplectic area. Note that, by Definition 2.13, the difference $f_i - f_j$ strictly *decreases* along the flowline. There are two important classes of flowlines in our context:

- (1) The flowline equation in (S^1, g_{flat}) for an immersed Betti Lagrangian $L = L_{\Lambda} \subset T^*S^1$ given by the Lagrangian projection of a Legendrian link $\Lambda := \{(\theta, y(\theta), z(\theta)\} \subset J^1S^1$ whose front in $S^1 \times \mathbb{R}$ has no cusps.
- (2) The flowline equation in a disk \mathbb{R}^2 for a Betti Lagrangian $L \subset T^*\mathbb{R}^2$ near a D_4^- -singularity. That is, near a branch point of the projection $L \to \mathbb{R}^2$ that, in the spatial wavefront in $\mathbb{R}^2 \times \mathbb{R}$ of the Legendrian lift of L, is a non-generic D_4^- singularity.

For case (1), we choose a base coordinate $\theta \in S^1$ and the *ij*-flowline equation for $\gamma : (-\epsilon, \epsilon) \longrightarrow S^1$ reads

(3)
$$\gamma'(s) + \left[(y_i(\theta) - y_j(\theta))\partial_\theta \right]_{|_{\gamma(s)}} = 0.$$

Here we denoted $\gamma'(s) = \gamma_*(\partial_s)$, so writing $\gamma'(s) = \theta'(s)\partial_\theta$ expresses (3) as $\theta'(s) + (y_i(\theta(s)) - y_j(\theta(s))) = 0$. Near a Reeb chord of the Legendrian link Λ , i.e. an immersed point of L_{Λ} , the local configuration of sheets reduces to that of the two Lagrangian strands

$$L_1 := (x, -x) \in T^*(-1, 1)$$
 and $L_2 := (x, x) \in T^*(-1, 1)$ meeting at $x = 0$.

In these local coordinates, the difference $y_1(x) - y_2(x) = 2x$ of the sheets y_1, y_2 gives the 12-flowline equation x'(s) - 2x(s) = 0 and the 21-flowline equation, which is x'(s) + 2x(s) = 0. These have solutions e^{2s} and e^{-2s} , respectively, and it follows that the 12-flowline equation has x = 0 as its unstable manifold, and the 21-flowline equation has x = 0 as its stable manifold. Assuming that L_1 and L_2 are all oriented in the direction ∂_x , the Reeb chord is positive if L_1 is above L_2 , and negative otherwise. This settles the local behavoir near a Reeb chord. Globally on $L_{\Lambda} \subset T^*S^1$, the trajectories must begin and end at some Reeb chord: Lemma 2.14 then implies that the difference function $z_i(\theta) - z_j(\theta)$ decreases along the trajectory. Here the sign of the difference function cannot be both negative at $s = -\infty$ and positive at $s = +\infty$. This describes the qualitative behavior of flowlines for case (1), which we summarize in the following:

Lemma 2.15. Let $L \subset T^*S^1$ be an immersed Betti Lagrangian $L_{\Lambda} \subset T^*S^1$ given by the Lagrangian projection of a Legendrian link $\Lambda := \{(\theta, y(\theta), z(\theta)\} \subset J^1S^1 \text{ whose front in } S^1_{\theta} \times \mathbb{R}_z \text{ has no cusps.}$ Consider a pair of points $c, c' \in S^1$ above each of which lies one crossing of L_{Λ} . Then:

- (1) There exist finitely many gradient trajectories beginning at c and terminating at c', up to translation.
- (2) Suppose each such ij-gradient trajectory is maximally well-defined an open interval $(a, b) = (a(c, c'), b(c, c')) \subset \mathbb{R}$ in the universal cover $\mathbb{R} \longrightarrow S^1$, up to \mathbb{Z} -translation. Then $\exists \epsilon = \epsilon(c, c') \in \mathbb{R}_+$ such that

$$z_i - z_j \neq 0$$
 if $\theta \in (a, a + \epsilon) \cup (b - \epsilon, b)$.

(3) $\exists T = T(\epsilon) \in \mathbb{R}_+$ a minimal escape time such that for any point $u \in (a + \frac{\epsilon}{2}, b - \frac{\epsilon}{2})$, an *ij-trajectory* γ with $\gamma(0) = u$ maps

$$\gamma((-\infty, -T)) = (a, a + \epsilon), \quad \gamma((T, \infty)) = (b - \epsilon, b),$$

if its orientation coincides with that of S^1 . Else, the same holds exchanging the two target intervals above. Here the constants $\epsilon, T \in \mathbb{R}_+$ do not depend on the choice of the lift of the trajectory.

For case (2) above, studying flowlines near a D_4^- -singularity, we proceed as follows. The base surface is $S = \mathbb{C}$ with coordinate $z \in \mathbb{C}$ and the Betti Lagrangian is $\Sigma = \operatorname{Re}\{w^2 - z = 0\} \subset T^*\mathbb{C}$. Note that this is the real part of the Lagrangian projection of the holomorphic Legendrian disk

$$\Lambda = \{(z, w; v) \in T^* \mathbb{C} \times \mathbb{C} : z = w^2, v = 2w^3/3\} \subset (J^1 \mathbb{C}, \xi_{hol}).$$

Here Λ is the Legendrian lift of the holomorphic simple cusp front $\{(w^2, 2w^3/3) : w \in \mathbb{C}\} \in \mathbb{C}_z \times \mathbb{C}_v$, where $\xi_{hol} = \ker\{dv - wdz\}$. The projection $L \longrightarrow \mathbb{C}_z$ is a 2-fold branched cover, branched at z = 0, and thus there are only two sheets and 12 or 21 flowlines. For such Σ , there are three gradient flowlines converging to z = 0, given by the three rays $\mathbb{R}_{\geq 0}$, $e^{\frac{2\pi}{3}}\mathbb{R}_{\geq 0}$ and $e^{-\frac{2\pi}{3}}\mathbb{R}_{\geq 0}$. For |z|large, these are respectively (rays over) the three Reeb chords at the Legendrian boundary, which is given by $\beta = \sigma_1^3$, cyclically understood. These three rays near a D_-^4 singularity are said to be the *initial trivalent rays*. In general, given be a Betti Lagrangian $L \subset T^*S$ and $b \in S$ a branched point of the projection $L \longrightarrow S$, L decomposes into its germ near the ramification point over b and the smooth sheets. Since the former germ comes from the real part of the holomorphic cusp, we refer to that connected component as the *cusp* component. We also refer to the gradient flowlines between the two sheets of the cusp component as cusp-cusp flowlines, and if the gradient flowlines are between the cusp component and a smooth sheet, we call them smooth-cusp flowlines. Thus, near a ramification point, only a cusp-cusp flowline in S is allowed to terminate at the associated branched point.

2.5. Asymptotics for conical ends (trapping lemma). Let us now explain how conical ends, as introduced in Definition 2.8, allow us to control the asymptotic behavior of flowlines as the

marked points $m \,\subset S$. Specifically, given the germ of a Lagrangian conical end, we prove a trapping property in Lemma 2.18 below: there exists a neighborhood of marked points such that the gradient trajectories in S cannot leave once inside. Technically, we first derive the trapping lemma for the conical metric $g = d\theta^2 + dr^2$ near the marked points, instead of the polar metric $dr^2 + r^2 d\theta^2$ used for the exact Betti Lagrangians in Definition 2.9.(ii). We then argue via Lemma 2.20 that there is a bijective correspondence between flowlines for the conical metric and those for the polar metric.

Consider the germ of a Lagrangian conical end $L = S^1 \times [1, \infty) \subset T^*D^2$ as in Definition 2.8, where we use the same notation $(\theta, r) \in S^1 \times [1, \infty)$ and $(x(\theta), y(\theta), z(\theta)) \in J^1S^1_{\theta}$ for its conical type; here the origin of D^2 is given by $r \to \infty$. Then the gradient flowline equation (2) takes the form

(4)
$$\gamma'(s) + \left[(y_i(\theta) - y_j(\theta))r\frac{d}{d\theta} + (z_i(\theta) - z_j(\theta))\frac{d}{dr} \right]_{|_{\gamma(s)}} = 0, \quad (\theta, r) \in S^1 \times [1, \infty)$$

To analyze (4), we first observe that the projection of equation (4) to the S^1_{θ} component becomes

$$\gamma'(s) + \left[(y_i(\theta) - y_j(\theta))r \frac{d}{d\theta} \right]_{|_{\gamma(s)}} = 0, \quad (\theta, r) \in S^1 \times [1, \infty),$$

which coincides with equation (3), up to the boosting r-coefficient: for a fixed $r \in \mathbb{R}$, its solutions behave as the gradient flowline equation for the Legendrian link $\Lambda = (s, y(s), z(s)) \subset J^1 S^1$ studied above. At an angle $\theta = \theta(s)$ with a Reeb chord, i.e. $y_i(\theta) = y_j(\theta)$, the corresponding solution of (4) becomes stationary in this S^1_{θ} -component, and thus gives us a Reeb chord ray, cf. Section 2.2. This determines the behavior in terms of the y_i -variables.

In particular, using Eq. (3) we can characterize the Reeb-positive cusp-free Legendrian links as follows.

Lemma 2.16. Let $\Lambda \subset (T^{\infty}S, \xi_{st})$ be a Legendrian braid link. Then Λ is Reeb-positive if and only if given any ij-trajectory, the difference function $z_i - z_j$ is positive near $s = -\infty$, and negative near $s = \infty$.

Proof. By definition, the Lagrangian projection of Λ has a positive crossing at an ij-Reeb chord if and only if the height difference function $z_i - z_j$ has Morse index 1. For (\Rightarrow) , suppose Λ is Reeb positive. If an ij-trajectory begins at c, since the neighborhood around c must be the unstable manifold of $z_i - z_j$, we must have $z_i > z_j$. Similarly, if an ij-trajectory terminates at a Reeb chord c, as Λ is Reeb positive, the stable manifold of c must be a point, and so we have $z_i < z_j$ at c. For (\Leftarrow), by contradiction: assume that there exists a negative crossing at c with $z_i > z_j$. In that case, the stable manifold of $z_j - z_i$ near c is necessarily the point c. Therefore, there exists an ji-trajectory terminating at c and with $z_j > z_i$. To understand the effect of the difference factor $(z^i - z^j)$ in equation (4), let us write $\gamma(s) = (\theta(s), r(s)) \in S^1 \times [1, \infty)$ and observe that the projection onto S^1_{θ} can be maximally extended to a solution of the ij-gradient flow-equation on S^1_{θ} , after orientation-preserving reparametrization. Denote by $P_{\gamma}(\theta)$ this projected ij-flowline on S^1_{θ} , traveling from a Reeb chord c' to a Reeb chord c, both Reeb chords of Λ . The sign of the derivative r'(s) is equal to the sign of the difference $-(z^i - z^j)(\gamma(s))$. In particular, since the difference function $(z^i - z^j)$ decreases along the projected trajectory $P_{\gamma(s)}, (z^i - z^j)$ decreases along the trajectory $\gamma(s)$ as well. Figure 5 provides a schematic picture on the behavior of the flowlines $\gamma(s)$.



FIGURE 5. The asymptotics of an *ij*-flowline $\gamma(s) = (r(s), \theta(s))$ based on the sign of $(z^i - z^j)$ at the Reeb chords. The case depicted in (1) occurs if $(z^i - z^j)$ goes from negative to negative. Case (2) occurs if $(z^i - z^j)$ goes from positive to negative. Case (3) if $(z^i - z^j)$ goes from positive to positive. Case (4) never occurs, which would correspond to $(z^i - z^j)$ going from negative to positive. The fact that the fourth case is excluded is the key ingredient behind the proof of Lemma 2.26 (the Trapping Lemma).

This leads to the following useful observation:

Lemma 2.17. Let $\gamma(s) = (r(s), \theta(s)), s \in [0, \infty)$ be an ij-flowline on $S^1 \times [1, \infty)$ solving equation (4) and suppose that r'(0) > 0. Then r(s) strictly increases and it asymptotes to a Reeb chord ray.

Proof. Since $\gamma(s)$ solves (4), the component $\theta(s)$ solves the r(s)-boosted gradient flowline equation for the Legendrian link Λ . Therefore, the difference $(z_i - z_j)$ strictly decreases along $\gamma(s)$. By equation (4), we have r'(0) > 0 if and only if $(z_i - z_j)(0) < 0$. Therefore, the hypothesis r'(0) > 0implies that r'(s) > 0 for all $s \in \mathbb{R}_+$, since $(z_i - z_j)(s)$ is negative at s = 0 and strictly decreases as a function of $s \in \mathbb{R}_+$.

Visually, Lemma 2.17 implies that, near a small enough neighborhood of a marked point, once a flowline start traveling up towards the marked point then it will not reverse course. The following trapping lemma builds on Lemma 2.17 to provide the more complete description that we need:

Lemma 2.18 (Trapping Lemma). Let $L \subset T^*D^2$ be the germ of a Lagrangian conical end over a generic cusp-free Legendrian link $\Lambda \subset (J^1S^1, \xi_{st})$, where the zero-section of the 1-jet space is identified with $T_0^{\infty}S^1$. Then there exist positive constants $T, R \in \mathbb{R}_+$ such that:

- (1) For the annular neighborhood $S^1 \times (R, 4R) \subset D^2$ of the origin, any trajectory $\gamma(s)$ passing through $S^1 \times [2R, 3R]$ at s = 0 must stay inside $S^1 \times [R, 4R]$ after time $\pm T$.
- (2) The domain of definition of either $\gamma(s)$ or $\gamma(-s)$ can be extended to $[0,\infty)$.
- (3) In the former case, r(s) strictly increases on $[T, \infty)$ and asymptotes to a Reeb chord ray. In the latter case, r(s) strictly decreases on $[-\infty, T)$.

That is, any trajectory γ passing through $S^1 \times [2R, 3R]$ must stay inside $S^1 \times [R, \infty)$ in at least one direction.

Proof. Consider the constants $\epsilon(c, c') \in \mathbb{R}_+$ for all the pairs $c, c' \in S^1$ associated to Reeb chords of the end Legendrian link Λ , as in Lemma 2.15. Let $\epsilon \in \mathbb{R}_+$ be their minimum and we choose T to be the corresponding minimal escape time. Choose some radii $R_0, R_1 \in \mathbb{R}_+, R_1 > R_0 \gg 1$ such that for any gradient trajectory passing through $S^1 \times [R_1, \infty)$, the time $\pm T$ image of the trajectory remains inside $S^1 \times [R_0, \infty)$. Such radii exist as there is a finite uniform upper bound on $|z_i - z_j|$ and trajectories satisfy equation (4). Now given an *ij*-trajectory γ passing through $S^1 \times [R_1, \infty)$ at time s = 0, we consider $P_{\gamma}(\theta)$ the trajectory on S^1_{θ} obtained by reparametrizing the S^1_{θ} -component of γ and consider $c \in S^1$ be the (angle associated to the) limiting Reeb chord at $s = -\infty$, while c' is the limiting Reeb chord at $s = +\infty$. This establishes (1), as the required R can be chosen from R_1 , and also shows (2).

For (3), since $z_i - z_j$ has to decrease along γ , it follows that both $(z_i - z_j)(c) < 0$ and $(z_i - z_j)(c') > 0$ cannot happen. By the choice of R_0 , the gradient flow γ still remains in $S^1 \times [R_0, \infty)$ after time $\pm T$. If $(z^i - z^j)(\theta(T)) < 0$, Lemma 2.17 implies that r(s) strictly increases, i.e. that γ never goes back. If else $(z^i - z^j)(\theta(T)) > 0$, it must be that $(z^i - z^j) > 0$ along $P_{\gamma} : (-\infty, T] \to S^1_{\theta}$ and then the backward ji-gradient flow will have $r'(\gamma(-s)) > 0$. To analyze the cases in the other intervals, we proceed similarly: e.g. if $P_{\gamma}(0) \in (a + \epsilon, b - \epsilon)$ then Lemma 2.15 implies that $(z^i - z^j)(\theta(T)) \neq 0$ and the same argument above applies. If $P_{\gamma}(0) \in (a, a + \epsilon]$, then there are two cases. Either $(z_i - z_j)(c') < 0$ and then $(z_i - z_j)(c')$ must stay negative, so $\gamma(s)$ will travel up as $s \to +\infty$. Or $(z_i - z_j)(c') > 0$, in which case the backward flow satisfies $(z_j - z_i)(c) < 0$ along it and so the backward flow will always travel up. The cases with $P_{\gamma}(0) \in (b - \epsilon, b]$ are deduced in the same way and (3) follows.

The characterization of asymptotics of trajectories for exact Betti Lagrangians is thus:

Proposition 2.19. Let $L \subset (T^*S, \lambda_{st})$ be an exact Betti Lagrangian and $\gamma : (a, b) \to S$ a gradient trajectory for L, $(a, b) \subset \mathbb{R}$ the maximal domain of definition of γ . Then:

(1) The trajectory γ cannot be periodic.

- (2) If a is finite, then $\gamma(a) \in S$ must be at a D_4^- -branch point of the projection $L \longrightarrow S$. If b is finite, the same applies to $\gamma(b) \in S$.
- (3) If $a = -\infty$, then $\gamma(s)$ asymptotes to a Reeb chord ray as $s \to -\infty$. Similarly, if $b = \infty$, then $\gamma(s)$ asymptotes to a Reeb chord ray as $s \to \infty$.

Proof. For (1), Lemma 2.14 implies that a closed trajectory would yield a smooth closed curve $\tau \subset L$ such that integral of the Liouville form along τ does not vanish. Since L is exact, this is a contradiction and thus there are no closed trajectories. For (2), showing that $\gamma(a)$ is a D_4^- -branch point if a is finite, we fix a large radius $R \in \mathbb{R}_+$ as in Lemma 2.18 and consider the complement $K \subset S$ of the union of all the cylinders $S^1 \times (R, \infty)$ near the marked points of S. We claim that a trajectory must either terminate at a finite end or leave the compact set K.

Indeed, suppose that image of the trajectory γ stays contained in K as $s \to a$. If a is finite, the flowline equation (2) and since L is embedded, it follows that $\lim \gamma(a)$ must be branch point of D_4^- -singularity. The flowtime needed to approach the D_4^- -singularity is finite, and thus we have a finite end, as required. If else $a = -\infty$, we need to argue that the trajectory escapes K, and Lemma 2.18 then determines its asymptotics. By the flowline energy formula of Lemma 2.14, it must be that $\gamma(s)$ lies in a small neighborhood of a D_4^- -branched point for $s \in \mathbb{R}_+$ large enough. Indeed, by energy quantization: the difference of the sheets have norm uniformly bounded below outside such neighborhood, and there is a minimal flow-time needed to travel from one boundary to another, but the flow-energy must be finite and so eventually the flowline that either ends at the branch point or coincides with one of the trivalent rays coming out of the D_4^- -singularity. The flowtimes for either of these events are finite, and thus this is a contradiction with $a = -\infty$ and staying within K. Thus the trajectory must leave the compact set K if $a = -\infty$, from which (3) follows after applying Lemma 2.18.

Finally, let us compare the behavior of trajectories for the polar metric $g = r^2 d\theta^2 + dr^2$ and the conical metric $g = d\theta^2 + dr^2$ near the marked points $\mathbf{m} \subset S$. By the flowline equation (4), trajectories are directed by the vector field of the form $\nabla f \partial_{\theta} + (f/r) \partial_r$ where $f = z^i - z^j$. We then have:

Lemma 2.20. The diffeomorphism $\phi : S^1 \times [0, \infty) \to S^1 \times [1, \infty)$ given by $(\theta, \tau) = \phi(\theta, r) = (\theta, e^{r^2/2})$ identifies the gradient flowlines for the metric $dr^2 + d\theta^2$ and the gradient flowlines for the metric $\tau^2 d\theta^2 + d\tau^2$.

Proof. The flowlines for the metric $\tau^2 d\theta^2 + d\tau$ are directed by $(\nabla f, \tau f) = \nabla f \partial_{\theta} + \tau f \partial_{\tau}$. Since $\phi_*(\partial_{\theta}) = \partial_{\theta}$ and $\phi_*(\partial_r) = r e^{r^2/2} \partial_{\tau}$, it follows that $\phi_*(\nabla f, f/r) = (\nabla f, e^{r^2/2} f) = (\nabla f, \tau f)$, as claimed.

For context, the conical metric $g = d\theta^2 + dr^2$ is better suited when cylindrizing the Betti Lagrangian Σ to a cylindrical Lagrangian filling, cf. Definition 2.9. This is, arguably, more natural from a geometric viewpoint and it fits better with the study of augmentations of the Legendrian contact

dg-algebra, but the analysis is a bit more elaborate. If $L \subset T^*S$ is an exact Betti Lagrangian, with conical ends, then the polar metric is better suited as it makes L uniformly bounded nears its asymptotic ends. Lemma 2.20 implies that we can switch between these two metrics when it comes to describing the qualitative behavior of asymptotics of flowlines.

2.6. Holomorphic Morse flowlines for meromorphic spectral curves. The analysis in Section 2.5 applies to conical ends, and thus to the exact Betti Lagrangians in Section 2.3.1. For the meromorphic spectral curves $\Sigma \subset (\mathbb{T}^*\mathbb{S}, \omega_{\mathbb{C}})$ in Section 2.3.2, with O(-1)-ends, a modification of the arguments in Section 2.5 is needed to study the Betti Lagrangians $e^{i\theta}\Sigma \subset (T^*S, \lambda_{st})$ given by their θ -parts. Consider the following holomorphic version of the flowline equation:

Definition 2.21. Let (\mathbb{S}, g) be Riemann surface endowed with a Kähler metric, $\theta \in S^1$, and write $\lambda_1, \ldots, \lambda_n$ for the holomorphic 1-forms on \mathbb{S} whose graphs give the multigraph $\Sigma \subset (T^*\mathbb{S}, \omega_{\mathbb{C}})$. By definition, a smooth map $\gamma : (-\epsilon, \epsilon) \longrightarrow \mathbb{S}$ is said to be a holomorphic *ij*-gradient flowline for Σ if it satisfies

(5)
$$\gamma'(s) + \overline{e^{-i\theta}g^{-1}(\lambda_i - \lambda_j)} = 0, \quad i, j \in [1, n],$$

which is said to be the *ij*-holomorphic gradient flow equation at phase θ .

The holomorphic flowline equation Eq. (5) is the real gradient flowline equation for the real parts of holomorphic functions $e^{i\theta}q_1, \ldots, e^{i\theta}q_n$, for some choices of local holomorphic primitives q_i of the holomorphic 1-forms λ_i , $i \in [1, n]$. In relation to the flow-energy, as in Lemma 2.14, note that $e^{i\theta}$ times the flow-energy equals the integral of the holomorphic Liouville 1-form $\lambda_{\mathbb{C}} \in \Omega^{1,0}(\mathbb{S})$ over the relative cycle $[\bar{\gamma}^j - \bar{\gamma}^i]$ in $\Sigma \subset T^*\mathbb{S}$.

In contrast to trajectories solving the gradient flowline equation in Definition 2.13, solutions of the *ij*-holomorphic flowline equations only depend on the conformal class of the metric g, not g itself. Indeed, given $\gamma : (-\epsilon, \epsilon) \longrightarrow S$, consider the following alternative differential equation for γ :

(6)
$$\operatorname{Im}(e^{-i\theta}(\lambda_i - \lambda_j)(\gamma'(s))) = 0, \qquad \operatorname{Re}(e^{-i\theta}(\lambda_i - \lambda_j)(\gamma'(s))) \le 0$$

Eq. (6) is said to be the ij-phase θ WKB equation at phase θ . Solutions to its first part, the equality, give rise to a line field on S whose integral curves are known as WKB lines in the literature. The second condition, the inequality, specifies an orientation for its parametrization, so the line field acquires a direction.

Remark 2.22. By replacing Σ with $e^{i\theta}\Sigma$, the WKB lines at phase θ become WKB phase at phase 0; phase $\theta = 0$ lines are known as horizontal WKB lines. Whenever we replace Θ with $e^{i\theta}\Theta$, the scalar factor $e^{i\theta}$ will rotate the anti-Stokes rays, for small enough θ . We implicitly assume onwards that whenever there is a rotation of the phase θ , we also rotate the anti-Stokes rays of the irregular class.

Since the solutions of (5) solve (6), it follows that the image of the trajectories for the holomorphic gradient flow equation do not depend on the choice of the Kähler metric but only on the conformal

structure, i.e. conformally changing the Kähler metric only changes the reparametrization of the solutions, not the integral curves themselves.

Example 2.23. (1) Solutions of Eq. (6) admit rather explicit forms in the local conformal coordinates $q_{ij} = q_i - q_j$ in a sectorial neighborhood of γ : in the q_{ij} -coordinate, $\lambda_i - \lambda_j = dq_{ij}$, and the equation transforms to $\text{Im}(e^{-i\theta}\gamma'(s)) = 0$, which is the straight line equation in the q_{ij} -plane.

(2) $(D_4^-$ -singularity) Consider a neighborhood of a ramification point $\Sigma \longrightarrow S$, which we model as $\Sigma = \{(z, w) \in T^*\mathbb{C} : w^2 - z = 0\}$ and locally there are only two indices $i, j \in [1, 2]$, with $\lambda_1 = \sqrt{z}dz$ and $\lambda_2 = -\sqrt{z}dz$. Near the branch point z = 0 in \mathbb{C} at phase $\theta = 0$, the line field solving Eq. (6) must be in ker Im $(\sqrt{z}dz)$. This line field coincides with that given by the real gradient $\nabla(\operatorname{Re} z^{3/2})$, and thus the WKB line equation of phase 0 defines the same singular foliation as in the case of real gradient flowlines, which we studied in Section 2.4. The leaves of the foliation are spanned by the maximal geodesics of $|z| dz^2$, and the critical locus of the foliation is given as before.

Varying the phase θ does not qualitatively affect this picture: the WKB singular line field is given by ker $I(e^{i\theta}\sqrt{z}dz)$, which coincides with the line field directed by $\nabla(e^{-i\theta}\text{Re}z^{3/2})$. Therefore, changing the phase θ rotates the initial trivalent rays in the anti-clockwise direction until it goes back to itself at $\theta = \pi$. By convention, we order the indices along the rays so that the flowlines are directed *outward*. Away from the branch point, as in (1), the analysis above can just be done by considering the locally defined coordinate $W = \int \sqrt{z}dz$, so that the 1-forms $\lambda_i = \pm \sqrt{z}dz$ reduce to $\pm dW$ and the WKB lines are geodesics in the W-plane.

2.7. Asymptotics for WKB trajectories. Let us study the asymptotics of the WKB trajectories, whose behavior is need to understand spectral networks for Betti Lagrangians associated to meromorphic spectral curves. In the case of conical ends (Section 2.5), we had the Legendrian link at infinity parametrized by $(x(\theta), y(\theta), z(\theta))$. In the case of the θ -part of a meromorphic spectral curve $\Sigma \subset T^*S$, the Legendrian link near the O(-1)-ends is given by the Stokes diagram. We now need to compare the holomorphic flowline equation near the poles to the Morse flowline equation associated to the cone over the Legendrian given by the Stokes diagram.

Consider a small neighborhood of a marked point $\mathbf{m} \subset \mathbb{S}$ parametrized by $z \in U \subset \mathbb{C}$, U an open disk near the origin, with the marked point given by z = 0. Let the associated irregular data Θ (cf. Example 2.5) at that marked point be given by

$$Q = \sum_{k_j \in \mathbb{Q}_{>0}} \frac{c_{k_j}}{k_j} z^{-k_j} \in \mathbb{C}((z^{1/\infty}))/z^{-1}\mathbb{C}[[z^{1/\infty}]], \quad c_{k_j} \in \mathfrak{t}_{reg} \subset \mathfrak{gl}_n.$$

In general, given a Puiseux series Q with finite polar part and convergent holomorphic part, its *meromorphic multigraph* $\Gamma_{dQ} \subset T^*\mathbb{C}$ is the union of the Lagrangian graphs of dq_1, \ldots, dq_n , where q_1, \ldots, q_n are branches of Q. Near a pole z = 0 for Q as above, there are sectors $\theta \in (\theta_0, \theta_1)$ for

the phase of $z = re^{i\theta}$, such that Γ_Q equals:

(7)
$$\Gamma_{dQ} = \left(\theta, -\sum_{k_j \in \mathbb{Q}} r^{-k_j} \operatorname{Im}(c_{k_j} e^{i\theta k_j}), r, -\sum_{k_j \in \mathbb{Q}} r^{-k_j - 1} \operatorname{Re}(c_{k_j} e^{i\theta k_j})\right) \subset T^* \mathbb{C}, \quad (r, \theta) \in \mathbb{R} \times (\theta_0, \theta_1).$$

Note that (7) is a generalization of the conical ends condition (1), arising naturally in the setting of asymptotics given by irregular classes.⁶

Example 2.24 (D_{-}^{4} -multigraph). This example continues Example 2.23 and aligns with Example 2.12 for the irregular class $\langle x^{3/2} \rangle$, which is the Stokes data for the D_{-}^{4} -singularity. For the Puiseux series Q associated to $\Sigma = \{w^{2} - z = 0\} \subset T^{*}\mathbb{C}$, the meromorphic multi-graph is parametrized as

(8)

$$(r,\theta) \longrightarrow \left(\theta, \pm \frac{3}{2}f(r)\cos(\frac{3\theta}{2}), r, \pm f'(r)\sin(\frac{3\theta}{2})\right) \in T^* \mathbb{R}^2_{r,\theta}, \quad \text{where } f(r) = \frac{2}{3}r^{3/2}, \quad (r,\theta) \in \mathbb{R} \times S^1$$

In fact, the Hamiltonian isotopy class of this Lagrangian is independent of the specific choice of function f(r), as long as it is positive and strictly increasing. Deforming f(r) to be linear at infinity recovers the conical end form.

Let us analyze the gradient equations near the O(-1) ends, i.e. a pole at z = 0. Consider the inversion $|z|^{-1}$ from $\mathbb{C}^* \simeq S^1 \times (0, \infty)$ to send $+\infty$ to 0 and pushforward the conical metric $|dz|^2$ to $|z|^{-2}|dz|^2$; the latter metric is $d\theta^2 + r^{-2}dr^2$ in polar coordinates. For the difference of the sheets to have growth at least z^{-1} , we need the following condition:

(9)
$$(dq_i - dq_j) - \frac{C}{z} = \frac{c_{ij}}{z^{k_{ij}}} + o(z^{-k_{ij}}) + (\text{holomorphic terms}),$$

for some constants C, c_{ij} where $C \neq 0$ only when $c_{ij} = 0$. In the case C = 0, we require that $c_{ij} \neq 0$ and $k_{ij} > 0$. By using the parametrization Eq. (7), the *ij*-gradient flow equations for the O(-1)-ends given by the irregular data Q read as follows:

(10)
$$\frac{dr}{ds} + r \operatorname{Re}\left(\frac{C}{z}\right) - \operatorname{Re}(c_{ij}e^{-ik_{ij}\theta})r^{-k_{ij}+1} + o(r^{-k_{ij}+1}) + o(r) = 0$$

(11)
$$\frac{d\theta}{ds} - \operatorname{Im}(C/z) - \operatorname{Im}(c_{ij}e^{-ik_{ij}\theta})r^{-k_{ij}} + o(r^{-k_{ij}}) = 0.$$

For instance, in the case $dQ(z) = z^{-1}dz$, we obtain straight rays going into the origin for Im(c) = 0, periodic circles for Re(c) = 0, and logarithmic spirals otherwise. In general, for $c_{ij} \neq 0$, the dominating contribution as $r \to 0$ comes from $c_{ij}e^{-ik_{ij}\theta}$ only and θ varies slowly while r increases fast as r is sufficiently small and the angle θ is close to the anti-Stokes rays. These are the qualitative features that we have established for the conical ends. Thus, for O(-1)-ends, we can

⁶Note that in order to represent irregular data with multiplicity, we would need to introduce additional terms of the form $d \log(z)$, for $t \in \mathfrak{t}_{req}$, and some holomorphic terms.

effectively repeat the previous arguments in Lemma 2.18 on minimal escape time and conclude that the ij-flowlines satisfy the trapping property (cf. Lemma 2.26), and trajectories also asymptote to the corresponding anti-Stokes rays, which coincide with the angles of Reeb chords.

Furthermore, the Stokes Legendrians without multiplicities further enjoy the following Reebpositivity property.

Proposition 2.25 (Stokes Legendrians are Reeb-positive). Let Θ be an irregular class without multiplicity, and Λ_{Θ} be the corresponding Stokes Legendrian. Then for r small enough, Λ_{Θ} is Reeb-positive.

Proof. Let Θ be an irregular class and let q_i denote the branches of Θ . The difference function is given by the real part of $q_i - q_j = \frac{1}{k_{ij}} c_{ij} z^{-k_{ij}} + o(z^{-k_{ij}})$. For r small enough, the dominating term comes from $\frac{c_{ij}}{k_{ij}} z^{-k_{ij}}$ and so the local gradient flow equation Eq. (4) reads

(12)
$$\frac{d\theta}{ds} - r^{-k_{ij}} \operatorname{Im}(c_{ij}e^{-i\theta k_{ij}}) + o(r^{-k_{ij}+1}) = 0.$$

Near $s = -\infty$, $\partial_{\theta}^2 \operatorname{Re}(q_i - q_j)$ over the limiting anti-Stokes direction has to be *negative*, because the limiting anti-Stokes direction has to be the *unstable* manifold of the difference function. However, the second derivative is computed to be equal to $k_{ij}r^{-k_{ij}}\operatorname{Re}(c_{ij}e^{-i\theta}k_{ij})$. So we see that the second derivative is negative if and only if the difference function is *positive*. Near $s = +\infty$, we can argue the same by replacing $\theta(s)$ with $\theta(-s)$. So by Lemma 2.16, Λ_{Θ} is Reeb positive. \Box

In this meromorphic case, there is also the following alternative approach to prove the necessary trapping lemma, using the cameral cover construction from [IM21; Kuw24]. Given $\Sigma \to \tilde{C}$ be a meromorphic spectral curve, its *cameral cover* is defined to be the Riemann surface

(13)
$$\Sigma^{cam} := \{ (\lambda_{\sigma(1)}(z)dz, ..., \lambda_{\sigma(n)}(z)dz) : z \in C, \sigma \in S_n \}.$$

For $1 \leq i < j \leq n$, its *intermediate cover* Σ^{ij} is the quotient of Σ^{cam} by the relation

(14)
$$(\lambda_{\sigma(1)}, \dots, \lambda_{\sigma(i)}, \dots, \lambda_{\sigma(j)}, \dots, \lambda_{\sigma(n)}) \sim (\lambda_{\sigma(1)}, \dots, \lambda_{\sigma(j)}, \dots, \lambda_{\sigma(i)}, \dots, \lambda_{\sigma(n)})$$

over all permutations σ satisfying $\sigma(i) < \sigma(j)$. Note that the quotient map $\Sigma^{cam} \to \Sigma^{ij}$ can be realized as the spectral cover of the quadratic differential defined by $(\lambda_{\sigma(i)} - \lambda_{\sigma(j)})^{\otimes 2}$, and that *ij*-trajectories of phase θ lift to trajectories of phase θ for the rank 2 spectral curve $\Sigma^{cam} \to \Sigma^{ij}$. Employing a cameral cover, we readily obtain the following:

Lemma 2.26. Let $\Sigma \subset T^*S$ be a meromorphic spectral curve with O(-1) ends. Then, for generic θ , there exists a neighborhood of the marked points $\mathbf{m} \subset S$ such that any WKB θ -trajectory entering this neighborhood cannot leave, and such trajectory asymptotes to an anti-Stokes ray.

Proof. From the O(-1) ends condition, it follows that the cover $\Sigma^{cam} \to \Sigma^{ij}$ has poles of order greater equal than 2. For quadratic differentials, the claimed statement follows from the literature, cf. [HM79; Str84]. Alternatively, the arguments in the proof of Lemma 2.18 work for poles of order $\ell \geq 3$ if one uses the Legendrian link defined by the function $\pm \cos((\ell - 2)\pi\theta/2)$. The case of poles

of order 2 follows from the standard classification of trajectories near it, as discussed after Eq. (10) or see [Str84, Chapter 7].

In line with Proposition 2.19, the asymptotics of WKB trajectories are summarized by:

Proposition 2.27. Let $L \subset (T^*S, \lambda_{st})$ be the θ -part of a meromorphic spectral curve $e^{i\theta}\Sigma \subset T^*\mathbb{S}$ with O(-1) ends at the marked point $\mathbf{m} \subset S$, and $\gamma : (a, b) \longrightarrow S$ a WKB trajectory for $e^{i\theta}\Sigma$. Then, for generic θ , we have that:

- (1) The trajectory γ cannot be periodic.
- (2) If a is finite, then $\gamma(a) \in S$ must be at a D_4^- -branch point of the projection $L \longrightarrow S$. If b is finite, the same applies to $\gamma(b) \in S$.
- (3) If $a = -\infty$, then $\gamma(s)$ asymptotes to an anti-Stokes ray of $e^{i\theta}\Sigma$ as $s \to -\infty$. (Similarly for $b = \infty$.)

Proof. Observe that any trajectory will lift to trajectory of some trajectory $\Sigma^{cam} \to \Sigma^{ij}$. For (2) and (3), the argument is the same as for Proposition 2.19, applied to the cameral cover lift, given that there are no recurrent trajectories ([Str84, Chapter 10.2]). To show that there are no periodic trajectories, note that such a periodic trajectories would lift to periodic trajectories of some $\Sigma^{cam} \to \Sigma^{ij}$. That said, for generic θ , complete quadratic differentials that are saddle-free do not admit periodic or recurrent trajectories. This completes the statement.

3. Morse spectral networks

This section introduces and develops first results on Morse spectral networks. The ingredients needed on flowtrees are discussed in Section 3.1 and the definition of Morse spectral networks is presented in Section 3.2. The proof of Theorem 1 is then established in Section 3.3, for the exact case, and in Section 3.4 for the meromorphic case. The section concludes with a discussion on BPS indices, in Section 3.5, and the local study of rigid flowtrees near a D_4^- -singularity, in Section 3.6.

3.1. Morse flowtrees and soliton classes. Flow trees arise in the study of gradient trajectories of tuples of Morse functions, see e.g. [Fuk97; FO97], [Rua06, Section 6] and [Abo11b, Section 2]. In particular, [Rua06] shows that certain counts of rigid holomorphic discs agree with counts of gradient flow trees in the context of Lagrangian Floer theory. In the framework of Legendrian submanifolds, [Ekh07] established a correspondence between the counts of rigid flow trees and boundary-punctured rigid pseudo-holomorphic disks governed by a Legendrian submanifold whose front has cusp-edge singularities. In the context of Betti Lagrangians, we use Section 2 to define and study Morse spectral networks in terms of flow trees. We also establish their existence, proving Theorem 1. Note that the fronts for the Legendrian lifts of Betti Lagrangians have D_{-}^4 -singularities, which are non-generic, and the techniques of [Ekh07] need to be modified appropriately. From the viewpoint of studying the BPS states in [GMN13b; GMN13a; GMN14], it will also be important to study relative homology classes associated to flow trees.

A tree will be a properly embedded planar graph $\Gamma \subset \mathbb{R}^2$ with possibly semi-infinite edges such that any two vertices are connected by exactly one finite path. Vertices of valence one are allowed and referred to as finite leaves. We consider trees endowed with a cyclic ordering of the edges at each vertex, as surfaces will be naturally oriented, and an orientation for each edge. By definition, a rooted tree will be an oriented tree with a choice of either a univalent vertex or a semi-infinite edge (but not both) such that the all edges of the tree point outward from the root. Following [Ekh07, Definition 2.10], we consider the following:

Definition 3.1. Let (M, g) be a Riemannian manifold, $L \subset T^*M$ a Lagrangian multigraph over rank *n* over (M, g), and $\Gamma \subset \mathbb{R}^2$ a connected tree. By definition, for a Morse flowtree for *L* is a continuous map $F : \Gamma \longrightarrow M$ such that:

(1) Flow-edge condition. The restriction $F|_e$ to each oriented edge $e \in E(\Gamma)$ is an injectively parametrized $\ell_1\ell_2$ -flowline for L, for some $\ell_1, \ell_2 \in [1, n]$. If the domain of definition of such trajectory is infinite, then we require e to be a semi-infinite edge.

We denote by $\overline{\gamma}_e^{\ell_i}$ its ℓ_i -cotangent lift, i = 1, 2, whose image lies in L, and write $\overline{\gamma}_e^{\ell_i}(v)$ for the image of a vertex $v \in \partial e$ as a limit point for $\overline{\gamma}_e^i(e)$.

(2) Balancing condition. Consider a vertex $v \in \Gamma$ with cyclically ordered edges (e_1, \ldots, e_s) and denote $\overline{\gamma}_{\ell} = \overline{\gamma}_{e_{\ell}}$. Then we require

$$\overline{\gamma}_{\ell}^{\ell_2}(v) = \overline{\gamma}_{\ell+1}^{\ell_1}(v),$$

cyclically in $\ell \in [1, s]$, and $\overline{\gamma}_{\ell}^2$ is oriented towards the point $\overline{\gamma}_{\ell}^{\ell_2}(v)$ whereas $\overline{\gamma}_{\ell+1}^{\ell_1}(v)$ is directed away from it.

- (3) Relative cycle condition. The relative cycles $\overline{\gamma}_e^{\ell_1} \overline{\gamma}_e^{\ell_2}$, where $e \in E(\Gamma)$ ranges over all edges, piece together to give an oriented relative cycle with endpoints above the univalent vertices.
- (4) Minimality. There are no 2-valent vertices $v \in \Gamma$ with edges e_1, e_2 such that $F(v) \notin \pi(K_L)$ and both are *ij*-flowlines with same set $\{i, j\}$. Similarly, there are no 2-valent vertices $v \in \Gamma$ such that $F(v) \in \pi(K_L)$ and both the *i*th and *j*th sheets extend smoothly over v.⁷
- (5) Special vertices. There is a chosen collection $S(\Gamma)$ of univalent vertices.

Definition 3.1.(5) is in line with partial flowtrees: the univalent vertices above are called special punctured in [Ekh07, Section 2.2.C]. Special punctures are used in the proofs of our arguments, but the type of trees relevant for the study of spectral networks, namely D_4^- -trees as in Definition 3.2, do not have any special punctures. The integral of λ over the oriented relative homology class

⁷2-valent vertices $v \in \Gamma$ such that $F(v) \in \pi(K_L)$ are allowed, but we force them to be switch-vertices in the terminology of [Ekh07].
in Definition 3.1.(3) is called the *flow-energy* of F. In the case of exact Betti Lagrangians, it is possible to discuss *positive* and *negative* punctures, as in [Ekh07]. That is, a vertex $v \in \Gamma$ is a positive puncture, resp. negative, if the height difference between the two Legendrian lifts of

$$\overline{\gamma}_{\ell+1}^{(\ell+1)_1}(v) = \overline{\gamma}_{\ell}^{(\ell)_2}(v)$$

is positive, resp. negative. By Stokes' theorem, the flow-energy of a tree is given by adding such difference of heights for all punctures of the tree. Lemma 2.14 implies that such quantity is positive, thus F contains at least one positive puncture, and the F travels from the positive puncture to the negative punctures.

3.1.1. D_4^- flowtrees. The type of Morse flowtrees that appear in the study of spectral networks, as related to Betti Lagrangians, are rather particular Morse flow trees. We now introduce this class of flowtrees in Definition 3.2 below.

The branch points of the projection $L \longrightarrow S$ from the Betti Lagrangian to the base will also sometimes referred to as D_4^- -singularities in S. Technically, the D_4^- -singularity is a front singularity for the projection of the Legendrian lift of a neighborhood of the ramification point in L: so this is a slight abuse of notation, but it is sometimes convenient. In the base surface (S, g), we assume that in a (small enough) neighborhood of the branch points in S, (S, g) is endowed with a local holomorphic coordinate, the metric g is locally Kähler, and the multigraph L above it is locally holomorphic. In the case of exact Betti Lagrangians, we also assume that the metric is of the polar form $r^2d\theta^2 + dr^2$ in a neighborhood of the marked points m.

Definition 3.2 $(D_4^-\text{-trees})$. Let $(\mathbf{S}, \boldsymbol{m}; \boldsymbol{\Lambda})$ be a Betti surface, L a Betti Lagrangian over (\mathbf{S}, g) for some Riemannian metric g and $z \in \mathbf{S}$ a point. By definition, a D_4^- flowtree for L is a rooted Morse flowtree $F : \Gamma \longrightarrow \mathbf{S}$ for L such that:

- (1) If the root of Γ is a vertex, then it must map to either a branch point, corresponding to a D_4^- -singularity, or to the fiber $T_z^* \mathbf{S}$ above z. In the former case, the unique edge adjacent to the root vertex is mapped out of one of the three initial ray flow-lines associated to the D_4^- -singularity in the outward direction.
- (2) If the root edge is a semi-infinite edge, then it must be asymptotic to a Reeb chord of a Legendrian link in Λ (See Proposition 2.19 and Proposition 2.27)
- (3) If an internal vertex v of Γ maps to a branch point, then there are no adjacent edges whose image near v is a cusp-cusp flowlines for that D_4^- singularity.
- (4) Non-root univalent vertices of Γ must map to branch points (at the D_4^- -singularities), and each corresponding adjacent edge maps to one of the three initial ray flow-lines associated to the D_4^- -singularity in the inward direction.

Given a D_4^- -flow tree, the associated relative cycle in L is said to be its *soliton class*. A D_4^- -flow tree is *rigid* if Γ if all non-univalent vertices are trivalent and none of them map to a D_4^- -singularity.

We often refer to D_4^- -flowtrees as D_4^- -trees, implicitly understanding that there are gradient flowline conditions as in Definition 3.2 and it is not just a combinatorial graph object.

Remark 3.3. By construction, a D_4^- -tree in fact defines a regular homotopy class, representing its soliton class. It is possible to develop the theory for relative homotopy classes on L, and not just relative homology classes.

In the physics literature [GMN13b; GMN13a; GMN14], soliton classes of D_4^- -trees have the following nomenclature:

- (i) For a D_4^- -tree with a root vertex mapping to a branch point in S, its soliton class is said to be a 4d BPS state.
- (ii) For a D_4^- -tree with a root vertex mapping to a fiber $T_z^* \mathbf{S}$, its soliton class is said to be a vanilla 2d-4d BPS state. The point $z \in \mathbf{S}$ above which the D_4^- starts is said to be the ultra-violet parameter for a certain 1/2-BPS surface defect. This surface defect is assumed to be massive in the infrared and have finitely many vacua: in our setting, this is equivalent to the puncture being positive and there existing finitely many intersections $L \cap T_z^* \mathbf{S}$, so-called vacua, above $z \in \mathbf{S}$.
- (iii) For a D_4^- -tree with a root edge, which must be semi-infinite and asymptote to a Reeb chord, we are not aware of any specific terminology in the physics context. They are certainly fundamental objects in the study of spectral networks, and produce Hamiltonian invariants for exact Betti Lagrangians: we refer to them as *augmented* D_4^- -trees. This notation is chosen due to its relation to the augmentation of the Legendrian contact dg-algebra, cf. Section 7.
- (iv) The framed 2d-4d BPS states in [GMN13b, Section 3.3] do not correspond to rigid D_4^- trees. These are associated to two points $z_1, z_2 \in \mathbf{S}$ and a path ρ between them, the latter defining a supersymmetric interface between the surface defects associated to $z_1, z_2 \in \mathbf{S}$. In our context, this is captured by the continuation strips from $L \cap T_z^* \mathbf{S}$ to $L \cap T_z^* \mathbf{S}$, where the path ρ is interpreted as a morphism between the fibers, cf. Section 5. Note that in [GMN13b, Section 3.3] it is written "It is generally believed that the defect [...] does not depend on the precise path ρ but only on its homotopy class. In the present paper we will take this as an assumption.". In our formalization, the properties of partially wrapped Fukaya categories readily imply that the (cobordism classes of) moduli of continuation strips are indeed only depending on the homotopy class of ρ , not the representative ρ itself, for any compactly supported deformation keeping the Floer datum and L fixed at ends.

As will be proven in Sections 4 and 5, a D_4^- -tree will correspond to the adiabatic limit of a punctured pseudoholomorphic disk bounded by the Betti Lagrangian $L \subset (T^*S, \lambda_{st})$. In the case that the root vertex maps to a branch point, there are no punctures and the absolute cycle in L defined by the boundary of such pseudoholomorphic disk coincides the soliton class of the tree. This case can occur if $L \subset (T^*S, \lambda_{st})$ is not exact and *does* occur for certain meromorphic Betti Lagrangians. The case that the root of a D_4^- -tree is a semi-infinite edge does occur for exact Betti Lagrangians: the D_4^- -tree is the adiabatic limit of a pseudoholomorphic disk with *one* positive puncture at the Reeb chord associated to the semi-infinite edge. The boundary of such pseudoholomorphic disk, understood as a relative cycle in $(L, \partial L)$, coincides with the soliton class of such D_4^- -tree. See Sections 4 and 5 for more details.

Remark 3.4. In the physics context of theories of class S, general 4d BPS states also include graphs which are not necessarily D_4^- -trees, cf. [GMN13b, Section 3.1]. There is a natural generalization of Definition 3.2 to the notion of D_4^- -graph, but it will not be used in this manuscript. In the framework of Floer theory, such general D_4^- -graph would correspond to a configuration of several pseudoholomorphic disks (with no punctured) bounded by the Betti Lagrangian, whose boundaries intersect each other.

3.2. Definitions on spectral networks. Let us introduce the definition of a spectral network associated to a Betti Lagrangian $L \subset (T^*S, \lambda_{st})$, where a Riemannian metric (S, g) has been fixed. First we introduce *pre*-spectral networks in Definition 3.5, discuss coherent extensions and the associated soliton classes. Then we define spectral networks in Definition 3.12.

Definition 3.5 (Pre-spectral networks). Let (S, g) be a Betti surface endowed with a Riemannian metric, and $L \subset (T^*S, \lambda_{st})$ a compatible Betti Lagrangian of rank n. By definition, a *pre-spectral network* \mathcal{F} compatible with (L, S, g) is a properly embedded finite directed graph $\mathcal{F} \subset S$ such that:

- (1) Each edge of \mathcal{F} lies outside K_L and is decorated with an ordered pair $(ij), i, j \in [1, n]$ distinct.
- (2) Each directed edge decorated with (ij) is an (ij)-gradient flowline for the multigraph $L \subset (T^*S, \lambda_{st})$.
- (3) There are four types of vertices allowed in \mathcal{F} : initial, interaction, non-interaction and inconsistent. These are depicted in Figure 6; note that this classification requires the decoration on the adjacent edges.
- (4) All branch points of the projection $L \to S$ are initial vertices of \mathcal{F} . At an initial vertex, all adjacent edges are decorated with (ij), where i, j index the sheets of L for the ramification point above the branch point.
- (5) The network \mathcal{F} is flow-acyclic, in the sense that it does not admit a directed cycle of edges $(\mathfrak{w}_1, \ldots, \mathfrak{w}_n)$ in \mathcal{F} , satisfying the following two conditions.
 - (a) For $1 \leq m \leq n-1$, if \mathfrak{w}_m and \mathfrak{w}_{m+1} meet at an inconsistent or a non-interaction vertex, then the two decorations agree.

(b) If \mathfrak{w}_m and \mathfrak{w}_{m+1} meet at an interaction vertex, then either the two decorations agree, or \mathfrak{w}_{m+1} is the outgoing edge decorated with (ik) in Fig. 6.

Edges of \mathcal{F} are also said to be *walls* for the pre-spectral network \mathcal{F} . An edge decorated with (ij) is often said to be an (ij)-wall. A *wall* which (asymptotically) ends at a marked point $\mathbf{m} \subset \mathbf{S}$ is said to be semi-infinite. Finally, in the meromorphic case, where (\mathbb{S}, Σ, g) has a Riemann structure, a WKB pre-spectral network \mathcal{F}_{θ} of phase θ is defined to be a pre-spectral network compatible with $(\operatorname{Re}(e^{i\theta}\Sigma), S, g)$.



FIGURE 6. The four local models for vertices in pre-spectral networks in Definition 3.5: initial, interaction, non-interaction and inconsistent. We refer to the two types of interaction vertices as creation and 6-valent. In all these models $i, j, k, l \in [1, n]$ are all different.

Vertices of \mathcal{F} are also referred to as joints, and a pre-spectral network is said to be *creative* if all the interaction joints are creation joints. Observe that creative pre-spectral networks are automatically flow-acyclic. For notational ease, we will always assume that a given WKB pre-spectral network has phase $0.^8$ Also, we will often omit explicitly writing the data of the Riemannian metric g in (S,g) and a Betti Lagrangian $L \subset (T^*S, \lambda_{st})$ will always implicitly have a compatible choice of such g.

⁸Because a WKB pre-spectral network of phase θ for Σ is the same object as the WKB pre-spectral network of phase 0 for $e^{i\theta}\Sigma$.

Remark 3.6. (1) Note that there are two different orientations on W, either seen as a spectral network or as a flowline graph. The orientation as a spectral network is outgoing to infinity: e.g. the three rays emerge out of a D_4^- -singularity. The orientation when seen as a flowtree is opposite, incoming from infinity: the three flowlines near a D_4^- -singularity convergence towards it.

(2) In the physics literature, annihilation type joints, opposite to creation joints, are also allowed as interaction vertices. This is because the BPS index vanishes and non-active walls are not drawn. These will not appear in our context, as flowlines canceling each other are accounted by themselves, e.g. cf. Section 7.3 and Figure 24 therein. \Box

The following result allows us to describe the pre-spectral networks in Definition 3.5 as unions of D_4^- flow-trees, as in Definition 3.2.

Proposition 3.7 (D_4^- -trees from pre-spectral networks). Let $L \subset (T^*S, \lambda_{st})$ be a Betti Lagrangian and \mathfrak{F} a compatible pre-spectral network. Consider a point $z \in \mathfrak{w}$ on a wall $\mathfrak{w} \in \mathfrak{F}$. Then:

- (1) There exists a D_4^- -tree whose root vertex maps to z and the flowline for its adjacent edge is mapped to \mathfrak{w} .
- (2) The entire image of such D_4^- -tree is contained in that of \mathfrak{F} .
- (3) The set of such flow-trees is finite and, if \mathcal{F} is creative, it contains a unique element⁹.

Proof. Consider a rooted (partial) flowtree Γ whose root edge maps to a wall of \mathcal{F} and the non-root univalent vertices map to the vertices of \mathcal{F} . Let us now explain an iterative procedure that extends Γ to a larger partial flow-tree Γ' by adding walls of \mathcal{F} to each of the non-root univalent vertices of Γ . This algorithm produces finitely many such extensions and the D_4^- -flowtree in the statement of Proposition 3.7 is then declared to be the set of maximal extensions of the partial flow-tree given by the flowline connecting z to the starting point of \mathfrak{W} .

The procedure is described as follows. Let v be a univalent vertex and \mathfrak{w} a wall of \mathcal{F} having v as the starting vertex:

- (1) If v maps to an initial vertex, then the procedure ends without doing anything.
- (2) If v maps to a non-interaction joint, add the ingoing wall labeled with the same pair of sheets as \mathfrak{w} .
- (3) If v maps to a hexavalent interaction joint. Let $\mathfrak{w}_{ij}, \mathfrak{w}_{jk}, \mathfrak{w}_{ik}$ be the ingoing edges and $\mathfrak{w}'_{ij}, \mathfrak{w}'_{jk}, \mathfrak{w}'_{ik}$ the outgoing edges. In the cases $\mathfrak{w} = \mathfrak{w}'_{ij}$ or $\mathfrak{w} = \mathfrak{w}'_{jk}$, we add the ingoing wall labeled with the same pair of sheets as \mathfrak{w} . If $\mathfrak{w} = \mathfrak{w}'_{ik}$, then we can either add the two edges \mathfrak{w}_{ij} and \mathfrak{w}_{jk} , or we add the single wall \mathfrak{w}_{ik} .
- (4) If v maps to a creation joint, and $\mathbf{w} = \mathbf{w}'_{ij}$ or $\mathbf{w} = \mathbf{w}'_{jk}$, do the same as above. In the case $\mathbf{w} = \mathbf{w}'_{ik}$, we add the two walls \mathbf{w}_{ij} and \mathbf{w}_{jk} .

 $^{^{9}}$ We can drop flow-acyclicity at the cost of changing the statement to the set of flow-trees being finite up to some energy cut-off.

(5) Finally, if v maps to an inconsistent joint, either $\mathfrak{w} = \mathfrak{w}'_{ij}$ or $\mathfrak{w} = \mathfrak{w}'_{jk}$, and so we can proceed on as above.

Since the spectral network \mathcal{F} is finite, and a vertex cannot reappear with the same flowline edge by the flow-acyclicity of \mathcal{F} , given any Γ the process will stop after finitely many steps. Since each step can produce more than one trees if and only if there are some hexavalent vertices involved, \mathcal{F} being creative implies that it can produce exactly one extension.

An example of a D_4^- -tree as Proposition 3.7 is depicted in Figure 7 (Center). We can use Proposition 3.7 to associate a D_4^- -tree to a wall $\mathfrak{w} \in \mathcal{F}$ without specifying a point $z \in \mathfrak{w}$. Indeed, since \mathfrak{w} is directed, we choose z to be the unique boundary point of \mathfrak{w} reached by flowing backwards along \mathfrak{w} . This would be the "negative end" of \mathfrak{w} , and note that it might be a marked point $\mathfrak{m} \subset \mathbf{S}$, in which case the D_4^- -tree is interpreted to have a root semi-infinite edge. In either case, the argument for Proposition 3.7 thus shows that there is a well-defined D_4^- -tree for \mathfrak{w} . By definition, a D_4^- -tree obtained from Proposition 3.7 is said to be a D_4^- -tree on the pre-spectral network \mathcal{F} .



FIGURE 7. (Left) A spectral network \mathcal{F} with a wall \mathfrak{w} , in clear orange, and a point $z \in \mathfrak{w}$, in pink. This wall \mathfrak{w} is a semi-infinite edge. (Center) The D_4^- -tree $\Gamma(z; \mathfrak{w})$ with a root vertex at z, in red. (Right) The D_4^- -tree $\Gamma(\mathfrak{w})$ in solid red, with a semi-infinite root edge asymptotic to a Reeb chord (in dashed red). In this example, one might take i = 1, j = 2 and k = 3. As with the above figures, yellow vertices are initial vertices of \mathcal{F} , the marked point in $\mathbf{m} \subset \mathbf{S}$ is drawn in orange and its associated Legendrian front in blue.

Definition 3.8. Let \mathcal{F} be a pre-spectral network, \mathfrak{w} a wall and $z \in \mathfrak{w}$ a point. Each D_4^- -tree obtained as in Proposition 3.7 is denoted by $\Gamma(z; \mathfrak{w}, \mathcal{F})$ and its soliton class is denoted by $\mathfrak{s}(z; \mathfrak{w}, \mathcal{F})$.

Remark 3.9. If $L \subset (T^*S, \lambda_{st})$ is an exact Betti Lagrangian, then the set of D_4^- -trees associated to a wall in Definition 3.8 contains a unique element for each wall \mathfrak{w} .

Note that creative pre-networks are rather robust objects, i.e. invariant under smooth isotopies of **S** and C^{∞} -small perturbations of the base metric (S, g).

The existence of *new Stokes' lines*, in the nomenclature of [BNR82], a.k.a. higher-order Stokes phenomenon in [HLOD04, Section 2] (see also [AKT01, Section 4]), implies that the pre-spectral networks in Definition 3.5 do not contain sufficient information to study spectral curves with irregular singularities, or Stokes local systems on a Betti surface, for rank three or higher. Newer walls for the pre-spectral networks must be introduced to account for all the possible D_4^- -trees dictated by a Betti Lagrangian. From this perspective, the role of inconsistent vertices is to give birth to the new Stokes' lines, producing interaction vertices. Inspired by the terminology in [GHK15, Definition 2.26], we introduce the following notion to capture the appearance of new Stokes' lines:

Definition 3.10. Let $\mathcal{F}, \mathcal{F}'$ be two pre-spectral networks compatible with (L, S, g) such that $\mathcal{F} \subset \mathcal{F}'$ is an inclusion¹⁰ of stratified 1-manifolds. By definition, \mathcal{F}' is said to be a consistent extension of \mathcal{F} if:

- (1) The vertex set of \mathcal{F} is contained in the vertex set of \mathcal{F}' ,
- (2) the inconsistent vertices in \mathcal{F} are interaction joints in \mathcal{F}' .

In such a situation, we denote $\mathcal{F} \to \mathcal{F}'$ for a consistent extension of \mathcal{F} .

Definition 3.10 is one step in an iterative process, as a consistent extension of a pre-spectral network might itself have inconsistent vertices. A first idea is to iterate this process ad infinitum. The only issue is that the flow-energy of the newly created D_4^- -trees might remain bounded, potentially leading to infinitely many D_4^- -trees below a finite flow-energy cut-off. For that, we introduce the following notion of a gapped sequence of consistent extensions, a terminology inspired by the notion of gapped A_{∞} -algebras from [Fuk+09a].

Definition 3.11. Let $\mathcal{F}_1 \to \mathcal{F}_2 \to \ldots$ be a sequence of consistent extensions of pre-spectral networks. Given a point $z \in \mathcal{F}_{m+1} - \mathcal{F}_m$, consider the minimum $\min_{m+1}(Z(\mathfrak{s}(z)))$ of the flow-energies for all D_4^- -trees $\Gamma(z; \mathfrak{w}, \mathcal{F}_{m+1})$, for \mathfrak{w} a wall in the pre-spectral network \mathcal{F}_{m+1} containing z. By definition, the sequence is said to be *gapped* if there exist positive constants $\hbar, M \in \mathbb{R}_+$ such that

$$\min_{m+1} \left(Z(\mathfrak{s}(z)) \right) > \lfloor m/M \rfloor \hbar$$

for all $z \in \mathcal{F}_{m+1} - \mathcal{F}_m$ and $m \ge M, m \in \mathbb{N}$.

Definitions 3.5, 3.10 and 3.11 allow us to introduce the definition of a spectral network:

¹⁰Note that this is different than a graph inclusion, as there might be more vertices in \mathcal{F}' than in \mathcal{F} .

ROGER CASALS AND YOON JAE NHO

Definition 3.12 (Spectral networks). Let $L \subset (T^*\mathbf{S}, \lambda_{st})$ be a Betti Lagrangian with (S, g) a chosen Riemannian metric. By definition, the Betti Lagrangian L is said to admit a Morse spectral network \mathcal{W} if there exists a gapped sequence $\mathcal{W}_1 \to \mathcal{W}_2 \to \ldots$ of consistent extensions of prespectral networks $\mathcal{W}_i, i \in \mathbb{N}$. By definition, a Morse spectral network is finite if $\mathcal{W}_i = \mathcal{W}_{i+1}$ for sufficiently large $i \in \mathbb{N}$. In the meromorphic case, Σ is said to admit a WKB spectral network \mathcal{W}_{θ} of phase θ if it admits a Morse spectral network consisting of WKB pre-spectral networks of phase θ .

Definition 3.12 is more general than the notion of spectral networks in the physics literature [GMN13b; GMN12; GMN13a; GMN14], thus the terminology Morse spectral network. That said, for notational ease, we often refer to Morse spectral networks in Definition 3.12 as *spectral networks* and refer to the spectral networks in the physics literature as WKB spectral networks, as they are always associated to meromorphic spectral curves.

In the introduction, as stated in Theorem 1, we claimed the existence of spectral networks compatible with a Betti Lagrangian $L \subset (T^*\mathbf{S}, \lambda_{st})$. The next two sections are devoted to the proof of Theorem 1. Specifically, Theorem 1.(i), for exact Betti Lagrangians, is established in Section 3.3. Theorem 1.(ii), for WKB spectral networks, is proven in Section 3.4.

Note that, thanks to the gapped condition, we can ensure that for a given energy level $E \in \mathbb{R}_+$, there exists $m \in \mathbb{N}$ such that any $z \in \mathcal{W}_{m+1} - \mathcal{W}_m$, satisfies $\min_{m+1}(Z(\mathfrak{s}(z))) > E$. This allows us to define the energy-filtration on spectral network.

Definition 3.13 (Energy filtration). Let \mathcal{F} be a spectral pre-network and $E \in \mathbb{R}_+$. By definition, $\mathcal{F}(E)$ is the set of points in \mathcal{F} such that $\min_{\mathcal{F}}(Z(\mathfrak{s}(z))) \leq E$. Similarly, for a spectral network $\mathcal{W} = (\mathcal{W}_1 \to \mathcal{W}_2 \to \ldots)$ we define $\mathcal{W}(E)$ to be $\mathcal{W}_m(E)$, where $m \in \mathbb{N}$ is the smallest number such that any $z \in \mathcal{W}_{m+1} - \mathcal{W}_m$ satisfies $\min_{m+1}(Z(\mathfrak{s}(z))) > E$.

Remark 3.14. By construction, for any $z \in W(E)$, any D_4^- flowtrees that end at z with energy less than E must be contained in W(E). By scanning according to the energy filtration in Definition 3.13, it follows that a Morse spectral network is the union of all possible D_4^- -trees. \Box

3.3. **Proof of Theorem 1.(i).** Let $L \subset (T^*S, \lambda_{st})$ be an exact Betti Lagrangian and (S, g) a metric adapted to L. The goal is to show that there exists a finite Morse spectral network W compatible with L and (S, g), as in Definition 3.12. In the local model for Betti Lagrangians near the D_4^- -ramification points of the projection $\pi : L \longrightarrow S$, the Lagrangian multigraph L can be (and is) assumed to be holomorphic over a neighborhood of the D_4^- -branch point in S, and the Riemannian metric (S, g) is taken to be Kähler near such neighborhood. Thus we can apply the local model for the three initial flowline rays near a D_4^- -singularity studied in Section 2.4, and consider the associated trajectory, maximally extended, for each of these three rays out of each D_4^- -branch point in S, using the asymptotics of trajectories (Proposition 2.19 and Proposition 2.27).

Let W_1 be the union of all these trajectories, i.e. for each D_4^- -branch point, consider the union of all the trajectories in S obtained by flowing out of the branch point from each of the three initial rays. This graph W_1 might nevertheless not define a pre-spectral network due to the following phenomena:

- (1) A trajectory from a D_4^- -branch point might pass through another D_4^- -branch point. Note that a trajectory might do that in a way that it flows towards a D_4^- -branch point entering it via its cusp-cusp sheets, or it might merely pass through it while having two smooth sheets, or a cusp-smooth sheet, above it. By exactness, such trajectories cannot enter a D_4^- -branch point via a cusp-cusp edge, so that case we have discarded, but the other two cases are still possible.
- (2) The trajectories forming W_1 might intersect each other in a non-transverse manner, e.g. tangentially.

In order to construct a pre-spectral network from W_1 , we must perturb the initial Riemannian metric (S, g) so that these two issues above do not occur. In this case of an exact Betti Lagrangian, such perturbation exists by [Ekh07, Section 3.2], which itself uses [Sma61]; the possible perturbations form an open dense set within the corresponding space of Riemannian metrics. We thus apply such perturbation to (S, g) and obtain a graph in S, which we still denote by W_1 . This graph W_1 , after perturbing g, does define a pre-spectral network which satisfies:

- (1) \mathcal{W}_1 has only initial, non-interaction and inconsistent vertices. That is, \mathcal{W}_1 has no interaction vertices. The initial vertices of \mathcal{W}_1 are in bijection with the D_4^- -branch points in S.
- (2) the edges of \mathcal{W}_1 adjacent to an initial vertices are given by the three possible trajectories out of the D_4^- -branch point.

The edges of W_1 are decorated automatically by the condition on the initial vertex in Definition 3.5: the three trajectories out of a D_4^- -branch point are decorated with (ij), where $i, j \in [1, n]$ are the two sheets joined at the ramification point above it. This decoration extends uniquely to all edges of W_1 because the incoming decorations at any of the vertices determine the outgoing ones, as in Figure 6. This W_1 is the first of the (to be) sequence of pre-spectral networks leading to the spectral network W.

Let us construct a first consistent extension $W_1 \to W_2$, as follows. For each inconsistent vertex of W_1 , consider the unique flowline that starts at the inconsistent vertex: this exists and it is well-defined due to the local model of the three sheets for the multigraph $L \subset (T^*S, \lambda_{st})$ above such an inconsistent vertex. Let W_2 be the graph in S defined by the union of all such flowlines; the decoration on the new edges of this graph are uniquely determined by the decorations in W_1 and Figure 6. Similar to the case of W_1 , the resulting graph in S might have non-transverse intersections and trajectories passing through a D_4^- -branch point which are not initial rays for that branch point. As before, we further perturb the Riemannian metric data so that such behavior is not present: the space of such perturbation is the intersection of two dense open sets, open and non-empty. Let us still denote by W_2 the result of such perturbation. As a graph, W_2 might appear to be a pre-spectral network: it is nevertheless possible that some of the new trajectories being introduced are periodic or, in general, violate the acyclicity condition Definition 3.5.(4). That is, there might be a trajectory in W_2 that starts at an inconsistent vertex of W_1 and ends at the same interaction vertex; e.g. entering it via one of the two existing incoming flowlines from W_1 . By Proposition 2.19, exactness of $L \subset (T^*S, \lambda_{st})$ implies that this cannot occur and thus W_2 defines a pre-spectral network with no periodic trajectories.

We iterate the construction in the previous paragraph, thus constructing a consistent extension $W_2 \to W_3$ whose new trajectories start or end at the inconsistent vertices of W_2 . In order to ensure that W_3 is a pre-spectral network, we perturb the metric data again and use Proposition 2.19. In particular, W_3 is flow-acyclic, because by our construction, W_3 network is creative. This argument proves flow-acyclicity of \mathcal{F}_3 . as above. Iterating this construction for $i \in \mathbb{N}$, we obtain a sequence $W_1 \to W_2 \to \ldots$ of consistent extensions of pre-spectral networks. Our task is now to ensure that this sequence is gapped. In the exact Betti Lagrangian case, we will establish an a priori bounds on the energy to prove finiteness. It is in line with the a priori inequalities used in Gromov-compactness results for pseudo-holomorphic strips. The following subsection is devoted to proving gapped and finite, thus establishing the necessary results to conclude the proof of Theorem 1.(i).

Remark 3.15. A type of behavior that might a priori occur is that of a spiraling sequence of creation vertices approaching a D_4^- -vertex, violating the finiteness condition for W. Figure 8 is an example of such a sequence.

3.3.1. Energy gaps in consistent extensions. Near each D_4^- -singularity $b \in L$, we have a holomorphic neighborhood where both the sheets of the Betti Lagrangian and a neighborhood $U_b \subset S$ of $\pi(b)$ are endowed with holomorphic coordinates. Let $z \in U_b$ be a local conformal coordinate and $\lambda_1, \ldots, \lambda_n$ define the smooth sheets of L over U_b . As in Example 2.5, consider the associated front given by the set of functions

$$\{\pm \operatorname{Re}(z^{3/2}), \operatorname{Re}(|z|\lambda_1(0)e^{i\theta}), \dots, \operatorname{Re}(|z|\lambda_n(0)e^{i\theta}), \}, z \in U_b.$$

By a C^{∞} -small perturbation $L \subset (T^*S, \lambda_{st})$, compactly supported near the ramification point, we can assume that this front is generic. That is, all the associated Reeb chords occur at different values of θ and it contains only double transverse points. We always choose a local model near a D_4^- -singularity satisfying these conditions and, inspired by [Ekh07, Section 3.1], we refer to this as satisfying preliminary transversality conditions at the D_4^- -singularity.

The spiraling behavior from Remark 3.15, which we want to ensure does not occur in an infinite manner, can be formalized as follows:

Definition 3.16 (Chain interactions). Locally in a neighborhood $U_b \subset S$ of a D_4^- -branch point, a *chain interaction tree* is a flow-tree $F : \Gamma \to U_b$ such that:



FIGURE 8. (Upper left) An initial vertex in a pre-spectral network of rank 3 with its adjacent three trajectories and a (13)-trajectory (from another vertex) interaction with a wall of this initial vertex. Notice that we have denoted the sheets as in [GMN13b; GMN13a; GMN14] and added a branch cut (in dashed blue) for the first and second sheets. (Upper center) A consistent extension of the upper left piece obtained by adding the (23)-wall, in red: its behavior is that of spiraling around the initial D_4^- vertex, approaching it. (Upper right onward) A sequence of consistent extensions where the newly created walls from inconsistent vertices spiral around the initial D_4^- -vertex, infinitely approaching it. As with the above figures, the yellow vertex is the initial vertex of \mathcal{F} and each new wall for a consistent extension is drawn in a different color: red, yellow, purple and green, in order of appearance.

- The distinguished out-going edge is a smooth-cusp flowline.
- All the internal vertices are trivalent and lie outside the origin. Furthermore, given an internal vertex, there is a unique out-going edge which is a smooth-cusp flowline, and there is exactly one in-going edge which is given by an initial ray at $\pi(b)$.

Let us show that chain interaction trees have uniformly bounded number of internal vertices.

Proposition 3.17. In the notation above, let $z \in U_b \subset S$ be a local coordinate centered the D_4^- branch point $\pi(b)$. Then, there exists two positive finite constants $\delta \in \mathbb{R}_+$ and $N \in \mathbb{N}$ such that any chain interaction trees with image contained in $\{z : |z| \leq \delta\} \subset U_b$ has its number of internal vertices bounded above by N. *Proof.* Consider a chain interaction tree in U_b and a sufficiently small circle S^1_{θ} giving the base for the front of the Legendrian at the D_4^- -singularity. We project the chain interaction tree to a flowtree on this circle S^1_{θ} , as follows. Given the tree, we totally order the vertices of Γ by the distance from its unique root vertex: then the in-going external edge coming from a smooth-singular flowline projects to a flowline of the same index on S^1_{θ} . In addition, each external edge coming from the Reeb chord ray meeting the smooth-singular flowline gives a two-valent internal vertex with a *negative puncture*. (For instance, if the incoming smooth-singular flowline *i j* interacts with a singular-singular flowline ik to create a smooth-singular flowline ik, then the projected flowtree is two-valent, with the in-going flowline ij and the out-going flowline ik.) Therefore, each such external edge gives rises to an internal *negative puncture*. The unique out-going external edge and the remaining unique in-going singular-smooth flowline edge may be either positive or negative. By [Ekh07, Section 3] and the preliminary transversality condition, a cusp-free generic Legendrian link has an upper bound on the number of internal vertices and edges, given a finite bound on the number of positive punctures, which in our case is at most 2. In conclusion, there is a uniform upper bound on the length of the chain interaction tree, as required.

Proposition 3.18. Let $\mathcal{W} = (\mathcal{W}_1 \to \mathcal{W}_2 \to ...)$ be a spectral network constructed as in Section 3.3. Then the spectral network \mathcal{W} is gapped and finite.

Proof. Let us show \mathcal{W} is gapped. First, apply Proposition 3.17 to obtain two positive constants $\delta \in \mathbb{R}^+$ and $N \in \mathbb{N}$ such that chain interaction trees contained in $U_{2\delta}$ neighborhoods of the D_4^- -singularities have lengths bounded above by N. By choosing $\delta \in \mathbb{R}^+$ small enough, we can and do assume that the neighborhoods $U_{2\delta}$ are a disjoint union of small balls containing all the D_4^- singularities. Second, choose a positive constant $\hbar \in \mathbb{R}_+$ smaller than the minimum of the flow-energies for all flowlines traveling from the boundary ∂U_{δ} to the boundary $\partial U_{2\delta}$. By Definition 2.7, Betti Lagrangians are weakly bounded and thus we can choose this minimal energy \hbar to be smaller than the minimal flow-energy needed to travel between the distinct neighborhoods $\partial U_{2\delta}$ of the D_4^- -singularities in S.

Let Γ be a rigid D_4^- tree with flow-energy $\leq E$. Suppose the non-univalent internal vertices avoid the D_4^- -singularities. Then we will show that $E > \lfloor (|V(\Gamma)|/N) \rfloor \hbar$, where N is the upper bound from Proposition 3.17. From this, it follows that W must be gapped, since D_4^- -trees in in $W_i - W_{i-1}$ are all rigid, and none of the non-univalent internal vertices get mapped to the branch points, and these trees have at least *i* internal vertices. To prove the claim, we make the observation that there are at most $\lfloor (|V(\Gamma)|/N) \rfloor$ maximal interaction chain trees contained in Γ , since by Definition 3.16, such trees have at most N internal vertices. Furthermore, by our choice of $\hbar \in \mathbb{R}_+$, the unique external in-going smooth-cusp edge in the interaction chain tree must carry flow-energy at least \hbar and thus a maximal chain interaction tree must also have flow-energy at least \hbar . Therefore, $E > \lfloor (|V(\Gamma)|/N) \rfloor \hbar$, as claimed. To argue that \mathcal{W} is finite, note that the primitive of λ_{st} restricted to the exact Lagrangian $L \subset (T^*S, \lambda_{st})$ is a smooth function on the compact domain given by the complement of the $U_{2\delta}$ -neighborhoods in S. It is therefore a bounded primitive and, by Lemma 2.14, the total energy of any D_4^- -tree must be uniformly bounded as well. Since \mathcal{W} is gapped, only finitely many D_4^- -trees can appear and thus \mathcal{W} is finite, on any precompact subset of S. Finally, in order to control finiteness of \mathcal{W} at infinity, we use Lemma 2.17 implies that the walls that intersect the boundary transversely, in the inward direction, cannot leave, and this condition holds for any other wall that is created at interaction joints. Therefore, we are reduced to the finiteness of flow-trees in J^1S^1 with a fixed number of negative punctures. Again, such flow-trees have finitely many internal vertices. Therefore, \mathcal{W} is finite everywhere.

3.4. **Proof of Theorem 1.(ii).** The structure of the argument is similar to the exact Betti case, as presented in Section 3.3. The only significant difference is that the only perturbations that we are allowed are those performed in the holomorphic setting and varying the phase θ . In particular, we can no longer guarantee that W is creative, since three flowlines may intersect at a single point. The sequence $W = (W_1 \rightarrow W_2 \rightarrow ...)$ of consistent extensions of pre-spectral networks is built in the same manner. As before, the issues to be addressed are non-transverse intersections and flow-acyclicity. For removing periodic trajectories, we use Proposition 2.27 in order to argue that the sequence of consistent extensions of pre-spectral networks produces no periodic trajectories (instead of Proposition 2.19). For the former, and the more general flow-acyclicity, we need a different argument, as [Ekh07, Section 3.2] is not a holomorphic perturbation. Lemma 3.19 below proves that tangencies can indeed be removed. Assuming that result, we apply Proposition 3.18, which also works in this setting, and conclude Theorem 1.(ii).

To argue that we can remove non-tranverse intersections, consider $z \in S$ a point in the base and let $\gamma(s)$ be an *ij*-flowline of phase 0 starting at $\gamma(0) = z$ and let $C \in \mathbb{R}_+$ be a constant. Define the charge map of γ at time *s* with initial charge *C* to be the constant $C + \int_0^s (\lambda^i - \lambda^j)(\gamma'(t))dt$. By definition, the flat coordinate¹¹ *W* on a neighborhood of γ in *S* is the analytic continuation of the charge map over a small sectorial neighborhood of γ avoiding the *ij*-branch points, so we have $W = C + \int (\lambda^i - \lambda^j)$ in such a neighborhood. The constant term *C* matters in the case γ is a new-born edge at an interaction joint. Consider the θ -deformation of γ given by the curve

(15)
$$s \to W^{-1}((C+s)e^{i\theta}),$$

so the θ -deformation has the same energy (the absolute value of charge) as γ at time s. By differentiating Eq. (15) with respect to θ , we deduce that the normal deformation vector at time s is given by $(C + s)/(\lambda^i - \lambda^j)$. Here the appearance of (C + s) in the expression in Eq. (15) is because charge-preserving deformations necessarily have their normal deformation affected by the energy. Let us now show that via θ -deformations we can indeed ensure transversality:

¹¹The map W is a flat conformal coordinate map in the sense that the pull-back of $\lambda^i - \lambda^j$ becomes dW in the W-coordinate.

Lemma 3.19. Let γ_1, γ_2 be a (ij)-flowline and (kl)-flowline of phase $\theta = 0$, $i, j, k, l \in [1, n]$ distinct, and suppose that γ_1, γ_2 have an isolated tangential intersection at a point $w \in S$. Then, for small enough generic $\theta \neq 0$, their θ -deformations intersect transversely.

Proof. Let W_{ij} and W_{kl} denote the charge maps with initial charges l_1 and l_2 , associated to γ_1 and γ_2 near z. The goal is to show that the variation vectors of γ_1 and γ_2 are $W_{ij}/(\lambda_i - \lambda_j)$ and $W_{kl}/(\lambda_k - \lambda_l)$ and that tangencies are stable if and only if the variation vectors agree.

We begin the proof. In the $z = W_{ij}$ -coordinate, we can write

$$\lambda_i - \lambda_j = dz, \quad \lambda_k - \lambda_l = \phi(z)dz,$$

for some holomorphic function ϕ defined in this z-domain. Then $\phi(z)$ is not constant because $\lambda_k - \lambda_l$ cannot be a constant multiple of $\lambda_i - \lambda_j$ everywhere, as w is an isolated tangential intersection point. As before, $W^{-1}((l_2 + s)e^{i\theta})$ gives rise to the θ -deformation of γ_2 . As we change θ , the initial points of γ_1 and γ_2 typically change too (unless they are initial flowlines themselves) and so the quantities $l_1 = l_1(\theta), l_2 = l_2(\theta)$ also depend on θ . Thus, our aim is to show that the map

(16)
$$F(s_1, s_2, \theta) = (s_1, s_2, \theta) \to ((l_1(\theta) + s_1)e^{i\theta}, W^{-1}((l_2(\theta) + s_2)e^{i\theta})) \in \mathbb{S} \times \mathbb{S}$$

is transverse to the diagonal for generic θ . This is achieved in two steps, as follows.

Step 1. Let us analyze the locus of points where the θ -deformations intersect tangentially. We claim that the θ -deformations can be only tangential over the tangency locus, where the *tangency* locus is defined to be the set where the imaginary part $\text{Im}(\phi) = 0$ vanishes. Indeed, suppose the θ -deformations are tangential, with the common tangency given by the vector v, and note that

$$v \in \ker \operatorname{Im}(e^{i\theta}) \Leftrightarrow v \equiv_{\mathbb{R}} e^{-i\theta}$$
, and so $v \in \ker(\operatorname{Im}(e^{i\theta}\phi)) \Leftrightarrow v \equiv_{\mathbb{R}} e^{-i\theta}/\phi$,

where $v \equiv_{\mathbb{R}} w$ denotes v, w being real colinear. Therefore, in this case, 1 and $1/\phi$ should be real colinear at the point of the tangency which is if and only if $\text{Im}(\phi) = 0$, as claimed. Therefore tangencies can only appear along $\text{Im}(\phi) = 0$.

Now, the tangency locus is a locally finite stratified manifold, see e.g. [Wal75]. Let us show that the tangency locus is smooth. For that, choose a smooth parametrized 1-dimensional stratum $\eta(t)$ of the tangency locus and note that differentiating the condition $\text{Im}(\phi) = 0$ leads to $\text{Im}(\phi'\eta'(t)) = 0$. This latter question is that of the flowline of ϕ' passing through l_1 . Then Picard-Lindelöf's uniqueness of solutions of differential equations implies that the tangency locus must be the entire flowline of ϕ' passing through l_1 and thus it is smooth.

Step 2. Let us compute the set of points in the tangency locus where the transversality of the map F also fails. By construction, the vector space given by $im(dF) + T \cdot diag$ is generated by the rows

of the 4×4 matrix:

$$\begin{pmatrix} e^{i\theta} & 0\\ 0 & e^{i\theta}/\phi\\ 1 & 1\\ \partial_{\theta}l_{1}e^{i\theta} + i(l_{1}+s_{1})e^{i\theta} & (e^{i\theta}/\phi)\partial_{\theta}l_{2} + i(l_{2}+s_{2})e^{i\theta}/\phi \end{pmatrix}$$

After an elementary transformation eliminate the terms containing $\partial_{\theta} l_1$ and $\partial_{\theta} l_2$, the determinant of the matrix is equal to

$$(1/\phi)((l_2+s_2)/\phi-(l_1+s_1))$$

Therefore the matrix is not invertible, and so transversality fails, if and only if

(17)
$$l_1 + s_1 = (l_2 + s_2)/\phi$$

This set is discrete. Indeed, since $z = (l_1 + s_1)e^{i\theta}$ and $W(z) = (l_2 + s_2)e^{i\theta}$, this set belongs to the zero set of the holomorphic function

$$z = W(z)/\phi(z)$$

which is discrete unless $W(z) = z\phi(z)$. This latter equality cannot occur though: differentiating both sides gives $\phi(z)dz = dW = (\phi(z) + z\phi'(z))dz$ and so $\phi'(z) = 0$ along the 1-stratum $\eta(t)$. This is a contradiction since $\phi(z)$ is holomorphic and not a constant function. Therefore, since the set of points where (17) holds (and so transversality fails) is finite, standard parametric transversality applies and we conclude that, for generic θ , the θ -deformations do not intersect tangentially. \Box

As argued above, at the beginning of Section 3.4, Theorem 1.(ii) then follows now that Lemma 3.19 is proven, after we sort out flow-acyclicity. To see that flow-acyclicity holds, we impose the genericity condition that for each inductive step, W_i consists only of vertices that are stable under generic infinitesimal perturbations of θ , and that there does not exist an interaction vertex such that the out-going *ik*-wall is an ancestor to one of the in-going walls (in other words, some of the in-going walls have multiplicities). Such vertices are not generic, since the normal deformation vector for the new WKB flowline at v has its norm greater than that of the out-going wall at v. We now proceed on by induction. Suppose the acyclicity condition holds for W_k but fails for W_{k+1} and let $(\mathfrak{w}_1, \ldots, \mathfrak{w}_n)$ be the directed cycle that satisfies conditions (a) and (b) in Definition 3.5, for which the decorations of \mathfrak{w}_1 and \mathfrak{w}_n agree. Since there are no periodic trajectories, this is possible only if \mathfrak{w}_1 is a wall contained in W_k , and so we see that there is some *i* such that $\mathfrak{w}_i \subset W_k$ is an ancestor to \mathfrak{w}_1 in W_k . However, then the terminal vertex of \mathfrak{w}_i fails to satisfy the genericity condition. Theorem 1 is thus established, as Theorem 1.(i) was concluded in Section 3.3.

3.5. Spectral networks and 2d-4d BPS indices. In the context of the supersymmetric field theories in [GMN13b; GMN12; GMN13a; GMN14], BPS states are certain irreducible unitary representations of a fixed super Lie algebra acting on a given Hilbert space. In this framework, the counts of BPS states are achieved by considering super traces of certain operators on this Hilbert space. In Floer theory, counts of pseudo-holomorphic disks are given by considering the (oriented) cobordism type of their moduli spaces. For D_4^- -trees, we have the choice of counting them either

way: by counting the pseudo-holomorphic strips that limit to them, or by treating them combinatorially via the spectral network \mathcal{W} and using supertraces; the results of this article show that these counts coincide, cf. Section 5.3. The Floer theoretic count is discussed in Section 5, and we now discuss the combinatorial count.

Let \check{S}, \check{L} be the unit tangent bundles of S and L, respectively. Let H be the homotopy class representing the sphere bundle fiber. Given $z \in \mathring{S}$, we write $z_i, i \in [1, n]$ for the lifts of z to \mathring{L} . We denote by $\mathbb{C}[\pi_1(\check{L})]$ the group algebra of the fundamental group of (relative) homotopy classes in \check{L} : as further discussed in Section 5, we do not explicitly include the set of endpoints in the notation, as it is implicitly understood by context. For instance, if we consider a soliton class $\mathfrak{s}(z;\mathfrak{w})$, it is implicitly understood that it is a relative homotopy class with endpoints in the lifts above $z \in S$.

Let \mathcal{F} be a pre-spectral network and \mathfrak{w} a wall. We denote by $t(\mathfrak{w})$, resp. $s(\mathfrak{w})$, the sphere bundle lift of the terminal point of \mathfrak{w} , resp. initial, given by the unit velocity vector of $\mathfrak{F}(\mathfrak{w})$ at the terminal point, resp. initial. Regard \mathbb{Z} as a category with a single object, with hom identified with \mathbb{Z} (as a set, not as an additive group), and the set of walls W(F) as a category with trivial morphisms.

Definition 3.20. Let \mathcal{F} be a finite pre-spectral network and $W(\mathcal{F})$ its set of walls. By definition, the vanilla 2d4d BPS index on \mathcal{F} is the functor

(18)
$$\mu : \mathbb{C}[\pi_1(\mathring{L})] \times W(\mathcal{F}) \to \mathbb{Z}$$

uniquely defined by the following constraints:

- (1) $\mu([\mathfrak{s}(z;\mathfrak{w})];\mathfrak{w}) = 1$ for \mathfrak{w} an initial wall.
- (2) $\mu(\cdot; \mathfrak{w})$ vanishes on all classes, except for the *H*-orbits of $\mathfrak{s}(z; \mathfrak{w}) \in {\mathfrak{s}(z; \mathfrak{w})}$ for $z \in \mathfrak{w}$.
- (3) $\mu(H\rho; \mathfrak{w}) = -\mu(\rho; \mathfrak{w}).$
- (4) μ is independent of the basepoint $z \in \mathfrak{w}$ for the classes in $\{\mathfrak{s}(z;\mathfrak{w})\}$.
- (5) Given a non-interaction vertex v with ingoing walls $\mathbf{w}_{ij}, \mathbf{w}_{kl}$ and outgoing walls $\mathbf{w}'_{ij}, \mathbf{w}'_{kl}$ let $v(\mathbf{w}_{ij})$ denote the unit velocity lift of v at \mathbf{w}_{ij} . Then

$$\mu(\rho; \mathfrak{w}_{ij}) = \mu(\rho; \mathfrak{w}'_{ij}),$$

for all $\rho \in \hom(v(\mathfrak{w}_{ij})_j, v(\mathfrak{w}_{ij})_i)$, and similarly for $\mu(\cdot; \mathfrak{w}_{kl})$ and $\mu(\cdot; \mathfrak{w}'_{kl})$.

(6) Let v be an interaction vertex with ingoing walls $\mathfrak{w}_{ij}, \mathfrak{w}_{jk}, \mathfrak{w}_{kl}$, outgoing walls $\mathfrak{w}'_{ij}, \mathfrak{w}'_{jk}, \mathfrak{w}'_{kl}$. and $v(\mathbf{w}_{ii}), v(\mathbf{w}_{ik}), v(\mathbf{w}_{ik})$ the corresponding unit velocity lifts. Then

(19)
$$\mu(\rho; \mathbf{w}'_{ij}) = \mu(\rho; \mathbf{w}_{ij})$$

(20)
$$\mu(\rho; \mathbf{w}'_{jk}) = \mu(\rho; \mathbf{w}_{jk})$$

 $\mu(\rho, \mathfrak{w}_{jk}) = \mu(\rho, \mathfrak{w}_{jk}),$ $\mu(\rho; \mathfrak{w}_{ik}) = \mu(\rho; \mathfrak{w}_{ik}) + \sum_{[\rho_1] \circ [\rho_2] = [\rho]} \mu(\rho_1; \mathfrak{w}_{ij}) \mu(\rho_2; \mathfrak{w}_{jk}),$ (21)

where ρ is a path such that the end points of the projection all lie at v, and the sum is over the mod 2H-orbits.

By definition, a wall of \mathcal{W} is said to be active if its BPS index is non-zero.

Remark 3.21. (1) Eq. (21) is known as the 2d Hori-Vafa wall-crossing formula. In the framework of spectral networks, for generic phase, the counts as captured by Definition 3.20 are given by a vanilla 2d-4d BPS index which is truly a 4d BPS index modified to accout for the 2d interactions. In the above we are able to restrict to pure 2d BPS indices because in our context they do coincide with the vanilla 2d4d BPS indices.

(2) In Section 5 it will be proven that Definition 3.20 is actually a count of pseudo-holomorphic strips for a fiber near that wall: it counts continuation strips oriented by the wall along a W-adapted path.

3.6. Spectral networks and rigid flowtrees. Flowtrees are often used in the context of Floer theory as a limiting model for configurations of pseudoholomorphic curves. To wit, [Fuk93, Chapter 1] studies such trees as a first model of the Fukaya A_{∞} -category, see also [FO97, Theorem 2.3] and [Fuk97, Chapter 1]. In the framework of Legendrian submanifolds, rigid flowtrees are studied in [Ekh07], compared to pseudo-holomorphic disks and used to provide certain computational models for the Legendrian contact dg-algebra, see also [RS20]. A first aspect in the Floer-theoretic understanding of spectral networks is its relation to such flowtrees, which we analyze in this subsection. Note that this is far from sufficient in order to prove our main results or compare to [GMN12; GMN13a; GMN14], and Sections 4, 5 and 6 proceed beyond these initial techniques. For now, we use the notation and concepts from [Ekh07] and establish the following result:

Proposition 3.22 (Augmented D_4^- -trees and rigid flowtrees). Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, $L \subset (T^*S, \lambda_{st})$ an exact Betti Lagrangian, (S, g) an adapted metric and $W \subset S$ a compatible creative Morse spectral network. Then, there exists a front-generic exact Lagrangian $L^{\dagger} \subset (T^*S, \lambda_{st})$ Hamiltonian isotopic to L, and a perturbed metric (S, g^{\dagger}) such that:

- (1) $L^{\dagger} = L$ and $g^{\dagger} = g$ outside a compact subset, and otherwise $L^{\dagger} \stackrel{C^{\infty}}{\approx} L$ and $g^{\dagger} \stackrel{C^{\infty}}{\approx} g$.
- (2) There is a bijective correspondence between augmented D_4^- -trees on W and the rigid flowtrees of (S, g^{\dagger}) with a single positive puncture on $\partial_{\infty} L^{\dagger}$.

The relation to spectral networks is Proposition 3.22.(2), whereas Proposition 3.22.(1) describe the type of perturbation that it is used, i.e. the Betti Lagrangian and the metric essentially only need to be modified near the branch points, equiv. the vertices of W.

Note that the results of [Ekh07] only apply to Legendrians with generic front singularities. The Legendrian lift of an exact Betti Lagrangian is *not* in that class because D_4^- -singularities are not generic real Legendrian singularities. Thus, to establish Proposition 3.22, we first need to understand rigid flowtrees near a D_4^- -singularity.

3.6.1. Trees near D_4^- -singularities. Consider the D_4^- -front singularity, as discussed in Case (2) of Section 2.4, cf. also [CZ22, Section 2.2.3]. Its germ can be parametrized as

$$\delta_4^-: \mathbb{R}^2_{u,v} \longrightarrow \mathbb{R}^3_{x,y,z}, \quad \delta_4^-(u,v) = \left(u^2 - v^2, 2uv, \frac{2}{3}(u^3 - 3uv^2)\right).$$

The singularity itself is at the origin of \mathbb{R}^3 and the front is invariant under $(2\pi/3)$ -rotation along the z-axis. Since $\delta_4^-(0, v) = \delta_4^-(0, -v)$, the three edges of A_1^2 -singularities are located at

$$E_0 := \{ (x, y, z) \in \mathbb{R}^3 : x < 0, y = z = 0 \} = \delta_4^-(0, v) = \mathbb{R}_{<0}, \quad v \in \mathbb{R} \setminus \{0\}$$

and its two additional images under this rotation, which are

$$E_1 := \{ (x, y, z) \in \mathbb{R}^3 : \sqrt{3}x = 2y, z = 0 \} = \delta_4^-(\sqrt{3}v, v) = e^{-2\pi i/3} \mathbb{R}_{<0}$$

$$E_2 := \{(x, y, z) \in \mathbb{R} : \sqrt{5x} = -2y, z = 0\} = \delta_4(-\sqrt{5v}, v) = e^{-\gamma} \mathbb{R}_{<0}$$

also with $v \in \mathbb{R} \setminus \{0\}$. The three initial rays discussed in Case (2) of Section 2.4 are

 $R_0 := \{ (x, y, z) \in \mathbb{R}^3 : x > 0, y = z = 0 \} = \mathbb{R}_{>0}, \quad R_1 := e^{-2\pi i/3} \mathbb{R}_{>0}, \quad R_2 := e^{2\pi i/3} \mathbb{R}_{>0},$

which lie exactly in between these three edges E_0, E_1, E_1 , cyclically alternating as $R_0, E_1, R_2, E_0, R_1, E_2$. The local lifts for these three non-generic flowtrees are depicted in Figure 9.



FIGURE 9. A non-generic model for the flowtrees near the D_4^- -singularity.

In a nutshell, the claim is that a generic C^{∞} -small compactly supported Hamiltonian perturbation of such non-generic D_4^- -front gives rise to a generic Legendrian front whose only rigid flowtrees are C^{∞} -close to the rays R_0, R_1, R_2 in the support of such perturbation, and otherwise coincide with them. This is the content of the following:

Lemma 3.23 (Sneaky trees). Let $\Pi \subset \mathbb{R}^3$ be the germ of a D_4^- -singularity, modeled as above, and $\Lambda(\Pi)$ its Legendrian lift. There exists a compactly supported C^{∞} -small Legendrian isotopy of $\Lambda(\Pi)$ such that the resulting Legendrian Λ^{\dagger} has a front Π^{\dagger} such that:

- (1) The Lagrangian projection of Λ^{\dagger} is embedded.
- (2) there are exactly three rigid flowtrees T_0, T_1, T_2 in $(\mathbb{R}^2_{x,y}, g_{st})$ for the generic Legendrian Λ^{\dagger} ,

- (3) Each T_i is C^{∞} -close to R_i and, outside a compact set, $T_i = R_i$, i = 1, 2, 3.
- (4) The compact support of the Legendrian isotopy can be taken to be arbitrarily small near the domain of the D_4^- -singularity.



FIGURE 10. (Left) The spatial front for a generic C^{∞} -small perturbation of the D_4^- front singularity. (Right) The view for this front from above, with orange encoding A_2 -cusp edges and dark blue edges indicating A_1^2 -crossing edges. Each of the three orange dots indicates a A_3 -swallowtail singularity.

Proof. Choose a C^{∞} -small Legendrian isotopy of $\Lambda(\Pi)$ which preserves the $(2\pi/3)$ -rotation symmetry of the front and let Π^{\dagger} be its front. Due to this symmetric choice and the fact that a flowtree between two sheets cannot pass through a crossing of those sheets, it suffices to analyze rigid flow trees in the sectorial region above E_1 and E_2 . A C^{∞} -small perturbation of D_4^- -singularity gives a front Π with three swallowtails as depicted in Figure 10. In Figure 10 (right) we choose this region betwee E_1, E_2 to be the bottom sector, so that R_0 is a vertical semirary from the center downwards. Since the perturbation is compactly supported and the gradient condition is local, any rigid flowtrees for Π coincide with those of Π^{\dagger} away from a compact set, and thus must coincide with R_0, R_1, R_2 . The argument now contains two parts:

- (1) the construction of a rigid Morse flow tree $\Gamma(D_4^-)$ coinciding with R_0 at the boundary,
- (2) showing that no other rigid trees exist in this local model.

For Part (1), a flowtree $\Gamma(D_4^-)$ is constructed following Figures 11 and 12. In Figure 11 (left) the flowtree is depicted in green, as seen from above, and Figure 11 (right) determines the relative homology class of its boundary by declaring that it geometrically intersects each relative cycles e_1, e_2, e_3 once, e_1, e_3 positively and e_2 negatively. We have also described $\Gamma(D_4^-)$ in Figure 12 as a movie, slicing the front in Figure 11 (left) with horizontal slices, from bottom to top. This flowtree $\Gamma(D_4^-)$ has an asymptotic positive puncture ρ at the boundary condition given by R_0 , no negative punctures, one *switch* (S) and one *end* (E). This is a legitimate flowtree by [Ekh07, Section 2]. We choose the perturbation of $\Lambda(\Pi)$ and perturb the metric q to q^{\dagger} such that the tree is transversely



FIGURE 11. (Left) The flowtree $\Gamma(D_4^-)$ as seen from above, with two black dots indicating its unique switch (S) and end (E). (Right) Three homology classes e_1, e_2, e_3 in the (Legendrian lift of the) front and the flowtree $\Gamma(D_4^-)$ seen from far above, so that the three swallowtails are contained in the central red dot.

cut out, which can be done by [Ekh07, Prop. 3.14]. The dimension of the moduli space for a general tree Γ is given in [Ekh07, Definition 3.4] as:

$$\dim(\Gamma) = -2 + I_u(\rho^+) - \left(\sum_{k=1}^{|\Gamma^-|} (I_s(\gamma_k^-) - n + 1)\right) + e(\Gamma) - s(\Gamma) - y_1(\Gamma),$$

where $I_u(\rho^+) = 2$ is the index at the positive puncture ρ^+ , i.e. the dimension $\dim(W^u(\rho^+))$ of the unstable manifold, $I_s(\gamma^-) = 2$ is the coindex at the negative puncture γ^- , i.e. the dimension $\dim(W^s(\gamma^-))$ of the stable manifold, $e(\Gamma)$ is the number of ends in Γ , $s(\Gamma)$ is the number of switches and $y_1(\Gamma)$ the number of Y_1 -type vertices. In the case of $\Gamma(D_4^-)$, the formula for the dimension yields

$$\dim(\Gamma) = -2 + I_u(\rho) + e(\Gamma) - s(\Gamma) = -2 + 2 + 1 - 1 = 0,$$

as there are no negative punctures nor Y_1 -vertices. This proves $\Gamma(D_4^-)$ is a *rigid* flowtree.

For Part (2), let us argue that no additional rigid flowtrees exist, which we deduce using the dimension formula for dim(Γ) above. First, for any germ of a Legendrian singularity, there exists a front with no Reeb chords. Hence, given that ρ is the only allowed boundary condition away from the singularity and in the sectorial region between E_1 and E_2 , there can be no negative punctures. By the classification of vertices in [Ekh07, Section 3] and the combinatorics from Figure 12, the rigid flowtrees must have one end, and therefore one switch. Since the switch can only occur as in $\Gamma(D_4^-)$, any rigid flowtree that matches R_0 actually coincides with $\Gamma(D_4^-)$. Finally, the combinatorics of the sheets imply that no two of the three rigid flowtrees, given by $\Gamma(D_4^-)$ and its images under the $(2\pi/3)$ -rotation, can interact with a Y_0 -vertex. Thus these are the only possible rigid flowtrees.



FIGURE 12. Slices of the rigid flowtree $\Gamma(D_4^-)$, drawn in red, in the germ of a D_4^- front singularity. The front is sliced horizontally, bottom to top. A *switch* (S) occurs at the leftmost diagram in the second row and an *end* (E) occurs at the center diagram in the third row.

Remark 3.24. These trees in the proof of Lemma 3.23 were first introduced by the first author in his work [CM18], and referred to as sneaky trees, due to their shape.¹²

3.6.2. Proof of Proposition 3.22. This is now a consequence of Lemma 3.23 and the bijection established in [Ekh07]. The only point left is appropriately choosing the neighborhoods of the D_4^- singularities, i.e. the initial vertices of \mathcal{W} , cut off the part of the rigid flow-trees that begin at such neighbourhoods, and arguing that the resulting partial flowtrees can be extended. For each D_4^- singularity $b \in S$, we choose a neighborhood $U_b(\delta) \subset S$ whose only initial vertex is b, there are no non-initial vertices of \mathcal{W} , and $\partial U_b(\delta)$ transversely intersects \mathcal{W} at the three initial rays associated to b. The perturbation needed is now obtained by locally perturbing according to Lemma 3.23 at each such $U_b(\delta)$. Since the rigid flow-trees must begin at ends, which are located at the cusp-edges, we see that we can cut-off rigid flow-trees along edges that first leave the $U_b(\delta)$. Note that the resulting partial flow tree in $U_b(\delta)$ can be extended to a global flowtree using Lemma 2.17. Such extension is rigid, as rigidity depends only on the vertices of the flowtree, and so it must have been the sneaky trees constructed in Lemma 3.23. Then the property in Part (1) holds by construction, and Part (2) holds by [Ekh07, Theorem 1.1.(b)].

Remark 3.25. (i) Technically, the perturbation in [Ekh07, Theorem 1.1] is global, and thus the spectral network W is perturbed by a C^{∞} -small smooth isotopy because of transversality. We

¹²A coarse chalkboard explanation of these trees can be found in the talk "Differential Algebra of Cubic Graphs" at the Harvard Center of Mathematical Sciences for "A Celebration of Symplectic Geometry 2017: 15 Years of JSG.", around minute 53.

implicitly absorb such smooth isotopy in our notation, as spectral networks are invariant under such isotopies.

(ii) We are implicitly using that the exact Betti Lagrangian L is Maslov 0, which will be shown in Lemma 5.13, which tells us that the topological index of the boundary must be zero. The topological dimension of such a disk (with single positive puncture) is zero only if the limiting Reeb chord has index 1 (and so the index of the limiting puncture is 2). This is if and only if the Reeb chord corresponds to a positive crossing, by the argument used in Section 2.5 (see also Fig. 5 for a heuristic explanation).

4. Adiabatic convergence for D_4^- -trees

The main object of this section is to prove the direction (\Leftarrow) of the characterization in Theorem 2. Namely, we show the contrapositive statement: points in the complement of the spectral network W do *not* have such pseudo-holomorphic strips through them and, for Theorem 2.(i), that for a point in the spectral network there are no such pseudo-holomorphic strips in any relative homology class except for possibly the soliton class. The precise statement is Theorem 4.2. The direction (\Longrightarrow) is proven in Section 5.

The image of a Betti Lagrangian $L \subset T^*S$ under the ε -scaling action on (T^*S, ω_{st}) is denoted $\varepsilon L \subset T^*S$. Throughout the manuscript, the expression *adiabatic limit* refers to the limit $\varepsilon \to 0$ and for this section we assume that L has either conical ends or is meromorphic with O(-1)-ends. The following notions formalize the scenarios with non-existence of pseudo-holomorphic strips in the adiabatic limit:

Definition 4.1. Let $(\mathbf{S}, \boldsymbol{m}, \Lambda)$ be a Betti surface, $L \subset T^*S$ a Betti Lagrangian adapted to (S, g) and J_g the almost complex structure associated to g. Consider a spectral network \mathcal{W} compatible with L, as in Definition 3.12. For a point $z \in S$ and the cotangent fiber $F_z = T_z^*S$:

- (1) The pair of Lagrangians (F_z, L) is said to be uniformly disk-free in the adiabatic limit if for any given energy $E \in \mathbb{R}_+$, there is some neighborhood $U_z(E) \subset S$ of z and a positive constant $\varepsilon_0(E, z) \in \mathbb{R}_+$ such that there is no non-constant J_g -holomorphic disk of energy below E between the Lagrangians εL and F_w , for all $w \in U_z(E)$ and $\varepsilon \in \mathbb{R}_+$ with $\varepsilon \in (0, \varepsilon_0(E, z)]$.
- (2) If $z \in W$ belongs to a wall, the pair of Lagrangians (F_z, L) is said to be uniformly Stokes in the adiabatic limit if for any given energy $E \in \mathbb{R}_+$, there is some neighborhood $U_z(E) \subset$ S of z and a positive constant $\varepsilon_0(E, z) \in \mathbb{R}_+$ such that there is no non-constant J_{g^-} holomorphic disk of energy below E bounded between the Lagrangians εL and F_w whose relative homology class is not of the parallel transport of the soliton class associated to z, for all $w \in U_z(E)$ and $\varepsilon \in \mathbb{R}_+$ with $\varepsilon \in (0, \varepsilon_0(E, z)]$.

That is, informally, (L, F_z) is uniformly disk-free if there are no pseudo-holomorphic strips bounded by $(\varepsilon L, F_w)$ for $\varepsilon \ll 1$ small enough and w close to z, for a given energy cut-off $E \in \mathbb{R}_+$. Similarly for uniformly Stokes. The use of the word uniformly in Definition 4.1 is to emphasize that the neighborhood $U_z(E)$ is independent of the adiabatic parameter ε . Since every statement in this section will be in the adiabatic limit as $\varepsilon \to 0$ we often drop the cue in the adiabatic limit, e.g. we simply refer to being uniformly disk-free or uniformly Stokes. The main result, implying the first part of Theorem 2, reads as follows:

Theorem 4.2 (Pseudo-holomorphic strips in spectral networks). Let $(\mathbf{S}, \boldsymbol{m}, \Lambda)$ be a Betti surface, $L \subset T^*S$ a Betti Lagrangian adapted to (S, g) and \mathcal{W} a compatible spectral network. Then

(1) If $z \in W^c$, then (F_z, L) is uniformly disk-free. (2) If $z \in W$, then (F_z, L) is uniformly Stokes.

The purpose of this section is to prove Theorem 4.2. The intuition for Part (1) is summarized as follows. If $z \in W^c$ was such that $(\varepsilon L, F_{z_{\varepsilon}})$ bounded a pseudo-holomorphic strip in (T^*S, J_g) , then we want to have such sequences of strips converge, as $\varepsilon \to 0$, to a D_4^- -tree in S passing through z. If such convergence statement held, then the results in Section 3 would force the D_4^- -tree to be contained in W and thus $z \in W$, reaching a contradiction. We remark that all the results we prove in this section rely on the relevant disks having uniformly bounded diameter (c.f. Lemma A.2), which follows from the geometric boundedness of L (Proposition A.1).

Of course the technical part is establishing convergence results for such pseudo-holomorphic strips with boundary conditions on $(\varepsilon L, F_z)$ in a manner that the limit is indeed a D_4^- -tree, or at least it is a D_4^- -tree away from a fixed small neighborhood of the branch points. Note that this is a much stronger result than just stating that the strips would converge to a flowtree, as such a weaker statement would provide no control on the trajectories associated to leaves and semi-infinite edges of the tree, and thus the resulting tree would typically *not* be a D_4^- -tree nor there would be any understanding of their behavior with regards to the initial rays at the branch points. In particular, a new aspect in the proof of Theorem 4.2 is that we must have control on the behavior of the degeneration of such pseudo-holomorphic strips as it approaches the boundary of a neighborhood of the branch points, even if $z \in W^c$ is typically far from any branch points. The precise convergence result that we establish reads as follows:

Proposition 4.3 (ε -strips converge to trimmed D_4^- -trees). Let $(\mathbf{S}, \boldsymbol{m}, \Lambda)$ be a Betti surface and $L \subset T^*S$ a Betti Lagrangian adapted to (S, g) with a compatible spectral network W. Suppose that $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ is an ε -strip sequence $u_{\varepsilon} : \Delta_2 \longrightarrow S$, with boundary of $(\varepsilon L, T_{z_{\varepsilon}}^*S)$. Then, after possibly considering a subsequence and a reparametrization of the domain Δ_2 , its limit

$$\lim_{\varepsilon \to 0} (u_{\varepsilon}, z_{\varepsilon}, \Delta_2, E) : \Gamma \longrightarrow S$$

exists and it is a (broken) trimmed D_4^- -tree. The tree is unbroken if z lies in the interior of $\mathcal{W}(E)$.

The concepts needed to state Proposition 4.3, including the notion of ε -strip sequences, their limits and broken trimmed D_4^- -trees, are introduced in Section 4.1 and Proposition 4.3 is then proven in Section 4.2. Proposition 4.3 is used to conclude Theorem 2 in Section 4.3.¹³

4.1. ε -strips and their limits. Let us first set up the geometric ingredients to discuss ε -strips and their limits. We endow its cotangent bundle (T^*S, ω) of (S, g) with the compatible almostcomplex structure J_g associated to ω and g via the Levi-Civita (Ehresmann) connection of the latter; J_g is also known as the Sasaki almost-complex structure.

We use the following type of domains for pseudo-holomorphic strips: fix $\xi \in (0, 1)$, $\xi \ll 1$, and consider a point $c = (c_1, ..., c_d) \in \mathbb{R}^d$. By definition, $\Delta_{d+1}(c_1, ..., c_d)$ is the subdomain of $(-\infty, \infty) \times [0, d+1] \subset \mathbb{C}$ given by removing d horizontal slits of vertical width ξ , each starting at the point (c_j, j) in the direction of $+\infty$, for $j \in [1, d]$. Each such horizontal slit is thus centered around the semi-infinite ray $[c_j, \infty) \times \{j\}$, see e.g. Figure 13. If the input position $c \in \mathbb{R}^d$ depends on a parameter $\varepsilon \in \mathbb{R}^+$ but c is implicitly understood (or arbitrary and not particularly relevant), then we denote such subdomain by $\Delta_{d+1}(\varepsilon)$. Such domains $\Delta_{d+1}(\varepsilon)$ are given the conformal structure inherited as subdomains of \mathbb{C} .

Definition 4.4. Let $(\mathbf{S}, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface and $L \subset T^*S$ a Betti Lagrangian adapted to (S, g). By definition, an ε -strip sequence $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ is a collection of J_g -holomorphic maps

$$u_{\varepsilon}: \Delta_m(\varepsilon) \longrightarrow T^*S$$

defined for all $\varepsilon \in \mathbb{R}_+$ small enough and such that:

(1) the image of u_{ε} is bounded by L and $F_{z_{\varepsilon}}$,

(2) area
$$(u_{\varepsilon}) < \varepsilon E$$

An additional piece of notation on such domains $\triangle_{d+1}(c)$: given a boundary component of $\partial \triangle_{d+1}(c)$ with both of its ends at $+\infty$, its *boundary minimum* of I the unique point with the smallest real part. Also, a subdomain in \triangle_{d+1} is said to be *horizontal* if it is of the form $[x_1, x_2] \times [y_1, y_2]$ with $[x_1, x_2] \times \{y_1\}$ and $[x_1, x_2] \times \{y_2\}$ lying on the boundary of \triangle_{d+1} . See Figure 13.

4.1.1. Combinatorial trees associated to domains \triangle_{d+1} . There is a (combinatorial) tree associated to each of the domains $\triangle_{d+1}(c_1, \ldots, c_d)$, constructed as follows.

Definition 4.5 (Trees associated to \triangle_{d+1}). Let $\triangle_{d+1}(c_1, \ldots, c_d) \subset \mathbb{C}$ be a domain, its associated tree $T := T(\triangle(c))$ is the rooted *d*-leaved tree whose interior vertices have valence at least three and satisfies:

¹³Note that it is nevertheless possible to upgrade Proposition 4.3 to show convergence to an actual (broken) D_4^- -tree. This would add even more technicality and pages to the manuscript and, for the sake of balance, we have not pursued this technical enhancement.



FIGURE 13. A depiction of the type of domain $\triangle_{d+1}(c)$ being used to describe the conformal structure. The domain is drawn in the strip with the slits highlighted in black, and the conformal structure in inherited from \mathbb{C} . The *x*-coordinate of the boundary minima is highlighted with a yellow vertical dashed line and the tree associated to the domain is depicted in green.

- (1) The vertices of T are in bijection with the values of $c \in \mathbb{R}^{d+1}$. These can be visually described by vertical rays passing through the boundary minima of $\Delta_{d+1}(c_1, \ldots, c_d)$.
- (2) The edges of T are in (natural) bijection with the connected components of the complement of such vertical rays in $\Delta_{d+1}(c_1, \ldots, c_d)$.
- (3) By definition, two edges of T share a vertex if and only if the corresponding components of those two edges are adjacent along the vertical ray corresponding to that vertex. \Box

See Figure 13 for an example of Definition 4.5. By definition, an open subdomain $D \subset \Delta_{d+1}$ is a *subdisk* if the part $\partial D \setminus (\partial \Delta_{d+1})$ of its boundary consist of vertical rays avoiding the boundary minima.¹⁴ As in Definition 4.5, subdisks $D \subset \Delta_{d+1}$ yield rooted (d+1)-leaved subtrees $T_D \subset T$ by introducing vertices at each of the edges whose corresponding components contain the vertical boundaries of ∂D . We denote by T/T_D the rooted (d+1)-leaved tree obtained by contracting the subtree $T_D \subset T$ associated to a subdisk $D \subset \Delta$.

We shall cover these domains \triangle by subdisk domains. We always implicitly assume that the vertical boundary segments of the components of such covers avoid the boundary minima, are all disjoint, and their intersections are also given by subdisk domains. In the same manner that a subdisk gave a subtree, an open cover $\triangle_m = D_0 \cup D_1$ by two subdisk domains D_0, D_1 gives an open cover of the associated tree: $T(\triangle) = T(D_0) \cup T(D_1)$. Note that D_0, D_1 are typically each disconnected.

4.1.2. Limits of ε -strip sequences. The goal of this subsection is to rigorously explain what it means for an ε -strip sequence, as in Definition 4.4, to converge to a trimmed D_4^- -tree. In particular, we refine the type of domain subdivision introduced in [Ekh07, Section 5.2], introduce adapted covers in Definition 4.6 and limits of ε -strip sequences in Definition 4.8.

¹⁴These subdisks are often disconnected, each connected component being topologically a disk.

For each branch point $b \in S$, we have fixed a neighborhood $U_b \subset S$ as in Section 3.3.1. We can and do choose such U_b to be disks of a certain radius r, denoting that by $U_b = U_b(r)$. Specifically, we fix two truncation parameters $\delta, \eta \in \mathbb{R}_+$ such that the disks $U_b(2\delta + \eta\delta)$ are disjoint and over each subdisk $U_b(2\delta)$, the piece of the Betti Lagrangian $L \cap T^*U_b(2\delta) \subset T^*S$ is holomorphic and the metric is the standard flat metric restricted from \mathbb{C} . We denote the union of such neighborhoods by

$$B_{\eta,\delta} := \bigcup_{b \in K_L} U_b (2\delta + \eta \delta).$$

By definition, a broken trimmed D_4^- -tree is the intersection of a disjoint union of D_4^- -trees in S with the domain $S \setminus B_{\eta,\delta}$ such that the flowlines entering each disk $U_b(2\delta + \eta\delta)$, if such flowlines exist, coincide with initial rays at $b \in S$. The tree is unbroken if there is a single D_4^- -tree component.

From now on, whenever we discuss the metric and the covariant derivative, we will assume that we are using the Sasaki metric, and the associated Levi-Civita connection. Given an ε -strip sequence $(u_{\varepsilon})_{\varepsilon}$, as in Definition 4.4, we want to understand its behavior in the adiabatic limit, i.e. as $\varepsilon \to 0$. As observed in [Ekh07, Lemma 5.6], $|\nabla u_{\varepsilon}| = O(\varepsilon)$ when restricted to a half-disk subdomain of fixed radius, that maps outside some fixed neighborhood of the caustics. However, near the caustics, we will not be guaranteed such a control. The idea is to subdivide the domains of each u_{ε} into open sets where we have some control of either $|\nabla u_{\varepsilon}|$ or the image of u_{ε} . As per usual application of gradient estimates, convergence for points where $|\nabla u_{\varepsilon}|$ remains uniformly $O(\epsilon)$ -bounded follows (c.f. [Flo89b, Theorem 2], [FO97, Proposition 9.7, Proposition 9.8]). The challenge is to find such subdivisions and control the behavior of the image when $|\nabla u_{\varepsilon}|$ blows up. The initial J_q -holomorphic strip, before performing any ε -scaling, has domain Δ_2 and maps to (T^*S, J_q) , with two punctures, as the boundary lies in L and a cotangent fiber F_z , for some $z \in S$. In order to obtain the subdivision, we will reparametrize that domain Δ_2 , as we ε -scale, by a sequence of ε -depending domains $\Delta_m(\varepsilon) := \Delta_m(c(\varepsilon)), m \ge 3$, where the parameters $c(\varepsilon)$ depend on the adiabatic parameter ε but the number $m \in \mathbb{N}$ does not. A first useful open cover for these domains is described as follows:

Definition 4.6 (Adapted covers for ε -strip sequences). Let $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ be an ε -strip sequence with z_{ε} converging to a point in $S \setminus B(2\delta + \eta\delta)$ as $\varepsilon \to 0$. By definition, an ε -sequence of open covers $\Delta_m(\varepsilon) = D_0(\varepsilon) \cup D_1(\varepsilon)$ is said to be adapted to $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ if the following holds for $\varepsilon \in \mathbb{R}_+$ small enough:

- (1) Each $D_i(\varepsilon)$ are subdisks, i = 1, 2.
- (2) The topology associated to the domains $D_i(\varepsilon)$ is independent of ε , i = 1, 2. That is, the order of the parameters $c(\varepsilon)$, the topology of the components of $D_i(\varepsilon)$, and the topology of their intersections (eventually) remains constant as $\varepsilon \to 0$.
- (3) $u_{\varepsilon}(D_0(\varepsilon))$ lies outside $B(\delta)$ and $|\nabla u_{\varepsilon}| = O(\varepsilon)$ for points in $D_0(\varepsilon)$.
- (4) $u_{\varepsilon}(D_1(\varepsilon))$ lies inside $B(2\delta)$.

In addition, a subdisk $W_0(\varepsilon) \subset D_0(\varepsilon)$ is said to give an adapted horizontal decomposition of $D_0(\varepsilon)$ if

- (i) W_0 has width $O(\log(\varepsilon^{-1}))$ and contains all the boundary minima contained in D_0 .
- (ii) The vertical boundary part of ∂W_0 is disjoint from the vertical boundaries of $D_i(\varepsilon)$, i = 1, 2, and its topology remains constant as $\varepsilon \to 0$.

A triple $(D_0(\varepsilon), D_1(\varepsilon), W_0(\varepsilon))$ is said to be adapted to $(u_\varepsilon, z_\varepsilon, \Delta_m(\varepsilon), E)$ if $\Delta_m(\varepsilon) = D_0(\varepsilon) \cup D_1(\varepsilon)$ is an adapted cover and $W_0(\varepsilon)$ is an adapted horizontal decomposition of $D_0(\varepsilon)$.

The geometric conditions on u_{ε} are Definition 4.6.(3),(4) and (i): on $D_0(\varepsilon)$ the norm of the derivative $|\nabla u_{\varepsilon}| = O(\varepsilon)$ remains controlled (constant with respect to the scaling measure), and the width condition $O(\log(\varepsilon^{-1}))$ on its subdomains $W_0(\varepsilon)$ ensures that the images $u_{\varepsilon}(W_0)$ converge to a point. In particular, the associated trees $T(\varepsilon)/T_{W_0(\varepsilon)}$ give models for the domain of a map that contracts the connected subdisks in W_0 to points.

Remark 4.7. The conditions Definition 4.6.(2) and (ii) imply that there is a fixed number of components for $D_0(\varepsilon)$, $D_1(\varepsilon)$ and $W_0(\varepsilon)$ in an adapted cover, independent of ε . In fact, the combinatorial trees associated to each such domains, as in Section 4.1.1, are well-defined and independent of ε . In particular, the combinatorial type of the tree Γ associated to $\Delta_m(\varepsilon)$ is independent of ε and so are the corresponding induced subtrees.

The adapted covers of Definition 4.6 are still not sufficient to conclude Proposition 4.3, i.e. that the ε -strip sequence converges to a trimmed D_4^- -tree. The remaining additional condition that we must require (and we will prove can be achieved) reads as follows:

Definition 4.8 (Limits of ε -strip sequences). The limit of an ε -strip sequence $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ is said to exist if there exists an adapted cover $(D_0(\varepsilon), D_1(\varepsilon), W_0(\varepsilon))$ such that the following two types of conditions hold:

- (1) There are three types of components of $D_0 \setminus W_0$: fiber, ghost or flowline. They are classified depending on their possible images under u_{ε} , as follows:
 - (i) (Fiber components) Some horizontal boundary components map to the cotangent fiber, with projections C^{∞} -converging to the point $\lim z_{\varepsilon}$ as $\varepsilon \to 0$.
 - (ii) (Ghost components) Horizontal boundaries map to the same sheets of L, with projections C^{∞} -converging to points.
 - (iii) (Flowline components) Horizontal boundaries map to different sheets of L if the domain has widths of size $O(\varepsilon^{-1})$ and the (projection of the) rescaling $u \circ (\varepsilon^{-1}) C^{\infty}$ -converges to a flowline. The flowlines are said to have zero length if the resulting limiting flowline is a point.
- (2) Let T be the tree associated to $\Delta_m(\varepsilon)$ and consider the (reduced) tree Γ^{red} obtained by removing the subtrees of $T/T_{W_0(\varepsilon)}$ corresponding to fiber components of $D_0 \setminus W_0$, contracting the subtrees corresponding to the ghost components of $D_0 \setminus W_0$, removing the subtrees

corresponding to components in D_1 , and removing any 2-valent vertices that connect the two flow-edges of the same index. Then, we require that Γ^{red} admits a set of edges such that:

- (i) (Edges map to initial D_4^- -rays) Each edge is such that there exists a vertical cut of the associated flowline component subdomain of $\Delta_m(\varepsilon)$ with u_{ε} mapping its right adjacent component inside $T^*B_{\eta,\delta}$, and the edge converges to one of the initial rays associated to a branch point, with the *inward* orientation.¹⁵
- (ii) (Maximality) For any flowline component $\Theta \subset D_0 \setminus W_0$ with non-zero flow, there exists an edge as in (i) that belongs to the right adjacent component of Θ .

In essence, proving Proposition 4.3 is tantamount to showing that, given the ε -strip sequence, there exists an adapted cover satisfying the conditions in Definition 4.8. The limiting object for the ε -strip sequence is technically not a flow-tree $F : \Gamma \longrightarrow S$ due to the presence of the ghost components and flowlines components with zero lengths in $\Delta_m(\varepsilon)$, cf. Definition 4.8.(1.*ii*) and (1.*iii*). For instance, an entire ghost component might map to a point in the adiabatic limit; these are said to be ghost edges of Γ . This matter is resolved combinatorially by contracting the edges of the tree Γ associated to $\Delta_m(\varepsilon)$ that correspond to these two types of components. The resulting tree is often referred to as the *reduced* tree, and one can effectively work with the reduced tree instead of the original Γ ; this is what is happening in Definition 4.8.(2).

If the limit of an ε -strip sequence exists, as in Definition 4.8, it is denoted by

$$\lim_{\varepsilon \to 0} (u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E) : \Gamma \longrightarrow S.$$

It is implicitly understood as a flow-tree with ghost components, so the map factors through its reduction Γ^{red} , as explicitly described in Definition 4.8.(2). In particular, the flowline edges of this limit are given by the adiabatic limit of the rescaling $u \circ (\varepsilon^{-1})$ along each flowline component of non-zero length.

Remark 4.9. The limit of an ε -strip sequence depends on the truncation parameters (δ, η) fixed initially and the corresponding trees $\Gamma = \Gamma_{\delta,\eta}$ might not be nested if one decreases the parameters $\eta, \delta \to 0$. That said, the associated reduced trees Γ^{red} can (and we will show will) be nested, in that $\Gamma^{red}_{(\delta,\eta)} \subset \Gamma^{red}_{(\delta',\eta')}$ for smaller truncation parameters (δ', η') . This is another technical reason for reduction.

4.2. **Proof of Proposition 4.3.** In this subsection we prove the convergence of ε -strip sequences to trimmed D_4^- -trees. In the initial setup, there are a number of technical steps which are mirroring [Ekh07, Section 5] and [Nho24, Section 5]; we thus simply refer the necessary parts of those manuscripts and continue focusing on the new part of the argument. Specifically, we must build an adapted cover as in Definition 4.8.

¹⁵Note that the spectral network \mathcal{W} is oriented opposite to the flowline direction.

First, in order to build such cover satisfying the component constraints from Definition 4.8.(1), we proceed as follows. By Lemmata 5.7, 5.13, 5.16 and 5.17 in [Ekh07], the domain $\Delta_m(\varepsilon)$ itself admits an adapted cover $(D_0(\varepsilon), D_1(\varepsilon), W_0(\varepsilon))$. Here the horizontal decomposition subdisk $W_0(\varepsilon)$ is obtained by considering subdisks which are neighborhoods of the boundary minima of the required width $O(\log(\varepsilon^{-1}))$.

Remark 4.10. (i) As far as we know, the construction of the domain-adapted cover in [Ekh07, Section 5] has a gap in the proof of [Ekh07, Lemma 5.4], in the claim that such balls of radius $\delta \varepsilon$ exist (his λ is our ε). It must be justified that there is a uniform upper bound on the boundary length of the images of u_{ε} outside of the union of the domains $T^*U_b(2\delta)$. This upper bound nevertheless exists, as it follows from the truncated reverse isoperimetric inequality that we prove in the appendix, cf. Theorem A.3.

(ii) Lemmatas 5.6 and 5.7 in [Ekh07] are stated only for exact Lagrangians. However, the only place the proof uses exactnesss is to establish the $O(\varepsilon)$ -estimate on energy. Since this condition is part of the definition of an ε -strip sequence, the proof goes through verbatim.

The components of $D_0(\varepsilon) \setminus W_0(\varepsilon)$ consist of horizontal subdomains of Δ_m of the three types required in Definition 4.8.(1). By [Nho24, Lemma 5.15], the fiber components C^{∞} -converge to the point lim z_{ε} . By [Nho24, Lemma 5.17], the ghost components also C^{∞} -converge to a point. By [Ekh07, Lemma 5.18] (see also [Nho24, Lemma 5.17]), the flowline components C^{∞} -converges to flowlines, after taking the ε -dependent reparametrization $u_{\varepsilon}(\varepsilon^{-1}s, \varepsilon^{-1}t)$. Therefore, the new contribution at this point is to construct the adapted cover such that Definition 4.8.(2) holds: controlling the behavior of the limit flowtree near the branch points.

Given the ε -strip sequence $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$, denote by Γ the tree associated to the domain $\Delta_m(\varepsilon)$, which is well-defined by the adapted cover constructed above, cf. Remark 4.7. Note that, as in [Nho24, Lemma 5.18], the ε -finiteness of energy implies that there cannot be flowline edges of infinite length. Thus, given the bound on the covariant derivative, any leaf in (the subtree associated to) the $D_0(\varepsilon)$ -region of Γ corresponds to either a ghost component or a fiber component. In particular, there can be no leaves that are flowline edges.Furthermore, the following lemma tells us that there exists at least one external D_1 -edge of Γ :

Lemma 4.11. There exists at least one external edge of Γ that corresponds to a horizontal component in $D_1(\varepsilon)$.

Proof. Suppose by contradiction that there is no such an external edge. By construction, this is only if the image of u_{ε} lies outside some fixed neighbourhood of the branch points, for small enough ε . It follows that for the given subsequence, the puncture removal step was unnecessary and $|\nabla u_{\varepsilon}| = O(\varepsilon)$. However, there is then a single horizontal strip component which necessarily must be a fiber component. Therefore, the image of u_{ε} converges to the point $\lim z_{\varepsilon}$ and that implies that the geometric energy must have been zero, which a contradiction.

Let us cut the domain $\Delta_m = \Delta_m(\varepsilon)$ along the vertical rays through the boundary minima of Δ_m that connect to a horizontal boundary component mapping to the fiber. We denote by $\bar{\Delta}_{m_1}, ..., \bar{\Delta}_{m_n}$ the resulting right-adjacent components and by Γ_{m_i} the induced tree in Γ associated to each such components. By the above, the $D_0(\varepsilon)$ -leaves of Γ_{m_i} are all ghost components.

Consider the set of vertices contained in Γ_{m_i} that are terminal points of a flowline edge of nonzero length; such a vertex is not 1-valent. These vertices are ordered by their x-coordinate, and a maximal vertex (for a given m_i) refers to such a vertex with highest x-coordinate within Γ_{m_i} . In particular, given two maximal vertices v, w, their right-adjacent components are necessarily disjoint. Let v be a maximal vertex, and let $\Theta_v = \Theta_v(\varepsilon)$ be the corresponding horizontal domain in $D_0(\varepsilon) \setminus W_0(\varepsilon)$. Then the component Θ_v associated to a maximal vertex satisfies the following:

Lemma 4.12. Let $v \in \Gamma$ be a maximal vertex as above. Then there exists a branch point $b(v) \in S$ such that the right-adjacent components of $\Theta_v(\varepsilon)$ are mapped into the neighborhood $T^*U_{b(v)}((2 + \eta)\delta)$.

Proof. Since $\Delta_m = D_0(\varepsilon) \cup D_1(\varepsilon)$ is adapted, the components in $D_1(\varepsilon)$ map into the region $T^*B_{\eta,\delta} \subset T^*S$. By maximality, it also follows that the horizontal domains of $D_0(\varepsilon) \setminus W_0(\varepsilon)$ right-adjacent to Θ_v are either ghosts, fiber components or flowline components of zero length. Since there are finitely many such components, their images C^{∞} -converge to points, and by construction the subdisk W_0 also converges to points. Since the neighborhoods $U_b(2\delta + \eta\delta)$ are all disjoint and the right-adjacent component to Θ_v is connected, it must map into one connected component $T^*U_b(2\delta + \eta\delta)$ of $T^*B_{\eta,\delta}$, which determines the branch point $b(v) = b \in S$.

Now we arrive at the crux of the argument: we must show that $\Theta_v(\varepsilon)$, possibly after refining with vertical cuts, is such that its right-adjacent component not only maps to $T^*U_{b(v)}(2\delta + \eta\delta)$ but does so converging near $T^*\partial(U_{b(v)}(2\delta + \eta\delta))$ to an initial flowline ray of the D_4^- -branch point $b(v) \in S$ in the adiabatic limit. This is the content of the following:

Proposition 4.13 (Behavior near D_4^- -points). Let $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ be an ε -strip sequence, $v \in \Gamma$ a maximal vertex and $b(v) \in S$ its associated branch point. Then, after possibly taking a subsequence, there exists a sequence of vertical cuts v_{ε} for the domain $\Theta_v(\varepsilon)$ such that the restriction $u_{\varepsilon}|_{\Theta_v^{\rightarrow}}$ of u_{ε} converges to one of the initial flowline rays of b(v), where Θ_v^{\rightarrow} denotes the right adjacent component of $\Theta_v(\varepsilon)$ after the cuts.

Proof. By construction, the open set $U_{b(v)}(2\delta + \eta\delta)$ is such that the metric g is flat, the associated Sasaki metric coincides with the standard flat metric on $\mathbb{C}^2 \cong T^*\mathbb{C}$, and the Betti Lagrangian $L \subset T^*S$ above this open set splits into the holomorphic cusp component and the smooth sheet components. Consider the limiting flowline γ_v associated to the component Θ_v . Since it is not a ghost edge, fiber component or zero-length flowline, it must (eventually) enter $B_{\eta,\delta}$, specifically the connected component containing $b(v) \in S$. Choose $\eta' \in (\eta/2, \eta)$ such that γ_v is transverse to the boundary $\partial B_{\eta',\delta}$ and let v(z) be the final intersection point at which γ_v is oriented inward. After passing to a subsequence, there is a sequence of vertical rays v_{ε} for Θ_v converging to the point v(z); fix such a sequence of rays. We claim that the energy $\int \lambda^*_{\mathbb{C}}(\varepsilon^{-1}(u_{\varepsilon}|_{v_{\varepsilon}}))$ of the ε -scaled u_{ε} converges to zero along these vertical cuts.

Indeed, write the restriction $u_{\varepsilon}|_{v_{\varepsilon}} = (z_{\varepsilon}, w_{\varepsilon})$ in coordinates and apply the gradient estimate $|\nabla z_{\varepsilon}| = O(\varepsilon)$ to obtain

(22)
$$\left| \int_{v_{\varepsilon}} \lambda_{\mathbb{C}}^* (\varepsilon^{-1} u_{\varepsilon}) |_{v_{\varepsilon}} \right| \le m \cdot \sup \left| \varepsilon^{-1} w_{\varepsilon} \right| \cdot \left| \nabla z_{\varepsilon} \right|.$$

Here the quantity m appears in the upper bound because the height of Θ_v is at most m. Since $|\varepsilon^{-1}w_{\varepsilon}|$ is uniformly bounded and $|\nabla z_{\varepsilon}|$ is of order $O(\varepsilon)$, the integral on the left converges to zero in the adiabatic limit, as claimed.

Let $v_{\varepsilon}(\text{top})$ and $v_{\varepsilon}(\text{bottom})$ denote the top and the bottom end-points of each vertical cut v_{ε} in Θ_{v} , $\Delta_{v_{\varepsilon}}$ the right-adjacent components to v_{ε} and $\partial^{h}(\Delta_{v_{\varepsilon}})$ the horizontal boundary of $\Delta_{v_{\varepsilon}}$. Since the restriction $u_{\varepsilon}|_{\Delta_{v_{\varepsilon}}}$ is J_{g} -holomorphic, with $g = g_{\text{st}}$, the pointwise symplectic area $\omega_{\mathbb{C}}^{*}u_{\varepsilon}$ over $\Delta_{v_{\varepsilon}}$ is real and positive, except possibly at finitely many points. Therefore, the symplectic area

(23)
$$\int_{\Delta_{v_{\varepsilon}}} \omega_{\mathbb{C}}^* u_{\varepsilon} \in \mathbb{R}_{+}$$

is real and positive, which will be used momentarily in Eq. (24). Independently, the restriction $u|_{\partial^h(\Delta_v)}$ extends to a connected curve in L and, since Θ_v is a flowline component, $\partial^h(\Theta_v)$ maps to distinct sheets of the Betti Lagrangian L. Given that the smooth sheets of L are disjoint over $B_{\eta,\delta}$, Lemma 4.12 implies that $u|_{\partial^h(\Delta_v)}$ must map to the holomorphic cusp component. Thus the flowline γ_v is a cusp-cusp flowline.

Let us finally deduce that γ_v coincides with an initial trivalent ray at b, as follows. By applying Stokes' theorem, the ε -scaled energy reads

$$\varepsilon^{-1} \int_{\Delta_{v_{\varepsilon}}} \omega_{\mathbb{C}}^* u_{\varepsilon} = \varepsilon^{-1} \int_{\partial^h \Delta_{\varepsilon}} \lambda_{\mathbb{C}}^* \partial u_{\varepsilon} - \varepsilon^{-1} \int_{v_{\varepsilon}} \lambda_{\mathbb{C}}^* u_{\varepsilon}|_{v_{\varepsilon}} = \pm \Big((z(v_{\varepsilon}(\operatorname{top}))^{3/2} - (z(v_{\varepsilon}(\operatorname{bottom}))^{3/2} \Big) - \varepsilon^{-1} \int_{v_{\varepsilon}} \lambda_{\mathbb{C}}^* u|_{v_{\varepsilon}}.$$

By Eq. (22), the second term on the right hand side of Eq. (24) converges to zero in the adiabatic limit $\varepsilon \to 0$. Since $(z(v_{\varepsilon}(\text{top})) \text{ and } (z(v_{\varepsilon}(\text{bottom})) \text{ converge to } z(v), \text{ the right hand side of Eq. (24)} \text{ converges to } \pm z(v)^{3/2}$. By Eq. (23), the left hand side of Eq. (24) is real and positive and thus $\pm z(v)^{3/2}$ is also real and positive. This can only occur if the (image of the) point v lied on one of the initial rays for its associated branch point b(v). Since the cusp-cusp flowlines form a singular foliation, the leaves of which lie outside b(v), except for the three trivalent rays, we see that the restriction $u_{\varepsilon}|\Theta_{v}$ must converge to one of the initial rays coming out of b(v) in the adiabatic limit.

ROGER CASALS AND YOON JAE NHO

Let us apply Proposition 4.13 to each maximal vertex v^i associated to the subdomains Δ_{m_i} for the initial adapted cover $(D_0(\varepsilon), D_1(\varepsilon), W_0(\varepsilon))$ at the beginning of the argument. Proposition 4.13 implies that these adapted covers can be refined to satisfy the condition in Definition 4.8.(2.i). The maximality condition Definition 4.8.(2.ii) follows from our choice of domains Δ_{m_i} and maximal vertices. In consequence, there exists an adapted cover for the given ε -sequence such that its limit exists.

Let $\{\Gamma\}$ be the set of all D_4^- -trees in \mathcal{W} with the univalent edges given by the initial edges of $(u_{\varepsilon}, z_{\varepsilon}, \Delta(\varepsilon), E)$, with energy less than E. The energy and the endpoint constraint imply that the set of such trees must be finite. By construction, each component $\lim_{\varepsilon \to 0} (u_{\varepsilon}, z_{\varepsilon}, \Delta(\varepsilon), E)$ must be a trimmed D_4^- -tree in $\{\Gamma\}$, since the initial edges in Proposition 4.13 are directed inward. Furthermore, the internal vertices of such trees can only appear at the joints of $\mathcal{W}(E)$. It follows that the D_4^- -tree is contained entirely in $\mathcal{W}(E)$. Furthermore, in the case z lies in the interior of $\mathcal{W}(E)$, the tree cannot be broken because their soliton classes cannot concatenate. Thus we have established Proposition 4.3.

4.3. **Proof of Theorem 4.2.** For Part (1), suppose that the point $z \in W^c$ is not uniformly diskfree. By definition, there exists an ε -strip sequence $(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ with $z = \lim z_{\varepsilon}$. Consider the spectral subnetwork $W(E) \subset W$ with energy at most E and fix the truncation parameters $\delta, \eta \in \mathbb{R}_+$ as in the proof of Proposition 4.3, so that the only vertices of W(E) that lie in $B_{\eta,\delta}$ are the initial trivalent vertices. Proposition 4.3 implies that, after possibly considering a subsequence, the limit $\lim(u_{\varepsilon}, z_{\varepsilon}, \Delta_m(\varepsilon), E)$ exists and it is a trimmed D_4^- -tree through z with energy below E. Furthermore, such a D_4^- tree must be entirely contained in W(E). It thus follows that $z \in W(E)$, which contradicts the initial hypothesis $z \in W^c$. Part (2) follows analogously.

5. FAMILY FLOER AND NON-ABELIANIZED LOCAL SYSTEMS

Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface and $L \subset (T^*S, \omega_{st})$ a Betti Lagrangian of rank n. The object of this section is two-fold. First, in the exact case, to construct two functors \mathbb{F} and Φ_W that carry GL₁-local systems on the Betti Lagrangian to GL_n-local systems on the base S and show that they are equivalent. Second, finish the proof of Theorem 2, which concerns both exact and meromorphic Betti Lagrangians. The construction of the functors depends on the existence of a compatible spectral network $\mathcal{W} \subset S$. In a nutshell:

- (1) The functor $\Phi_{\mathcal{W}} : \operatorname{Loc}_{1}^{\dagger}(L) \longrightarrow \operatorname{Loc}_{n}^{\dagger}(S)$ is constructed by considering a variant of D_{4}^{-} -trees, with the parallel transport of image local system being modified according to the interaction of a given path in S with the spectral network \mathcal{W} . It is constructed in Section 5.1.¹⁶
- (2) The functor $\mathbb{F} : \operatorname{Loc}_1^{\dagger}(L) \longrightarrow \operatorname{Loc}_n^{\dagger}(S)$ is constructed by considering continuation maps between the Family Floer cohomology groups of the Lagrangian L with the cotangent fibres

¹⁶Technically, $\Phi_{\mathcal{W}}(V)$ has only parallel transport defined for \mathcal{W} -paths. In contrast, $\mathbb{F}(V)$ has parallel transport for all paths.

of T^*S . It is constructed in Section 5.2.

(3) A natural transformation between these two functors is established in Section 5.3. The proof of Theorem 2 is then completed in Section 5.4.

Note that the projection $L \longrightarrow S$, given by restricting to L the projection $T^*S \longrightarrow S$ to the zero section, is a branch cover and thus the push-foward of a local system on L does not give a local system on S, just a constructible sheaf on S. The point of these two functors above is that they conceptually explain how to correct such constructible sheaf on S (coming from the push-forward of a local system on L) back to a local system on S. Each approach presents a different type of correction: Φ_W uses the detour paths coming from W to correct the lack of flatness, in line with [GMN13b; GMN13a; GMN14], whereas \mathbb{F} uses pseudo-holomorphic strip counts. As one of the main themes of this article is the comparison between spectral networks and Floer theory, we shall explain how the former is an expression of the adiabatic limit of the latter.

In the above, $\operatorname{Loc}_k(M)$ denotes the dg-category of rank-k local systems of \mathbb{C} -modules on a smooth manifold M, and $\operatorname{Loc}_k^{\dagger}(M)$ denotes the dg-category of rank-k twisted local systems of \mathbb{C} -modules on a smooth manifold M. We recall that a twisted local system on M is a local system on the unit cotangent bundle $T^{\infty}M$ of M whose monodromy around the spherical cotangent fiber is given by $-\operatorname{Id}$. The unit cotangent bundle will also be denoted \mathring{M} and its projection to the base by $\pi : \mathring{M} \longrightarrow M$. There are equivalent descriptions of local systems of \mathbb{C} -modules, e.g. as locally constant sheaves, as modules $\operatorname{Fun}(\Pi(M), \mathbb{C}$ -mod) over the fundamental ∞ -groupoid $\Pi(M)$ of M, or as \mathbb{C} -modules over chains $C_{-*}(\Omega M; \mathbb{C})$ on the base loop space; e.g. cf. [Lur17, Appendix A]. For an open surface $M = K(\pi_1(M), 1)$ and $\mathring{M} \cong M \times S^1$, so (twisted) local systems effectively translates to representations $\pi_1(M, m) \longrightarrow \operatorname{GL}(V)$ of the fundamental group of M, where V is a \mathbb{C} -vector space and $m \in M$ a base point, which will be implicitly understood. Such representations are equivalent to modules over the group ring $\mathbb{C}[\pi_1(M)]$, and the functors \mathbb{F} and Φ_W will essentially be described as

$$\mathbb{F}, \Phi_{\mathcal{W}} : \mathbb{C}[\pi_1(\mathring{L})] \operatorname{-mod} \longrightarrow \mathbb{C}[\pi_1(\mathring{S})] \operatorname{-mod}.$$

In fact, both \mathbb{F} , Φ_W will be described locally (for short paths) and thus we shall be considering paths with varying endpoints. So our precise choice for describing $\operatorname{Loc}_k(M)$ will be that of the fundamental ∞ -groupoid. That is, for this manuscript, a local system $\mathcal{L} \in \operatorname{Loc}_k(M)$ consists of the data of a stalk \mathcal{L}_m at every point $m \in M$, where \mathcal{L}_m is a k-dimensional \mathbb{C} -vector space, and an isomorphism $\mathcal{L}(\gamma) : \mathcal{L}_x \longrightarrow \mathcal{L}_y$ for any continuous path $\gamma : [0,1] \longrightarrow M$, $\gamma(0) = x, \gamma(1) = y$, which only depends on the homotopy class of γ . The structure group of the associated bundles for these local systems will always be $\operatorname{GL}_n(\mathbb{C})$, if we are in a given rank n.

Remark 5.1. (1) The branch cover $\pi : L \longrightarrow S$ lifts to a branch cover $\mathring{\pi} : \mathring{L} \longrightarrow \mathring{S}$. In this case, there are identifications $\mathring{L} \cong L \times S^1$, $\mathring{S} \cong S \times S^1$, and the latter map reads $\mathring{\pi} = (\pi, id)$.

(2) Intuitively, the functors $\mathbb{F}, \Phi_{\mathcal{W}} : \mathbb{C}[\pi_1(\mathring{L})]$ -mod $\longrightarrow \mathbb{C}[\pi_1(\mathring{S})]$ -mod will be described by giving a group algebra morphism $\mathbb{C}[\pi_1(\mathring{S})] \longrightarrow \mathbb{C}[\pi_1(\mathring{L})]$. That is, they are described by explaining how to lift paths in \mathring{S} to paths in \mathring{L} . Rigorously, a path in \mathring{S} with two endpoints $z, z' \in \mathring{S}$ will lift to a collection of paths in \mathring{L} whose endpoints are allowed to be in a set of 2n points (the pre-images of z, z'). Such paths are not morphisms in the fundamental groupoid of \mathring{L} : they would roughly be morphisms between direct sums of objects, which only make sense once we consider \mathbb{C} -modules. This is a technical reason to describe the functors between the module categories.

(3) For an open surface M, there is a dg-equivalence $Loc_k(M) \simeq Loc_k^{\dagger}(M)$ once a spin structure is chosen; cf. Appendix A.3 or [CL22, Appendix B.2]. Since protected spin characters are not the scope of this article, this effectively implies that \mathbb{F}, Φ_W can be understood as functors carrying a GL_1 -local system on the Betti Lagrangian $L \subset T^*S$ to a rank GL_n -local system on the base S. \Box

Due to Remark 5.1.(2), we adopt the following notational conventions. First, $\pi_1(\mathring{S})$ will stand for the relative homotopy classes $\pi_1(\mathring{S}, \mathfrak{t}_S)$, where \mathfrak{t}_S is a collection of points in S, often understood implicitly by context. Second, consider the functor $f^* : \operatorname{Loc}^{\dagger}(L) \longrightarrow \operatorname{Loc}^{\dagger}(S)$ that is the pull-back of a group algebra morphism $f : \mathbb{C}[\pi_1(\mathring{S})] \longrightarrow \mathbb{C}[\pi_1(\mathring{L})]$ and $V \in \operatorname{Loc}^{\dagger}(L)$. Given a path $\gamma \subset S$ with

$$f(\gamma) = \sum_{i=1}^{g} a_i \tau_i, \quad \tau_i \in \pi_1(\mathring{L}, \mathfrak{t}_L), a_i \in \mathbb{C}$$

a linear combination of relative homotopy classes in $\pi_1(\mathring{L}, \mathfrak{t}_L)$, with the points in \mathfrak{t}_L projecting to those in \mathfrak{t}_S , we denote

$$f^*(V)(\gamma) = \sum_{i=1}^g a_i V(\tau_i)[\tau_i]$$

to indicate that the parallel transport $f^*(V)(\gamma)$ of $f^*(V)$ along γ is given by

$$\sum_{i=1}^{g} a_i V(\tau_i)$$

and zero otherwise, and that the map f^* in fact admitted the homotopy refinement f which (further) keeps track of the relative homotopy classes $[\tau_i]$, a linear combination of which gives the image of γ . This notation is used in the following subsections, c.f. e.g. Definition 5.6 and Corollary 5.22.

Remark 5.2. Alternatively, we can add the dependence on base points \mathfrak{t}_S and \mathfrak{t}_L explicitly in the notation, as in Section 3.5. Such notation becomes rather excessive and we have leaned towards having such endpoints of relative homotopy classes implicitly understood. This is not a particular challenge: the sets are always given a point in S, or \mathring{S} , where we will consider the cotangent fiber, and its lifts in the Betti Lagrangian $L \subset T^*S$, or \mathring{L} .

5.1. The functor Φ_{W} : non-abelianization of local systems. The goal of this section is to define the functor

$$\Phi_{\mathcal{W}}: \operatorname{Loc}_{1}^{\dagger}(L) \longrightarrow \operatorname{Loc}_{n}^{\dagger}(S),$$

using a finite spectral network \mathcal{W} compatible with a Betti Lagrangian $L \subset T^*S$. It formalizes the constructions in [GMN13b; GMN13a; GMN14], where such types of local systems were studied in the context of providing coordinates for the moduli of flat connections on Riemann surfaces, see [GMN13b, Section 10.1]. In particular, the images $\Phi_{\mathcal{W}}(V)$ of (twisted) local systems $V \in \operatorname{Loc}_1^{\dagger}(L)$ encode the formal generating functions of framed 2d-4d states from [GMN13b] after evaluating the spin character at y = 1. The local systems $\Phi_{\mathcal{W}}(V)$ will be described by giving their stalks $\Phi_{\mathcal{W}}(V)_z$ at $z \in \mathring{S}$ and the associated parallel transport isomorphisms $\Phi_{\mathcal{W}}(V)(\rho) : \Phi_{\mathcal{W}}(V)_z \longrightarrow \Phi_{\mathcal{W}}(V)_{z'}$ along certain paths $\rho : [0, 1] \longrightarrow \mathring{S}$ with $\rho(0) = z, \rho(1) = z'$.

Remark 5.3. For notational ease, we often denote by $z \in \mathring{S}$ points in \mathring{S} , even if z also denoted points in S. Given a regularly parametrized path in S that starts at $\rho(0) = z \in S$, the unit cotangent lift via the velocity vector $\rho'(0)$ at z gives a unique point in \mathring{S} projecting to z. That is the type of point we also denote $z \in \mathring{S}$, as it is uniquely determined by and determines $z \in S$ once the path is chosen.

The stalks $\Phi_{W}(V)_{z}$ at $z \in \mathring{S}$ are simple to describe for points $z \in \mathring{S}$ with $\pi(z) \notin K_{L}$ not a branch point. By definition, we choose

$$\Phi_{\mathcal{W}}(V)_z := V_{z_1} \oplus \cdots \oplus V_{z_n},$$

where $z_1, \ldots, z_n \in \mathring{L}$ are the lifts to \mathring{L} of the *n*-preimages of $\pi(z) \in S$ via $\pi : L \longrightarrow S$.¹⁷ The requirement that $\Phi_W(V) \in \operatorname{Loc}^{\dagger}(S)$ determines the stalks at all $z \in \mathring{S}$. If $V \in \operatorname{Loc}_1^{\dagger}(S)$, V_{z_i} is a rank-1 \mathbb{C} -module and $\Phi_W(V)_z$ is therefore a rank- $n \mathbb{C}$ -module.

The parallel transport isomorphisms $\Phi_{\mathcal{W}}(V)(\rho) : \Phi_{\mathcal{W}}(V)_z \longrightarrow \Phi_{\mathcal{W}}(V)_{z'}$ are more interesting and depend on the spectral network \mathcal{W} . In order to describe them, we shall define parallel transport for an open dense set of paths ρ that interact generically with \mathcal{W} , as follows.

Definition 5.4 (W-adapted paths). Let $I \subset \mathbb{R}$ be a finite closed interval. An immersed path $\rho: I \to \mathring{S}$ is:

- (i) W-adapted, if its projection $\pi(\rho): I \longrightarrow S$ is immersed and transverse to $\mathcal{W} \subset S$.
- (ii) free, if the image of $\pi(\rho)$ is embedded and disjoint from \mathcal{W} .
- (iii) short, if its the unit-velocity lift of an embedded path in S intersecting \mathcal{W} transversely and exactly once.

Any relative homotopy class with endpoints in \mathcal{W}^c is represented by a \mathcal{W} -adapted path, as in Definition 5.4. Given a \mathcal{W} -adapted path, we can trivialize the sheets of L along the path $\pi(\rho)$: we

 $^{^{17}}$ Since V is a twisted local system, all choices of lifts are isomorphic.

denote by ρ^i be the lifts of ρ to \check{L} , $i \in [1, n]$; they are given by lifting $\pi(\rho)$ to n lifts in L and then considering the unit-velocity lifts to \mathring{L} .

Let $\rho : [-\varepsilon, \varepsilon] \to \mathring{S}$ be a short W-adapted path (with its projection) intersecting W at a point $z = \rho(0)$ in an (ij)-wall \mathfrak{w} . Let $\mathfrak{S}(z; \mathfrak{w}) := {\mathfrak{s}(z; \mathfrak{w})}$ be the set of soliton classes at z: each $\mathfrak{s}(z; \mathfrak{w})$ defines a regular homotopy class with relative endpoints z^j and z^i , oriented from the former to the latter. We denote by $\mathring{\mathfrak{s}}(z)$ be the unit-velocity lift of $\mathfrak{s}(z; \mathfrak{w})$ to \mathring{L} and set the signs $\operatorname{sgn}(\rho)(z) = 1$ if the frame $\langle W'(z), \rho'(0) \rangle$ is positively oriented, and -1 otherwise. We need the following deformations for short paths. If $\operatorname{sgn}(\rho)(z) = 1$, then we deform $\pi(\rho)|_{[-\varepsilon,0]}$ near 0 so that it becomes tangential to W at 0 and directed against the orientation on W: denote by ρ_- the lift to \mathring{S} of such resulting path. Similarly, let ρ_+ be the lift to \mathring{S} of a deformation of $\pi(\rho)|_{[0,\varepsilon]}$ near 0 so that it becomes tangential to W at 0, now in the same direction as that of W. The following definition formalizes the detour paths in the physics literature, cf. e.g. [GLM15, Section 3.2] or [Lon18].

Definition 5.5 (Detour paths). Let $\rho : [-\varepsilon, \varepsilon]$ be a short \mathcal{W} -adapted path with $\operatorname{sgn}(z) = +1$ and ρ_{\pm} its two deformations. By definition, the regular homotopy class $[\mathfrak{s}(\rho)]$, with relative endpoints $\rho^i(-\varepsilon)$ to $\rho^j(\varepsilon)$, is given by

$$[\mathring{\mathfrak{s}}(\rho)] := \left[\left(\rho_+ \right)^i \circ \mathring{\mathfrak{s}}(z) \circ \left(\rho_- \right)^j \right].$$

For $\operatorname{sgn}(z) = -1$, the regular homotopy class $[\mathring{\mathfrak{s}}(\rho)]$ is defined to be $[\mathring{\mathfrak{s}}(\rho)] := [\overline{\rho}_{-}^{i} \circ \mathring{\mathfrak{s}}(z) \circ \overline{\rho}_{+}^{i}]$, where $\overline{\rho}$ denotes the inverse path.

Intuitively, the representative $(\rho_+)^i \circ \mathfrak{s}(z) \circ (\rho_-)^j$ of $[\mathfrak{s}(\rho)]$ in Definition 5.5 is (the lift of) a path that starts as ρ , when it hits z it takes a detour going around the soliton $\mathfrak{s}(z;\mathfrak{w})$, and then goes back to z and continues the path ρ . These detour paths are the crucial correction so as to define Φ_W in a manner that $\Phi_W(V)$ is indeed a local system on \mathring{S} , and not merely a constructible sheaf. For instance, at a D_4^- -singularity, these detour paths correct the non-trivial monodromy around a branch point to be the identity. Specifically, the parallel transport isomorphisms for $\Phi_W(V)$ are defined as follows:

Definition 5.6 (Parallel transport for Φ_{W}). Let $V \in \text{Loc}_{1}^{\dagger}(L)$, (W, μ) and indexed spectral network, and $\rho : I \longrightarrow \mathring{S}$ a W-adapted path. By definition, the parallel transport isomorphisms $\Phi_{W}(V)(\rho)$ are defined as follows:

(i) If ρ is free, then

$$\Phi_{\mathcal{W}}(V)(\rho) := \sum_{i=1}^{n} V(\rho^{i})[\rho^{i}],$$

where the sum runs over the *n* lifts ρ^i of ρ to L, $i \in [1, n]$.
(ii) If ρ is short, with its projection intersecting \mathcal{W} at z, then

$$\Phi_{\mathcal{W}}(V)(\rho) := \sum_{i=1}^{n} V(\rho^{i})[\rho^{i}] + \sum_{\mathfrak{s}(z)\in\mathfrak{S}(z;\mathfrak{w})} \mu(\mathfrak{s}(z))[\mathfrak{s}(\rho)]V(\mathfrak{s}(\rho)),$$

where the second sum runs over all the soliton classes $\mathfrak{s}(z; \mathfrak{w})$ associated to $z \in \mathcal{W}$.

(iii) If ρ is given as a composition $\rho = \rho_1 \circ \rho_2 \circ \ldots \circ \rho_n$, each ρ_i either short or free, then

$$\Phi_{\mathcal{W}}(V)(\rho) = \Phi_{\mathcal{W}}(V)(\rho_1) \circ \Phi_{\mathcal{W}}(V)(\rho_2) \circ \ldots \circ \Phi_{\mathcal{W}}(V)(\rho_k).$$

Definition 5.6 allows us to define parallel transport for all W-adapted paths, by declaring parallel transport along a homotopy class to be independent of a representative. For instance, if a class is represented by a free path, then Definition 5.6.(i) applies. The contributions from the second summand of Definition 5.6.(ii) are said to come from *detour paths*, in line with Definition 5.5. It is possible to verify combinatorially by hand that Definition 5.6 yields a well-defined local system, independent of the representative of a given (twisted) homotopy class of a path, see e.g. [GMN13b, Section 5.6]. Given the Floer-theoretic framework we develop, this property will be rather a consequence of interpreting W in terms of pseudo-holomorphic strips.

Remark 5.7. A brief explanation of the homotopy invariance from a Floer-theoretic viewpoint is as follows. Given a W-adapted path ρ , we can consider the exact Lagrangian isotopy $T^*_{\pi(\rho(t))}S$ given by moving the cotangent fibres along the path ρ . After an appropriate choice of (a family of) almost complex structures, we consider continuation strips with boundary on L and $T^*_{\pi(\rho(t))}S$ and, using the data of $\rho(t)$ and the (twisted) local system V, we orient the moduli space of such strips. The cobordism type of such moduli space depends only on the twisted isotopy class of ρ , and thus invariance follows once one shows that the formal sum of the boundaries of such continuation strips coincides with $\Phi_W(V)(\rho)$.

Note that Definition 5.6 suffices to define parallel transport for paths with endpoints in the dense complement \mathcal{W}^c . This is in line with the current understand of wall-crossing phenomena and related constructions in the literature. In contrast to see, the Floer theoretic parallel transports are well-defined for paths with endpoints on \mathcal{W} .

Due to the presence of Reeb chords at infinity, given by the data Λ of the Betti surface, we also have an *infinite* version of W-adapted paths. By definition, an infinite path $\rho : (-\infty, \infty) \to \mathring{S}$ is said to be W-adapted if ρ is W-adapted over any interval $[-N, N], N \in \mathbb{N}$, with a uniformly finite number of intersection points with W, and its projection $\pi(\rho(t)), t \to \pm \infty$ asymptotes to radial Reeb chord rays near a subset of marked points. The discussion above all apply to these types of paths as well.

Given a Reeb chord \mathfrak{r} of a Legendrian link in Λ at a marked point $m \in \mathfrak{m}$. By Proposition 2.19, there exists a small sectorial neighborhood

$$S_{\varepsilon}(\mathfrak{r}) \cong \{ (e^{i\theta}, r) \in S^1 \times \mathbb{R} : \theta \in [\theta_0 - \varepsilon, \theta_0 + \varepsilon], r > R \},\$$

with $R \in \mathbb{R}_+$ large enough, such that the Reeb chord \mathfrak{r} corresponds to θ_0 and $r \to \infty$ and $S_{\varepsilon}(\mathfrak{r})$ contains only the edges of \mathcal{W} that are asymptotic to \mathfrak{r} . Let $\rho(\mathfrak{r}) : (-\infty, \infty) \longrightarrow S$ be the infinite path given by smooth the boundary $\partial S_{\varepsilon}(\mathfrak{r})$, with $\rho(\infty) = \rho(-\infty) = m$, which we can assume intersects transversely \mathcal{W} . Such infinite paths $\rho(\mathfrak{r})$ are said to be the Reeb paths of \mathfrak{r} . The following result establishes that the morphism of the functor $\Phi_{\mathcal{W}}$ over a Reeb path, even with its homotopy refinement, is an invariant of the Hamiltonian isotopy class of $L \subset (T^*S, \lambda_{st})$, up to compactly supported Hamiltonian isotopy.

Theorem 5.8 (Invariance of soliton classes for Reeb paths). Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, \mathfrak{r} a Reeb chord of a component of $\boldsymbol{\Lambda}$ near a marked point with Reeb path $\rho(\mathfrak{r})$, and $L \subset (T^*S, \lambda_{st})$ an exact Betti Lagrangian endowed with a compatible spectral network $W \subset S$. Then $\Phi_W(V)(\rho(\mathfrak{r}))$ is an invariant of L, for any $V \in Loc^{\dagger}(L)$ and up to compactly supported Hamiltonian isotopy.

Theorem 5.8 will be a consequence of the results we establish in Section 5.2 and Section 5.3, c.f. Theorem 5.24.

5.2. The functor \mathbb{F} : Family Floer and local systems. The goal of this section is to define a functor

$$\mathbb{F}: \operatorname{Loc}_1^{\dagger}(L) \longrightarrow \operatorname{Loc}_n^{\dagger}(S)$$

using Floer-theoretic methods, and independently of a choice of a spectral network in S. For that, we use a version of Family Floer cohomology. Family Floer cohomology was initially pioneered by K. Fukaya, see e.g. [Fuk02], along with [Abo14] and references therein. In brief, adapting this framework to exact Betti Lagrangians allows us to define a type of Floer cochain complexes $CF^{\bullet}(T_z^*S, L)$, recording Lagrangian intersections between $L \subset T^*S$ and the cotangent fiber $T_z^*S \subset T^*S$, and continuation maps between these as $z \in S$ varies. Such maps are defined by studying certain (cobordism type of) moduli spaces of pseudo-holomorphic strips. In particular, given a path between $z, z' \in S$, it is possible to construct a parallel transport from $CF^{\bullet}(T_z^*S, L)$ to $CF^{\bullet}(T_z^*S, L)$, which is what ultimately yields the functor \mathbb{F} . This section presents the necessary details to construct Family Floer cohomology for Betti Lagrangians and use it to define \mathbb{F} . For the rest of the section, we will assume that L is an exact Betti Lagrangian.

5.2.1. Framework for Floer cohomology. Let $L \subset (T^*S, \omega_{st})$ be a Betti Lagrangian. The difference between Betti Lagrangian and other types of Lagrangians previously studied in the literature (e.g. closed cases, or partial wrappings in Liouville sectors) is the existence of horizontal ends for the former, c.f. the conical and O(-1)-ends in Section 2. These ends naturally appear when studying spectral curves, and thus we now develop the Floer-theoretic techniques for them. In this context, the type of Hamiltonian isotopies that we employ is described as follows: **Definition 5.9.** A Hamiltonian $H: T^*S \to \mathbb{R}$ is said to be *cylindrical* if it is of the form $H = h \cdot r$ outside the radius-R disk bundle D_R^*S , for some sufficiently large $R \in \mathbb{R}_+$, where $h: \mathring{S} \to \mathbb{R}$ is a smooth function on the unit sphere bundle and $r: T^*S \to S$ is the (radial) distance coordinate to the zero section. Such a Hamiltonian H is said to have compact *horizontal support* if the projection $\pi(\operatorname{supp}(dH))$ lies in a pre-compact subset of S. Families of Hamiltonians parametrized over a compact manifold are said to be *uniformly horizontally supported* if these Hamiltonians are themselves horizontally supported over a compact set and are uniformly cylindrical.

Let Z be the Liouville vector field on T^*S uniquely determined by $\iota_Z \omega_{st} = \lambda_{st}$. By definition, an ω_{st} -compatible almost complex structure J is cylindrical if $L_Z J = J$ for large enough r. As in [Ekh+13], we deform the Sasaki almost complex structure to make it cylindrical via

(25)
$$J_{cyl} := \begin{bmatrix} 0 & \beta(r)^{-1}g^{-1} \\ -\beta(r)g & 0 \end{bmatrix}$$

where $\beta : [1, \infty) \to \mathbb{R}_+$ is a smooth increasing positive function such that $\rho(r) \equiv 1$ near r = 1 and $\rho \equiv r$ near infinity. We onwards implicitly assume that the chosen (families of) cylindrical almost complex structure J agree with J_{cyl} outside the cotangent bundle over a pre-compact subset of S.

Remark 5.10. For simplicity, we provide the construction of \mathbb{F} for exact Betti Lagrangians. The key result obtained (e.g. Theorem 2) for the WKB case is similar: the presence of O(-1)-ends, rather than conical or cylindrical ends, requires modifying some of the arguments but the qualitative descriptions remains the same, as illustrated by several of the results proven above in both settings. However, in the current paper, we will not construct the full functor \mathbb{F} for the non-exact case, though we essentially compute it for infinitesimally short line segments.

Let $\mathcal{J}(T^*S)$ denote the space of compatible almost complex structures on (T^*S, ω_{st}) . To introduce the precise Floer data needed to construct Floer cochain complexes, we use the following interplay between the certain Lagrangians $P \subset T^*S$ and almost complex structures:

Definition 5.11. Let $J_s \in \mathcal{J}(T^*S)$ be a 1-dimensional family of almost complex structures, with real parameter $s \in \mathbb{R} \subset \mathbb{C}$. By definition, a Lagrangian $P \subset T^*S$ is said to be *uniformly* geometrically bounded with respect to J_s if it is geometrically bounded with respect to the family of metrics induced by J_s and ω_{st} . A 1-dimensional family of Lagrangians P_s is said to be uniformly geometrically bounded if given the associated movie totally real submanifold $\{(P_s, s)\} \subset (T^*S \times \mathbb{C}, \omega_{st} \oplus \omega_{st})$, there exists a constant $C \in \mathbb{R}_+$ and a J_s -compatible symplectic form ω_{P_s} such that (P_s, s) is Lagrangian and the triple $(\omega_{P_s}, J_s, (P_s, s))$ is geometrically bounded.

A summary result of the needed outcome from the literature on Lagrangian Floer theory reads as follows:

Proposition 5.12 (Floer Complexes). Let $P, K \subset (T^*S, \omega_{st})$ be a transverse pair of spin Lagrangians, each endowed local systems V_P , V_K of \mathbb{C} -modules and with chosen spin structures. The following holds:

(1) Let $J_{P,K}: [0,1] \to \mathcal{J}(T^*S)$ be a generic time-dependent family of cylindrical almost complex structures¹⁸ such that P and K are uniformly geometrically bounded. Then, there exists a cochain complex

 $(CF^*((P, \mathbf{V}_P), (K, \mathbf{V}_K); J_{P,K}), \partial),$

known as the Lagrangian Floer complex, generated by the intersection points in $P \cap K$ and whose differential is obtained by counting the 0-dimensional part of the moduli space of $J_{P,K}$ -holomorphic strips bounded by P and K.

(2) The chain-homotopy type of the complex in (1) is independent on the choice of $J_{P,K}$: in particular, a homotopy from $J_{P,K}$ to $J'_{P,K}$, for which P and K are uniformly geometrically bounded, induces a quasi-isomorphism

 $(CF^*((P, \mathbf{V}_P), (K, \mathbf{V}_K); J_{P,K}), \partial) \longrightarrow (CF^*((P, \mathbf{V}_P), (K, \mathbf{V}_K); J'_{P,K}), \partial).$

(3) Let $K_s : K \times [0,1] \to T^*S$ be a Lagrangian isotopy, with associated local systems \mathbf{V}_{K_s} , such that either its movie is uniformly geometrically bounded, or the isotopy generating K_s is a composition of a compactly supported Hamiltonian isotopy with a global Hamiltonian isotopy fixing P. Then $\{K_s\}_s$ induces a continuation quasi-isomorphism

 $c: (CF^*((P, \mathbf{V}_P), (K_0, \mathbf{V}_{K_0}); J_{P,K}), \partial) \longrightarrow (CF^*((P, \mathbf{V}_P), (K_1, \mathbf{V}_{K_1}); J_{P,K}), \partial),$

which is compatible with concatenation, and depends on $\{K_s, V_{K_s}\}$ only up to chain homotopy.

(4) If P, K are both Maslov zero Lagrangians, then the differential of (1) vanishes and thus

$$(CF^*((P, \mathbf{V}_P), (K, \mathbf{V}_K); J_{P,K}), \partial) \cong \bigoplus_{x \in P \pitchfork K} (\hom(\mathbf{V}_P(x), \mathbf{V}_K(x))).$$

is the \mathbb{C} -module generated by the intersection points, endowed with the zero differential. \Box

Proof. For P, K closed Lagrangians and $\mathbf{V}_P, \mathbf{V}_K$, the Lagrangian Floer complex and its invariance were established in [Flo88]. A more contemporary treatment which includes the construction and invariance of such complexes for exact and monotone Lagrangians is [Oh15, Chapter 16]. The case were the local systems $\mathbf{V}_P, \mathbf{V}_K$ are non-trivial is addressed in [Kon95]. This addresses Parts (1) and (2) in these cases. The method of moving Lagrangians, for Part (3), was introduced in [Oh93], with [Oh15, Chapter 14.4] providing the necessary energy estimates. For P, K (and their movies) not necessarily compact Lagrangians, but geometrically bounded, the statement follows by using the monotonicty inequalities introduced in [AL94, Section 5], cf. also the arguments surrounding the proof of [GPS20, Proposition 3.19]. For Part (4), with P, K Maslov 0, the vanishing of the differential is a consequence of the virtual dimension of the moduli space of strips between P and K: it necessarily has to be -1 and so, for regular Floer data, it must be empty.

¹⁸This is sometimes known as the Floer data.

The case of interest in this manuscript is $P = T_z^*S$ a cotangent fiber transverse to L, and K = L a Betti Lagrangian. For Proposition 5.12.(4) to apply, we must ensure the hypothesis of Maslov 0 for L. It is verified as follows:

Lemma 5.13. A Betti Lagrangian $L \subset T^*S$ is Maslov 0.

Proof. By definition, the metric g induces a complex structure on S, and L is holomorphic over some neighborhood of the D_4^- -singularities. Now, over any pre-compact subset, the smooth sheets of the rescaled $\varepsilon L \subset T^*S$ uniformly C^{∞} -converge to the zero section, outside of some fixed neighborhood of the D_4^- -singularities. Since the zero section is holomorphic and L is holomorphic near the branch points, εL is arbitrarily close to being holomorphic over any precompact subset of S as $\varepsilon \to 0$. Therefore the phase function of εL is arbitrarily close to zero over any pre-compact subset of Sand thus L is Maslov 0 (cf. also [Nho24, Section 6.2.1]).

The computation of the Floer complexes in Proposition 5.12.(1) and the continuation maps in Proposition 5.12.(3) is typically challenging: for reasonably generic Lagrangians P, K, there is no explicit description of the continuation maps beyond their definition. This in part is caused by a difficulty in explicitly computing the (cobordism type of the moduli spaces of) pseudo-holomorphic strips involved in these maps. An important conceptual point is that in our setting we are able to use a spectral network W adapted to a Betti Lagrangian $L \subset (T^*S, \omega_{st})$ in order to succeed at explicitly computing such pseudo-holomorphic strips, at least in the adiabatic limit. For instance, the arguments in Section 4 ensure that, adiabatically, there are no pseudo-holomorphic strips bounded by εL and T_z^*S for $z \in W^c$, cf. Theorem 4.2.(1), which is the key ingredient behind showing that the continuation strips for $(CF^*(T_z^*S, L), \partial)$ with $z \in \rho$ moving along a W-free path ρ are "trivial" (we will make this notion of "trivial strips" precise using Lemma 5.28 and the discussion afterward).

Remark 5.14. (1) The dependence on spin structures in Proposition 5.12 can be more compactly repackaged by passing to sphere bundles, after choosing the Floer data. This is related to the map Φ_{W} in Section 5.1 being defined on homotopy classes of paths on the unit sphere bundle; see also Remark 5.1.(3).

(2) We often write $CF^*(P, K)$ for the Lagrangian Floer complex, due to the invariance Proposition 5.12.(2), and implicitly understanding ∂ and the local systems if the context allows for that. For instance, if $P = T_z^*S$ is a cotangent fibre, the local system V_K will always be trivial.

5.2.2. The stalks of $\mathbb{F}(V)$. A naive application of Proposition 5.12 leads to the following ansatz for constructing $\mathbb{F} : \operatorname{Loc}_1^{\dagger}(L) \longrightarrow \operatorname{Loc}_n^{\dagger}(S)$. Given a local system $V \in \operatorname{Loc}_1^{\dagger}(L)$ of rank-1 \mathbb{C} -modules, we define the stalk $\mathbb{F}(V)_z$ of $\mathbb{F}(V)$ at a point $z \in S$ to be \mathbb{C} -module underlying $CF(T_z^*S, L; V)$. The following first considerations then appear:

(1) $\mathbb{F}(V)_z$ is well-defined as a cochain complex. Thus considering just its underlying \mathbb{C} -module is not invariant unless the differential vanishes.¹⁹

¹⁹In which case the \mathbb{C} -module coincides with its cohomology and is invariant.

- (2) $\mathbb{F}(V)_z$ is a rank-*n* \mathbb{C} -module if $L \subset T^*S$ is a Betti Lagrangian only if $z \notin K_L$ is not a branch point.
- (3) The stalk $\mathbb{F}(V)_z$ and the resulting local system $\mathbb{F}(V)$ are independent of the adiabatic parameter ε , up to isomorphism. However, Φ_W and the comparison to spectral network certainly requires passing to the adiabatic limit $\varepsilon \to 0$.

The three consideration are intertwined, and they are all solved at once by considering $z \in W^c$ and introducing an adiabatic parameter $\varepsilon \to 0$. This is achieved by considering the stalk $\mathbb{F}(V)_z$ to be \mathbb{C} -module underlying $CF(T_z^*S, \varepsilon L; V)$ for ε small enough. This is well-defined and addresses the above thanks to the following fact, which allows us to choose the Floer datum to be J_{cyl} .

Proposition 5.15. Let $z \in W^c$ and $U_z \subset S$ be the disk-free neighborhood. Then, there exists $\varepsilon(z) \in \mathbb{R}_+$ the moduli space of non-constant J_{cyl} -holomorphic strips between εL and T_w^*S is empty for any $w \in U_z$.

Proof. Choose $\varepsilon \in \mathbb{R}_+$ small enough such that εL lies in the unit disk bundle. Then, the integrated maximum principle argument (see e.g. [Nho24, Lemma 3.28]) implies that there exists a bijective correspondence between J_{cyl} -holomorphic strips bounded by εL and T_z^*S , and J_g -holomorphic strips bounded εL and T_z^*S . Indeed, the integrated maximum principle guarantees that the images of the J_{cyl} and J_g -holomorphic strips lie within the unit disk bundle, and $J_{cyl} \equiv J_g$ coincide inside this disk bundle. Then Theorem 4.2 implies the statement.

Remark 5.16. The dependence of such putative stalk $\mathbb{F}(V)_z$ (more precisely, the Lagrangian L) on the adiabatic parameter ε can also effectively be removed, at the cost of introducing ε -dependence on the Floer datum instance. Indeed, it follows from [SW06, Remark 1.3] that the rescaling $v(s,t) = \varepsilon^{-1} \cdot u(s,t)$ gives a bijective correspondence between J_g -holomorphic discs bounded by εL and T_z^*S , and $J_{\varepsilon^{-1}g}$ -holomorphic discs bounded by L and T_z^*S .

Let $V \in \text{Loc}^{\dagger}(L)$, then the above discussion, including Proposition 5.15 and Remark 5.16, leads to the following definition for the stalks of $\mathbb{F}(V) \in \text{Loc}^{\dagger}(S)$:

Definition 5.17 (Stalks for $\mathbb{F}(V)$). Let $V \in \text{Loc}^{\dagger}(L)$ and $z \in \mathring{S}$ be such that $\pi(z) \in W^{c}$ and $\varepsilon \in \mathbb{R}_{+}$. The stalk $\mathbb{F}(V)_{z}$ is defined to be

(26)
$$\mathbb{F}(V)_z := CF(T_z^*S, \varepsilon L) = \bigoplus_{i=1}^n V_{z^i}.$$

where $z^i \in \mathring{L}$ are the lifts of z. That is, the stalk at z is the Lagrangian Floer complex defined by the rescaled Betti Lagrangian εL and the cotangent fiber $T^*_{\pi(z)}S \subset T^*S$, where all the elements are concentrated in degree 0 and the differential is trivial.

Two comments on Definition 5.17:

- (1) We use of the sphere bundles \check{S} and \check{L} in order to have signs over \mathbb{Z} (and thus \mathbb{C}) welldefined. As said above, spin structures could be used instead when working with the Betti Lagrangian L and cotangent fibres in T^*S . An argument in favor of using sphere bundles is that one can refine this framework to quantum local systems, where the q-variable, keeping track of the protected spin character y in [GMN13a, Section 3.2] (see also [GMN12, Section 4.2]), is now the value of the monodromy of the local system along the sphere fiber. This restricts to twisted local systems when $q \to -1$, which is our setup, but it is also valuable to develop the theory without specializing the q-variable and thus we setup the groundwork for such generalization to be rather natural.
- (2) The dependence on ε is emphasized as a reminder that computations take place in the adiabatic limit $\varepsilon \to 0$, where W can be characterized Floer-theoretically. It is nevertheless not strictly necessary, which can be argued as follows. First, L is a uniformly bounded multigraph and thus there exists a radius $R \in \mathbb{R}_+$ such that the Betti Lagrangian $L \subset D_R^*S$ lies within the disk bundle of T^*S of radius R. Second, consider a smooth family $\beta_s(r), s \in [\varepsilon, 1]$, of deformation functions such that
 - (i) Each $\beta_s(r)$ is constant for $r \leq R$ and are uniformly linear at infinity $r \to \infty$,
 - (ii) $\beta_{\varepsilon} = \varepsilon$ for $r \in [0, 2R)$ and $\beta_{\varepsilon} = r$ for sufficiently large r,
 - (iii) $\beta_1 = 1$ for $r \in [0, 3R)$.

If $J_{cyl}^s = J_g^{\beta_s}$ denotes the associated family of deformed Sasaki almost complex structures, then J_{cyl}^s is such that $J_{cyl}^{\varepsilon} = J_{\varepsilon^{-1}g}$ for $r \leq R$ and $J_{cyl}^1 = J_g$ for $r \leq R$. Also, the family is uniformly geometrically bounded and of generalized contact type, in the sense that $r^2\beta_s^{-1}$ are all J_{cyl}^s -plurisubharmonic. Therefore, we can apply Proposition 5.15 and Remark 5.16 and identify $CF(T_z^*S, (\varepsilon L, \mathbf{V}_{\varepsilon L}), J_{cyl}) \cong CF(T_z^*S, (L, \mathbf{V}_L), J_{cyl}^{\varepsilon})$, the latter being defined using J_{cyl}^{ε} . Note that interpolating $CF(T_z^*S, L; J_{cyl}^s)$ from $s = \varepsilon$ to s = 1 can be done using standard techniques because there are no moving boundaries involved.

(3) The stalks for $\mathbb{F}(V)_z$ are defined for points in $z \in \mathcal{W}$. Similarly, the parallel transport in Section 5.2.3 is also defined for paths with endpoints on \mathcal{W} . The choices of Floer data, which are done homotopy-coherently, allow for such well-definedness on the entirety of S.

The stalks for $\mathbb{F}(V)$ given as in Definition 5.17, we now move to the parallel transport for the twisted local systems $\mathbb{F}(V)$. As with Section 5.1, the parallel transport isomorphisms contain the more subtle and interesting information, in this case that of continuation strips.

5.2.3. The parallel transport of $\mathbb{F}(V)$. Let $V \in \operatorname{Loc}_{1}^{\dagger}(L)$ be a twisted GL_{1} -local system. In Section 5.2.2, we explained the choice of stalks for $\mathbb{F}(V)$, which consisted of $\mathbb{F}(V)_z = CF(T_z^*S, \varepsilon L) =$ $\bigoplus_{i=1}^{n} V_{z^{i}}$ for a point $z \in \mathring{S}$ whose projection to S is not a branch point of the projection $L \longrightarrow S$. Given a path $\rho: [0,1] \longrightarrow \mathring{S}$, with $z = \rho(0)$ and $z' = \rho(1)$, we now want to construct a parallel transport isomorphism $\mathbb{F}(V)(\rho) : \mathbb{F}(V)_z \longrightarrow \mathbb{F}(V)_{z'}$. This will uniquely determine the image twisted local system $\mathbb{F}(V) \in \operatorname{Loc}_n^{\dagger}(S)$. We will first construct a local system $\mathbb{F}_{\epsilon}(V)$ for each adiabatic parameter $\epsilon > 0$ with the stalks as in Definition 5.17, for points contained outside some ε -dependent nested neighbourhood of \mathcal{W} , that converges to \mathcal{W}^c as $\varepsilon \to 0$. These local systems will come with the property that for all $0 < \varepsilon' < \varepsilon$, there is a unique natural isomorphism $\mathbb{F}_{\epsilon}(V)$ and $\mathbb{F}_{\epsilon'}(V)$, which stabilizes to the identity map over morphisms with end-points on \mathcal{W}^c . We then define $\mathbb{F}(V)$ as the direct limit local system. Intuitively, the parallel transport for $\mathbb{F}_{\epsilon}(V)$ is a generalization of the continuation maps in Proposition 5.12.(3). The continuations maps in Proposition 5.12.(3)cannot be applied in our case because the (moving) fibers T_z^*S are not geometrically bounded. We will consider instead deformed continuations strips, which introduce correction terms to the standard continuation maps that account for the failure of geometric boundedness of the moving fiber. Such parallel transport is constructed as follows.

Let $H_s: T^*S \times [0,1] \longrightarrow \mathbb{R}$ be a uniformly horizontally supported time-dependent Hamiltonian isotopy generating the fiber transport along the *inverse* path of $\bar{\alpha}$ of α , X_s its Hamiltonian vector field and ψ_s its flow, and consider the infinite strip $\mathcal{Z} := \mathbb{R}_s \times [0,1]_t \subset \mathbb{C}$. Choose a smooth, increasing function $l: (-\infty, \infty) \to [0, \infty)$ such that l = 0 near $-\infty$ and l = 1 near $+\infty$. We call such functions elongation functions. Let $J_0(t)$ and $J_1(t)$ be a family of almost complex structures obtained by compactly supported perturbations of J_{cyl} , and let J(s, t) be a family of almost complex structures such that

- (i) $J(s,t) = J_0(t)$ for $s \ll 1$ and $J(s,t) = J_1(t)$ for $s \gg 1$,
- (ii) J(s,t) is also given as a compactly supported perturbation of J_{cyl} .

Such family of almost complex structures is said to be *admissible*.

Definition 5.18. Let $\alpha : [0,1] \longrightarrow S$ be a path, $L \subset T^*S$ a Betti Lagrangian, and J(s,t) be an admissible family of almost complex structures. A map $u : \mathbb{Z} \to T^*S$ is said to be a *deformed* continuation strip for α and L if it solves the following partial differential equation:

(27)
$$\begin{cases} \partial_s u + J(s,t)\partial_t u - l'(s)X_{l(s)}(u) = 0\\ u(s,1) \subset L\\ u(s,0) \subset T^*_{\bar{\alpha}(l(s))}S\\ \lim_{s \to \infty} u(s,t) \in T^*_{\alpha(1)}S \cap L\\ \lim_{s \to -\infty} u(s,t) \in T^*_{\alpha(0)}S \cap L. \end{cases}$$

For paths $\alpha : [a, b] \to S$, the deformed continuation strip for α will mean Eq. (27) using the path $\alpha(bt + (1 - t)a)$ instead.

In Eq. (27), the condition $u(s,0) \subset T^*_{\bar{\alpha}(l(s))}S$ is a moving boundary condition and, in a sense, the term $l'(s)X_{l(s)}(u)$ accounts for that. Indeed, up to composing with the Hamiltonian flow ψ of H_s , the deformed continuation strips in Definition 5.18 are essentially pseudo-holomorphic strips: taking the gauge transform $\tilde{u} := \psi_{l(s)}^{-1} u$ and $\tilde{J} := \psi^* J(s, t)$, we recover the homogeneous pseudoholomorphic curve equation $\partial_{\tilde{J}}\tilde{u} = 0$ with moving boundary condition on L (instead of the fibers). By the exponential estimate [Oh15, Proposition 14.1.5], the solutions of (27) are asymptotic to the intersection points of L and $T^*_{\alpha(1)}S$, as $s \to -\infty$, and of L and $T^*_{\alpha(0)}S$, for $s \to \infty$. By [Nho24, Proposition 3.37], the solutions of the gauge transforms have bounded diameter, and this diameter can be chosen uniformly for a uniform family of Hamiltonian isotopies. Thus the moduli space of solutions of (27) admits a Gromov compactification. By the transversality arguments in [Sei08b, Chapter 2], the solutions of (27) are transversely cut-out for a generic choice of J and the 0-dimensional part of the moduli space of solutions of (27) gives a finite set. Given a generic family J(s,t) of admissible almost complex structures, we denote by $M(\alpha, J(s,t))$ the moduli space of solutions of Eq. (27). The cobordism type of $M(\alpha, J(s, t))$ is invariant under deformations of J(s,t), l(s,t) and α . Observe that since we are in cohomological grading, the induced continuation map sends a generator in $L \pitchfork T^*_{\alpha(0)}S$ at $s = +\infty$ to $L \pitchfork T^*_{\alpha(1)}S$ at $s = -\infty$. This explains why we are using $\bar{\alpha}$, and the inverse of the Hamiltonian isotopy instead.

The parallel transports for \mathbb{F} are defined by counting the solutions of (27): intuitively, it is the number of points in $M(\rho, J(s, t))$. Since only the cobordism type of $M(\rho, J(s, t))$ is invariant, the count must be such that a canceling pair of points contributes zero to the count. Specifically, to count solutions of (27) we must invariantly orient $M(\rho, J(s, t))$.

For that, we first use the following fact, often used in Floer theory: given a bordered²⁰ Riemann surface $C \subset \mathbb{C}$ and a Lagrangian sub-bundle $E \longrightarrow \partial C$ of a bundle $\mathbb{C}^2 \longrightarrow C$, a boundary condition on the Lagrangian sub-bundle E induces the Cauchy-Riemann operator

$$\overline{\partial}: C^{\infty}(C, \mathbb{C}^2, E) \to C^{\infty}(C, \Omega_C^{0,1} \otimes \mathbb{C}^2),$$

where $C^{\infty}(C, \mathbb{C}^2, E)$ denotes the space of C^{∞} -sections of \mathbb{C}^2 whose restriction to the boundary lies in E.

Remark 5.19 (On orientations). 1. Let U_2/O_2 be the Lagrangian Grassmannian of \mathbb{C}^2 , $\rho: S^1 \to U_2/O_2$ a loop of index 0, and E_{ρ} be the Lagrangian sub-bundle of \mathbb{C}^2 over S^1 given by ρ . If the index of the induced Cauchy-Riemann operator $\overline{\partial}$ on (D^2, S^1) operator is 2, then ker $(\overline{\partial}) = \rho(1)$ and the homotopy class of the trivialization of $(\mathbb{C}^2, E) \simeq (\mathbb{C}^2, \mathbb{R}^2)$ determines the orientation on

²⁰For our purposes, it is sufficient to consider C being the disk, the strip \mathcal{Z} , or the half plane.

det $\bar{\partial}$; cf. e.g. [Fuk+09b, Chapter 6].

2. The homotopy class of the trivialization is detected by $\pi_2(U_2/O_2)$: the second Whitney class defines a map $w_2 : \pi_2(U_2/O_2) \to \mathbb{Z}_2$ and we can identify $\pi_2(U_2/O_2) \simeq \mathbb{Z}\langle H \rangle$. Under the natural homomorphism $\pi_2(U_2/O_2) \to \pi_1(O_2)$, H is the generator that gets mapped to the family of orthonormal frames on \mathbb{R}^2 , whose lift to the sphere bundle $\mathbb{R}^2 \times S^1$ winds around the the unit sphere fiber once. The orientation of $\overline{\partial}$ depends only on the mod 2H class; declaring the orientation induced by the trivial loop to be +1, the orientation induced by H is the opposite one [Fuk+09b, Chapter 6.2].

Technically, the stalks as defined in Definition 5.17 implicitly use an orientation of the determinant line bundle of the associated linearized Cauchy-Riemann operator, which would themselves be tensoring the stalks V_{z^i} in the Floer complex. Let us precise on the choice of these orientations, which thus trivialize these line bundles and legitimize our choice of stalk in Definition 5.17. Near each intersection point in $L \pitchfork T_z^*S$, we use a specific linearized Cauchy-Riemann operator of the constant half-disk at each $z^i \in L \pitchfork T_z^*S$, described as follows. The configuration tuple $(T^*S; L, T_z^*S, \omega_{st})$ is locally isomorphic to $(\mathbb{C}^2; \langle u_1, u_2 \rangle, i\mathbb{R}^2_{(x,y)}, \omega_{std})$, where u_1, u_2 is a unitary frame. Consider U the unitary matrix with columns u_1, u_2 and the path of unitary matrices

$$\phi: [0,1]_t \longrightarrow U_2, \quad \phi(t) := i \exp(-t \log(U)),$$

each of which acts on \mathbb{C}^2 as an isometry by left multiplication. Consider the path of linear Lagrangians

$$L_t := \phi_t(\langle u_1, u_2 \rangle) \subset T^* \mathbb{C}^2$$

from $\langle u_1, u_2 \rangle$ to $i\mathbb{R}^2$. The path ϕ_t takes an orthonormal basis on $\langle u_1, u_2 \rangle$ to an orthonormal basis on $i\mathbb{R}^2$ and the square phase of this as a path in the Lagrangian Grassmannian U_2/O_2 is $e^{-2\pi t}$ and so the grading of the path becomes $-2t, t \in [0, 1]$. The path L_t defines a linearized Cauchy-Riemann operator of the constant half-disk at each $z^i \in L \pitchfork T_z^*S$ by the moving boundary condition on L_t .

We now discuss the specific trivialization to choose on the moving boundary part $T^*_{\alpha(s)}S$. Given a W-adapted path ρ , the metric g on S defines a complex structure I and we can consider the frame $F = \langle \rho(a), I\rho(a) \rangle$ at $T_{\pi(\rho(a))}S$. Choose the frame on the tangent space of $T^*_zS \pitchfork L$ to be the one given by the image of $\exp(\log(U))$ on the dual frame $F^* = \langle \rho^*(a), I^*\rho^*(a) \rangle$. The trivial spin cover over the frames $(\phi_t \circ \phi_1^{-1})(F)$ gives an orientation on the linearized Cauchy Riemann operator, which is then identified with the trivial line bundle. This is our trivialization, and thus we conclude the discussion on orienting the moduli spaces $M(\rho, J(s, t))$ of deformed continuation strips, solving Eq. (27).

We can now construct the continuation maps without the assumption that the Lagrangian K in Proposition 5.12.(3) is geometrically bounded, which is needed in this framework of spectral networks as we must apply it to $P = T_z^*S$ a cotangent fiber. The result reads as follows:

Proposition 5.20 (Continuation of CF^* under moving fiber). Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, $L \subset (T^*S, \lambda_{st})$ a Betti Lagrangian, \mathcal{W} a compatible spectral network and $V \in Loc^{\dagger}(L)$. Then:

(1) (Well-defined) Given $\rho : [0,1] \longrightarrow \mathring{S}$ be a W-adapted path from $z_0 = \rho(0)$ and $z_1 = \rho(1)$, the signed count of points in the (oriented cobordism type of the) moduli spaces $M(\rho, J(s,t))$ of solutions to Eq. (27) determines a continuation map

$$\mathbb{F}_{\varepsilon}(V)(\rho): CF^*(T^*_{\pi(z_0)}S, (\varepsilon\Sigma, V)) \to CF^*(T^*_{\pi(z_1)}S, (\varepsilon\Sigma, V))$$

which is independent of the representative of the smooth homotopy class ρ .

(2) (Twisted equivariance) For any $\rho : [0,1] \longrightarrow \mathring{S}$ be a W-adapted path, the continuation map in (1) satisfies

$$\mathbb{F}_{\varepsilon}(V)(H\rho) = -\mathbb{F}_{\varepsilon}(V)(\rho),$$

where H is the homotopy class of the cotangent circle.

(28)

(3) (Concatenation morphism) Given two W-adapted path ρ_1, ρ_2 , the continuation map in (1) satisfies

$$\mathbb{F}_{\varepsilon}(V)(\rho_1 \circ \rho_2) = \mathbb{F}_{\varepsilon}(V)(\rho_1) \circ \mathbb{F}_{\varepsilon}(V)(\rho_2).$$

Proof. (of Proposition 5.20) It suffices to orient the moduli $M(\rho, J(s, t))$ using the twisted local system $V \in \text{Loc}^{\dagger}(L)$ such that it becomes cobordism invariant with respect to ρ . Indeed, once that is established, the cobordism argument in Floer theory, cf. e.g. [Sei08b], concludes invariance (1) and the concatenation property (3). To orient, consider a deformed continuation strip u traveling between two lifts z_0^i to z_1^j in \mathring{L} , i.e. with positive punctures on these two points, the lower boundary mapped to L and the upper boundary to the moving fibers along ρ . Choose a sphere bundle lift of the path $u(s,1), s \in (-\infty,\infty)$, so that it agrees with $\phi_1^{-1}(\rho(a))$ at z_a^i , for all a=0 and a=1. The induced metric on S gives a framing of S along u(s,1). Considering the two points $z_0^i, z_1^j \in L$ as constant half-disks, we can glue them to the ends of the strip u and obtain the glued disk $z_0^i \sharp u \sharp z_1^j$. As discussed above, this data defines a linearized Cauchy Riemann operator over the disk $z_0^i \sharp u \sharp z_1^j$ which has index 2. Its orientation depends on the choice of the trivialization of the pull-back of the tangent bundles of the Lagrangian spaces: our choice for orientation is to concatenate the sphere bundle lift of u(s,1) traveling through L, the paths $(\phi_t \circ \phi_1^{-1})(F)$ and then (the dual frame given by) the sphere bundle path ρ traveling through the moving fibers. This choice is independent of the sphere bundle lift of u(s, 1) and, by construction, twisting the path ρ by H reverses the sign of the orientation, which established (2).

We now construct the local systems \mathbb{F}_{ε} and \mathbb{F} . Firstly, make a choice of the regular Floer datum²¹ for each pair $(T_z^*S, L), z \in S$, so that we get the stalk given by the (abstract) Floer cohomology groups $HF(T_z^*S, L)$. Furthermore, the parallel transport is well-defined once the Floer datum has been chosen at each intersection point, since the moduli space of solutions of Definition 5.18 is well-defined, up to cobordism. Thus, regardless of the choice of the regular Floer datum, the family Floer twisted local system, regarded as a functor $\Pi_1(\mathring{S}) \to \operatorname{Vect}_n$, is well-defined, up to unique

²¹This datum must also contain a time-dependent Hamiltonian term in the case that z is a branch point, c.f. [Sei08b, Section 8(e)].

natural isomorphism.

The dependence on the adiabatic parameter ε is incorporated as follows. By Theorem 4.2, for any precompact subset $U \subset S - W$ there exists $\varepsilon_U > 0$ such that for all $0 < \varepsilon < \varepsilon_U$, the Floer datum J_{cyl}^{ε} for $(T_z^*S, L, J_{cyl}), z \in \overline{U}$, is regular. Choosing J_{cyl}^{ε} to be the regular Floer datum for (T_z^*S, L) for z outside \overline{U} , we obtain a local system, which we consider of as a functor $\mathbb{F}_{\varepsilon}(V) : \Pi_1(\mathring{S}) \to Vect_k$. Furthermore, by our description of J_{cyl}^{ε} , given any $U \subset U'$, and scaling stability (Lemma 5.31) the induced natural isomorphism between $\mathbb{F}_{\varepsilon}(V)$ and $\mathbb{F}_{\varepsilon'}(V)$ acts as the identity on morphisms that represent paths that are contained fully in U, for $\varepsilon' < \min(\varepsilon_{U'}, \varepsilon)$.

Definition 5.21. The family Floer cohomology local system \mathbb{F} is the direct limit local system $\lim_{t \to \infty} \mathbb{F}_{\varepsilon}$.

The discussion above implies that \mathbb{F} is well-defined up to unique isomorphism. By scaling stability, whenever we compute parallel transport maps of \mathbb{F} , we may compute it for ε small enough. The results of the current section then can be regarded as showing that the local system \mathbb{F} coincides with $\Phi_{\mathcal{W}}(V)$, over objects that lie outside \mathcal{W}^c .

We conclude this section with a discussion on how the continuation maps in Proposition 5.20 can be enhanced to record (twisted) homotopy classes, while remaining invariant. Specifically, let $M(\rho; J(s,t), V)$ denote the moduli space $M(\rho, J(s,t))$ equipped with the orientation (from V) as in the proof of Proposition 5.20. For each deformed continuation strip $u \in M(\rho; J(s,t), V)$, we denote by $\mathfrak{s} u$ the sphere bundle lift of u(s, 0) and by $\sigma(\mathfrak{s} u, \rho)$ the sign of $\det(D\bar{\partial}_J u)_{\mathfrak{s} u, \rho}$, the oriented determinant line of the linearized Cauchy-Riemann operator obtained using $\mathfrak{s} u$ and ρ .

Corollary 5.22 (Homotopy refinement). In the notation above, the homotopy refinement of the deformed continuation map in Proposition 5.20 is

(29)
$$\mathbb{F}(V)(\rho) = \sum_{u \in M(\rho; J(s,t), V)} \sigma(\mathring{s}u, \rho) V(\mathring{s}u)[u(-s, 1)],$$

and it only depends on the homotopy class of ρ , and is compatible with concatenation.

An advantage of Eq. (29) over Eq. (28) is that the former is invariant under the *H*-action, as an *H*-twist in the sphere bundle lift $\mathfrak{s} u$ is canceled out by the parallel transport $V(\mathfrak{s} u)$. We shall momentarily show in Section 5.3 that the parallel transport $\mathbb{F}(V)$ as in Eq. (29) yields, in the adiabatic limit, the homotopy refinement of $\Phi_{\mathcal{W}}(V)$.

5.2.4. Ancillary discussion on spin structures. The (twisted) Floer complexes constructed above used the sphere bundles \mathring{S} , \mathring{L} instead of working directly with S, L, each endowed with spin structures. Two reasons are that, with this groundwork, it is reasonably direct to improve the above discussion to include the protected spin characters y-variable in [GLM15] and, more technically, it allows us to package the discussion on the dependence on spin structures directly in the sphere bundle classes. The former is achieved by keeping track of the homotopy class on the sphere bundle, without forcing the twisted evaluation $y \to -1$. Indeed, it is possible to show that the solutions of Eq. (27) are immersed and thus have canonical lifts to the unit sphere bundles, where their (relative) homotopy classes can be recorded.

To compare the Floer complexes constructed above using the sphere bundles, cf. Proposition 5.20, with the Floer complexes one would define using S, L and their spin structures, we proceed as follows. First, suppose that $\tilde{\mathfrak{s}}, \mathfrak{s}, \beta$ are all trivial near $L \Leftrightarrow T_z^*S$, and that the frame bundle of L has its torsor at $\overline{z} \in T_z^*S \Leftrightarrow L$ so that it is equal to the one given by the ϕ_t^{-1} image of some chosen frame $\langle e_1, e_2 \rangle$ on $i\mathbb{R}^2$. Taking the trivial spin cover over the framing $(\phi_t \circ \phi_1^{-1}) \langle e_1, e_2 \rangle$ gives a path of spin structures on L_t and the induced orientation is the trivial one. Then consider the sphere bundle lift of the path α and the sphere bundle lift of the path L_t , as prescribed by \mathring{S} and $\tilde{\mathfrak{s}}$. The orientation on the induced linearized Cauchy-Riemann operator then determines the sign and it coincides with that of the holonomy of $\mathbb{F}(V)$ along u(s, 0). As a consequence of Theorem 5.24 and this discussion, we can summarize the outcome as follows:

Corollary 5.23. Let (S, \mathfrak{s}) and $(L, \tilde{\mathfrak{s}})$ be endowed with spin structures, and consider $\mathbf{V} \in Loc(L)$ and $V \in Loc^{\dagger}(L)$ such that $\tilde{\mathfrak{s}}^*V = \mathbf{V}$. Then the family Floer local system obtained using \mathbf{V} is the \mathfrak{s}^* -pullback of $\mathbb{F}(V)$.

5.3. Equivalence between \mathbb{F} and $\Phi_{\mathcal{W}}$. The aim of this section is to prove the following result, comparing the functors \mathbb{F} and $\Phi_{\mathcal{W}}$, i.e. the twisted local systems obtained from Family Floer and those obtained via the combinatorics of spectral networks:

Theorem 5.24 (\mathbb{F} and $\Phi_{\mathcal{W}}$ are equivalent). Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, $L \subset (T^*S, \lambda_{st})$ an exact Betti Lagrangian, (S, g) a compatible metric and \mathcal{W} an adapted Morse spectral network. Then the two functors

$$\mathbb{F}, \Phi_{\mathcal{W}}: Loc_1^{\dagger}(L) \longrightarrow Loc_n^{\dagger}(S)$$

are equivalent. In particular, the combinatorial corrections from detour paths on W coincide with the Floer-theoretic corrections from deformed continuations strips.

To compare these two functors it suffices to focus on the objects, as morphisms naturally commute with the construction. The statement of Theorem 5.24 also holds for the enhancements of those functors that keep track of specific (relative) homotopy classes in the unit sphere bundles. Theorem 5.24 is proven by analyzing the behavior of \mathbb{F} in the adiabatic limit $\varepsilon \to 0$. Namely, understanding the deformed continuation strips from Eq. (27) with one boundary condition on εL and the other (moving) boundary condition on cotangent fibers as $\varepsilon \to 0$. Since \mathcal{W} controls the adiabatic behavior of pseudo-holomorphic strips, there are two regions to consider:

(1) In the complement \mathcal{W}^c of the spectral network, the parallel transport of $\Phi_{\mathcal{W}}$ is trivial. Therefore, we need to show that the deformed continuation strips defining the parallel transport for \mathbb{F} are also trivial if the path lies in \mathcal{W}^c . (2) For W-adapted short paths, we must show that the adiabatic limit of deformed continuation strips across a wall of W is exactly given by the trivial strips and strips whose flowtree limit coincides with the detour paths introduced in Section 5.1.

In order to prove these facts above, we start with a quantitative refinement of Eq. (27), estimating the behaviors of deformed pseudo-holomorphic strips in terms of their length. Specifically, consider $z \in S$ such that T_z^*S is transverse to the Betti Lagrangian L and fix a unit tangent vector $v \in \mathring{S}$ base at z. Employing the Riemannian metric (S, g), consider the path $\alpha_v(s) = \exp_z(sv)$ starting at z in the direction of v. For s sufficiently small, α_v is an embedding and $T^*_{\alpha_v(s)}S$ remains transverse to L. This latter condition implies in particular that there are no D_4^- -singularities in a small enough neighborhood of the image of α_v . We model such a path by considering local coordinates (x_1, x_2) near $z = x_1 + ix_2$ so that $\alpha_v(s) = (s, 0)$, and then the (local) Hamiltonian $H^{v}: T^{*}\mathbb{R}^{2}_{x_{1},x_{2}} \longrightarrow \mathbb{R}, \ H^{v}(x_{1},x_{2};p_{1},p_{2}) = -p_{1} \text{ generates the transport of } T^{*}_{z}S \text{ along } \bar{\alpha}_{v}(s).$ To achieve compactly supported behavior, localizing this fiber transport along $\bar{\alpha}_v(s)$, let us cut-off H^v to a Hamiltonian $b(x_1, x_2)H^v$ with $b(x_1, x_2)$ a bump function concentrated at $\alpha_v(s)$, i.e. $b(x_1, x_2)$ is constant equal to 1 in a neighborhood of $\alpha_v(s)$ and vanishes outside a slightly larger neighborhood. To simplify, we still denote by H^v the cut-off Hamiltonian, denote by X_v its Hamiltonian vector field and by ψ_v its flow. The system Eq. (27) defining deformed continuation strips can now be specialized to paths $\alpha_v(s)$, where dependence on the unit tangent vector v is acquired and H^v is fixed.

Definition 5.25. Let $\varepsilon \in \mathbb{R}_+$ and J(s,t) be an ε -admissible family of almost complex structures on \mathcal{Z} , and consider an elongation function l(s) and a length $d \in \mathbb{R}_+$. By definition, a map $u_d : \mathcal{Z} \longrightarrow T^*S$ is said to be a continuation strip along $\alpha_v(s) : [0,d]_s \to \mathbf{S}$ if it solves the system

(30)
$$\begin{cases} \partial_s u_d + J(s,t)\partial_t u_d - d \cdot l'(s)X_v(u_d) = 0\\ u_d(s,1) \subset L\\ u_d(s,0) \subset T^*_{\bar{\alpha}_v(d\cdot l(s))}S\\ \lim_{s \to -\infty} u_d(s,t) \in T^*_{\alpha_v(d)}S \cap L\\ \lim_{s \to +\infty} u_d(s,t) \in T^*_zS \cap L. \end{cases}$$

We often denote $u(s,t) = u_d(s,t)$, understanding the dependence on d implicitly.

To ease notation, given a value for the adiabatic parameter $\varepsilon \in \mathbb{R}_+$, we assume that we have fixed an *s*-dependent deformation function $\rho_s(r)$, as in Section 5.2.1, and an ε -admissible family of almost complex structures refers to an admissible family of almost complex structures obtained by deforming $J_{cyl}^{\rho_{\varepsilon}}$, instead of J_{cyl} . Our first task is choosing the elongation function l(s) such that solutions to Eq. (30) of small enough length have controlled behavior. This is the content of the following:

Proposition 5.26. In the notation for Eq. (30), there exists a choice of an elongation function l(s) and a constant $d_v \in \mathbb{R}_+$ such that the following holds:

86

- (1) The continuation strips $u_d(s,t)$ along $\alpha_v(s)$ of length $d \in (0, d_v)$ whose limiting endpoints lie in the same smooth sheet of L cannot leave the d_v -neighborhood of their endpoints.
- (2) For the continuation strips $u_d(s,t)$ along $\alpha_v(s)$ of length $d \in (0, d_v)$ whose limiting endpoints lie in different smooth sheets of L, there exists a sequence of lengths $d_n \to 0$ such that the associated family u_{d_n} of strips Gromov converges to a broken J(s,t)-pseudo-holomorphic strip bounded between L and T_z^*S .

Proof. For Part (1), let $u = u_d(s,t)$ be a solution of Eq. (30) and consider its gauge transform $\tilde{u} = \psi_{l(s)}^{-1} u$. Then the symplectic area of \tilde{u} is

$$\int_{\mathcal{Z}} \tilde{u}^* \omega = -\int_{u(s,1)} \lambda_{st} - d \cdot \int H^v(u(s,1)) ds = W((\psi_1^{-1} \circ u)(-\infty,0)) - W(u(+\infty,0)) - d \cdot \int H^v(u(s,1)) ds,$$

where $dW = \lambda_{st}|_L$ is a primitive of the Liouville form along L. Along $\alpha_v(s)$, let h > 0 be such that over some small neighbourhood of $\alpha_v(s)$, the smooth sheets of ϵL have distance at least h > 0apart, with respect to the Sasaki metric. For each $z \in \alpha_v(s)$, let $\bigcup B_{h/2}(z^i) \subset T^*S$ denote the small neighbourhood of the intersection points between T_z^*S and L. Let $\ell \in \mathbb{R}_+$ be a constant so that J_{cyl} -holomorphic half-disks with energy less than $\ell/2$ intersecting $\bigcup_i B_{h/4}(z^i)$ cannot leave the larger neighbourhood $\bigcup_i B_{h/2}(z^i)$, for all $z \in \alpha_v(s)$.

Choose an elongation function l(s) so that l'(s) is supported on $[0, \ell]$. By the hypothesis that the endpoints lie on the same sheet, the limiting endpoints of the gauged transformed strips have to be the unique intersection point of T_z^*S and L lying on the same sheet as the endpoints of u_d . By Eq. (31), the symplectic area (a.k.a. energy) of the gauge-transformed strip is of the form O(d) for for small enough. Therefore, the graph of $\psi_{d\cdot l(s)}^{-1}(u(s,t))$ on the moving part has energy O(d) + l.

In addition, for d small enough, the symplectic 2-form making the movie $(\psi_{d\cdot l(s)}^{-1}(L), s)$ Lagrangian can be made arbitrarily close to $\omega_{st} \oplus \omega_{st}$ and the metric induced by $\psi_{d\cdot l(s)}^*J(s,t)$ can be made arbitrarily close to the correspondingly standard metric. Thus, [Nho24, Proposition 3.37] implies that the graph of $\psi_{l(s)}^{-1}(u(s,t))$ on the moving part cannot leave the $(3\ell)/2$ -neighborhood of the intersection points between L and T_z^*S for small enough d, which concludes Part (1). For Part (2), it suffices to verify that the symplectic area of the family of solutions (30) admits a uniform positive lower bound. This follows from the fact that $u_d(s,t)$ travels between distinct sheets of L (see [Nho24, Eq.6.1.10]). Therefore, the Gromov compactness argument developed in [Nho24, Section 6.1.5] applies and $u_d(s,t)$ Gromov converges to a broken J(s,t)-pseudo-holomorphic strips bounded between L and T_z^*S for a subsequence of lengths $d_n \to 0$.

By definition, a continuation strip u is said to be *small* if it is as in Proposition 5.26.(1).

Remark 5.27. Note that there is a difference between Proposition 5.26 and [Nho24, Proposition 6.11]: in the result above, the Hamiltonian isotopy does not necessarily fix the neighborhood of the

intersection points. In order to correct this matter, we need to be very careful with the choice of the elongation function l.

After a small perturbation of J(s,t), we can and do assume that the moduli space of same sheet continuation strips consist of small continuation strips, and that it is transversely cut-out.

Lemma 5.28. In the notation above, for small enough $d \in \mathbb{R}_+$, the moduli space of small continuation strips of length d is cobordant to a singleton moduli space, consisting of a single strip whose boundary on L may be identified with a lift of the arc $\alpha_v(s), s \in (0, d_v)$.

Proof. Near z, a smooth sheet of L can be described a graph of a smooth function f. As observed in [Flo89b], we may regard f as a (locally-defined) Hamiltonian on T^*S and write $\varepsilon \cdot \operatorname{gr}(f)$ as the image of the zero section under the time ε -flow Ψ of f. Taking the inverse of the fibre-preserving global Hamiltonian isotopy Ψ_{ε}^{-1} , the configuration ($\varepsilon L, T_z^*S, J_{cyl}$) transforms into $(Z, T_z^*S, \Psi_{\varepsilon}^{-1}(J_{cyl}))$. Under this transformation, the perturbation term in Eq. (27) turns into $(d\Psi_{\varepsilon}^{-1})_*(X_v(u_d))$. Since we are only interested in small continuation strips, whose diameter we can a priori control, using the monotonicity argument in the proof of [Nho24, Proposition 6.11], we are free to interpolate the perturbation term from $(d\Psi_{\varepsilon}^{-1})_*X_v$ to $l'(s)X_v$. We claim that the solutions of the new equation are all constant, up to gauge-transformation. This leads to a cobordism of the moduli space of small continuation strips, whose other end consists of a single strip.

To verify the claim, consider the solutions of Eq. (27) with J(s,t) replaced by $\psi_{\varepsilon}^* J(s,t)$ and the boundary condition L replaced by the zero section. Since the Hamiltonian fixes the zero section, we are back to the situation of [Nho24, Proposition 6.11] where the Hamiltonian fixes the Lagrangians near the intersection points. Therefore, after taking the gauge-transform $\Psi_s^{-1}u(s,t)$, the strip becomes *constant*. As claimed, we may identify its boundary on L with a lift of the arc $\alpha_v(s), s \in (0, d_v)$ to the smooth sheet $\operatorname{gr}(f)$.

Strips as in Lemma 5.28 are said to be *trivial strips*.

We introduce the following notions, analogous to Definition 4.1:

Definition 5.29. Let $L \subset (T^*S, \lambda_{st})$ be a Betti Lagrangian and \mathcal{W} an adapted finite creative Morse spectral network.

- (i) A point $z \in S$ is said to be *continuation-free* in the adiabatic limit if there exists $\varepsilon(z) \in \mathbb{R}_+$ and a neighbourhood U_z such that for any $\varepsilon \in (0, \varepsilon(z))$, (T_w^*S, L) is J_{cyl}^{ε} disk-free for $w \in U_z$, and there exists a length $d(z, \varepsilon) \in \mathbb{R}_+$ and an elongation function $l(s, \epsilon)$ such that for all points $w \in U_z$ there are no J_{cyl}^{ε} -holomorphic continuation strips that are not small along arcs of the form $\alpha_v : [0, d]_s \to S$, for $d \in (0, d(z, \varepsilon))$, $w \subset U_z$ and $v \in T_w S$ a unit tangent vector at w
- (ii) A point $z \in W$ different from a vertex is said to have Stokes transport in the adiabatic limit if there exists $\varepsilon(z) \in \mathbb{R}_+$ and a length $d(z) \in \mathbb{R}_+$ such that for any $\varepsilon \in (0, \varepsilon(z))$, there

exists an elongation function $l(s, \varepsilon)$ for which there are *no* large J_{cyl}^{ε} -holomorphic strips for L along the arc $\alpha_v(s) : [-d(z), d(z)]_s \to S$, that do not belong to the path detour class along $\alpha_v(s)$, where v denotes the outward normal vector to \mathcal{W} at z.

Note that vertices of \mathcal{W} do not need to be considered in Definition 5.29 because the behavior near them has been fixed by the local models. We must now establish the analogue to Theorem 4.2 for continuation strips, which reads as follows:

Proposition 5.30. Let $L \subset (T^*S, \lambda_{st})$ be a Betti Lagrangian and W an adapted finite creative Morse spectral network. Then the following holds:

- (i) If $z \in W^c$, then z is continuation-free in the adiabatic limit.
- (ii) If $z \in W$ and not a vertex, then z has Stokes transport in the adiabatic limit.

Proof. For Part (1), fix a precompact open subset of W^c whose closure does not intersect W. We argue by contradiction: suppose that there is a sequence of points z_n , unit tangent vectors v_n of z_n , with lengths $d(z_n)$ converging to zero but such that there are J_{cyl}^{ε} -holomorphic continuation strips which are not small; for L and along arcs $\alpha_{v_n}(z_n)$. By Proposition 5.26 such a sequence has a subsequence that converges to a broken J_{cyl} -holomorphic strip. By the integrated maximum principle and the equality $J_{cyl}^{\varepsilon} = J_{\varepsilon^{-1}g}$ along the disk bundle containing L, J_{cyl}^{ε} -holomorphic strips have to necessarily be $J_{\varepsilon^{-1}g}$ -holomorphic strips. Then Remark 5.16 and Theorem 4.2 together imply that such broken $J_{\varepsilon^{-1}g}$ -holomorphic strips cannot exist, thus a contradiction. This establishes Part (1).

For part (2), let U_z be the Stokes neighbourhood of z. By Theorem 4.2, we see that there exists some $\varepsilon(z) > 0$ such that for all $\varepsilon \in (0, \varepsilon(z))$, the only disks in $(T_w^*S, \varepsilon L)$ must be parallel transports of the soliton classes. We claim that the maximal arc $\alpha_v(s): [-d_z, d_z]_s \to S$ contained in such a neighbourhood is the desired one. To see this, shrink $\varepsilon(z)$ so that $(T^*_{\alpha_v(0)}S, L)$ and $(T^*_{\alpha_v(1)}S,L)$ are all J^{ε}_{cyl} disk-free, for all $\varepsilon \in (0,\varepsilon(z))$. Then we can combine Proposition 5.26 and the previous argument to conclude that there exists an elongation function l(s) such that for each $t \in [-d_v(s), d_v(s)]$, there exists $d_t > 0$ such that along $\alpha_v(t+s) : [0, d]_s \to S$, for $d \in (0, d_t)$, the large J_{cul}^{ε} -continuation strips must have the homotopy class of the detour path across z (deformed to have the end-points of $\alpha_n(t+s)$, in a straightforward way). Indeed, this follows from Theorem 4.2 because the path homotopy class of its boundary must converge, as $d \to 0$, to broken chains of $\mathfrak{s}(z)$, none of which can concatenate. Then, using the compactness of $\alpha_v(s)$, we can split $\alpha_v(s)$ into finitely such segments. We can now perturb slightly to make sure the continuation strips are all transversely cut-out. By Gromov compactness, having no large continuation strips of certain homotopy class is a condition invariant under small enough perturbations of J(s,t). We then apply the "solitons cannot concatenate" argument again to conclude that the glued continuation strips must be of the path detour class. We now turn off the perturbation, and argue by Gromov compactness again, to conclude that for some elongation function, the homotopy class must also be the path detour class, concluding (2).

In addition, a simpler argument leads to the following type of stability under ε -scaling:

Lemma 5.31 (Scaling stability). For $z \in W^c$, there exists $\varepsilon(z) \in \mathbb{R}_+$ such that for all $\varepsilon', \varepsilon \in (0, \varepsilon(z)), \varepsilon' < \varepsilon$, the continuation map $CF(T_z^*S, L; J_{cyl}^e) \to CF(T_z^*S, L; J_{cyl}^{e'})$ is the identity map.

Proof. The proof is similar to Part (1) in Proposition 5.30. Take $\varepsilon(z)$ as in Theorem 4.2 so that $(\epsilon L, T_z^*S)$ is disk-free for all $\varepsilon \in (0, \varepsilon(z))$. By the previous discussion, we obtain a family J_{cyl}^s such that $J_{cyl}^s = J_{s^{-1}g}$ in the disk bundle for $\varepsilon' \leq s \leq \varepsilon$. Fix an elongation function l, then for each $\varepsilon' < a' < a < e$, the continuation map from $CF(T_z^*S, L; J_{cyl}^a)$ to $CF(T_z^*S, L; J_{cyl}^{a'})$ is given by counting pseudo-holomorphic strips with the boundary conditions

(32)
$$\begin{cases} \partial_s u + J(s,t)\partial_t u = 0\\ u(s,1) \subset L\\ u(s,0) \subset T_z^*S\\ \lim_{s \to \pm \infty} u(s,t) \in T_z^*S \cap L, \end{cases}$$

1

using the family almost complex structures J(s,t) that are given by small perturbations of the family $J_{cyl}^{l(s)a+[1-l(s)]a'}$. Since the equation is homogeneous, the strips that travel between the same lifts of T_z^*S must all be constant, and the strips that travel between distinct lifts of T_z^*S must have energy a priori bounded below by some small $h \in \mathbb{R}_+$. The same argument as in Proposition 5.30 implies that for |a'-a| small enough, the continuation strips are all constant, and so they induce the identity map. The rest follows from concatenation compatibility, from which we conclude that the continuation map from $s = \varepsilon$ to $s = \varepsilon'$ must be the identity map.

Let us finally establish the equivalence between \mathbb{F} and $\Phi_{\mathcal{W}}$:

Proof of Theorem 5.24. Let us first compare the parallel transports for a path of the form $\rho(s) = \alpha_v(s)$ starting at $z \in W^c$ and s small enough. Since any $\Phi_W(V)$ has trivial parallel transport for $V \in \operatorname{Loc}^{\dagger}(L)$, c.f. Definition 5.6.(i), we must argue that the same holds for $\mathbb{F}(V)$, for ε small enough. Fix a small neighbourhood $U_{\rho} \subset W^c$ of ρ . By Theorem 4.2, there exists some $\varepsilon(\rho) > 0$ such that for all $0 < \varepsilon < \varepsilon(\rho)$, $(T_z^*S, \varepsilon L), z \in U_{\rho}$ is disk-free. Fix $0 < \varepsilon < \varepsilon(\rho)$, then by Section 5.2.3, the parallel transport of the latter is given by the moduli spaces of deformed continuation strips, cf. Proposition 5.20. By Lemma 5.28 and Proposition 5.30 the corresponding moduli spaces of continuation strips for such a path away from W is cobordant to the moduli space of trivial strips. For the orientation, see e.g. [Nho24, Lemma 6.14]. This implies that the count of deformed continuation strips is that of trivial strips, and therefore the parallel transport of $\mathbb{F}(V)$ along such $\alpha_v(s) \subset W^c$ equals that of $\Phi_W(V)$. By concatenation compatibility, we conclude that the parallel transports agree over any free W-adapted path.

For each $z \in W$, we can consider a W-adapted short path of the form $\rho(s) = \alpha_v(s)$, intersecting W at z, just as in Definition 5.29. By Gromov compactness, we can perturb by a small amount so

91

that the moduli space of large continuation strips is transversely cut-out, and so that the boundary homotopy classes belong to the path detour class. Let u(s,t) be the resulting continuation strip along such path: we must show that the assignment $\mathfrak{s}(z;\mathfrak{w}) \to \sigma(\mathfrak{s}u,\alpha_v(s))$ (where the count is defined using the homotopy refinement Corollary 5.22) gives a functor with the properties as in Definition 3.20. For each vertex $v \in \mathcal{W}$, introduce nearby points $b \in \mathcal{W}$ on each edge adjacent to v and for each such b, consider the corresponding short path arcs $\alpha_{\nu}^{(b)}(s)$, intersecting W in the positive direction. We now proceed on inductively, by starting with the points close to the initial walls, and then following the birth of new walls along \mathcal{W} . First, for these points b close enough to the initial rays of the D_4^- -vertex v, direct computation gives the count $\sigma(\mathfrak{s} u, \alpha_v(s)) = 1$, e.g. c.f. [GMN13b, Section 5.6, Figure 21]. It thus suffices to show that $\sigma(\mathfrak{s} u, \alpha_v(s))$ satisfies the same properties as in Definition 3.20 near the creation joints of \mathcal{W} . At the same time, that these properties are satisfied is a consequence of Proposition 5.20 and the fact that the trivial loop encircling the vertex of \mathcal{W} is contractible, and thus the associated (cobordism class of) moduli space of continuation strips consists of constant strips (one per intersection point), up to cobordism. The cancellation of the concatenation of non-trivial continuation strips along the trivial loop is encoded in the 2d Hori-vafa formula Definition 3.20. Therefore, since the counting functions are the same, the two local systems $\Phi_{\mathcal{W}}(V)$ and $\mathbb{F}(V)$ must also be isomorphic.

Theorem 5.8 now follows from Theorem 5.24:

Proof of Theorem 5.8. By Theorem 5.24, the parallel transport $\Phi_{\mathcal{W}}(V)(\rho(\mathfrak{r}))$ is equivalent to that of $\mathbb{F}(V)(\rho(\mathfrak{r}))$. In the adiabatic limit, the deformed continuation strips contributing to this latter parallel transport converge to D_4^- -flowtrees with semi-infinite root edges asymptotic to \mathfrak{r} , and the counts are in fact equal. Indeed, by [EHK16, Lemma 5.12] and [EHK16, Theorem 1.5], there exists a bijective correspondence between rigid flow-trees for (a perturbed and neck-stretched version of)²² L and rigid index-0 pseudo-holomorphic disks (adjusted, in the sense of [EHK16, Section 3.2]) with boundary on that exact Betti Lagrangian, all constrained to be asymptotic to \mathfrak{r} . In general, the cobordism type of the latter moduli space of pseudo-holomorphic disks is not an invariant because of bifurcations given by index -1 disks, see e.g. [EHK16, Lemma 3.13]. That said, Lemma 5.13 shows that Betti Lagrangians $L \subset (T^*S, \lambda_{st})$ have Maslov index 0 and thus such index-(-1) disks cannot appear. Therefore, the cobordism type of this moduli space is an invariant up to compactly supported Hamiltonian isotopy and, consequently, so is the count of D_4^- -trees asymptotic to \mathfrak{r} .

5.4. **Proof of Theorem 2.** The object of this subsection is to conclude the proof of Theorem 2. The direction (\iff) of the characterization in Theorem 2 was established in Section 4, c.f. Theorem 4.2. We must thus show the implication (\Longrightarrow) . This will be first done for exact Betti Lagrangians in Section 5.4.1, using the results of this section (c.f. Proposition 5.30) and then Section 5.4.2 is devoted to the meromorphic WKB case. To align with Definition 4.1, the existence of a pseudo-holomorphic disk in the adiabatic limit can be made precise as follows:

²²This modification of L is denoted by L^{\dagger} in [Ekh07; EHK16].

Definition 5.32. The pair of Lagrangians (T_z^*S, L) bounds a pseudo-holomorphic disk in the adiabatic limit if there exists a constant $\varepsilon_0 \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon_0)$ there exists a neighborhood $U_{\varepsilon}(z) \subset S$ and a point $w \in U_{\varepsilon}(z)$ in that neighborhood for which $(\varepsilon L, T_w^*S)$ bounds a non-constant pseudo-holomorphic disk.

The results established thus far, specifically Theorem 4.2 and Proposition 5.30, now suffice to conclude Theorem 2 for exact Betti Lagrangians.

5.4.1. Proof of Theorem 2 for exact Betti Lagrangians. Let $z \in W$ be a point in a wall of the spectral network, we must argue that there exists a pseudo-holomorphic strip between T_z^*S and L in the adiabatic limit. By Theorem 4.2, given any neighborhood $U \subset S$ of z, we can find a nested neighborhood $W \subset U$ and a constant $\varepsilon_0 \in \mathbb{R}_+$, depending on U, such that the pair of Lagrangians $(T_w^*S, \varepsilon L)$ is disk-free for all points $w \in U \setminus W$ and $\varepsilon \in (0, \varepsilon_0)$. Consider a W-adapted short path α_v with its endpoints in $U \setminus W$ and intersecting W and z. By the same argument as in the proof of Theorem 5.24, there exists some $\varepsilon'_0 \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon'_0)$ the Floer-theoretic parallel transport $\mathbb{F}(V)(\mathring{\alpha}_v(s))$ equals that of $\Phi_W(V)(\mathring{\alpha}_v(s))$, where $\mathring{\alpha}_v(s)$ denotes the lift to the unit cotangent bundle \mathring{S} . The latter parallel transport is non-trivial, because the 2d-4d BPS counting indices from Section 3.5 are non-zero at all walls of W, as W is creative. Thus the former parallel transport $\mathbb{F}(V)(\mathring{\alpha}_v(s))$ is non-trivial. This implies the existence of a point $w \in W$ such that $(T_w^*S, \varepsilon L)$ is not disk-free, as otherwise Proposition 5.30 would imply $\Phi_W(V)(\mathring{\alpha}_v(s))$) was trivial. In conclusion, as the adiabatic parameter $\varepsilon \to 0$ tends to zero, this concludes the existence of a pseudo-holomorphic strip between L and T_z^*S in the adiabatic limit, as required.

5.4.2. Proof of Theorem 2 for meromorphic spectral curves. There are a few differences between the case of a meromorphic spectral curve $\Sigma \subset (T^*S, \omega_{\mathbb{C}})$ and an exact Betti Lagrangian $L \subset (T^*S, \lambda_{st})$. That said, when studying pseudo-holomorphic curves with respect to a generic meromorphic spectral curve $\Sigma \subset (T^*S, \omega_{\mathbb{C}})$, we will now establish results showing that the behaviors that occur are as if we were under an exactness assumption. In that sense, the proof follows closely that of the exact case and we only highlight the key ingredients. First, we have the following strengthening of the Stokes condition in Definition 4.1.

Lemma 5.33. Let $z \in W$ be a non-initial point. Then, for every $E \in \mathbb{R}_+$, there exists a neighbourhood $U_z(E) \subset S$ of z and $\varepsilon_0(E, z) \in \mathbb{R}_+$ such that there are no non-constant J_g -holomorphic strips bounded between $\varepsilon \Sigma$ and T_w^*S of energy less than E, for $w \in W^c(E)$.

Proof. From Theorem 4.2, such pseudo-holomophic strips must have the homology class of the parallel transport of the solitons with energy less than E. Since the integral of the holomorphic Liouville 1-form along the boundary of such a strip must be real and positive, z must lie on $\mathcal{W}(E)$.

The following notation is useful:

Definition 5.34. Let $L \subset T^*S$ be a Betti Lagrangian. By definition, L is said to be disk-free in the adiabatic limit, if for every $E \in \mathbb{R}_+$ there exists $\varepsilon_0(E) \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon_0(E))$, the scaled Lagrangian εL does not bound a pseudo-holomorphic disk of energy less than equal to E.

Then the same argument used for Theorem 4.2 implies the following variation, which tells us that Σ is tautologically unobstructed, up to some energy. In other words, we can run the usual Floer theoretic machinery (see c.f [Sei15, Section 8(c)]) up to some energy cut-off.

Proposition 5.35. Let Σ be a meromorphic spectral curve with a WKB spectral network W. Then Σ is disk-free in the adiabatic limit.

Proof. Let u_{ε} be a sequence of somewhere injective disks of energy less than E with boundary on $\varepsilon \Sigma$. Introduce three punctures on the boundary to fix the conformal structure and apply the same adiabatic degeneration argument as in Theorem 4.2. By Section 2.7, the walls of W never return to the branch points and thus the initial edge condition implies that the disks cannot close up. Thus $\varepsilon \Sigma$ must be disk-free for small enough ε .

The scaling relations is as follows: the adiabatic cut-off $\varepsilon_0(E)$ tends to zero as the energy E is sent to infinity, i.e. $\varepsilon_0(E) \to 0$ for $E \to \infty$. The necessary variant of Proposition 5.26 can be discussed as follows.

Consider an energy cut-off $E \in \mathbb{R}_+$. The gapped condition implies that there exists a small enough parameter $h \in \mathbb{R}_+$ such that $\mathcal{W}(E)$ and $\mathcal{W}(E+h)$ are isotopic as stratified 1-manifolds with boundary. For each vertex v of the energy-truncated $\mathcal{W}(E)$, choose a Stokes neighborhood and a bump function ρ . Fix a neighborhood of v contained in that of Proposition 5.35, with diameter $C\delta$ for some constant $C \in \mathbb{R}_+$. From Proposition 5.35, the pair $(T_z^*S, \varepsilon\Sigma)$ is disk-free for $z \in B_{C\delta}(v) \setminus \mathcal{W}(E)$. Choose C such that $C\delta$ is less than the injectivity radius of v consider $\alpha(s)$ a geodesic line segment contained in $B_{C\delta}(v)$.

Let \mathfrak{w} be an (ij)-wall of $\mathcal{W}(E)$, $W := \int \lambda^i - \lambda^j = x + iy$ be the conformal flat coordinate and (p^x, p^y) dual coordinates to (x, y); here W is defined by a line integral from the given point to the end variable point. For $z \in \mathfrak{w}$, we write $\alpha_z(s)$ for the path given by $s \to z+is$ in the W-coordinate. Let $\sup_{\rho} |L|$ be the supremum of the norm of the sheets of L, over a small neighbourhood of ρ that contains the support of the generating Hamiltonian H^v . Then observe that the Hamiltonian term H^v in Eq. (31) is less than $\epsilon \sup_{\rho} |L|$. The analogue of Proposition 5.26 then reads:

Proposition 5.36. In the notation above, there exists $\varepsilon_0(z, E) \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon_0(z, E))$ the following holds:

(1) There exists $d(z,\varepsilon) \in \mathbb{R}_+$ and an elongation function such that, given a free W-adapted line segment $\alpha(s)$ contained in $B_{C\delta}$ of length ε , the moduli of J_g -holomorphic continuation strips of energy less than $\varepsilon(E + \hbar)$ are all small.

ROGER CASALS AND YOON JAE NHO

(2) For an interior point $z \in W(E) \cap B_{C\delta}$, there exists $\delta(z,\varepsilon) \in \mathbb{R}_+$ and an elongation function such that along the short path $\alpha_z(s)$, $s \in [-\delta(z,\varepsilon), \delta(z,\varepsilon)]$, any large J_g -holomorphic continuation strips of energy less than $\varepsilon(E + \hbar)$ must have the homotopy class of a soliton at z.

Proof. The argument follows that of Proposition 5.26. For (1), choosing the elongation function as in Proposition 5.26, since the Hamiltonian term contribution in the energy formula is of size $d \cdot |\sup_{\rho} L| \epsilon \cdot \int l'(s) ds$, for d small enough any pseudo-holomorphic strip bounded by $(T_z^*S, \varepsilon \Sigma)$ which comes from breaking off a continuation strip must have energy less than εE . Therefore, for any $s_0 \in [0, 1]$, there exists $\eta(s_0) \in \mathbb{R}_+$ such that the moduli of J_g -continuation strips along $[s - \eta(s_0), s + \eta(s_0)]$ is cobordant to the moduli of trivial strips. For (2), a similar argument applies: the key point is that for J_g , we can fix the homotopy classes of continuation strips with energy bounded above by $\leq E + \delta$.

Since there are finitely many vertices in $\mathcal{W}(E)$, let us choose $\delta \in \mathbb{R}_+$ such that Lemma 5.33 and Proposition 5.36 hold. Given an (ij)-wall \mathfrak{w} , which is compact, we choose a (small) neighborhood $U_{\mathfrak{w}} \subset S$ of \mathfrak{w} so that Lemma 5.33 holds for all $w \in U_{\mathfrak{w}} \setminus \mathcal{W}(E)$. We also (slightly) truncate \mathfrak{w} off near the starting ends if \mathfrak{w} begins at the initial vertices of $\mathcal{W}(E)$. The following proposition has exactly the same proof as Proposition 5.36:

Proposition 5.37. In the notation above, there exists $\varepsilon_0(v) \in \mathbb{R}_+$ such that, for all $\varepsilon \in (0, \varepsilon_0(v, E))$, there exists $\delta(\mathfrak{w}, \varepsilon) > 0$ such that the following conditions hold for strips along the path $s \to z + is$, $s \in [-\delta(\mathfrak{w}, \varepsilon), \delta(\mathfrak{w}, \varepsilon)]$:

- (1) For any $z \in U_{\mathfrak{w}} \setminus \mathcal{W}$, the J_g -holomorphic continuation strips of energy less than $\varepsilon(E + \hbar)$ are all small.
- (2) For any $z \in \mathfrak{w} \cap U_{\mathfrak{w}}$, then the large J_g -holomorphic continuation strips of energy less than $\varepsilon(E + \hbar)$ all have the homotopy class of a soliton at z.

In general, J_g -holomorphic continuation strips are not going to be regular in this content. Fortunately, given the energy bound, all the relevant moduli space of continuation strips are cobordant with the given energy cut-off (here the perturbation of the almost complex structure must be chosen in a way that it agrees with J_g on the boundary mapping to L, to avoid boundary bubbling, c.f [Sei15, Section 8(c)]). Indeed, the energy formula for short enough arc implies that the dominant term in the energy is $\int \lambda_{st}$ over the boundary mapping to εL , the Hamiltonian term being proportionately small, which is invariant up to homotopy. Therefore, the energy of the perturbed continuation strip is determined by the boundary homotopy class.²³ It follows that we can regularize the moduli space by choosing some unspecified perturbation of J_g to make the strips regular. The following terminology is convenient:

 $^{^{23}}$ See also the energy estimates in [Fuk+09b, Section 5.3.5]; in their terminology, the Hofer norm of the Hamiltonian isotopy as observed by the disk is small enough.

Definition 5.38. Let \mathfrak{w} be an active wall of W(E) and $z \in \mathfrak{w}$. By definition, z supports a disk in the adiabatic limit if for any $E \in \mathbb{R}_+$ there exists $\varepsilon(z) \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon_0(z))$ there exists $\delta(z, \varepsilon) \in \mathbb{R}_+$ such that the moduli space of large J_g -holomorphic continuation strip along $\alpha_z(s), |s| < \delta(z, \varepsilon)$, of energy less than εE is non-empty and its regularization is not cobordant to the empty set.

By definition, a wall \mathfrak{w} is said to support a non-constant disk in the adiabatic limit if the set of points $z \in \mathfrak{w}$ that support a disk in the adiabatic limit is non-empty.

Proof of Theorem 2 in meromorphic case. First, consider a wall \mathfrak{w} of W(E). Let us show that the set of points on \mathfrak{w} that support a disk in the adiabatic limit is open and closed. Indeed, openness is a consequence of Gromov compactness and regularity of strips being an open condition. For closedness, note that [Nho24, Proposition 4.4] shows the set is non-empty. Consider then a sequence of points $z_n \to z$ such that each support a disk in the adiabatic limit. For large enough $n \in \mathbb{N}$, z_n belongs to the Stokes neighborhood of z and, by assumption, the moduli space of non-diagonal J_g -holomorphic continuation strips along $\alpha_{z_n}(s)$ is non-empty. Concatenate the horizontal arc coming from the top of $\alpha_z(s)$ to the arc $\alpha_{z_n}(s)$, and further concatenate it with the horizontal arc connecting to the bottom of $\alpha_z(s)$. The resulting path is isotopic to the arc $\alpha_z(s)$. Now, any two moduli space of J_g -holomorphic continuation strips with given energy less than $\varepsilon(E + \hbar)$ are cobordant, and since the moduli space of large J_g -holomorphic continuation strips along $\alpha_{z_n}(s)$.

Second, we discuss vertices of the spectral network. Suppose that the active ingoing walls all support disks in the adiabatic limit. We claim that this is also the case for the active outgoing walls. Indeed, by Proposition 5.36, for a small neighborhood of v the moduli space of J_g -holomorphic continuation strips for the short paths across the outgoing walls is virtually cobordant to the gluing of the ingoing moduli spaces. The count of these gluings is controlled by the index in Definition 3.20, and so we are reduced to the exact situation.

6. Spectral Networks and 4d Fukaya Categories

In this section we relate spectral networks and Fukaya categories in 4-dimensions, in particular proving Theorem 4. Specifically, we show how exact Betti Lagrangians can be understood as objects of a Fukaya category in two related ways, cf. Sections 6.1 and 6.3, and how the parallel transport rules dictated by a spectral network \mathcal{W} are in fact encoding μ_2 -type morphisms for the underlying A_{∞} -structures naturally appearing in Floer theory, cf. Theorem 6.3, proven in Section 6.2, and Theorem 6.12, proven in Section 6.3. An explicit description of the associated Yoneda A_{∞} -module over $\mathfrak{W}(T^*S, \Lambda)$ is also provided in Section 6.4. See [GPS20] for references and context on (partially) wrapped Fukaya A_{∞} -categories. In brief, the present section enhances the Floer-theoretic characterization of spectral networks and non-abelian parallel transport, as presented in Sections 4 and 5, to the A_{∞} -categorical framework. In this section, we will focus our attention on exact Betti Lagrangians.

6.1. The Yoneda module of an exact Betti Lagrangian. Let $\mathfrak{W}(T^*S)$ denote the wrapped Fukaya A_{∞} -category of the open Liouville sector (T^*S, λ_{st}) , and mod_k the dg-category of chain complexes of k-modules, understood as an A_{∞} -category with higher μ_k vanishing, $k \geq 2$. A Betti Lagrangian $L \subset (T^*S, \omega_{st})$ does not define an object in $\mathfrak{W}(T^*S)$, due to the nature of its asymptotic ends near the marked points, cf. Section 2. Thus, even if we endow a Betti Lagrangian Lwith a local system²⁴ $V \in \operatorname{Loc}(L)$, it is a priori unclear how Betti Lagrangians and their spectral networks fits within the context of Fukaya categories. There are at least two solutions to address this, depending on whether we want to preserve the geometry of L at infinity or we allow geometric modifications of L at infinity. The first approach is presented in this section and the second in Section 6.3.

For the first approach, the algebraic idea is simple: if $C \in Ob(\mathcal{C})$ is an object in a category, then Hom_{\mathcal{C}} $(\cdot, C) \in Fun(\mathcal{C}^{op}, \operatorname{mod}_k)$ defines a module over \mathcal{C} . This module is often known as the Yoneda module. Since a Betti Lagrangian $L \notin Ob(\mathfrak{W}(T^*S))$ is not an object, as it does not have compact horizontal support, we cannot merely construct a Yoneda module in this manner. Nevertheless, given a Betti Lagrangian (L, V) endowed with a local system $V \in \operatorname{Loc}(L)$, we now construct an A_{∞} -module over $\mathfrak{W}(T^*S)$ which behaves as if it was the Yoneda module of an object given by (L, V). Intuitively, its value on objects $P_1, \ldots, P_k \in Ob(\mathfrak{W}(T^*S))$ is given by counting pseudoholomorphic polygons with boundary on the Lagrangians P_1, \ldots, P_k and the Betti Lagrangian L. Consider the Floer cohomology groups $HF^*(P, (L, V))$ as in Section 5, the result reads as follows:

Proposition 6.1. Let $L \subset (T^*S, \omega_{st})$ be a Betti Lagrangian endowed with a local system $V \in Loc(L)$. Then there exists an A_{∞} -module $\mathcal{Y}_{(L,V)} : \mathfrak{W}(T^*S) \longrightarrow mod_k$ with cohomologies

$$H^*(\mathcal{Y}_{(L,V)}(P)) \cong HF^*(P,L),$$

where $P \subset T^*S$ is any exact cylindrical Lagrangian with compact horizontal support.

Proof. The intuition is to formally add the Betti Lagrangian $L \subset T^*S$ as the most negative object in $\mathfrak{W}(T^*S)$. We provide the necessary details, as follows. By [GPS20, Prop. 3.43], the wrapped Fukaya A_{∞} -category $\mathfrak{W}(T^*S)$ can be obtained as the strictly unital directed A_{∞} -category of a certain (terminal) decorated poset $\overrightarrow{P}(S)_{ter}$ for (T^*S, ω_{st}) localized at the set of continuation morphisms, as discussed in Section 3.8 in ibid. To begin, consider a decorated poset $\overrightarrow{P}(S)$, not necessarily terminal, which satisfies the following conditions:

 $^{^{24}}$ Since [GPS20] uses local systems, not twisted, we use spin structures for Betti Lagrangians to orient. The relation between twisted local systems and spin structures is discussed in Appendix A.3. We can use Corollary 5.23 to pass from spin structure count to twisted count. Alternatively, one could develop [GPS20] for twisted local systems.

- (a) Each Lagrangian P_k in the poset is transverse to L, and each family of almost complex structure can be extended to an almost complex structure on T^*S , that agrees with J_{cyl} outside some uniform compact base.
- (b) For each chain $p_0 > p_1 > \ldots > p_k$, the moduli space of pseudo-holomorphic polygons bounded by the (ordered) tuple of Lagrangians $(P_{p_0}, P_{p_1}, \ldots, P_{p_k}, L)$ is also transversely cut-out and compatible with gluing.

Such decorated poset exists after perturbing the almost complex structure. This requires the fact that the relevant pseudo-holomorphic polygons do not escape off to infinity. Namely, that given the exact Betti Lagrangian L, the (pairwise transverse) cylindrical Lagrangians $P_1, \ldots, P_k \subset T^*S$ and $\{J_s\}$ a family of almost complex structures on T^*S , we need that any J_s -holomorphic polygon with boundary conditions given by (P_1, \ldots, P_k, L) has uniformly bounded diameter, and the same holds with moving Lagrangian boundary conditions on P_1, \ldots, P_k given by positive isotopies. This geometric condition does hold in our case: L is contained in a finite-radius disk bundle, it is uniformly geometrically bounded with respect to J_s and, being exact, the energy of such polygons is a priori bounded by the action of the intersection points. Therefore, the monotonicity argument in [Nho24, Proposition 3.29] applies to guarantee such condition.

We then construct an A_{∞} -module $\mathcal{Y}_{(L,V)}^{pre}$ over the poset in $\overrightarrow{P}(S)$ with its A_{∞} -operations given by counting pseudo-holomorphic polygons between the Lagrangians $\{P_i\}$ in $\overrightarrow{P}(S)$. For such chain of Lagrangians $\{P_i\}$, we define the map

$$\mu_{k-1|1}: \mathcal{O}(P_1, P_2) \otimes \ldots \mathcal{O}(P_{k-1}, P_k) \otimes \mathcal{Y}_{(L,V)}^{pre}(P_k) \longrightarrow \mathcal{Y}_{(L,V)}^{pre}(P_1), \quad P_1 > P_2 \ldots > P_k$$

by counting pseudo-holomorphic (k + 1)-gons with boundary conditions given by $(P_1, ..., P_k, L)$, with \mathcal{O} as in Eq. (3.52) in ibid. We set the A_{∞} -module map to be equal to zero otherwise, except for the case $\mu_{1|1} : \mathcal{Y}_{(L,V)}^{pre}(P) \longrightarrow \mathcal{Y}_{(L,V)}^{pre}(P)$ where $\overrightarrow{P}(P, P) = \mathbb{Z}$ acts as the identity. The above defines $\mathcal{Y}_{(L,V)}^{pre}$ as A_{∞} -module over the poset in $\overrightarrow{P}(S)$, we need to argue that it descends to the localization $\mathfrak{W}_{\overrightarrow{P}(S)}$ by continuation morphisms. For that, note that given any positive isotopy of cylindrical Lagrangians $\{K_t\}, t \in [0, 1]$, the continuation map $CF(K_0, L) \to CF(K_1, L)$ is invertible. Indeed, since L lies in a finite disk bundle, L and K_t do not intersect at infinity, thus the conclusion follows from the argument in [GPS20, Lemma 3.21] and the final assertion in [GPS20, Lemma 3.26]. It also follows that $C^*(\mathcal{Y}_{(L,V)}^{pre}(P)) \simeq CF^*(P, L)$.

It now suffices to relate $\mathfrak{W}_{\overrightarrow{P}(S)}$ to $\mathfrak{W}(T^*S)$. For that, consider the category of decorated posets which satisfy (a) and (b) above, are downward-closed and duplicate-free. Denote its terminal object by $\overrightarrow{P}(S,L)_{ter}$, cf. [GPS20, Lemma 3.42]. The inclusion $\overrightarrow{P}(S,L)_{ter} \subset \overrightarrow{P}(S)_{ter}$ of posets induces an inclusion functor between the associated localized categories, and thus $\mathfrak{W}_{\overrightarrow{P}(S)}$ admits a tautological inclusion functor into $\mathfrak{W}(T^*S)$, which is in fact an equivalence. By precomposing $\mathcal{Y}_{(L,V)}^{pre}$ with the inverse of this equivalence, we obtain the required A_{∞} -module $\mathcal{Y}_{(L,V)}$. **Remark 6.2.** We use the following convention for composition and wrapping: the μ_2 -maps are given as

$$\mu_2: CF(A, B) \otimes CF(B, C) \to CF(A, C)$$

and, when wrapping is involved, a morphism $A \to B$ has its domain A more positive than its codomain B, i.e. positive wrapping gives $CF(A^w, A) \otimes CF(A, B) \to CF(A^w, B)$. In particular, the Betti Lagrangian $L \subset T^*S$, e.g. a spectral curve, is always taken on the right slot of $CF(\cdot, \cdot)$, as it is the most negative object. Similarly, in the partially wrapped setting, the positive push-off of the (cylindrized) Betti Lagrangian is the most negative object, in that small negative wrapping makes it fall into $\partial_{\infty}L$.

In the same line as the equivalence established in Section 5.3, cf. Theorem 5.24, the A_{∞} -modules $\mathcal{Y}_{(L,V)}$ in Proposition 6.1 are closely related to the non-abelianized local systems, as follows. First, for technical reasons, we normalize the Riemannian metric (S,g) so that $\operatorname{inj}(g) > 2$, which is achieved after a constant metric rescaling $g \to c^2 g$; this is necessary for the proof of Proposition 6.6 to estimate action. Now, by definition, two points $z, w \in (\mathbf{S}, g)$ in a Betti surface with a spectral network $\mathcal{W} \subset S$ are said to be \mathcal{W} -adapted if $\operatorname{dist}_g(z,w) < \operatorname{inj}(g)/20$ and the minimal geodesic α_{zw} between z, w is a \mathcal{W} -adapted path (cf. Definition 5.4). Since the metric (S,g) is complete, we can and do choose w in terms of z such that the energy functional on the path space $\Omega_{z,w}$ between z and w is a Morse functional, cf. [Mil63, Theorems 14.2, 18.1 & 18.2]. This ensures the transversality of intersections of cotangent fibres in (T^*S, ω_{st}) after wrapping by the geodesic flow.

We implicitly identify $\operatorname{Hom}_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S) := CF^*(T_z^*S, T_w^*S)$ with chains on the path space $\Omega_{z,w}S$, see [Abo12]. In fact, $CF^*(T_z^*S, T_z^*S)$ is A_∞ -quasi-isomorphic to $C_{-*}(\Omega_z S)$ endowed with the A_∞ -refined Pontryagin product. The reason we focus on cotangent fibers is that [Abo11a] shows that any cotangent fiber $T_z^*S \subset (T^*S, \omega_{st})$ generates $\mathfrak{W}(T^*S)$, and thus we have an A_∞ -quasi-equivalence $\operatorname{Tw}(\mathfrak{W}(T^*S)) \simeq \operatorname{Tw}(CF^*(T_z^*S, T_z^*S))$. We denote by

$$\mu_{1|1} : \operatorname{Hom}_{\mathfrak{W}(T^*S)}(T^*_zS, T^*_wS) \otimes \mathfrak{Y}_{(L,V)} \longrightarrow \mathfrak{Y}_{(L,V)}$$

the structure map of the A_{∞} -module $\mathcal{Y}_{(L,V)}$ in these degrees evaluated at two cotangent fibers. The categorical statement relating Floer theory to non-abelianized local system reads as follows:

Theorem 6.3. Let (S,g) be a Betti surface, $L \subset (T^*S, \omega_{st})$ a Betti Lagrangian, $V \in Loc(L)$, $W \subset S$ an adapted spectral network and $z, w \in S$ an W-adapted pair. Then the minimal geodesic generator $[\alpha_{zw}] \in Hom_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S)$ satisfies

(33)
$$\mu_{1|1}([\alpha_{zw}], \cdot) = \Phi_{\mathcal{W}}(V)(\alpha_{zw}).$$

In brief, Eq. (33) equates a map given by Floer-theoretically counting pseudo-holomorphic strips, on its left, to the topological parallel transport given by the spectral network, on the right. Note that the left hand side of the isomorphism Eq. (33) can be identified with $\mathcal{Y}_{(L,V)}([\alpha_{zw}])$, understanding $\mathcal{Y}_{(L,V)}$ as being applied to a morphism in $\mathfrak{W}(T^*S)$. Thus the structure map $\mu_{1|1}$ appearing in Theorem 6.3 categorically (and cohomologically) describes the parallel transport map from z to w when its domain is evaluated at $\operatorname{Hom}_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S) \otimes \mathcal{Y}_{(L,V)}(T_z^*S)$. Theorem 6.3 will be proven in a moment, in Section 6.2, after we discuss a useful corollary.

Given the A_{∞} -quasi-equivalence $\operatorname{Tw}(\mathfrak{W}(T^*S)) \simeq \operatorname{Tw}(CF^*(T_z^*S, T_z^*S))$, the right hand side is a model for $\operatorname{Loc}(S) \simeq C_{-*}(\Omega_z S)$ -mod. Thus, a local system in $\operatorname{Loc}(S)$ can be understood as a $\mathfrak{W}(T^*S)$ -module. It is in general challenging to explicitly present A_{∞} -modules over $\mathfrak{W}(T^*S)$, as it involves solving partial differential equations, significant preliminary Floer data to be specified and homotopy (co)limits need to be computed. An advantageous consequence of Theorem 6.3 is that we can explicitly describe the class of A_{∞} -modules over $\mathfrak{W}(T^*S)$ arising as non-abelianized local system via the A_{∞} -quasi-equivalence above. Specifically, under these identifications, we have the following

Corollary 6.4. Let (S, g) be a Betti surface, $L \subset (T^*S, \omega_{st})$ a Betti Lagrangian, $V \in Loc(L)$, and $\mathcal{W} \subset S$ a spectral network adapted to L. Then, under the A_{∞} -quasi-equivalence $\mathfrak{W}(T^*S)$ -mod $\cong C_{-*}(\Omega_z S)$ -mod,

$$\mathcal{Y}_{(L,V)} \simeq \Phi_{\mathcal{W}}(V),$$

as A_{∞} -modules over $\mathfrak{W}(T^*S)$. That is, the A_{∞} -module $\mathcal{Y}_{(L,V)} : \mathfrak{W}(T^*S) \longrightarrow mod_k$ is homotopic to the A_{∞} -module corresponding to $\Phi_{\mathcal{W}}(V) : Loc(S) \longrightarrow mod_k$.

6.2. Proof of Theorem 6.3 (Floer $\mu_{1|1}$ is spectral transport). We need to compute the wrapped Floer complex using a model that is also suitable for computing the $\mu_{1|1}$ -maps in the statement of Theorem 6.3. For that, we introduce such a model in Section 6.2.1, then compute the structure maps in Section 6.2.3, and conclude the argument in Section 6.2.4.

6.2.1. A model for computing the wrapped Floer complex. Given two nearby points $z, w \in S$, the goal of this subsection is to establish that appropriately chosen positive isotopies applied to T_z^*S exactly lead to a degree-0 cycle in $\operatorname{Hom}_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S)$ represented by the minimal geodesic α_{zw} between z and w, with this morphism complex given by direct limits as in [GPS20].

Recall that an exact isotopy $\{P\}_{t\in[0,1]}$ of cylindrical Lagrangians is said to be positive (at infinity), denoted by $P_0 \rightsquigarrow P_1$, if $\alpha(\partial_t(\partial_\infty P_t)) > 0$, where $\partial_\infty P_t$ denotes the boundary at infinity of P_t and α is a given contact form. Note that a positive isotopy $P_0 \rightsquigarrow P_1$ induces an element $c \in CF(P_1, P_0)$, called the continuation element, cf. [GPS20, Section 3.3]. The relationship between continuation elements and continuation maps is as follows. If $P_0 \rightsquigarrow P_1$ and $K \subset T^*S$ is a Lagrangian with K, P_0, P_1 pairwise transverse, then the continuation map

$$c(P_t): CF(P_0, K) \to CF(P_1, K)$$

associated to the continuation element $c \in CF^0(P_1, P_0)$ is chain homotopic to the evaluation $\mu_2(c, \cdot)$ of the map

$$\mu_2: CF(P_1, P_0) \otimes CF(P_0, K) \to CF(P_1, K)$$

of the A_{∞} -structure at the continuation element c. In fact, this even holds if we fix P and move K by a negative isotopy. Indeed, [GPS20, Lemma 3.26] proves this for K, P having compact horizontal support; for K an exact Betti Lagrangian, the same argument works because of the existence of uniform diameter bounds, as discussed in the proof of Proposition 6.1. Thus we can and do implicitly switch between continuation elements and continuation maps.

In general, the wrapped Floer cohomology groups defined in [GPS20, Section 3.4] are given as the direct limit

$$HW^*(P,K) := \lim_{(P \rightsquigarrow P^w)^+} HF^*(P^w,K)$$

over (cofinal) sequences of positive isotopies. Our task is to choose a suitable such cofinal sequence, which we explicitly present as a sequence $P_0 \rightsquigarrow P_1 \rightsquigarrow P_1 \rightsquigarrow \cdots$ of Lagrangians. We often denote $HF^*(P, K) = HW^*(P, K)$ when it is clear by context that wrapping must be taken: we always consider a positive wrapping of P, in the sense above, though one can equivalently negatively wrap K. To define our cofinal sequence, it is convenient to introduce the following notation:

Definition 6.5 (Wrapped fibres and fibre wrappings). A cylindrical Lagrangian $F \subset (T^*S, \omega_{st})$ is said to be a wrapped fibre if there exists a point $z \in S$ and a positive isotopy $T_z^*S \rightsquigarrow F$ supported on the complement of the unit disk bundle D^*S . In this case, the point $z \in S$ is said to be the base point of F. By definition, a positive isotopy $\{F_t\}_t$ is called a fibre wrapping if F_t is a wrapped fibre for each t, and $\{F_t\}$ are uniformly cylindrical and have uniform horizontal support. In this case, the path of base points of $\{F_t\}$ in S is said to be its support arc.

In order to prove Theorem 6.3 we need a cofinal sequence that relates to minimal geodesics: intuitively, we will consider a cofinal sequence given by fibre wrapping such that the associated continuation elements map to the minimal geodesics. For that, we need to cover T^*S by sectorial covers, as defined in [GPS24b, Section 12.5], and we achieve that by modifying the given metric (S,g), as follows. Given the spectral network $W \subset S$ adapted to (S,g), we consider a finite good cover $\{U_{\alpha}\}$ of **S** such that each U_{α} contains at most one vertex of W, and the open sets U_{α} have their closures contained in S, except for a distinguished open set for each puncture of **S**, which we choose to be contained in the region where the metric is conical. We call the open sets of the former type the "internal" open sets. Then, we choose a collar neighborhood of each $\partial U_{\alpha} \subset S$ with a diffeomorphism to $S^1 \times (0, 1)$ such that:

- ∂U_{α} is mapped to $S^1 \times \{1/2\},\$
- $\partial U_{\eta}, \eta \neq \alpha$, and the edges of \mathcal{W} intersect the neighborhood $S^1 \times (0,1)$ of ∂U_{α} as an arc of the form $\{p\} \times (0,1), p \in S^1$ a point.

With this cover chosen, we modify the metric (S, g) near each such collar of ∂U_{α} so it is of the form $g_{S^1} + dt \otimes dt$, where g_{S^1} is the standard round metric on S^1 . This can be achieved by an arbitrarily C^0 -small compactly supported perturbation of g in a manner that $\mathcal{W} \subset S$ stays invariant, up to smooth isotopies supported away from the boundaries of these disks U_{α} . Note that for two W-adapted points $z, w \in W^c$, the length of the second shortest geodesic will be bounded below by the minimal injectivity radius inj(q) of this perturbed metric. For the rest of this section, we assume the metric (S, q) has been modified as just described, and still denote this perturbed metric by (S, q).

The sequence of fibre wrappings that we use to prove Theorem 6.3, whose continuation elements limit to the minimal geodesics, is constructed in (the proof of) the following result:

Proposition 6.6 (Fibre wrappings and minimal geodesics). Let $F_z, F_w \subset (T^*S, \omega_{st})$ be wrapped fibres based at $z, w \in S$, two close enough generic points. Then, there exists a sequence of fibre wrappings

$$F_z \rightsquigarrow F_1 \rightsquigarrow F_2 \rightsquigarrow \ldots$$

with the base point of F_1 equal to z such that:

- (1) $HW^*(F_z, F_w) \cong \lim_{i \to \infty} HF(F_i, F_w),$ (2) the image in $HW^0(F_z, F_w)$ of the continuation elements in $HF^0(F_i, F_w)$ is the class $[\alpha_{zw}] \in$ $C_0(\Omega_{z,w})$ represented by the minimal geodesic α_{zw} between z, w.

Proposition 6.6.(2) is the part that requires a new type of argument, whereas Proposition 6.6.(1)can be reasonably obtained by modified existing arguments in the literature, as the proof will show, cf. Lemma 6.8 and 6.9 below.

6.2.2. Proof of Proposition 6.6. First, we need to introduce the following notion of cofinality:

Definition 6.7. Let $P_t \subset (T^*S, \omega_{st})$ be a positive isotopy of cylindrical Lagrangians, $t \in [0, \infty)$, and $\{\overline{\mathbf{S}}_{\mathbf{t}}\}\$ a nested family of truncations of S such that $P_t \subset \overline{\mathbf{S}}_t$ and $\bigcup \overline{\mathbf{S}}_t = \mathbf{S}$. By definition, a such a pair $(P_t, \overline{\mathbf{S}}_t)$ is said to be ind-cofinal if:

- (1) For any isotopy $P_0 \to P^w$, there exists some $t \in \mathbb{R}_+$ and a positive isotopy $P^w \rightsquigarrow P_t$ such that the composition of $P_0 \to P^w$ with $P^w \rightsquigarrow P_t$ is homotopic to a positive isotopy $P_0 \rightsquigarrow P_t$. Furthermore, both the isotopy and the homotopy can be taken to be supported in the interior of $T^*\mathbf{S}_t$.
- (2) Any two positive isotopies $P_0 \rightsquigarrow P_t$ coincide, up to homotopy, after composing with $P_t \rightarrow$ P_s , for some large s > t. Furthermore, the homotopy (of positive isotopies) can be taken to be supported in the interior of $T^*\overline{\mathbf{S}}_{\mathbf{s}}$.

A positive isotopy P_t is said to be ind-cofinal if there exists $\{\overline{\mathbf{S}}_t\}$ such that $(P_t, \overline{\mathbf{S}}_t)$ is ind-cofinal.

As in the non-limit case, the ind-cofinality in Definition 6.7 of a positive isotopy P_t can be guaranteed by forcing the boundaries $\partial_{\infty} P_t$ to move fast enough in a Reeb direction in the ideal contact boundary of (T^*S, ω_{st}) . In precise terms:

Lemma 6.8. Let $P_t \subset (T^*S, \omega_{st})$ be a positive isotopy. Then, P_t is ind-cofinal if there exists a contact form α on the unit sphere bundle, coinciding with the restriction of λ_{st} outside a compact set, such that

(34)
$$\int_0^\infty \inf_{\partial_\infty P_t} \alpha(\partial_t(\partial_\infty P_t)) dt = \infty.$$

In particular, the time-t the geodesic flow at infinity applied to a Lagrangian $P \subset (T^*S, \omega_{st})$ gives an ind-cofinal family.

Proof. It suffices to modify [GPS20, Lemma 3.29]. Since the metric is conical at infinity, the associated geodesic flow is complete. By the hypothesis Eq. (34), we can reparametrize the family P_t in the *t*-variable such that $\alpha_{st}(\partial_t(\partial_\infty P_t)) \geq 2$, α_{st} being the restriction of λ_{st} to the ideal contact boundary, and the horizontal support of P_t is contained in $\overline{\mathbf{S}}_t = \mathbf{U}_{\mathbf{r} > \mathbf{f}(t)}^{25}$ for some function $f : \mathbb{R}_+ \to \mathbb{R}_+$ such that f(t) > 3t.

Since $\alpha_{st}(\partial_t(\partial_\infty P_t)) \geq 2$, then

$$P^0 \rightsquigarrow \phi_{-t}(P^t)$$

is a positive isotopy because $\alpha_{st}(\partial_t(\partial_{\infty}(\phi_{-t}P_t))) \geq 1$, where ϕ_t is the geodesic flow. Independently, we can use the geodesic flow on the inverse isotopy $P^w \to P^0$ to also make it positive and, since the metric is conical at infinity, we can ensure $\phi_t(T^*\overline{\mathbf{S}}_{\mathbf{s}}) \subset \overline{\mathbf{S}}_{\mathbf{s}+\mathbf{t}}$. Now that $P^w \to \phi_t(P^0)$ has been made positive, note that

$$\phi_t(P^0) \rightsquigarrow (\phi_t \circ \phi_{-t})(P^t) = P^t$$

is positive because $P^0 \rightsquigarrow \phi_{-t}(P^t)$ is positive. This establishes property Definition 6.7.(1). For Definition 6.7.(2) the argument is the same as GPS, with the truncation being preserved.

Lemma 6.8 gives an ind-cofinal sequence, as in Definition 6.7, and the first part Proposition 6.6.(1) is then a consequence of the following lemma:

Lemma 6.9. Let P and K be Lagrangians in T^*S with compact horizontal support. Let $\{P_j\}$ be an ind-cofinal sequence with $P_0 = P$, then

$$HW(P, K) = \lim_{j \to \infty} HF(P_j, K)$$

Proof. The sequence P_i regarded as a decorated poset is cofinal in the sub-poset of the final decorated poset P_{final} given by those P_p that belong to the wrapping category of P_0 , c.f. [GPS20, Section 3.4]. Therefore, the natural map $HF(P_j, K) \to HW(P, K)$ is an isomorphism in the limit.

Let us now focus on establishing Proposition 6.6.(2). The difficulty here is naming the necessary geometric objects in a homotopically coherent manner. To wit, we must be able to describe the geodesic generator $\alpha_{z,w}$ within the framework of sectorial coverings of [GPS20], which is itself a

²⁵Here r is the one over the distance to the puncture, so puncture corresponds to $r = \infty$.

highly non-explicit setting. For that, consider the finite good cover from Section 6.2.1 and the correspondingly perturbed metric (S, g). Then the Liouville sectors $\{T^*\overline{U}_{\alpha}\}_{\alpha}$ form a sectorial covering of $T^*\overline{\mathbf{S}}^{26}$ and we can consider the associated wrapped Fukaya categories $\mathfrak{W}(T^*\overline{U}_{\alpha})$ on each of these sectors. In this particular case, using the argument similar to [GPS24a, Section 5.6], we can directly compute wrapped chain complexes as follows. For an internal open set U_{α} , and each $z \in U_{\alpha}$, we consider a cofinal fibre wrapping $F_{z,t}$ of T_z^*S with the support arc given by z, such that for t large enough, the ideal boundary of $F_{z,t}$ is given by some inward push-off of the conormal of ∂U_{α} . These $F_{z,t}$ give an explicit description of $\mathfrak{W}(T^*\overline{U_{\alpha}})$ since

$$HF^*(T_z^*S, T_{z'}^*S) = HF^0(T_z^*S, T_{z'}^*S) \cong \mathbb{Z}[\gamma_{zz'}], \quad \forall z, z' \in U_\alpha.$$

In particular, we can identify $HF^*(T_z^*S, T_{z'}^*S)$ with the unique path homotopy class in $\pi_1(U_\alpha; z, z')$. Furthermore, if $z, z', z'' \in U_\alpha$, then the A_∞ -structure map is

$$\mu_2: HF^*(T_z^*S, T_{z'}^*S) \otimes H^*(T_{z'}^*S, T_{z''}^*S) \to H^*(T_z^*S, T_{z''}^*S), \quad \mu_2([\gamma_{zz'}], [\gamma_{z'z''}]) = [\gamma_{zz''}].$$

Note also that we have inclusion functors $(i_{\alpha})_* : \mathfrak{W}(T^*\overline{U_{\alpha}}) \to \mathfrak{W}(T^*S)$ and [GPS24b, Theorem 1.35] implies that the natural functor

(36)
$$\operatorname{hocolim}(i_{\alpha})_{*} : \operatorname{hocolim}_{\alpha_{1},\dots,\alpha_{k}} \mathfrak{W}(T^{*}U_{\alpha_{1}\dots\alpha_{k}}) \to \mathfrak{W}(T^{*}S)$$

is a pre-triangulated equivalence, where $U_{\alpha_1,\ldots,\alpha_k} := U_{\alpha_1} \cap \ldots \cap U_{\alpha_k}$ and the homotopy category of the homotopy colimit is naturally identified with the fundamental groupoid of S.

Proof of Proposition 6.6. As said above, Part (1) follows from Lemma 6.8. For Part (2), let $x, y \in U_{\alpha}$ such that y is within the distance $\frac{1}{20}$ inj(g) from x. By the prior discussion, we have a generator $[\gamma_{xy}]$ that represents the minimal geodesic in the homotopy colimit 36. We must now compute the image of $[\gamma_{xy}]$ under the functor hocolim $(i_{\alpha})_*$ and show that, for careful choices, the minimal geodesic generator indeed gets preserved. First, we claim that there are fibre wrappings $\{F_k\}_k$ of T_x^*S such that

- (i) the generators of $CF(F_k, T_y^*S)$ correspond to geodesics between T_x^*S and T_y^*S ,
- (ii) the wrapping continuation map $CF(F_k, T_y^*S) \to CF(F_{k+1}, T_y^*S)$ preserves the minimal geodesic element.

Intuitively, these can be obtained by considering globally positive Hamiltonians H_k that are linear at infinity but quadratic in a large compact part of the cotangent bundle and using them to flow T_x^*S . Part (2) follows from the existence of such fibre wrappings, so let us prove the claim, as

 26 Using the product decomposition near the boundary, we obtain the projection structure

(35)
$$Nbh^{Z}(\partial T^{*}(U_{\alpha})) \simeq T^{*}(\partial U_{\alpha}) \times T^{*}[0.5, 1]$$

where we identify $T^*[0.5,1) \simeq \mathbb{C}_{0.5 \leq Re(z) < 1}$. Unlike the setting in [GPS20], here we equip \mathbb{C} with the Liouville structure obtained from $\mathbb{C} \simeq T^*\mathbb{R}$. The argument [GPS20, Lemma 2.41] that prevents pseudo-holomorphic curves from escaping off the boundary applies for almost complex structures on \mathbb{C} obtained from the conically deformed Sasaki metric on $T^*\mathbb{R}$ coming from the Euclidean metic. Therefore, we may impose the constraint that the almost complex structure on T^*U_{α} coincides with J_{cyl} near the boundary.

follows.

Let $\{\ell_i\} \to +\infty$ be an increasing sequence of positive real numbers, with $\ell_1 > 2$, such that no ℓ_i is in the length spectrum of geodesics between x and y and the minimal geodesic is the only geodesic with length less than ℓ_1 . (This is possible because we normalized so that $\operatorname{inj}(g) > 2$.) For each $k \ge 0$, consider $\delta_k \in \mathbb{R}_+$ such that $[\ell_k - \delta_k, \ell_k + \delta_k]$ is disjoint from the length spectrum and, for each δ_k , choose a smooth increasing positive function $H_k : [1, \infty) \to [0, \infty)$ such that

- $H'_k(r) = 0$ on $[1, 1 + \frac{1}{80}],$
- $H_k(r) = \frac{1}{2}r^2$ on $r \in [1 + \frac{1}{40}, \ell_k \delta_k],$
- $H_k(r) = \overline{\ell}_k \cdot r$ for $r \in [\ell_k, \infty)$.
- With respect to k, H_k is fixed on $[1, \ell_1]$ and the derivatives of H_k form an increasing sequence of functions.

These Hamiltonians H_k are linear at infinity and quadratic on large compact part. We then declare the sequence of fibre wrappings $\{F_k\}$ to be time-1 image of T_x^*S under the Hamiltonian flow of H_k . The family $\{F_k\}$ is cofinal because under a (large) compactly supported deformation, it becomes the linear Hamiltonian $\ell_k r$ and thus the cofinality criterion Lemma 6.8 applies. Let us argue that it satisfies (i) and (ii).

By the observation in [Sei08a, Section 3(c)], the Hamiltonian chords of H_k between T_x^*S and T_u^*S are of the form $(r, x(\ell s))$, where x is the sphere bundle lift of a unit-speed geodesic, ℓ is its length, and r > 0 is such that $H'_k(r) = \ell$. By our condition inj(g) > 2, given a non-minimal geodesic of length ℓ we must have $\ell > 1$. Since the interval $[\ell_k - \delta_k, \ell_k + \delta_k]$ is disjoint to the length spectrum, the Hamiltonian chords lie on the interval where H_k is quadratic and are thus in oneto-one correspondence with non-minimal geodesics of length bounded above by ℓ_k . Furthermore, there is a unique Hamiltonian chord $[\gamma_{xy}]$ representing the minimal geodesic in the region $[1, \ell_1]$. This concludes (i). For (ii), consider the continuation map $CF(F_k, T_y^*S) \to CF(F_{k+1}, T_y^*S)$. Since H_k is fixed on $[1, \ell_1]$, the minimal geodesic element gives a constant continuation strip: let us argue that this is the only continuation strip. Indeed, let u be a homogeneous continuation strip for $CF(F_k, T^*_y S) \to CF(F_{k+1}, T^*_y S)$ mapping a geodesic x^- of length ℓ to the minimal geodesic γ . By the geometric energy formula and global positivity of the Hamiltonians, we must have $A(x^{-}) \geq A(\gamma)$. In contrast, [Sei08a, Section 3(c), Eq.(3.12)] implies that the action of the first nonminimal geodesic is $-\frac{1}{2}\ell^2$, so $A(x^-) = -L^2/2$, and we had $|A(\gamma)| < L^2/2$. This is a contradiction and thus the only strip must be constant, hence the minimal geodesic $[\gamma_{xy}]$ is preserved under wrappings.

6.2.3. Continuation maps for fibre wrappings. In order to continue our proof of Theorem 6.3, we need to compute $\mu_{1|1}(c, \cdot)$, where c is a continuation element. To employ the family Floer techniques from Section 5.2, we use the following facts to relate the maps $\mu_{1|1}(c, \cdot)$ with continuation maps (c.f [GPS20, Lemma 3.26]). First, an analogue of Theorem 4.2 for fibre wrappings:

104

Lemma 6.10. Let F be a wrapped fibre whose base-point is contained in the complement of W. Then there exists $\varepsilon_0(F) \in \mathbb{R}_+$ such that there are no non-constant J_{cyl} -holomorphic strips between F and εL if $\varepsilon \in (0, \varepsilon_0(F))$.

Proof. Let z be the base-point of F, then since $F = T_z^*S$ on D_1^*S , the action of the intersection points are of size $O(\varepsilon)$. Therefore, monotonicity tells us that the J_{cyl} -holomorphic strips cannot escape D_1^*S .

The analogue of Proposition 5.30 for fibre wrappings reads as follows. For brevity, we suppress the discussion on the necessity to make the right choice of the elongation functions.

Proposition 6.11. Let $\{F_t\}, t \in [0,1]$, be a fibre wrapping and $W \subset S$ a spectral network.

- (1) If the support arc of $\{F_t\}$ lies outside W, then there exists a family J(s,t) of almost complex structures such that the moduli space of (F_t, L) -continuation strips is cobordant to the moduli space of trivial strips.
- (2) If the support arc of $\{F_t\}$ intersects W at an ij-wall, then there exists a family J(s,t) of almost complex structures such that the moduli space of (F_t, L) -continuation strips that do not travel from the jth sheet to the ith sheet is cobordant to that of trivial strips.

Proof. Proposition 5.30 implies that there exists $\varepsilon_0(\{F_t\}) \in \mathbb{R}_+$ such that for all $\varepsilon \in (0, \varepsilon_0(\{F_t\}))$ there are no non-constant J_{cyl} -holomorhic strips between εL and F_t . We can then apply the Gromov compactness argument as in the proof of Proposition 5.26 to conclude that the moduli space of continuation strips is cobordant to the moduli of trivial strips.

Since the scaling parameter ε in Lemma 6.10 only depends on the base point, a consequence of Proposition 6.11 is that wrappings at infinity do not affect L. That is, if $F \rightsquigarrow F'$ is a fibre wrapping with the same base point, then the continuation map $CF(F, L) \rightarrow CF(F', L)$ is the identity map. This is an analogue of the invariance statement in Proposition 5.12.

6.2.4. Concluding Theorem 6.3 and Corollary 6.4. For Theorem 6.3, it suffices to show that $\mathcal{Y}_{(L,V)}$ is the the non-abelianization module, as a module over the cohomology subcategory generated by the wrapped fibres, which we identify with the fundamental groupoid. For that, note that given any W-adapted minimal geodesic, Proposition 6.6 and Proposition 6.11 guarantee the upper-triangular form of $\mathcal{Y}_{(L,V)}([\alpha_{zw}])$. Then the same argument as in the proof of Theorem 5.24 and the wall-crossing formula imply that the parallel transport map must be equal to that of $\Phi_W(V)$.

For Corollary 6.4, Lemma 5.13 implies that L is Maslov 0 and thus the chain complex $\mathcal{Y}_{(L,V)}(F)$ for a wrapping fibre F must be cohomologically concentrated in degree zero. By homological perturbation, comparing to the non-abelianization modules, we can perturb $\mathcal{Y}_{(L,V)}$ to be defined over the subcategory of the wrapping fibres. Since the latter subcategory generate the wrapped Fukaya category, the statement follows from Theorem 6.3.

6.3. Cylindrized Betti Lagrangians and spectral networks. As discussed in Section 6.1, a Betti Lagrangian $L \subset (T^*S, \omega_{st})$ does not give an element of the wrapped Fukaya category $\mathfrak{W}(T^*S)$, nor its partially wrapped modification $\mathfrak{W}(T^*S, \Lambda)$. The reason is the asymptotic behavior of Lat infinity, which is not compatible with the definitions of these categories. In Section 6.1, we showed that a Betti Lagrangian, endowed with $V \in \operatorname{Loc}(L)$, does nevertheless give an A_{∞} -module $\mathcal{Y}_{(L,V)}$ over $\mathfrak{W}(T^*S)$ which behaves as if it was the Yoneda module of an object in $\mathfrak{W}(T^*S)$, and in turn we related the $\mu_{1|1}$ -maps of such A_{∞} -modules to the non-abelianization framework of spectral networks, cf. Theorem 6.3. An alternative approach to comparing Betti Lagrangians and Fukaya categories to spectral networks is to allow ourselves to modify the Betti Lagrangian $L \subset (T^*S, \omega_{st})$ at infinity so that in fact it defines an object in a Fukaya category. The present section develops this approach, defining such cylindrizations and showing that the non-abelianized parallel transport from spectral networks is captured by μ_2 -map of the A_{∞} -structure associated to such cylindrization.

Specifically, Section 6.3.1 explains how to deform L to a cylindrical Lagrangian $L_{\circ} \subset (T^*S, \omega_{st})$ with boundary on the Legendrian links associated to the Betti surface $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$. Denote by $\mathfrak{W}(T^*S, \boldsymbol{\Lambda})$ the partially wrapped Fukaya category of (T^*S, ω_{st}) partially stopped at $\boldsymbol{\Lambda}$, as constructed in [GPS24b]. By construction, the cylindrization L_{\circ} defines an object in $\mathfrak{W}(T^*S, \boldsymbol{\Lambda})$ once endowed with a local system $V \in \operatorname{Loc}(L)$. We denote by

$$\mathcal{Y}(L_{\circ}, V) : \mathfrak{W}(T^*S, \Lambda) \to \mathrm{mod}_k$$

its associated Yoneda A_{∞} -module and consider its component

$$\mu_2: HW(T_z^*S, T_w^*S) \otimes HW(T_z^*S, L_\circ) \to HW(T_w^*S, L_\circ),$$

where $HW(T_z^*S, T_w^*S) = \operatorname{Hom}_{\mathfrak{W}(T^*S,\Lambda)}(T_z^*S, T_w^*S)$ are now morphisms in $\mathfrak{W}(T^*S, \Lambda)$, not $\mathfrak{W}(T^*S)$. That said, given a \mathcal{W} -adapted pair $z, w \in S$, we had the minimal geodesic α_{zw} which defined a class in $\operatorname{Hom}_{\mathfrak{W}(T^*S)}(T_z^*S, T_w^*S)$. In order to transfer it to a class in $\operatorname{Hom}_{\mathfrak{W}(T^*S,\Lambda)}(T_z^*S, T_w^*S)$, we consider a subset $S_{int} \subset S$ such that the front projections of the Legendrian links in Λ are all contained in $S \setminus S_{int}$ and S retracts to S_{int} : we obtain S_{int} by removing a sufficiently large disk neighborhood of each marked point. Then there a natural inclusion functor $i_* : \mathfrak{W}(T^*S_{int}) \to \mathfrak{W}(T^*S,\Lambda)$, and we can consider the image $i_*[\alpha_{zw}]$ of (the class defined by) α_{zw} under this functor.

For the cylindrized Betti Lagrangian L_{\circ} , now in the context of the partially wrapped Fukaya category $\mathfrak{W}(T^*S, \Lambda)$, the relation between the A_{∞} -structure and non-abelianized parallel transport for spectral networks is as follows:

Theorem 6.12. Let (S,g) be a Betti surface, $L \subset (T^*S, \omega_{st})$ a Betti Lagrangian, $V \in Loc(L)$, $W \subset S$ an adapted spectral network and $z, w \in S$ an W-adapted pair. Then the minimal geodesic $\alpha_{zw} \in \Omega_{z,w}$ satisfies

(37)
$$\mu_2(i_*[\alpha_{zw}], \cdot) = \Phi_{\mathcal{W}}(V)(\alpha_{zw})$$

Theorem 6.12 is momentarily proven in Section 6.3.2, after describing L_{\circ} in Section 6.3.1. A consequence of Corollary 6.4 and Theorem 6.12 is the relationship between $\mathcal{Y}_{(L,V)}$ and $\mathcal{Y}_{(L_{\circ},V)}$, which reads as follows:

Corollary 6.13. There is a homotopy

$$\mathcal{Y}_{(L_\circ,V)} \circ i_* \simeq \mathcal{Y}_{(L,V)}$$

of A_{∞} -modules over $\mathfrak{W}(T^*S)$.

6.3.1. Cylindrization of Betti Lagrangians. Let $L \subset T^*S$ be a Betti Lagrangian on a Betti surface $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$, we want to modify L near infinity (around the punctures of S) to obtain a Lagrangian $L_{\circ} \subset T^*S$ which is a Lagrangian filling of the Legendrian link in $(T^{\infty}S, \xi_{st})$ defined by $\boldsymbol{\Lambda}$ and defines an object in $\mathfrak{W}(T^*S, \boldsymbol{\Lambda})$. The cylindrization procedure is in line with the modification for weave fillings in [CL22].

From a symplectic topological viewpoint, without taking \mathcal{W} into account, the cylindrization process is as follows. Choose a collection S of circles in S, one circle per puncture chosen so that it encloses the puncture and is sufficiently close to it. The positive conormal lift of such collection S is a Lagrangian submanifold that cleanly intersects the zero section of T^*S . Let $C \subset T^*S$ be the Lagrangian surgery resolving that clean intersection, which is an exact Lagrangian that coincides with the zero section away from the punctures, and with the positive conormal lift of the circles near the punctures. Since $C \subset T^*S$ is Lagrangian and cylindrical at infinity, there exists a Weinstein neighborhood $i_{cyl}: D^*C \to T^*S$ for a small enough disk bundle D^*C , which coincides with the disk bundle of T^*S away from the punctures. Since C is diffeomorphic to S, the Betti Lagrangian $L \subset T^*S$ can be identified with a homonymous Lagrangian $L \subset T^*C$ which can be assumed to belong to D^*C .

Definition 6.14. The cylindrization L_{\circ} of L is defined to be $i_{cul}(L) \subset (T^*S, \lambda_{st})$.

To incorporate compatibility with a given spectral network \mathcal{W} for L, we require that the circles in S are chosen enclosing each puncture close enough such that:

- (1) \mathcal{W} is transverse to the circles,
- (2) each circle is contained in the trapping neighborhood, cf. Lemma 2.26, for the corresponding puncture,
- (3) no vertex of \mathcal{W} gets mapped in the cylindrical region between each circle and its corresponding puncture.

This can be ensured by choosing the circles in S close enough to the punctures and generically. Note that Lemma A.2 implies that L is uniformly bounded with respect to g, and so any (F_z, L) disk for z away from the circles must have its image contained in a similar region, truncated away from the punctures. **Remark 6.15.** Note that we had to scale L down so that $L \subset D^*C$ lied inside the Weinstein neighborhood. This is possible while being compatible with W because the metric is of the form $r^2 d\theta^2 + dr^2$ near the punctures.

The cylindrization L_{\circ} in Definition 6.14 defines an object in $\mathfrak{W}(T^*S, \Lambda)$ after applying a small positive push-off at infinity. We always implicitly understand a Lagrangian filling of a Legendrian Λ in $T^{\infty}S$ as giving an object in $\mathfrak{W}(T^*S, \Lambda)$ in this manner.

6.3.2. Proof of Theorem 6.12 (Floer μ_2 is spectral transport). The argument is in line with the family Floer techniques developed in Section 5 and follows the same steps as Section 6.2, now suitably modified in the partially wrapped setting. Let $\overline{\mathbf{S}}(S)$ denote the unique compact connected component of the complement in S of the circles in S. First, by geometric boundedness, the following locality lemma holds:

Lemma 6.16. Let F_z be a wrapped fibre, with basepoint $z \in \overline{\mathbf{S}}(S)$, whose boundary does not intersect Λ . Then, any (L_{\circ}, F_z) -disk belongs to a region of the form $T^*\overline{\mathbf{S}}(S')$, for S' another compatible collection of circles. Similarly, the same statement holds for any small enough negative push-offs of L_{\circ} whose boundary stays disjoint to ∂F_z .

Lemma 6.16 implies that the remainder of the proof of Theorem 6.12 can be the same as the proof of Theorem 6.3, except for the fact that the cohomology module is now defined as the direct limit of $CF(T_z^*S, L_{\alpha}^i)$ as $i \to \infty$. We thus need to take into consideration of wrapping morphisms

 $CF(T^*_zS,L^i_\circ) \to CF(T^*_zS^w,L^i_\circ), \quad CF(T^*_zS,L^i_\circ) \to CF(T^*_zS,L^{i+1}_\circ).$

This does not affect the computations overall, as a combination of the argument in Lemma 6.16 with the Gromov compactness used in Proposition 6.11 implies the following fact:

Lemma 6.17. There exists a sequence of positive push-offs $\{L_{\circ}^{i}\}_{i}$, $\{F_{z}^{j}\}_{j}$ such that:

(1) For $z \in \overline{\mathbf{S}}(S)$ and for i - 1 > j, the continuation maps

 $CF_{\Lambda}(F_z^j, L_{\circ}^i) \to CF_{\Lambda}(F_z^{j+1}, L_{\circ}^i), \quad CF(F_z^j, L_{\circ}^i) \to CF_{\Lambda}(F_z^j, L_{\circ}^{i+1})$

are the identity map.

(2)
$$HF_{\Lambda}(F_z, L) = \lim_{j \to \infty} HF(F_z^j, L) = \lim_{i \to \infty} HF(F_z, L^i)$$

Here we denoted CF_{Λ} when morphisms are in $\mathfrak{W}(T^*S, \Lambda)$, i.e. partially stopped at Λ .

The remaining ingredient for the proof of Theorem 6.12 is a description of the image morphisms $i_*(HW(T_x^*S, T_y^*S))$ under the inclusion $i_*: \mathfrak{W}(T^*S) \to \mathfrak{W}(T^*S, \Lambda)$. It is provided by the following proposition:

Proposition 6.18. Let $x, y \in \overline{\mathbf{S}}(S)$ be two generic points at distance less than $\frac{1}{20}inj(g)$. Then the inclusion map

$$i_*: HF^*(T^*_xS, T^*_yS) \to HF^*_\Lambda(T^*_xS, T^*_yS)$$
SPECTRAL NETWORKS AND BETTI LAGRANGIANS

is an isomorphism. In fact, for $[\alpha_{xy}]$ the representative of the minimal geodesic, $i_*[\alpha_{xy}] = [\alpha_{xy}]$.

Proof. In brief the core of the proof is using an action filtration argument combined with the fact that, since the metric is conical near the punctures, geodesics in the truncated region $\overline{\mathbf{S}}(S)$ that enter the neighborhood of the punctures cannot leave.²⁷ Let us provide the details of the argument.

First, we must construct a contact form α for $(T^{\infty}S, \xi_{st})$ whose Reeb chords correspond to geodesics between x and y while being stopped at Λ , so as to be consistent with the partial wrapping in $\mathfrak{W}(T^*S, \Lambda)$. In precise terms, we claim that there exists a contact 1-form $\alpha \in \Omega^1(T^{\infty}S)$ for $\xi_{st} = \ker \alpha$ such that:

- (i) The Reeb flow of α is complete,
- (ii) $\alpha = \lambda_{st}|_{T^{\infty}S}$ outside a small standard neighborhood of Λ ,
- (iii) α -Reeb chords between ∂T_x^*S and ∂T_y^*S are in bijection with λ_{st} -chords from ∂T_x^*S to ∂T_y^*S .

The neighborhood in (ii) is referred to as the stop region of α . Let us argue that such α exists, as follows. By construction, Λ lies in a small neighborhood of the positive conormal of the circles in S. In such a neighborhood, the geodesic flow for (S, g) points outward, transverse to the boundary of $\overline{\mathbf{S}}(S)$. By [GPS20, Lemma 3.29], there exists a contact form satisfying (i), (ii) with the stop region lying in this neighborhood. For this choice of neighborhood, (iii) also holds. Indeed, an α -Reeb chord from x to y is a geodesic chord away from the stop region. Now, on the one hand, if it entered such neighborhood of Λ , the geodesic must enter the boundary in the transverse direction and then the conical metric structure implies that the original geodesic could not have returned to y. On the other hand, for the α -Reeb chord to return to y, it must leave the stopping region but the only way the chord can leave it is by exiting the neighborhood of the positive conormal of S. In that case the geodesic flow would further send the chord off to infinity. Therefore, geodesics between xand y could not have entered such neighborhood of the stop boundary and no α -Reeb chord could have done the same. Thus (iii) holds and we can choose a contact form α with the properties above.

Second, we proceed with an action filtration argument, in line with Proposition 6.6. By our choice of contact form above, the α -Reeb chords are in bijection with geodesics between x and y. By genericity of x, y, we can assume that the length of the geodesics are all distinct, the energy functional on the path space is Morse, and all the geodesics stay outside the stop collar region. We now consider two variations on the family of Hamiltonians from the proof of Proposition 6.6, one for CF^* and one for CF^*_{Λ} .

(1) For CF^* , the wrapped case with no stop at Λ , let $h_{\overline{\mathbf{S}}} : T^{\infty}\overline{\mathbf{S}}(S) \to \mathbb{R}$ be a smooth, nonnegative function on the unit sphere bundle of $\overline{\mathbf{S}}(S)$ with ∇h_{st} supported on the stop region, such that $h_{\overline{\mathbf{S}}}\lambda_{st}|_{T^{\infty}S}$ has complete Reeb flow. Let H_n, ℓ_n be as in the proof of Proposition 6.6, consider

 $^{^{27}}$ Also, there is no conceptual importance to the factor 1/20: it is just there for action estimates.

the linear-at-infinity Hamiltonians $h_{\overline{\mathbf{S}}} \cdot H_n$ and let F_n denote the time 1-image of T_y^*S under the corresponding Hamiltonian flows. The Hamiltonian chords of $h_{\overline{\mathbf{S}}} \cdot H_n$ all lie outside the stop region, and so the argument in Proposition 6.6 implies that the Hamiltonian chords correspond to non-minimal geodesic of length ℓ less than ℓ_n , and their actions are given by $-\ell^2/2$.

(2) For CF_{Λ}^* , write the contact form α above as $\alpha = h_{stop}\lambda_{st}$ and let F_n^{stop} be the time-1 image of T_x^*S using the quadratic Hamiltonians $h^{stop} \cdot H_n$. By construction, the inequality $h_{\overline{\mathbf{S}}} \leq h_{stop}$ holds and so there is a positive isotopy $F_n \rightsquigarrow F_n^{stop}$ and an induced continuation map $CF^*(F_n, T_y^*S) \to CF^*(F_n^{stop}, T_y^*S)$. By Property (*iii*) above, these two groups coincide as vector spaces. Since the actions are given by $-\ell^2/2$, and the length spectrum is discrete, we can consider a basis $[\gamma_1], [\gamma_2], \ldots, [\gamma_k]$, by ordering by length the geodesics of length bounded by ℓ_n .

The last step is to argue that the continuation map is upper-triangular with respect to the basis $[\gamma_1], [\gamma_2], \ldots, [\gamma_k]$. Indeed, let $[\gamma_i]$ be a generator of $CF^*(F_n, T_y^*S)$ and consider the image $\sum n_j[\gamma_j]$ under the continuation map. By the geometric energy formula, we have $A(\gamma_i) \leq A(\gamma_j)$ if $i \leq j$ and a continuation strip from γ_i to itself must be constant. Therefore, the action formula implies that this image must be of the form $[\gamma_i] + \sum_{j < i} n_{ij}[\gamma_j]$. Thus the continuation map $CF(F_n, T_y^*S) \rightarrow CF(F_n, T_y^*S)$.

 $CF(F_n^{stop}, T_y^*S)$ is upper-triangular, with identity on the diagonal, and so it must be a quasiisomorphism. Furthermore, the following diagram commutes, up to chain homotopy:

In consequence, the induced map $WF(T_x^*S, T_y^*S) \to WF_{\Lambda}(T_x^*S, T_y^*S)$ must also be a quasiisomorphism. Finally, the minimal geodesic generator is preserved under these maps by arguing as in the proof of Proposition 6.6.

Proof of Theorem 6.12. By Proposition 6.18, i_* is cohomologically fully faithful, and the minimal geodesic elements are preserved under the inclusion functor. Consider the cohomology module of $\mathcal{Y}(L_{\circ}, V)$ over i_* restricted to the subcategory given by the cotangent fibres which is istelf identified with the fundamental groupoid. The cohomology module $\mathcal{Y}(L_{\circ}, V)$ with the parallel transport morphisms is defined by the direct limit

$$\lim_{i \to \infty} CF(T_z^*S, L^i).$$

By Lemma 6.17, such wrappings induce identity maps on the Floer chain complexes $CF_{\Lambda}(F_z^j, L^i)$ along arcs away from the spectral network \mathcal{W} . So the local system given by (the cohomology module of) $\mathcal{Y}(L_o, V)$ has trivial parallel transport over arcs that do not cross \mathcal{W} . The same argument as in Proposition 6.11.(2) implies that the associated parallel along an arc crossing W must be as in the wall-crossing formula for non-abelianized local systems. The statement then follows by applying the argument used for Theorem 6.3.

Proof of Corollary 6.13. By Theorem 6.12, $\mathcal{Y}(L_{\circ}, V) \circ i_* \simeq \mathcal{Y}(L)$ over the cohomology subcategory of cotangent fibres. Since the inclusion functor i_* is cohomologically fully faithful, and the cotangent fibres generate $\mathfrak{W}(T^*S)$, the statement follows from the same argument as the proofs of Theorem 6.3 and Corollary 6.4.

6.4. Wrapping up the description of $\mathcal{Y}_{(L_{\circ},V)}$. In Section 6.3 we introduced the Yoneda A_{∞} module $\mathcal{Y}_{(L_{\circ},V)} : \mathfrak{W}(T^*S,\Lambda) \longrightarrow \operatorname{mod}_k$ associated to the cylindrization $L_{\circ} \subset (T^*S,\lambda_{\mathrm{st}})$ of a Betti Lagrangian L. In a nutshell, Theorem 6.12 and Corollary 6.13 give a description of this A_{∞} module on the A_{∞} -subcategory given by the image of the functor $i_* : \mathfrak{W}(T^*S_{int}) \longrightarrow \mathfrak{W}(T^*S,\Lambda)$, which is generated by the internal cotangent fiber. The goal of this section is to complete the description of the A_{∞} -module $\mathcal{Y}_{(L_{\circ},V)}$ on $\mathfrak{W}(T^*S,\Lambda)$. In order to describe the whole of $\mathcal{Y}_{(L_{\circ},V)} : \mathfrak{W}(T^*S,\Lambda) \longrightarrow \operatorname{mod}_k$, we construct in Proposition 6.19 a set of generators of $\mathfrak{W}(T^*S,\Lambda)$, containing the internal cotangent fiber, where we can compute the values of $\mathcal{Y}_{(L_{\circ},V)}$.

6.4.1. Generation by interior and infinity fibers. Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, and denote by $\Lambda_i \subset (T^{\infty}S, \xi_{st})$ the Legendrian link associated to the marked point $m_i \in \boldsymbol{m}$. For simplicity, we shall henceforth assume that Λ_i has finitely many Reeb chords, each at a different angle, and that it is Reeb-positive (for instance, in the case of Stokes Legendrians without multiplicities Proposition 2.25, or the positive braid links described in Remark 2.2). This includes many of the most important classes, including the case where each Legendrian in $\boldsymbol{\Lambda}$ is isotopic to the cylindrical closure of a positive braid. We consider the following Lagrangians in T^*S , all of which yield objects in $\mathfrak{W}(T^*S, \Lambda)$ once appropriately decorated and partially wrapped:

- (1) An internal cotangent fiber T_z^*S , for some point $z \in S$,
- (2) The cotangent fiber F_{m_i} near the puncture corresponding to m_i . It is given by the cotangent fiber at any point in S arbitrarily close to the marked point $m_i \in \mathbf{S}$ such that the geodesic between that point and m_i does not intersect Λ_i , i.e. it is the cotangent fiber at infinity of m_i , past the Legendrian link Λ_i . We denote $F_{\mathbf{m}} := \{F_1, \ldots, F_{|\mathbf{m}|}\}$ the set of such fibers at the punctured infinity of S.
- (3) The collection of linking disks \mathbb{D}_i associated to the Legendrian Λ_i . That is, each $\mathbb{D}_i := \{D_{i,1}, \ldots, D_{i,c_i}\}$ is the disjoint union of the linking disks of Λ_i , as in [GPS24b, Section 5.3], $c_i = |\pi_0(\Lambda_i)|$, one for each component of the Legendrian link Λ_i . See Figure 14.



FIGURE 14. A front for the Legendrian Λ_i : in this picture we assume Λ_i has three components, one per strand. The three linking disks $D_{i,1}, D_{i,2}$ and $D_{i,3}$ for Λ_i are depicted, with the short Reeb chords between them highlighted in purple. The large linking disk \mathcal{D} , homologically obtained from coning the small linking disks along the short Reeb chords, is depicted in orange to their right.

By [GPS24b, Theorem 1.14], the set of Lagrangians $\{T_z^*S, \mathbb{D}_1, \ldots, \mathbb{D}_{|\boldsymbol{m}|}\}$ generates $\mathfrak{W}(T^*S, \Lambda)$. By Section 6.3, the spectral network \mathcal{W} allows us to describe $\mathcal{Y}_{(L_o,V)}$ at the generator T_z^*S , by choosing $z \in \mathcal{W}^c$ and using that the inclusion induces a fully faithful functor $\mathfrak{W}(T^*S) \to \mathfrak{W}(T^*S, \Lambda)$. The value of $\mathcal{Y}_{(L_o,V)}$ at the linking disks in \mathbb{D}_i is

$$\mathcal{Y}_{(L_{\circ},V)}(D_{i}) = \mathbb{Z}.$$

Indeed, since $\partial^{\infty} L_{\circ} = \Lambda$, D_i intersects L_{\circ} once. Since the same holds for a small enough positive push-off L_{\circ}^+ at the puncture m_i , $CF^*(D_i, L_{\circ}^+) = \mathbb{Z}$. Nevertheless, the value of $\mathcal{Y}_{(L,V)}$ on morphisms $\operatorname{Hom}_{\mathfrak{W}(T^*S,\Lambda)}(D_{i,j}, D_{i,k})$ between such linking disks, even for j = k, contains subtler information, encoding augmentations of Reeb chords at infinity which, in terms of \mathcal{W} , correspond to augmented D_4^- -trees.²⁸ In contrast, $\mathcal{Y}_{(L_{\circ},V)}$ vanishes at the fibers F_m at infinity. Therefore, a description of $\mathcal{Y}_{(L_{\circ},V)}$ will follow if we can prove that $\mathfrak{W}(T^*S,\Lambda)$ is generated by $\{T_z^*S, F_m\}$. This is the content of the following Floer-theoretical result, which is independent of spectral networks:

Proposition 6.19 (Generation by internal and infinity cotangent fibers). Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface, $\mathfrak{W}(T^*S, \boldsymbol{\Lambda})$ the associated partially wrapped Fukaya category, $L \subset (T^*S, \omega_{st})$ a Betti Lagrangian and $V \in Loc(L)$. Then $\mathfrak{W}(T^*S, \boldsymbol{\Lambda})$ is generated by $\{T_z^*S, F_{\boldsymbol{m}}\}$.

Proof. Let $T_z^*S \subset T^*S$ be an internal cotangent fiber and choose a puncture $m \in \mathbf{m}$ with its associated fiber F_m and Legendrian link Λ_m . Denote by $\mathbb{D}_m := \{D_{p_1}, \ldots, D_{p_n}\}$ a collection of linking disks chosen such that D_{p_i} locally links Λ_m at the point $p_i \in \Lambda_m$ and the ordered points $p_1, \ldots, p_n \in \Lambda_m$ are located one per each strand of Λ_m and all in the same angle.²⁹ Consider the

²⁸These morphism groups between linking disks can themselves be difficult to describe.

²⁹We can and do assume that there is no Reeb chord at that angle.

large linking disk \mathcal{D}_m associated to Λ_m , which is defined as in Figure 14. The wrapping exact triangle in [GPS24b, Theorem 1.10] gives the exact triangle

$$(38) T_z^*S \to F_m \to \mathcal{D}_m$$

in $\mathfrak{W}(T^*S, \Lambda)$. Eq. (38) implies that $\{T_z^*S, F_m\}$ generate the same category as $\{T_z^*S, \mathcal{D}_{m_1}, \ldots, \mathcal{D}_{m_{|m|}}\}$.³⁰

The next step is to show that this latter collection $\{T_z^*S, \mathcal{D}_{m_1}, \ldots, \mathcal{D}_{m_{|m|}}\}$ generates $\mathfrak{W}(T^*S, \Lambda)$, which we do by comparing the large linking disk \mathcal{D}_m to the linking disks D_{p_1}, \ldots, D_{p_n} at Λ_m . This comparison is obtained by iterating the wrapping exact triangle for short Reeb chords, which gives the homological description of \mathcal{D}_m as

$$\mathcal{D}_m \cong \operatorname{cone}(\ldots \operatorname{cone}(\operatorname{cone}(D_{p_1} \to D_{p_2})[-1] \to D_{p_3})[-1]) \ldots \to D_{p_n})[-1].$$

as objects in the dg-category $\operatorname{Tw}(\mathfrak{W}(T^*S,\Lambda))$ of twisted complexes of $\mathfrak{W}(T^*S,\Lambda)$. Here the morphisms $D_{p_i} \to D_{p_{i+1}}$ being coned are given by the short Reeb chords, see e.g. Figure 14. In fact, by Lemma 6.21.(1) and (2), we actually have a quasi-isomorphism

(39)
$$\mathfrak{D}_m \cong D_{p_1} \oplus D_{p_2} \oplus \ldots \oplus D_{p_n}$$

again in $\operatorname{Tw}(\mathfrak{W}(T^*S,\Lambda))$. Indeed, this is a consequence of functoriality of mapping cones and homological perturbation, as follows. By [Sei08b, Lemma 3.30], exact triangles are sent to exact triangles and in Section 3(e) of ibid., given a different cocycle representative with the same cohomology class, a non-canonical isomorphism is constructed between the abstract mapping cones, that are identified with Yoneda modules of mapping cones. Thus given an A_{∞} -equivalence $f : CF(\mathcal{D}_m, \mathcal{D}_m) \to$ $H^0(\mathcal{D}_m, \mathcal{D}_m)$ which induces an isomorphism $\operatorname{Tw}(f) : \operatorname{Tw}(CF(\mathcal{D}_m, \mathcal{D}_m)) \to \operatorname{Tw}(H^0(\mathcal{D}_m, \mathcal{D}_m))$, we can directly use the functoriality of mapping cones to conclude that

$$\operatorname{Tw}(f)(\operatorname{cone}(\mathfrak{r})) = \operatorname{cone}(\operatorname{Tw}(f)^1(\mathfrak{r})),$$

where \mathfrak{r} is the short Reeb chord and use that the morphism $\operatorname{Tw}(f)^1(\mathfrak{r})$ vanishes.

By [GPS24b, Theorem 1.14], $\operatorname{Tw}(\mathfrak{W}(T^*S, \Lambda))$ is generated by the interior fiber T_z^*S and the objects $D_{p_1} \oplus D_{p_2} \oplus \ldots \oplus D_{p_n}$, taken over all Λ_m ranging over the marked points $m \in \mathbf{m}$. Therefore Eq. (39) implies that the large linking disks \mathcal{D}_m , ranging over the marked points $m \in \mathbf{m}$, along with the interior fiber T_z^*S , also generate $\operatorname{Tw}(\mathfrak{W}(T^*S, \Lambda))$.

Remark 6.20. In Proposition 6.19, generation is defined as having quasi-equivalent categories of modules over twisted complexes, as in e.g. [GPS20; GPS24a]. Given our use of cones, we must work in a triangulated framework, and we use the enhancement of twisted complexes; see also [Sei08b, Chapter 1].

Note that there is no canonical choice of grading for the linking disks D_{p_i} . By definition, we refer to the grading on D_{p_i} as the *geometric* grading if $CF^*(D_p, L_\circ)$ is concentrated in degree 0. For a Legendrian knot and a linking disk, this geometric grading is the grading that is invariant under

³⁰It also follows from Eq. (38) that $\mathcal{Y}_{(L_{\circ},V)}(\mathcal{D}_m) \cong \mathbb{Z}^n$.

contact isotopies of (the boundary of) the linking disk around the Legendrian.

Let us summarize useful properties of the linking disks. Intuitively, these follow from the fact that the surgery models in [BEE12, Section 6.1] and [ENS18, Lemma 6.4] show that the dynamics around a (punctured) handle can be taken to be the geodesic flow inside of the handle and the given Reeb flow outside. In more detail:

Lemma 6.21 (Properties of linking disks). In the notation above, the following properties hold:

- (1) (Locality) Let m_1, m_2 be two distinct punctures with Legendrian links $\Lambda_{m_1}, \Lambda_{m_2}$ and D_1, D_2 corresponding linking disks. Then $HW^*(D_1, D_2) \cong 0$.
- (2) Let \mathbb{D} be a collection of linking disks for the Legendrian links Λ , each equipped with the geometric grading. Then its endomorphism algebra $HW^*(D, D)$ is concentrated in non-negative degree and, if links are Reeb-positive, then it is concentrated in the zero degree.
- (3) Let $D_{p_i}, D_{p_{i+1}}$ be two linking disks at the same puncture, as above. Then the short Reeb chords $D_{p_i} \to D_{p_{i+1}}$ have cohomological grading 1.

Proof. Part (1) Locality follows from the fact that the geodesic flow near the positive unit conormal cannot escape the neighborhood of the punctures. For Part (2), we translate the surgery partial wrapping introduced in [EL23] to the partial wrapping in [GPS20]. That is, we consider an open Liouville sector W_{Λ} obtained by attaching the disk cotangent bundle of $\Lambda \times [1, \infty)$, the sectorial analogue of a punctured handle attachment. The Reeb chords of cotangent fibres in W_{Λ} are computed using [EL23, Lemma 88], up to some action cut-off, from which it follows that the Reeb chords all have non-negative degrees.³¹ This open Liouville sector W_{Λ} can be truncated to give a closed Liouville sector $W_{\Lambda}(T)$ given by attaching the disk cotangent bundle of $\Lambda \times [1,T]$ instead, for a large enough $T \gg 1$. The inclusion functor induced from the inclusion of Liouville sectors $W_{\Lambda}(T) \to W_{\Lambda}(T'), T < T'$, is cohomologically fully faithful for T sufficiently large enough. Therefore, it suffices to compute the wrapped Floer cohomology in $W_{\Lambda}(T)$. Now, $W_{\Lambda}(T)$ is deformation equivalent to the closed Liouville sector obtained by removing a standard 1-jet neighborhood of Λ in the boundary, c.f. [GPS20, Example 2.15], the wrapped Floer cohomology of which is isomorphic to the partially wrapped Floer cohomology in [GPS20], by [GPS24b, Corollary 3.9. Therefore, partially wrapped Floer cohomology must also be concentrated in non-negative degrees.

³¹Strictly speaking, the form of the metric used in ibid. is $f(r)d\theta^2 + dr^2$ for some smooth, positive function f(r), with the properties as in the discussion in Section B.3 in ibid., in order to ensure that the geodesics do not travel arbitrarily deep into infinity. In our case, the boundary is a circle and so using the conical metric is sufficient.

For Part (3), we first present the full argument for the rank 2 case. Define the grading on D_{p_2} so that the short Reeb chord lies in hom⁰ (D_{p_i}, D_{p_2}) . For this grading

$$\hom(\operatorname{cone}(D_{p_1} \to D_{p_2})[-1], L) = \hom(\mathcal{D}_l, L)$$

with the large linking disk \mathcal{D}_l , connecting sum D_{p_1} and D_{p_2} , being equipped with the geometric grading. In particular, $D_{p_2}[-1]$ is the graded Lagrangian object equivalent to the linking disk at p_2 equipped with the geometric grading. Therefore, since

$$\hom^*(D_{p_1}, D_{p_2}[-1]) = \hom^{*-1}(D_{p_1}, D_{p_2})$$

the short Reeb chord is a degree +1 element in hom^{*} $(D_{p_1}, D_{p_2}[-1])$, as claimed. The higher rank case follows from the exactly same argument.

6.4.2. A concluding remark. There are more topological descriptions of $\mathfrak{W}(T^*S)$ and $\mathfrak{W}(T^*S, \Lambda)$, not involving pseudo-holomorphic curves. As stated above, $\mathfrak{W}(T^*S)$ -mod_k is quasi-equivalent to the category of local systems Loc(S) and thus perfect modules $\operatorname{Perf}(\mathfrak{W}(T^*S))$ is equivalent to its compact objects Loc^c(S). More generally, [GPS24a, Theorem 1.1] shows that the category of perfect modules $\operatorname{Perf}(\mathfrak{W}(T^*S, \Lambda))$ is equivalent to the category $\operatorname{Sh}^c_{\Lambda}(S)$ of compact objects in $\operatorname{Sh}_{\Lambda}(S)$, the category of sheaves with singular support contained in Λ . Independently, since S is a surface in the study of Floer theory and spectral networks, one can also approach the generalization from $\mathfrak{W}(T^*S)$ to $\mathfrak{W}(T^*S, \Lambda)$ as the generalization from local systems to Stokes local systems. We refer to [Boa21] for a detailed description of the latter.

We conjecture that the relation between partially wrapped Fukaya categories and spectral networks, as discussed above, is compatible with Stokes local systems. In an admittedly unimpressive degree of detail, we briefly sketch what we expect, as follows. Given a Betti Lagrangian $L \subset T^*S$ and a compatible spectral network $W \subset S$, we expect the existence of the following commutative diagram:

(40)
$$\operatorname{Loc}^{c}(L) \xrightarrow[\Phi_{W}^{St}]{\mathfrak{M}} \operatorname{St}(S, \Lambda)$$

where the notions in the diagram are:

(1) $\operatorname{St}(S, \Lambda)$ is the category of Stokes local systems, as described in [Boa21, Section 8], with the Stokes diagrams given by Λ .³² Technically, ibid. defines Stokes local systems given irregular data – which is naturally algebraic – but the same definitions allow for the notion

³²The category might be needed to be dg-enhanced, derived and consider compact objects to fit the diagram above, but all meaningful content is in ibid.

of a Stokes local system associated to any Legendrian link isotopic to (the lift of) a front given by the cylindrical closure of a positive braid.

- (2) The Yoneda functor \mathcal{Y} is given by $V \mapsto \mathcal{Y}_{(L_{\circ},V)}$, as discussed in Section 6.3 and Section 6.4.1 above.
- (3) The functor $\Phi_{\mathcal{W}}^{St}$ is essentially the functor $\Phi_{\mathcal{W}}$ constructed and studied in Section 5. The only difference is that $\Phi_{\mathcal{W}}(V)$ was described as a local system, and $\Phi_{\mathcal{W}}^{St}(V)$ should be a Stokes local system. As discussed in [GMN12; GMN13a], the non-abelianization process actually produces a Stokes local system, so this adjustment on the codomain is obtained by the same methods.
- (4) The functor \mathcal{M} can be described explicitly as follows. Given a Stokes local system $V_{St} \in$ $St(S, \Lambda)$, the A_{∞} -module $\mathcal{M}(V_{St}) : \mathfrak{W}(T^*S, \Lambda) \longrightarrow \text{mod}_k$ is determined by

$$\mathcal{M}(V_{St})(T_z^*S) = V_z, \quad \mathcal{M}(V_{St})(F_\infty) = 0,$$

where T_z^*S is an internal fiber, F_∞ any fiber at infinity and V is the underlying local system of V_{St} . By Proposition 6.19, this determines \mathcal{M} on objects. Note that for any $z, w \in S$ two \mathcal{W} -adapted points, we will also have $\mathcal{M}(V_{St})(i_*[\alpha_{zw}]) = V([\alpha_{zw}])$.

The commutativity of Diagram (40) is a categorical analogue of the results in Section 5, comparing the non-abelianiation functor Φ_W to the Family Floer functor \mathbb{F} .

Remark 6.22. (1) Note that [Boa21] establishes equivalent descriptions of $St(S, \Lambda)$, comparing to Stokes graded and Stokes filtered local systems. The category of Stokes filtered local systems is much closer to $Sh_{\Lambda}(S)$. In contrast, the Floer-theoretic context has generators that directly give a grading, provided by the Lagrangian intersections, and not just the filtration more naturally associated to $Sh_{\Lambda}(S)$. Therefore, the passage from $Sh_{\Lambda}(S)$ to $\mathfrak{W}(T^*S, \Lambda)$ provides, in a sense, a symplectic topological viewpoint on the splitting result [Boa21, Theorem 1.1], which produces a canonical Stokes grading from the filtration.

(2) It would be desirable to also have a conceptual description of \mathcal{M} , not given in terms of explicit generators of $\mathfrak{W}(T^*S,\Lambda)$. Specifically, we expect \mathcal{M} to be an equivalence and its inverse should be given by an enhancement of the Family Floer methods in Section 5.2.

By Theorem 6.12, Diagram (40) commutes if the Legendrian Λ at infinity is Reeb-positive. This is the case, for instance, if the Legendrian links at infinity are all cylindrical closures of positive braids, which includes the case of any irregular data with no multiplicities. In this Reeb-positive case, it also follows that the non-abelianization functor is injective on objects: it injectively sends local systems on L to Stokes local systems on S, which was conjectured by the original works [GMN12; GMN13a]. Indeed, given $V, V' \in \text{Loc}(L)$, wrapped cohomology $HW^*((L_{\circ}, V), (L_{\circ}, V'))$ is isomorphic to the Morse cohomology of L twisted by $V^* \otimes V'$ coefficients. In particular, its degree-0 group is non-zero if and only if V and V' are isomorphic, from which injectivity follows.

7. Spectral networks and weaves

Weaves were introduced in [CZ22], in the context of contact and symplectic topology. Since then, they have also been productively used in the study of cluster structures, see e.g. [Cas+23; CW24; Cas+25]. The object of this section is to present a first relation between weaves and spectral networks, in particular discussing Theorem 5. The necessary ingredients on weaves are presented in Section 7.1, and the combinatorial construction of a spectral network $W_{\rm w}$ from a Demazure weave ${\rm w}$ is given in Section 7.2. We then conclude with Section 7.3 presenting a number of explicit examples and computations. Note that, given the rigorously established bridge between weaves and cluster algebras, cf. [Cas+25], this construction of $W_{\rm w}$ provides a precise account on how cluster coordinate arise in spectral networks, in line with some predictions from supersymmetric computations, see e.g. [Nei14]. In addition, since [Cas+23; CW24; Cas+25] give explicit descriptions of Donaldson-Thomas transformations in terms of weaves, this connection between weaves and spectral networks therefore gives a direct method to compute the BPS spectrum generator.

7.1. Weaves on Betti surfaces. Let $(S, \boldsymbol{m}, \boldsymbol{\Lambda})$ be a Betti surface of rank n and let β_i be a positive (cyclic) braid word in n-strands whose cylindrical closure gives the link Λ_i at the puncture $m_i \in \boldsymbol{m}, i \in [1, |\boldsymbol{m}|]$. The definition of a weave in [CZ22] is adapted to the context of Betti surfaces as follows:

Definition 7.1 (Weaves). A weave \mathfrak{w} in (S, \mathbf{m}, Λ) is a properly embedded graph in S with edges decorated by permutations in $W(G) \cong S_n$ such that:

(1) There are only three types of vertices for \mathfrak{w} , as depicted in Figure 15:



FIGURE 15. The three types of vertices allowed in a weave \mathfrak{w} . (Left) Trivalent vertex, where all edges are decorated with the same permutation $s_i \in W(G)$. (Center) Hexavalent vertex, with the edge decorations alternating between s_i, s_j with |i - j| = 1. (Right) Tetravalent vertex, with edge decorations as drawn with |i - k| > 1.

(2) The asymptotics of the graph \mathfrak{w} at a puncture $m_i \in \mathfrak{m}$ are such that the permutations decorating its edges coincide with the corresponding positive cyclic braid word β_i , for all $i \in [1, m]$.

The asymptotic condition in Definition 7.1.(2) is drawn in Figure 16. In figures, we often adopt the convention that colors blue and red are adjacent, and blue and green are not adjacent, see e.g. Fig. 15. By convention, when drawing in small rank $n \in \mathbb{N}$, we typically use blue for s_1 , red for s_2 and green for s_3 .



FIGURE 16. A weave \boldsymbol{w} near a circle ∂S_i around a puncture $m_i \in \boldsymbol{m}$, according to Definition 7.1.(2). The decorations on the edges of \boldsymbol{w} are written as s_{i_j} , and we use colors to emphasize that they might be different. The cyclic braid word being spelled is $\beta_i = s_{i_1}s_{i_2}s_{i_3}\ldots s_{i_\ell}$.

In this section, we focus on the case of the Betti surface $(S^2, \{\infty\}, \beta\delta(\beta))$ given by a once punctured 2-sphere $S = S^2$, with the braid around the unique puncture $\{\infty\}$ being the form $\beta\delta(\beta)$.³³ In higher rank $G = SL_n$, these Betti surfaces already lead to many interesting cases, including Bers-Nevins-Roberts [BNR82] and all braid varieties [Cas+25], thus all double Bruhat cells, positroids and double Bott-Samelson cases. To study weaves and spectral networks in $(S^2, \{\infty\}, \beta\delta(\beta))$, we use the class of Demazure weaves introduced in [Cas+24, Section 4], cf. also [Cas+25, Section 4]. Recall that a Demazure weave $\mathfrak{w} : \beta \longrightarrow \delta(\beta)$ for $\beta = \sigma_{i_1} \cdots \sigma_{i_\ell}$ is a weave in \mathbb{R}^2 drawn vertically top-to-bottom such that

(i) There are two types of semi-infinite edges: north and south. The northern semi-infinite edges coincide with the vertical upwards semi-rays $\{j\} \times \mathbb{R}_+$ outside of a compact set, $j \in [1, \ell]$, and the souththern semi-infinite edges coincide with the vertical downwards

118

³³We can and will assume that $\delta(\beta) = w_0$, see e.g. [Cas+25].

SPECTRAL NETWORKS AND BETTI LAGRANGIANS

semi-rays $\{k\} \times \mathbb{R}_+, k \in [1, \ell(\delta(\beta)).$

- (ii) The weave \mathfrak{w} is never tangent to the horizontal lines $\mathbb{R} \times \{y\}, y \in \mathbb{R}$.
- (iii) The only vertices of \mathfrak{w} are as literally as drawn in Fig. 15 when \mathfrak{w} is scanned top-to-bottom. That is, the trivalent vertex cannot go from s_i on top to $s_i s_i$ at the bottom: it must always go from $s_i s_i$ at the top to s_i at the bottom.

Here we identify $S^2 \setminus \{\infty\} \cong \mathbb{R}^2$ via a diffeomorphism and choose Cartesian coordinates $(x, y) \in \mathbb{R}^2$. Demazure weaves are a diagrammatic description of a sequence of braid words where the only allowed moves are braid moves and the nil-Hecke type move $\sigma_i^2 \to \sigma_i$, the sequence starts with β , corresponding to a horizontal slice in positive large *y*-height, and ends with $\delta(\beta)$, corresponding to a horizontal slice in negative large *y*-height. For the purpose of constructing spectral networks, we use the following modification of a Demazure weave:

Definition 7.2 (Bending of Demazure weaves). Let $\mathfrak{w} : \beta \longrightarrow \delta(\beta)$ be a Demazure weave, as depicted in Figure 17 (left). By definition, the right bending of \mathfrak{w} is the weave in \mathbb{R}^2 drawn as in Figure 17 (right).



FIGURE 17. (Left) A general Demazure weave $\mathbf{w} : \beta \to \delta(\beta)$. (Right) The right bending of \mathbf{w} . The use of the bent weave, as depicted on the right, is that it can be seen as an embedded exact Lagrangian filling of the (-1)-closure of $\beta\delta(\beta)$.

The relevance of Definition 7.2 is that the bent Demazure weave represents a spatial wavefront whose Legendrian lift is such that its Lagrangian projection in $(\mathbb{R}^4, \omega_{st})$ an exact embedded Lagrangian filling of the (-1)-closure $\Lambda_{\beta\delta(\beta)} \subset (\mathbb{R}^3, \xi_{st})$. This follows from the original construction of weaves in [CZ22] and the observation that bending a Demazure weave \mathfrak{w} does not introduce any Reeb chords. It is the starting idea behind many recent results connecting symplectic topology and cluster algebras, cf. e.g. [CL22; Cas+23; Cas+25; Cas+24]. In short, the bending of a Demazure weave $\mathfrak{w} \subset \mathbb{R}^2$ describe an exact Betti Lagrangian $L_{\mathfrak{w}}$ for the Betti surface $(S^2, \{\infty\}, \beta\delta(\beta))$.

Remark 7.3. (1) The need for bending is to ensure there are no concave ends. A Demazure weave \mathfrak{w} , without bending, would naturally yield an exact embedded Lagrangian cobordism from $\Lambda_{\delta(\beta)}$, in the concave end, up to Λ_{β} , in the convex end. The bending effectively fills the concave end, yielding a Lagrangian filling of Λ_{β} . We plan to study non-empty concave ends in future work as well.

(2) The theory of weaves, as developed in [CZ22], works in an arbitrary Betti surface. For instance, [CZ22, Section 3.1] builds weaves from N-triangulations of surfaces, whose spectral networks will match (and generalize) the level N lift spectral networks from [GMN14]. Although it is possible to define a generalization of Demazure weaves to any Betti surface, with higher genus and punctures, the wealth of Demazure weaves in $S^2 \times \{\infty\}$ and their associated spectral networks, as momentarily built in Section 7.2, already covers many interesting examples and applications, so we focus on those.

7.2. Spectral networks for Demazure weaves. By Section 7.1, bending a Demazure weave $\mathfrak{w} \subset \mathbb{R}^2$ gives an exact Betti Lagrangian $L_{\mathfrak{w}} \subset (T^*\mathbb{R}^2, \lambda_{st})$ for the Betti surface $(S^2, \{\infty\}, \beta\delta(\beta))$. The goal is to construct a Morse spectral network $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ for $L_{\mathfrak{w}}$ from the data of the weave \mathfrak{w} .

Definition 7.4 (Augmentation forest of \mathfrak{w}). Let $\mathfrak{w} \subset \mathbb{R}^2$ be the right bending of a Demazure weave with ℓ trivalent vertices. By definition, the augmentation forest $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ of \mathfrak{w} is the spectral network built as follows:

- (1) Start with an empty prespectral network $\mathcal{F}_0 = \mathcal{W}_0 = \emptyset$. Scan the trivalent vertices of \mathfrak{w} bottom-to-top, indexing by $i \in [1, \ell]$ in this bottom-to-top order for the vertices: at the *i*th trivalent weave vertex v_i of \mathfrak{w} , create three directed flowlines as in Figure 18.(1).
- (2) Flow each of the three edge flowlines in Figure 18.(1) from v_i as follows. Flowlines (a) and (b) continue following the weave edges up until their reach the top of the weave. These flowlines go straight through both 4 and 6-valent vertices and continue up on the left of a trivalent, as depicted in Figure 18.(2); the permutation labels of the flowline change by conjugation with the labels of the weave lines. The flowline (c) is extended to the right, through weave edges, changing its labeling accordingly, until it reaches a weave edge whose permutation label coincides with that of the flowline (at that stage). Right before that point (to its left), the edge (c) is extended upward following that weave edge until it reaches the top. The result of these extensions is a prespectral network \mathcal{F}_i , which is obtained by adding

SPECTRAL NETWORKS AND BETTI LAGRANGIANS

these flowlines to the spectral network \mathcal{W}_{i-1} .

- (3) Consider the prespectral network \mathcal{F}_i and add additional flowlines to create a consistent extension of \mathcal{F}_i . The only new flowlines that need to be created appear at the intersection of the (new) flowlines both at v_i with the flowlines of \mathcal{W}_{i-1} . Then declare \mathcal{W}_i to be the resulting spectral network.
- (4) Declare $\mathcal{W}_{\mathfrak{w}}$ to be \mathcal{W}_{ℓ} , the result of applying the above process to all trivalent vertices of \mathfrak{w} , scanned bottom to top.



FIGURE 18. (1) The local model of the flowlines in the augmentation forest $\Theta_{\mathfrak{w}} \subset \mathbb{R}^2$ near a trivalent vertex. (2) The local models for the flowline behaviour near trivalent vertices and hexavalent vertices north of their source trivalent vertex.

The spectral network $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ in Section 7.2 is constructed combinatorially from \mathfrak{w} , it is just a directed graph with labels, a priori without any analytic meaning or relation to D_4^- -flowtrees. The adiabatic degeneration results in Section 3 along with the spatial front associated to a weave \mathfrak{w} , as described in [CZ22], imply that $\mathcal{W}_{\mathfrak{w}}$ is in fact obtained by adiabatically degenerating the pseudoholomorphic strips associated to the augmentation induced by the Lagrangian filling $L_{\mathfrak{w}}$ of the (-1)-closure of $\beta\delta(\beta)$. We record this fact as follows, which implies Theorem 5:

Proposition 7.5. Let $\mathfrak{w} : \beta \to w_0$ be a Demazure weave for a braid word β with $\delta(\beta) = w_0$. Consider the associated embedded exact Lagrangian filling $L_{\mathfrak{w}} \subset T^*\mathbb{R}^2$ of $\Lambda_{\beta w_0}$ and the induced augmentation $\varepsilon_{\mathfrak{w}} : \mathcal{A}(\Lambda_{\beta w_0}) \longrightarrow k[H_1(L_{\mathfrak{w}})]$ for its Legendrian contact dg-algebra. Then, the spectral network $\mathcal{W}_{\mathfrak{w}} \subset \mathbb{R}^2$ is the union of all the D_4^- -trees obtained by adiabatically degenerating the rigid pseudo-holomorphic strips contributing to $\varepsilon_{\mathfrak{w}}$.

Definition 7.4 and Proposition 7.5 provide a wealth of interesting spectral networks directly related to Floer theory. In particular, given the rigorously establish existence of and weave calculus for cluster structures, they yield a versatile set of tools for computing BPS states. For instance, the Donaldson-Thomas transformations for Demazure weaves are explicitly described in [Cas+25, Section 8.3], cf. also [CW24, Section 5.4], which readily yield the BPS spectrum generator, a.k.a. BPS monodromy, for the associated physical system. **Remark 7.6.** (1) In the meromorphic case, some of the D_4^- -trees in $W_{\mathfrak{w}}$ with a positive puncture at a Reeb chord in $\Lambda_{\partial_{\mathfrak{w}}}$ align with the BPS solitons described by the (non-degenerate) open finite webs in [GMN13b, Section 3.2].

(2) The Lusztig cycles of a Demazure weave \mathfrak{w} are introduced in [Cas+25, Section 4], see also [CW24, Section 3]. Intuitively, in the meromorphic case, the mutable Lusztig cycles relate to $W_{\mathfrak{w}}$ in that they should correspond to finite webs representing 4d BPS states, as in [GMN13b, Section 3.1], for a degenerate spectral network related to (but in a different phase than) the non-degenerate network $W_{\mathfrak{w}}$. This relation remains to be explored in detail.

7.3. Explicit computations of spectral networks from Demazure weaves. In this subsection, we discuss several cases of Definition 7.4 and Proposition 7.5 in detail, establishing the comparison between the flowlines of spectral networks and the adiabatic degenerations of pseudoholomorphic strips.

7.3.1. A 2-stranded example in detail. Consider the braid word $\beta = \sigma_1^6$ in 2-strands and the Legendrian dg-algebra \mathcal{A}_{β} of the (-1)-closure of $\beta\delta(\beta)$. Let us denote its degree-0 Reeb chords by the variables $z_6, z_5, z_4, z_3, z_2, z_1, w_1$, read left-to-right, with w_1 corresponding to the crossing of $\delta(\beta) = \sigma_1$. This labeling is drawn in Figure 19. Consider the Demazure weave $\mathbf{w} : \beta \longrightarrow \delta(\beta)$ depicted in Figure 19 (left). In the language of pinching sequences, the Lagrangian filling $L_{\mathbf{w}}$ is obtained by pinching the crossings in the Lagrangian projection associated to z_1, z_3, z_4, z_5, z_2 , in this order. A computation, as in [CN22], shows that $L_{\mathbf{w}}$ gives the augmentation

$$\varepsilon_{\mathfrak{w}}: \mathcal{A}_{\beta} \longrightarrow k[H_1(L_{\mathfrak{w}}, T)]$$

uniquely determined by its non-zero values:

$$\begin{aligned} \varepsilon_{\mathfrak{w}}(z_{6}) &= \frac{1}{s_{2}s_{3}^{2}s_{4}^{2}s_{5}^{2}} - \frac{1}{s_{5}} \\ \varepsilon_{\mathfrak{w}}(z_{5}) &= s_{5} - \frac{1}{s_{4}} \\ \varepsilon_{\mathfrak{w}}(z_{5}) &= s_{5} - \frac{1}{s_{4}} \\ \varepsilon_{\mathfrak{w}}(z_{4}) &= s_{4} - \frac{1}{s_{3}} \\ \varepsilon_{\mathfrak{w}}(z_{3}) &= s_{3} \\ \varepsilon_{\mathfrak{w}}(z_{3}) &= s_{3} \\ \varepsilon_{\mathfrak{w}}(z_{2}) &= s_{2} - \frac{1}{s_{1}} - \frac{1}{s_{3}} + \frac{1}{s_{3}^{2}s_{4}} - \frac{1}{s_{3}^{2}s_{4}^{2}s_{5}} \end{aligned}$$

$$\begin{aligned} \varepsilon_{\mathfrak{w}}(t_{1}) &= -\frac{1}{s_{1}s_{2}s_{3}s_{4}s_{5}} \\ \varepsilon_{\mathfrak{w}}(t_{2}) &= s_{1}s_{2}s_{3}s_{4}s_{5}. \end{aligned}$$

where t_1, t_2 are marked points $T = \{t_1, t_2\}$ at the (right at the) right of w_1, t_i in the *i*th strand, counting from the bottom, and $s_i \in H_1(L_{\mathfrak{w}}, T)$ is the relative cycle Poincaré dual to the trivalent vertex whose associated top-right edge corresponds to z_i .

By construction, $\varepsilon_{\mathfrak{w}}$ counts rigid pseudo-holomorphic strips and records their (relative) homology classes. Therefore, there ought to be a precise correspondence between the non-zero terms in $\varepsilon_{\mathfrak{w}}(z_i)$,

and $\varepsilon_{\mathfrak{w}}(w_1)$, and the rigid flowtrees of the spectral network $\mathcal{W}_{\mathfrak{w}}$ for $L_{\mathfrak{w}}$ which are asymptotic to z_i , and w_1 . This is indeed the case: $\mathcal{W}_{\mathfrak{w}}$ is drawn in Figure 19 (right), and each term of $\varepsilon_{\mathfrak{w}}$ above exactly corresponds to such flowtrees. For instance, the flowtrees in the spectral network that are coloured not black correspond to the terms of

$$\varepsilon_{\mathfrak{w}}(z_2) = s_2 - \frac{1}{s_1} - \frac{1}{s_3} + \frac{1}{s_3^2 s_4} - \frac{1}{s_3^2 s_4^2 s_5},$$

as they are asymptotic to the Reeb chord z_2 . Specifically, the green flowline gives the term $-s_3^{-1}$, the yellow gives $s_3^{-2}s_4$, the orange $-s_3^{-2}s_4^{-2}s_5^{-1}$, the purple gives s_2 and the rightmost cardinal flowline $-s_1^{-1}$.



FIGURE 19. (Left) A Demazure weave \mathfrak{w} for $\beta = \sigma_1^6$, with its Reeb chords at the (top) boundary marked with a pink dot and labeled. (Right) The spectral network $\mathcal{W}_{\mathfrak{w}}$ compatible with $L_{\mathfrak{w}}$, as built above.

7.3.2. Mutation in 2-strands. Similar to the example above, this correspondence between terms of the augmentation $\varepsilon_{\mathfrak{w}}$ and the flowlines in $\mathcal{W}_{\mathfrak{w}}$ works in the same lin for other 2-stranded braids and their Demazure weaves. The mutation in spectral networks can be analyzed in the local model of $\beta = \sigma_1^3$. The two sides of a weave mutations are depicted in Figure 20, \mathfrak{w}_1 on the left and \mathfrak{w}_2 on the right.

If we label the degree-0 Reeb chords associated to the crossings of $\beta\delta(\beta) = \sigma_1^4$ by z_3, z_2, z_1, w_1 , reading left-to-right, and use $T = \{t_1, t_2\}$ as above, the two augmentations

$$\varepsilon_{\mathfrak{w}_i}: \mathcal{A}_{\sigma_1^3} \longrightarrow k[H_1(L_{\mathfrak{w}_i}, T)], \quad i = 1, 2,$$



FIGURE 20. The two sides of a weave mutation, \mathfrak{w}_1 and \mathfrak{w}_2 , and their associated spectral networks $\mathcal{W}_{\mathfrak{w}_1}$ and $\mathcal{W}_{\mathfrak{w}_2}$.

are given by

$$\begin{split} \varepsilon_{\mathfrak{w}_{1}}(z_{3}) &= \frac{1}{s_{1}s_{2}^{2}} - \frac{1}{s_{2}} & \varepsilon_{\mathfrak{w}_{2}}(z_{3}) &= -\frac{1}{s_{2}} \\ \varepsilon_{\mathfrak{w}_{1}}(z_{2}) &= s_{2} & \varepsilon_{\mathfrak{w}_{2}}(z_{2}) &= s_{2} - \frac{1}{s_{1}} \\ \varepsilon_{\mathfrak{w}_{1}}(z_{1}) &= s_{1} - \frac{1}{s_{2}} & \varepsilon_{\mathfrak{w}_{2}}(z_{1}) &= s_{1} \\ \varepsilon_{\mathfrak{w}_{1}}(w_{1}) &= -\frac{1}{s_{1}} & \varepsilon_{\mathfrak{w}_{2}}(w_{1}) &= \frac{1}{s_{1}^{2}s_{2}} - \frac{1}{s_{1}} \\ \varepsilon_{\mathfrak{w}_{1}}(t_{1}) &= -\frac{1}{s_{1}s_{2}} & \varepsilon_{\mathfrak{w}_{2}}(t_{1}) &= -\frac{1}{s_{1}s_{2}} \\ \varepsilon_{\mathfrak{w}_{1}}(t_{2}) &= -s_{1}s_{2}. & \varepsilon_{\mathfrak{w}_{2}}(t_{2}) &= -s_{1}s_{2}. \end{split}$$

Note that in the z-coordinates of the braid variety $X(\sigma_1^3)$, the mutable cluster variable A_1 for the torus chart $T_{\mathfrak{w}_1} \subset X(\sigma_1^3)$ associated to $L_{\mathfrak{w}_1}$ is $A_1 = z_2$, and the frozen is $z_2z_3 - 1$. The quiver $Q_{\mathfrak{w}_2}$ has a unique arrow from the frozen to the mutable. After mutation, the mutable cluster variable A_2 for the torus chart $T_{\mathfrak{w}_2} \subset X(\sigma_1^3)$ associated to $L_{\mathfrak{w}_2}$ is $A_2 = z_3$, and the frozen $z_3z_2 - 1$. The quiver $Q_{\mathfrak{w}_1}$ has a unique arrow from the mutable to the frozen. By the mutation formula, we must have that the mutated variable A'_1 of $A_1 = z_2$ is such that

$$A_1 A_1' = 1 + (z_2 z_3 - 1)$$

and thus indeed $A'_1 = A_2 = z_3$. The corresponding mutation in the spectral networks $\mathcal{W}_{\mathfrak{w}_1}$ and $\mathcal{W}_{\mathfrak{w}_2}$, i.e. its change in the flowtrees, is drawn in Figure 20. The change in the flowlines, seen as adiabatic limits of pseudo-holomorphic strips, matches the difference in the augmentation $\varepsilon_{\mathfrak{w}_1}$ and $\varepsilon_{\mathfrak{w}_2}$ and, using z-coordinates, it becomes the standard cluster mutation.

7.3.3. The Berk-Nevins-Roberts spectral network. The remarkable article [BNR82] first introduced the new Stokes lines, born at certain intersections of two ordinary Stokes lines and studied them



FIGURE 21. (Left) The weave \mathfrak{w}_{bnr} obtained by considering the front of the Legendrian lift of the real part of the Bers-Nevins-Roberts spectral curve Σ_{bnr} . (Right) The spectral network $\mathcal{W}_{\mathfrak{w}_{bnr}}$ according to Section 7.2.

in the context of the WKB asymptotics, c.f. Example 2.12. These flowlines only appear at higher rank $n \ge 3$, and mark a crucial difference between the rank-2 case and higher ranks. The specific example studied in [BNR82] corresponds to the spectral curve

$$\Sigma_{bnr} := \{ (z, w) \in T^* \mathbb{C} : w^3 - 3w + x = 0 \}.$$

The weave \mathfrak{w}_{bnr} associated to its real part $\operatorname{Re}(\Sigma)$ is depicted in Figure 21 (left). It is already depicted with the bending, so that it is apparent that can be seen as a bent Demazure weave for the braid word $\beta = \sigma_2 \sigma_1 \sigma_2 \sigma_1 \sigma_2$. Note that the associated spectral network \mathcal{W}_{bnr} , depicted in [BNR82, Figure 1] was analyzed in [Kat+15, Section 5] from the perspective of harmonic maps to buildings.

From the Floer-theoretic perspective, we label the Reeb chords of $\beta\delta(\beta) = \sigma_2\sigma_1\sigma_2\sigma_1\sigma_2(\sigma_1\sigma_2\sigma_1)$ as in Figure 21: z_1, z_2, z_3, z_4, z_5 correspond to the crossings of β , reading right-to-left, and w_1, w_2, w_3 to those of $\delta(\beta)$, also reading right-to-left. The spectral network $\mathcal{W}_{\mathfrak{w}_{bnr}}$ associated to \mathfrak{w}_{bnr} according to Section 7.2 is depicted in Figure 21 (right). We have also drawn $\mathcal{W}_{\mathfrak{w}_{bnr}}$ on its own in Figure 22, where the two *new* Stokes lines, born at creation vertices, are depicted in purple. Here we can readily see that the spectral network $\mathcal{W}_{\mathfrak{w}_{bnr}}$ obtained by adiabatic degeneration of pseudoholomorphic strips coincides with the original spectral network \mathcal{W}_{bnr} from [BNR82] associated to Σ_{bnr} .



FIGURE 22. The spectral network $\mathcal{W}_{\mathfrak{w}_{bnr}}$, which indeed coincides with the spectral network \mathcal{W}_{bnr} from [BNR82]. In this figure, the two new Stokes line are highlighted in purple, and the two trivalent D_4^- -vertices are circled in dashed yellow.

7.3.4. An example of a cancellation pair. Let us consider a local model for a cancellation pair coming from applying a Reidemeister III move twice to $\beta = \sigma_2 \sigma_1 \sigma_2$. This corresponds to the sequence of braid words $\sigma_2 \sigma_1 \sigma_2 \rightarrow \sigma_1 \sigma_2 \sigma_1 \rightarrow \sigma_2 \sigma_1 \sigma_2$. First, we will consider one Reidemeister III move $\sigma_2 \sigma_1 \sigma_2 \rightarrow \sigma_1 \sigma_2 \sigma_1$ on its own, as realized by the weave \mathbf{w}_{R3} in Figure 23. This is understood as a local move that can be inserted anywhere in a weave where there is a subword $\sigma_{i+1}\sigma_i\sigma_{i+1}$.



FIGURE 23. The four degenerations of pseudoholomorphic strips contributing to the dg-algebra morphism induced by a Reidemeister III move on a Legendrian front. In (4), the bifurcation occurs as a creation vertex from the flowlines in (1) and (2), the traces of which are drawn in dashed gray in (4).

The embedded exact Lagrangian cobordism associated to \mathfrak{w}_{R3} gives a dg-algebra morphism Ψ_{R3} between the Legendrian contact dg-algebra of the convex end, which has the piece $\sigma_2\sigma_1\sigma_2$, to the concave end, which contains $\sigma_1\sigma_2\sigma_1$. Denote by z_1, z_2, z_3 the Reeb chords next to the crossings associated to $\sigma_2\sigma_1\sigma_2$, and keep the same notation for those in $\sigma_1\sigma_2\sigma_1$, as depicted in Figure 23.(1). This morphism Ψ_{R3} is computed in [CN22, Section 4.1.2] and, locally in this model, reads as follows:

$$\Psi_{R3}(z_1) = z_3, \quad \Psi_{R3}(z_3) = z_1, \quad \Psi_{R3}(z_2) = z_2 \pm z_1 z_3,$$

where we do not specify the sign, cf. [CN22] for the choices that give precise signs. Each of these four terms exists because of a pseudoholomorphic strip. Their adiabatic degenerations, in the form of flowtrees, are depicted in Figure 23. Namely, Figure 23.(1) explains $\Psi_{R3}(z_3) = z_1$, 23.(2) explains $\Psi_{R3}(z_1) = z_3$, 23.(3) is the tree corresponding to the term z_2 in $\Psi_{R3}(z_2) = z_2 \pm z_1 z_3$ and the D_4^- -tree in 23.(4) corresponds to the term $z_1 z_3$.

Let us now perform two consecutive Reidemeister moves, inverses of each other. The corresponding weave \mathbf{w}_c is depicted in Figure 24.(1). By the candy twist move in [CZ22, Figure 19], this weave \mathbf{w}_c is equivalent to the trivial weave with three strands labeled s_1, s_2, s_1 and no crossings. Therefore, by Hamiltonian invariance of the associated exact embedded Lagrangian cobordisms, it must be that the induced dg-algebra map given by \mathbf{w}_c is the identity and there *has to* be a cancelation account for the term z_1z_3 above appearing and disappearing.



FIGURE 24. (1) The weave corresponding to two consecutive Reidemeister III moves $\sigma_2\sigma_1\sigma_2 \rightarrow \sigma_1\sigma_2\sigma_1 \rightarrow \sigma_2\sigma_1\sigma_2$. (2) The associated spectral network, which coincides with the adiabatic D_4^- -tree degeneration of the pseudoholomorphic strips corresponding to the induce dg-algebra maps. (3) The spectral network on its own, with the two canceling terms highlighted in purple. (4) The result of deleting the walls of the spectral network in (3) that cancel each other. In this case, the interaction vertex that is not creation would be of *annihilation* type. The actual spectral network in (3) witnesses both canceling flowlines, and their cancelation is seen in the BPS index. More intuitively, the D_4^- -tree corresponding to the upper purple flowline has a self-intersection point, unlike the tree for the purple flowline, hence its sign is opposite.

This cancellation is indeed seen in the spectral network $\mathcal{W}_{\mathfrak{w}_c}$, which is depicted in Figure 24.(2). The two flowlines, with the same asymptotic end, that cancel each other are highlighted in purple in Figure 24.(3). In the physics literature, this configuration is sometimes depicted as in Figure 24.(4), where the parts of the walls of the canceling pair that have vanishing BPS index are not

drawn: this is, with phases appropriately understood, related to the *setting sun* model in [GMN13b, Section 7.4].

7.3.5. A more elaborate 3-stranded example. Consider the 3-stranded braid word $\beta = (\sigma_2 \sigma_1)^3 \sigma_2$, so that $\beta \delta(\beta) = (\sigma_2 \sigma_1)^5 \sigma_2$ and the (-1)-closure of $\beta \delta(\beta)$ is the max-tb (3,2)-torus knot, i.e. the max-tb trefoil presented in 3-strands. Consider its right inductive weave \mathbf{w} , as depicted in Figure 25 (left), and its associated embedded exact Lagrangian filling $L_{\mathbf{w}}$. This example illustrates how the correspondence between the flowlines of the spectral network $\mathcal{W}_{\mathbf{w}}$ and adiabatic degenerations of the pseudoholomorphic strips contributing to the augmentation $\varepsilon_{\mathbf{w}}$ induced by $L_{\mathbf{w}}$. It is particularly helpful in understanding how terms can (and do) cancel, locally as in Section 7.3.4, in a globally interesting example.



FIGURE 25. (1) The right inductive weave \mathfrak{w} for $\beta = (\sigma_2 \sigma_1)^3 \sigma_2$. The four trivalent vertices are labeled v_1, v_2, v_3, v_4 , the Reeb chords associated to crossings in β are denoted z_1, \ldots, z_7 and those for $\delta(\beta)$ are denoted w_1, w_2, w_3 . (2) The first prespectral network \mathcal{F}_1 , consisting of the three flowlines starting at the trivalent D_4^- -vertex v_1 . (3) The second prespectral network \mathcal{F}_2 , consisting of the flowlines starting at the two D_4^- -vertices v_1, v_2 . Notice that it is *not* consistent, as there are 4-valent vertices in \mathcal{F}_2 which should be creation vertices. A consistent extension is drawn in Figure 26.(4).

First, from the Floer-theoretic side, we label the Reeb chords as in Figure 25 (left). In these coordinates, the Lagrangian filling $L_{\mathfrak{w}}$ gives the augmentation

$$\varepsilon_{\mathfrak{w}}: \mathcal{A}_{\beta} \longrightarrow k[H_1(L_{\mathfrak{w}}, T)]$$

which determined by its non-zero values:

(41)

$$\varepsilon_{\mathfrak{w}}(z_{7}) = -\frac{s_{3}}{s_{2}s_{4}^{2}} - \frac{1}{s_{4}} \qquad \qquad \varepsilon_{\mathfrak{w}}(w_{3}) = \frac{s_{1}}{s_{2}} - \frac{1}{s_{3}} \\ \varepsilon_{\mathfrak{w}}(z_{6}) = \frac{s_{2}}{s_{1}s_{3}^{2}} - \frac{1}{s_{3}} + \frac{1}{s_{1}s_{3}s_{4}} \qquad \qquad \varepsilon_{\mathfrak{w}}(w_{2}) = -\frac{1}{s_{2}} \\ \varepsilon_{\mathfrak{w}}(z_{5}) = \frac{s_{2}s_{4}}{s_{1}s_{3}^{2}} - \frac{s_{4}}{s_{3}} \qquad \qquad \varepsilon_{\mathfrak{w}}(w_{2}) = -\frac{1}{s_{2}} \\ \varepsilon_{\mathfrak{w}}(z_{4}) = s_{4} \qquad \qquad \varepsilon_{\mathfrak{w}}(z_{4}) = s_{4} \\ \varepsilon_{\mathfrak{w}}(z_{3}) = s_{3} \qquad \qquad \varepsilon_{\mathfrak{w}}(z_{1}) = -\frac{1}{s_{1}s_{3}} \\ \varepsilon_{\mathfrak{w}}(z_{2}) = s_{2} + \frac{s_{3}}{s_{4}} \qquad \qquad \varepsilon_{\mathfrak{w}}(t_{1}) = -\frac{1}{s_{1}s_{3}} \\ \varepsilon_{\mathfrak{w}}(z_{1}) = s_{1} - \frac{s_{2}}{s_{3}} - \frac{1}{s_{4}} \qquad \qquad \varepsilon_{\mathfrak{w}}(t_{3}) = -s_{2}s_{4}.$$

where t_1, t_2, t_3 are marked points $T = \{t_1, t_2, t_3\}$ at the (right at the) right of w_1, t_i in the *i*th strand, counting from the bottom. As above, $s_i \in H_1(L_{\mathfrak{w}}, T)$ denotes the relative cycle Poincaré dual to the trivalent vertex whose associated top-right edge corresponds to z_i .

Second, independent of the Floer-theoretic computation, we consider the spectral network $\mathcal{W}_{\mathfrak{w}}$ associated to the weave \mathfrak{w} , as built in Section 7.2. The final spectral network $\mathcal{W}_{\mathfrak{w}}$ is drawn in Figure 28. The process to obtain it is depicted in several steps and described as follows:

- (i) The first prespectral network \mathcal{F}_1 is depicted in Figure 25.(2), where the three walls of \mathcal{F}_1 are drawn, being born at the trivalent D_4^- -vertex v_1 .
- (ii) The second prespectral network \mathcal{F}_2 is depicted in Figure 25.(3), where three additional walls are born at the D_4^- -vertex v_2 , in addition to those for \mathcal{F}_1 are drawn.
- (iii) Since the prespectral network \mathcal{F}_2 not consistent, we proceed by considering a consistent extension \mathcal{F}_3 , as drawn in Figure 26.(4). It is obtained by adding four flowlines to \mathcal{F}_2 , resulting in four creation vertices. The prespectral network \mathcal{F}_3 is consistent.
- (iv) The next prespectral network \mathcal{F}_4 is depicted in Figure 26.(5), where the three flowlines born in the D_4^- -vertex v_3 are added. Such \mathcal{F}_4 is inconsistent, and thus we consider its



FIGURE 26. Prespectral networks in the steps towards building $\mathcal{W}_{\mathfrak{w}}$. In (4) we added flowlines to make the network in Figure 25.(3) consistent. In (5) the flowlines born from v_3 are added and (6) depicts the associated consistent extension.

consistent extension \mathcal{F}_5 , which is drawn in Figure 26.(6). The only flowline needed to be added to \mathcal{F}_4 for consistency is depicted in pink, which itself is caused by the flowline born in v_3 highlighted in green interacting with a flowline born in v_2 .

(v) The final step is to add the three flowlines born at v_4 and the additional flowlines to make the resulting prespectral network consistent. The resulting spectral network is $\mathcal{W}_{\mathfrak{w}}$, which is drawn in Figure 27 (left), on top of the weave \mathfrak{w} . It is drawn on its own in Figure 27 (right).

The spectral network \mathcal{W}_{w} , with all its flowlines, is drawn in Figure 27 (right), but some of these flowlines cancel each other. That is, when considering counts of BPS states or pseudoholomorphic strips, there are pairs of such flowlines where each of the two flowlines contributes the same term, but they do so with opposite signs. Specifically, this occurs in this example in two cases: the two pink flowlines in Figure 27 (right) cancel each other, and two of the yellow flowlines (the two ones born lower than the third one) also cancel each other. We have depicted the spectral network obtained by removing such flowlines in Figure 28, where the two canceling yellow lines are drawn dashed and the two pink flowlines have been erased.



FIGURE 27. (Left) The spectral network $\mathcal{W}_{\mathfrak{w}}$ on the weave \mathfrak{w} . (Right) The spectral network $\mathcal{W}_{\mathfrak{w}}$ drawn on its own. The colors for some of the flowlines indicate that they are creation flowlines: those that share a color are created by (repeated) intersections of the same two flowlines. Some pairs of flowlines of the same color cancel each other, e.g. the two pink flowlines cancel each other.

We conclude by noting that the flowlines for the spectral network $\mathcal{W}_{\mathfrak{w}}$, as drawn in Figure 28, indeed match with the terms of the augmentation $\varepsilon_{\mathfrak{w}}$ as described in Eq. (41).



FIGURE 28. The spectral network $\mathcal{W}_{\mathfrak{w}}$ associated to the weave \mathfrak{w} . The two dashed flowlines, drawn in yellow, cancel each other. The (solid) flowlines in yellow, green and blue are the flowlines born in creation vertices in the last step of the process, after adding the flowlines born in v_4 and making the resulting prespectral network consistent.

APPENDIX A. ASPECTS OF THE GEOMETRY OF PSEUDO-HOLOMORPHIC CURVES

This appendix collects three technical results that are used in the manuscript: geometric boundedness of meromorphic spectral curves, a type of reverse isoperimetric inequality and a brief discussion on spin structures.

A.1. Geometric boundedness of meromorphic spectral curves. In this section, we show that the meromorphic spectral curves that satisfy the O(-1)-end conditions are geometrically bounded, in the sense of [GS16, Definition 2.7]. This is achieved by showing that the C^{ℓ} -norm of the second fundamental form is uniformly bounded. We use the Sasaki metric obtained from metrics of the form $|z|^{-2}|dz|^2$ near the poles. This metric is geometrically bounded outside T^*K for some compact subset $K \subset C$, since the metric $|z|^{-2}|dz|^2$ is flat. This is used in Sections 4, 5 and 6. The precise result reads as follows:

Proposition A.1. Let $\Sigma \subset T^*C$ be a meromorphic spectral curve. Then the smooth sheets of the scaled curves $\varepsilon \Sigma \subset T^*C$ near infinity are uniformly geometrically bounded, uniformly with respect to ε , for any $\varepsilon \in \mathbb{R}_+$.

Proof. The argument is essentially finding an appropriate coordinate system where geometric boundedness is apparent. First, by construction, the germ of Σ at infinity can be written as

$$z \to (z, c_1 t^{-k} + c_2 t^{-k+1} + \ldots + c_{k-n} t^{-n}), \quad \text{for some } n < k, \quad t^n = z.$$

Choose a branch cut for the logarithm and consider the flat conformal coordinate $W := \int z^{-1} dz = \log(z)$; we can and do assume that |z| < 1, so that W has *negative* real part. Under this coordinate transformation, the smooth sheets of the meromorphic spectral curve transform into

$$(W, c_1 \exp((-k/n + 1)W) + \ldots + c_k).$$

Set m = k/n and note that $|\exp((-m+1)W)|$ converges to $+\infty$ as $W \to -\infty$ because m > 1. Also, since the metric $|z|^{-2}|dz|^2$ transforms into $|dW|^2$, the metric on $\mathbb{C}^2_{W,dW}$ is the standard flat metric. By denoting $Q := \exp(c_1(-m+1)W)$, and using the branch cut, we can write $\log(Q) = c_1(-m+1)W$. This coordinate Q is the one we will use to prove geometric boundedness, as follows. Reparametrizing the sheets in the Q-coordinate we obtain

$$(b_0 \log(Q), b_1 Q + O(Q^{1/(k-1)}))$$

for some $0 < \alpha < 1$ and non-zero constants b_0, b_1 . We now show that curves in \mathbb{C}^2 of the form

 $(\log(Q), Q)$

are geometrically bounded over the entire complex plane, with the more general case with the $O(Q^{1/(k-1)})$ term follows similarly. Consider a curve of the form $(\log(Q), Q)$. If $|Q| \leq 1$ then re-write $W = \log(Q)$ so that $|Q| \leq 1$ is equivalent to W having negative real part. The geometric boundedness of the curve for $|Q| \leq 1$ then follows from the geometric boundedness of the curve (W, e^W) for $\operatorname{Re}(W) < 0$.

If $|Q| \ge 1$, note that given a holomorphic curve of the form (f(z), g(z)) in \mathbb{C}^2 , the tangent vectors are given by (f'(z), g'(z)), (if'(z), ig'(z)) and the normal vectors are given by $(-\overline{g'(z)}, \overline{f'(z)})$ and $(-i\overline{g'(z)}, i\overline{f'(z)})$. Specializing to our case, the tangent vectors are given by (1/Q, 1) and (i/Q, i), and the normal vectors are given by $(-1, 1/\overline{Q})$ and $(-i, i/\overline{Q})$. The induced metric on the curve is $(|Q|^{-2} + 1)|dQ|^2$. Therefore, the non-vanishing Christoffel symbols are given by

$$\Gamma^Q_{QQ} = (1 + |Q|^{-2})^{-1} (\bar{Q}^{-1}),$$

$$\Gamma^{\bar{Q}}_{\bar{Q}\bar{Q}} = (1 + |Q|^{-2})^{-1} (Q^{-1}),$$

and the second fundamental form reads

$$B(\partial_z, \partial_z) = \left[-Q^{-2}/(1+|Q|^{-1})^{3/2}\right](-1, 1/\bar{Q}).$$

A direct computation implies that the derivatives of the second fundamental form, the Christofell symbols, and their derivatives are of the form $O(|Q|^{-1})$. The norm of ∂_z is uniformly positive, and the derivatives are again $O(|Q|^{-1})$. Therefore, C^{ℓ} norm of the second fundamental form must

be bounded as well. In addition, these bounds are uniform with respect to ε because we can reparametrize $(\log(Q), \epsilon Q) = (\log(P) - \log(\epsilon), P)$ and use the boundedness of $(\log(Q), Q)$.

A consequence of Proposition A.1 is the following a priori diameter estimate for pseudoholomorphic disks with boundaries on Betti Lagrangians and conormals:

Lemma A.2. Let S be a Betti surface and $L \subset T^*S$ a Betti Lagrangian. If L is not exact, suppose that it is meromorphic near the punctures. Let $K \subset S$ be a precompact submanifold of S containing $\pi(K_L)$ in its interior. Then there exists $\varepsilon_0 = \varepsilon_0(M, g, L, K) \in \mathbb{R}_+$ such that for any $\varepsilon \in (0, \varepsilon_0)$ and J_g -holomorphic half-disk $u : (A_1, \partial^+A_1) \to (D^*M, D^*(K) \cup \varepsilon L)$ with energy less than εE , the image of u is contained in the disk bundle $D^*N_{CE}(K)$, where $C \in \mathbb{R}_+$ depends only on (M, g, L, K).

Proof. By Proposition A.1, smooth sheeets of $\varepsilon \Sigma$ are uniformly geometrically bounded near infinity. Thus an argument as in [AL94, Chapter V] implies that that given any enery cut-off $E \in \mathbb{R}_+$, there is a constant $C_0 \in \mathbb{R}_+$ such that pseudo-holomorphic half-disks with boundary on a *single* smooth sheet of $\varepsilon \Sigma$ satisfy the estimate. Since $\varepsilon \Sigma$ is multi-sheeted, we set $C = \max(\rho^{-1}, 1) \cdot C_0$ and iterate. For the monotonicity argument, we need to choose a small neighborhood of points on L such that its intersection with εL is contractible and connected. The size of such a neighborhood will be $\rho\varepsilon$ and shrinking the size of the neighborhood affects the diameter given by monotonicity estimate by $\rho^{-1}\varepsilon^{-1}$. Since ρ^{-1} is fixed, the diameter is bounded by CE, as required.

A.2. Truncated reverse isoperimetric inequality. Reverse isoperimetric inequalities in the context of pseudoholomorphic curves bounded by Lagrangians appeared in [GS14], cf. also [Duv16]. As pointed out in Section 4, we believe an inequality of this type is also required in the arguments of [Ekh07] and, for completeness, we include a proof here.

Let X be a compact almost Kähler manifold and K a union of finitely many balls in X. Let $L \subset X$ be a Lagrangian submanifold and r_0 the radial injectivity radius of L outside K. We consider r_0 to be small enough such that outside of K, the squared radial distance function $h := \rho^2 : D_{r_0}^* L \to \mathbb{R}$ is strictly plurisubharmonic with respect to J. For $\delta \in \mathbb{R}_+$, we write $N_{\delta}(K)$ for the δ -neighborhood of K. The necessary reverse isoperimetric inequality reads as follows:

Theorem A.3. In the notation above, there exist a choice of $\delta \in \mathbb{R}_+$ and a constant $T \in \mathbb{R}_+$ such that any *J*-holomorphic curve

$$u: (S, \partial S) \to (X, L \cup K)$$
 with energy $\frac{1}{2} \int |du|^2 \le E$

has boundary length outside $N_{2\delta}(K)$ bounded above by TE.

Proof. At core, the argument uses the coarea formula to slice in terms of the level sets of the distance function ρ and then use the Lagrangian boundary condition to study bounds for the length. We have structured the proof in four steps: summarizing, we will choose an appropriate bump function β around K and define functions a(r) which coarsely measures the area outside K.

Then we show that the inequality

$$ra'(r) \ge a(r)\frac{1-Dr}{1+Cr}$$

for some constants $C, D \in \mathbb{R}_+$ that only depend on (X, K, L, r_0) but not on u. The required bound on length will follows from that.

First, let us construct the support function $\beta : X \to \mathbb{R}$, which typically depends on u, though the resulting constants C, D will not be affected. Fix a reference constant $\delta \in (0, r_0/2)$ and $\delta_u \in (0, \delta)$. We then choose a non-negative smooth function $\beta : X \to \mathbb{R}_{\geq 0}$ such that

- (1) β vanishes on $N_{\delta_0}(K)$,
- (2) β is strictly positive on $N_{\delta_0}(K)^c \partial N_{\delta_0}(K)$,
- (3) β is equal to 1 outside $N_{2\delta}(K)$.

Choose δ_u above generically enough so that u is transverse to $\partial N_{\delta}(K)$, and sufficiently close to δ so that the corresponding bump function β has its derivative bounded above by $C\delta^{-1}$, for some constant $C \in \mathbb{R}_+$, independent of δ_u . The transversality condition is needed because in Step 3 below we will need $u^{-1}(\partial N_{\delta}(K))$ to have *finite* length, although its length does not contribute to our inequality.

Set $N := N_{\delta_0}(K)$ for the δ_0 -neighborhood of K and choose a strictly decreasing sequence $\delta_m \to \delta_u$ so that ρ/β is Lipshitz on the δ_m -neighborhood $N_m := N_{\delta_m}(K)$. The functions a(r) and $l^{\beta}(r)$ are defined, for almost every real number, as follows.

(42)
$$a(r) := \int_{\{\rho \le \beta r\} \cap u \cap N^c} dA \qquad a_m(r) := \int_{\{\rho \le \beta r\} \cap u \cap N^c_m} dA$$

(43)
$$l^{\beta}(r) := \int_{\{\rho = \beta r\} \cap u \cap N^c} \beta dl. \qquad l^{\beta}_m(r) := \int_{\{\rho = \beta r\} \cap u \cap N^c_m} \beta dl.$$

Here, the integral is taken with respect to the induced metric on the image of u. The reason why $a_m(r)$ and $l_m^\beta(r)$ are introduced is that we may apply the co-area formula. Since the integrand is always non-negative, by the monotone convergence theorem, we can freely replace the integral and limit.

Step 1. We show that the function $a_m(r)$ is absolutely continuous and satisfies

(44)
$$a_m(r) \ge \int_0^r \frac{l_m^\beta(\tau)}{1+\tau C} d\tau \quad \text{and} \quad a'_m(r) \ge \frac{l_m^\beta(r)}{1+rC} \quad \text{a.e.}$$

We deduce Eq. (44) from the coarea formula, as follows. Since β is positive on the complement N^c , it follows that ρ/β is Lipschitz on N_m^c for any $m \in \mathbb{N}$. Applying the coarea formula, we obtain (45)

$$a_m(r) = \int_{\{\rho \le \beta r\} \cap u \cap N_m^c} dA = \int_{\{\rho \le \beta r\} \cap u \cap N_m^c} \frac{1}{\left|\nabla(\rho/\beta)\right|} \cdot \left|\nabla(\rho/\beta)\right| dA = \int_0^r \int_{\{\rho = \beta \tau\} \cap u \cap N_m^c} \frac{1}{\left|\nabla(\rho/\beta)\right|} dl d\tau.$$

Since $|\nabla \rho| = 1$, in the region $\rho = \tau \beta$ we have the estimate

(46)
$$\left|\nabla(\rho/\beta)\right| \leq \frac{1}{\beta} |\nabla\rho| + \frac{\rho|\beta'|}{\beta^2} \leq \frac{1}{\beta} \left(|\nabla\rho| + \tau |\beta'| \right) \leq \frac{1 + \tau C}{\beta}.$$

Thus (44) follows from from (45) and (46). Since $a_m(r)$ is a absolutely continuous non-negative function with non-negative derivative, it follows that $a(r) = \lim_{m\to\infty} a(r)$ is also absolutely continuous. Furthermore, by the monotone convergence theorem applied with respect to δ_m ,

(47)
$$a(r) \ge \int_0^r \frac{l^{\beta}(\tau)}{1+\tau C} d\tau \quad \text{and} \quad a'(r) \ge \frac{l^{\beta}(r)}{1+rC} \quad \text{a.e.}$$

Step 2. Recall that we denote $h = \rho^2$ for the square of the distance function to L. Let us show the inequality

$$rl^{\beta}(r) \ge \frac{1}{2} \int_{\{\rho \le \beta r\} \cap u \cap N^c} dd^c h.$$

The key observation is that, because β vanishes near $\partial(N_{\delta}(K))$, we can suppress the contribution to the length coming from $u \cap \partial(N_{\delta}(K))$. For almost every r, we have the following:

(48)
$$rl_{m}^{\beta}(r) = \int_{\{\rho=\beta r\}\cap u\cap N_{m}^{c}} r\beta dl \ge \int_{\{\rho=\beta r\}\cap u\cap N_{m}^{c}} \frac{1}{2} \langle \nabla h, \nu \rangle dl$$
(40)
$$= \int_{\{\rho=\beta r\}\cap u\cap N_{m}^{c}} \frac{1}{d} dr h = \int_{\{\rho=\beta r\}\cap u\cap N_{m}^{c}} \frac{1}{d} dr h \int_{\{\rho=\beta r\}\cap u\cap N_{m}^{c}} \frac{1}{d} dr h dr h dr h$$

(49)
$$= \int_{\{\rho=\beta r\}\cap u\cap N_m^c} \frac{1}{2} d^c h = \frac{1}{2} \int_{\{\rho\leq\beta r\}\cap u\cap N_m^c} dd^c h - \int_{\{\rho\leq\beta r\}\cap u\cap\partial N_m} d^c h$$

Here we have parametrized the oriented smooth curve $\{\rho = \beta r\} \cap u \cap N_m^c\}$ via l(t) and took its unit normal $\nu(t) = -J(du \circ l(t)) | du \circ l'(t) |^{-1}$. We are also using the fact that the critical points of u do not lie on l(t) for generic r. For the inequality in (48), we use that $|\nabla \rho| = 1$ and, from $\langle \nabla h, \nu \rangle dl$ to $d^c h$, we used

$$\langle \nabla h, \nu \rangle dl = \langle \nabla h, \nu \rangle \left| du(l'(t)) \right| l'(t) dt = -l'(t) dh \circ (J \circ du \circ \nu(t)) = d^c h.$$

The critical points of u do not contribute because they are discrete, which is measure 0. The first equality in (49) is more involved. For that, we first use Stokes' theorem to obtain

(50)
$$\int_{\{\rho \leq \beta r\} \cap u \cap N_m^c} dd^c h = \int_{\{\rho \leq \beta r\} \cap u \cap \partial N_m} d^c h + \int_{\{\rho \leq \beta r\} \cap \partial u \cap N_m^c} d^c h + \int_{\{\rho = \beta r\} \cap u \cap N_m^c} d^c h.$$

Here, $d^c h = 0$ on $\{\rho \leq \beta r\} \cap \partial u$ since this set is contained in L. This explains why the second term in (50) vanishes. We now need to show that the term $\int_{\{\rho \leq \beta r\} \cap u \cap \partial N_m} d^c h$ converges to zero as $m \to 0$. For that, note that the set is contained in $u^{-1}(\partial N_m) = u^{-1}(\partial N_{\delta_m}(K))$, whose length is

136

uniformly bounded for sufficiently large $m \in \mathbb{N}$. This is the only place where u being transverse to $\partial(N_{\delta}(K))$ is used. Therefore, we have

(51)
$$\int_{\{\rho \le \beta r\} \cap u \cap \partial N_m} d^c h \le \int_{\{\rho \le \beta r\} \cap u \cap N_m^c} \frac{1}{2} \langle \nabla h, \nu \rangle dl \le \sup_{\partial N_{\delta_m}(K)} |\beta| r_0 \cdot \operatorname{Length}(u^{-1}(\partial N_m(K))).$$

Now, since $\sup_{\partial(N_{\delta_m}(K))} |\beta|$ converges to zero as $m \to \infty$, this term uniformly converges to zero as $m \to \infty$. The remaining term $\frac{1}{2} \int_{\{\rho \leq \beta r\} \cap u \cap N_m^c} dd^c h$ in (49) converges to $\int_{\rho \leq \beta r \cap u \cap K^c} dd^c h$ by monotone convergence theorem, since $dd^c h$ is *J*-plurisubharmonic that the integrand is necessarily non-negative. In consequence, we obtained the claimed inequality

(52)
$$rl^{\beta}(r) = \lim_{m \to \infty} rl^{\beta}_{m}(r) \ge \frac{1}{2} \int_{\{\rho \le \beta r\} \cap u \cap N^{c}} dd^{c}h.$$

Step 3. Let us prove the inequality

$$\frac{1}{2}dd^ch \ge (1-Dr)\omega, \quad \text{ for some } D \in \mathbb{R}_+.$$

Indeed, for each point $p \in L$, we can trivialize the metric and bring (X, L, J) to the configuration $(\mathbb{C}^n_{(x,y)}, \mathbb{R}^n_x, J)$ such that $h = \sum_{i,j} a_{ij} y^i y^j + O(|y|^3)$, $a_{ij}(x) = \delta_{ij} + O(|x|^2)$, J - i = O(|y|) and $g = g_{std} + O(|y|^2) + O(|x|^2)$. In particular, $\frac{1}{2}dd^ch$ restricted to x = 0 is $\omega_{std} + O(|y|)$ and g restricted to x = 0 is $g_{std} + O(|y|^2)$. It follows that $\omega = \omega_{std} + O(|y|)$ and so $\omega = dd^ch + O(|y|)$. For sufficiently small y, it follows that $dd^ch \ge (1 - Dr)\omega$.

Step 4. Let us combine the inequalities thus far to show that

$$ra'(r) \ge a(r)\frac{1-Dr}{1+Cr}$$

and conclude the proof of the truncated reverse isoperimetric inequality. Step 3 implies that $\lim rl^{\beta}(r) \geq (1 - Dr)a(r)$ for some $D \in \mathbb{R}_+$. Combining (47) and (48)–(49), we obtain the inequality

(53)
$$ra'(r) \ge a(r)\frac{1-Dr}{1+Cr}.$$

This implies the inequality

(54)
$$\frac{d}{dr}\log\left(a(r)\cdot\frac{Cr+1}{r(1-Dr)}\right) \ge 0,$$

which implies that the function

$$r \rightarrow \frac{a(r)}{r(1 - Dr)^{-(C+D)/D}}$$

is non-decreasing. For $r \ll \min(D^{-1}, C^{-1})$ small enough, choose a constant $T \in \mathbb{R}_+$ greater than $(1 - Dr)^{-(C+D)/D}$ so that

(55)
$$T\frac{a(r)}{r} \ge \lim_{s \to 0} \frac{\operatorname{Area}(u; \{\rho \le s\} \cap N_{2\delta}(K)^c)}{s} \frac{Cs+1}{(1-sD)}.$$

This implies $a(r)T \ge rl$, as required.

A.3. Local systems and twisted local systems. In Section 5 we directly work with twisted local systems. Since [GPS20] is presented only for (untwisted) local systems, Section 6 is written in the context of local systems. In the framework of Betti Lagrangians, being spin, there is an equivalence between twisted local systems and local systems. To unify both approaches, we briefly set up the non-abelianization functor Φ_W in terms of spin structures and local systems, instead of twisted local systems. The notation follows that of the manuscript. Let $\alpha \subset S$ be a path transverse to W. The difference line bundle $\beta = \tilde{\mathfrak{s}} - \pi^* \mathfrak{s}$ can be understood as a $O(1; \mathbb{R})$ -local system on the trivial bundle over $L - K_L$. Such local systems are almost-flat, in that their monodromy along a small loop encircling a point in K_L is -Id.

As in Section 5.1, we will define a functor

$$\Phi_{\mathcal{W}}^{\mathbf{V},\beta} : \operatorname{Loc}_1(L) \longrightarrow \operatorname{Loc}_n(S).$$

We will adopt the same notational conventions as introduced in Section 5, using genuine paths. As before, we will write α^i to denote the cotangent lifts of the paths. The analogue of Definition 5.6 reads as follows.

Definition A.4. Let $\mathbf{V} \in \text{Loc}_1(L)$ be a $\text{GL}_1(\mathbb{C})$ -local system on L and $\alpha : [0,1] \to \mathbf{S}$ a path transverse to \mathcal{W} . Let $\tilde{\mathfrak{s}}, \mathfrak{s}$ be spin structures on L and S, respectively, and β be their difference line bundle.

(i) If α is represented by a free W-adapted path, then

$$\Phi_{\mathcal{W}}^{\mathbf{V},\beta}(\alpha) = \sum_{i=1}^{n} \Phi^{\mathbf{V},\beta}(\alpha^{i})[\alpha^{i}],$$

where the sum runs over the *n* lifts α^i to $L, i \in [1, n]$.

(ii) If α is short and intersects \mathcal{W} at z, and ρ^u the unit velocity sphere bundle lift, then

$$\Phi_{\mathcal{W}}^{\mathbf{V}\otimes\beta}(\alpha) = \sum_{i=1}^{n} \Phi^{\mathbf{V}\otimes\beta}(\alpha^{i})[\alpha^{i}] + \sum_{\mathfrak{s}(z;\mathfrak{w})\in\mathfrak{S}(z;\mathfrak{w})} \mu(\mathring{\mathfrak{s}}(z))\Phi^{\mathbf{V}\otimes\beta}\big(\pi\big(\mathring{\mathfrak{s}}(\rho^{u})\big)\big)[\pi\big(\mathring{\mathfrak{s}}(\rho^{u})\big)\big]$$

(iii) If α is given as a composition $\alpha = \alpha_1 \circ \alpha_2 \circ \ldots \circ \alpha_n$, where each α_i is either short or free, then

$$\Phi_{\mathcal{W}}^{\mathbf{V},\beta}(\alpha) = \Phi_{\mathcal{W}}^{\mathbf{V},\beta}(\alpha_1) \circ \Phi_{\mathcal{W}}^{\mathbf{V},\beta}(\alpha_2) \circ \cdots \circ \Phi_{\mathcal{W}}^{\mathbf{V},\beta}(\alpha_k).$$

138

The spin structure $\tilde{\mathfrak{s}}$ can be seen as a fibrewise double cover $\tilde{\mathfrak{s}} : \overset{\perp}{L} \to \overset{\perp}{L}$. Then the pull-back $\tilde{\mathfrak{s}}^* \mathbf{V}$ has monodromy Id along H, and so using $\tilde{\mathfrak{s}}$ to choose a trivialization $\overset{\perp}{L} \simeq \overset{\perp}{L} \times U(1)_{\theta}$, and pulling back $\tilde{\mathfrak{s}}^* V$ by the U(1)-section $\Sigma \to \overset{\perp}{L}$, we can regard $\tilde{\mathfrak{s}}^* V$ as a local system on L. Conversely, given a rank-1 flat connection ∇ on the trivial bundle of L, we can use $\tilde{\mathfrak{s}}$ to trivialize $\overset{\perp}{L} \simeq L \times U(1)_{\theta}$ and consider the twisted connection $\nabla + \pi i d\theta$. The pull-back under the fibrewise square map gives $\nabla + 2\pi i d\theta$, which has trivial monodromy along the U(1) component. The same construction recovers ∇ .

Similarly, the spin structure \mathfrak{s} trivializes the sphere bundle on S. Let α be a W-transverse path, and let $\rho^{\mathfrak{s}}(\alpha)$ be the lift of α via the trivialization $\mathfrak{s} : \mathring{S} \simeq S \times U(1)$. By construction, $\rho^{\mathfrak{s}}(\alpha)$ is a W-adapted path. We identify the stalk of $\tilde{\mathfrak{s}}^*V$, regarded as a local system on L, with that of V. The relation with Φ_W in Section 5 reads:

Proposition A.5. $\Phi_{\mathcal{W}}(V)(\rho^{\mathfrak{s}}(\alpha)) = \Phi_{\mathcal{W}}^{\tilde{\mathfrak{s}}^*V,\beta}(\alpha)$

Proof. It suffices to verify the formula for free paths and short paths. For free paths,

$$\Phi_{\mathcal{W}}^{\tilde{\mathfrak{s}}^*V\otimes\beta}(\alpha) = \sum_{i=1}^n \Phi^{\tilde{\mathfrak{s}}^*V\otimes\beta}(\alpha^i)[\alpha^i] = \sum_{i=1}^n \Phi^V(\rho^{\mathfrak{s}}(\alpha)^i)[\alpha^i],$$

since $\beta = \tilde{\mathfrak{s}} - \pi^* \mathfrak{s}$. For short paths,

$$\mu(\mathring{\mathfrak{s}}(z))\Phi^{\widetilde{\mathfrak{s}}^*\mathbf{V}\otimes\beta}\big(\pi\big(\mathring{\mathfrak{s}}(\rho^u)\big)\big[\pi\big(\mathring{\mathfrak{s}}(\rho^u)\big)\big] = \mu(\mathring{\mathfrak{s}}(z))\Phi^{\mathbf{V}}\big(\mathring{\mathfrak{s}}(\rho^{\mathfrak{s}}(\alpha))\big)\big[\pi\big(\mathring{\mathfrak{s}}(\rho(\alpha))\big)\big]$$

since $\mathfrak{s}(\rho^u)$ and $\mathfrak{s}(\rho^{\mathfrak{s}}(\alpha))$ differ by $\langle \beta, \partial \mathfrak{s}(\rho^u) \rangle H$ modulo 2H.

Finally, given a local system $\mathbf{V} \in \text{Loc}(L)$ and $V \in \text{Loc}^{\dagger}(L)$ a twisted local system on \mathring{L} such that $\widetilde{\mathfrak{s}}^*V = \mathbf{V}$, we can define the non-abelianization of $\widetilde{\mathfrak{s}}^*V$ to be $\mathfrak{s}^*\Phi_{\mathcal{W}}(V)$. This is consistent with the main body of the manuscript by Proposition A.5.

References

- [Abo11a] Mohammed Abouzaid. "A cotangent fibre generates the Fukaya category". In: Adv. Math. 228.2 (2011), pp. 894–939. ISSN: 0001-8708,1090-2082. DOI: 10.1016/j.aim. 2011.06.007. URL: https://doi.org/10.1016/j.aim.2011.06.007.
- [Abo11b] Mohammed Abouzaid. "A topological model for the Fukaya categories of plumbings". In: J. Differential Geom. 87.1 (2011), pp. 1–80. ISSN: 0022-040X,1945-743X. URL: http://projecteuclid.org/euclid.jdg/1303219772.
- [Abo12] Mohammed Abouzaid. "On the wrapped Fukaya category and based loops". In: J. Symplectic Geom. 10.1 (2012), pp. 27–79. ISSN: 1527-5256,1540-2347. URL: http://projecteuclid.org/euclid.jsg/1332853049.
- [Abo14] Mohammed Abouzaid. "Family Floer cohomology and mirror symmetry". In: Proceedings of the International Congress of Mathematicians—Seoul 2014. Vol. II. Kyung Moon Sa, Seoul, 2014, pp. 813–836. ISBN: 978-89-6105-805-6; 978-89-6105-803-2.

- [AG01] V. I. Arnold and A. B. Givental. "Symplectic geometry [MR0842908 (88b:58044)]".
 In: Dynamical systems, IV. Vol. 4. Encyclopaedia Math. Sci. Springer, Berlin, 2001, pp. 1–138.
- [AKT01] Takashi Aoki, Takahiro Kawai, and Yoshitsugu Takei. "On the exact steepest descent method: a new method for the description of Stokes curves". In: J. Math. Phys. 42.8 (2001), pp. 3691–3713. ISSN: 0022-2488,1089-7658. DOI: 10.1063/1.1368138. URL: https://doi.org/10.1063/1.1368138.
- [AL94] Michèle Audin and Jacques Lafontaine, eds. Holomorphic curves in symplectic geometry. Vol. 117. Progress in Mathematics. Birkhäuser Verlag, Basel, 1994, pp. xii+328.
 ISBN: 3-7643-2997-1. DOI: 10.1007/978-3-0348-8508-9. URL: https://doi.org/10.1007/978-3-0348-8508-9.
- [Ale+24] Anton Alekseev et al. "WKB asymptotics of stokes matrices, spectral curves and rhombus inequalities". In: Comm. Math. Phys. 405.11 (2024), Paper No. 269, 68.
 ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-024-05133-0. URL: https://doi.org/10.1007/s00220-024-05133-0.
- [Arn90] V. I. Arnold. Singularities of caustics and wave fronts. Vol. 62. Mathematics and its Applications (Soviet Series). Kluwer Academic Publishers Group, Dordrecht, 1990, pp. xiv+259. ISBN: 0-7923-1038-1. DOI: 10.1007/978-94-011-3330-2. URL: https://doi.org/10.1007/978-94-011-3330-2.
- [AS19] Alberto Abbondandolo and Felix Schlenk. "Floer homologies, with applications". In: Jahresber. Dtsch. Math.-Ver. 121.3 (2019), pp. 155–238. ISSN: 0012-0456,1869-7135.
 DOI: 10.1365/s13291-018-0193-x. URL: https://doi.org/10.1365/s13291-018-0193-x.
- [BB04] Olivier Biquard and Philip Boalch. "Wild non-abelian Hodge theory on curves". In: *Compos. Math.* 140.1 (2004), pp. 179–204. ISSN: 0010-437X,1570-5846. DOI: 10.1112/ S0010437X03000010. URL: https://doi.org/10.1112/S0010437X03000010.
- [BEE12] Frédéric Bourgeois, Tobias Ekholm, and Yasha Eliashberg. "Effect of Legendrian surgery". In: Geom. Topol. 16.1 (2012). With an appendix by Sheel Ganatra and Maksim Maydanskiy, pp. 301–389. ISSN: 1465-3060,1364-0380. DOI: 10.2140/gt. 2012.16.301. URL: https://doi.org/10.2140/gt.2012.16.301.
- [BNR82] H. L. Berk, William McCay Nevins, and K. V. Roberts. "New Stokes' line in WKB theory". In: J. Math. Phys. 23.6 (1982), pp. 988–1002. ISSN: 0022-2488,1089-7658. DOI: 10.1063/1.525467. URL: https://doi.org/10.1063/1.525467.
- [Boa14] P. P. Boalch. "Geometry and braiding of Stokes data; fission and wild character varieties". In: Ann. of Math. (2) 179.1 (2014), pp. 301–365. ISSN: 0003-486X,1939-8980.
 DOI: 10.4007/annals.2014.179.1.5. URL: https://doi.org/10.4007/annals.2014.179.1.5.
- [Boa21] P. P. Boalch. "Topology of the Stokes phenomenon". In: Integrability, quantization, and geometry. I. Integrable systems. Vol. 103.1. Proc. Sympos. Pure Math. Amer. Math. Soc., Providence, RI, [2021] ©2021, pp. 55–100. ISBN: 978-1-4704-5591-0. DOI:

10.1090/pspum/103.1/01832. URL: https://doi.org/10.1090/pspum/103.1/01832.

- [Bri19] Tom Bridgeland. "Riemann-Hilbert problems from Donaldson-Thomas theory". In: Invent. Math. 216.1 (2019), pp. 69–124. ISSN: 0020-9910,1432-1297. DOI: 10.1007/ s00222-018-0843-8. URL: https://doi.org/10.1007/s00222-018-0843-8.
- [BS15] Tom Bridgeland and Ivan Smith. "Quadratic differentials as stability conditions". In: *Publ. Math. Inst. Hautes Études Sci.* 121 (2015), pp. 155–278. ISSN: 0073-8301,1618- 1913. DOI: 10.1007/s10240-014-0066-5. URL: https://doi.org/10.1007/s10240-014-0066-5.
- [Cas22] Roger Casals. "Lagrangian skeleta and plane curve singularities". In: J. Fixed Point Theory Appl. 24.2 (2022), Paper No. 34, 43. ISSN: 1661-7738,1661-7746. DOI: 10.1007/ s11784-022-00939-8. URL: https://doi.org/10.1007/s11784-022-00939-8.
- [Cas+23] Roger Casals et al. Demazure weaves for reduced plabic graphs (with a proof that Muller-Speyer twist is Donaldson-Thomas). 2023. arXiv: 2308.06184 [math.CO]. URL: https://arxiv.org/abs/2308.06184.
- [Cas+24] Roger Casals et al. "Algebraic weaves and braid varieties". In: Amer. J. Math. 146.6 (2024), pp. 1469–1576. ISSN: 0002-9327,1080-6377.
- [Cas+25] Roger Casals et al. "Cluster structures on braid varieties". In: J. Amer. Math. Soc. 38.2 (2025), pp. 369–479. ISSN: 0894-0347,1088-6834. DOI: 10.1090/jams/1048. URL: https://doi.org/10.1090/jams/1048.
- [CG22] Roger Casals and Honghao Gao. "Infinitely many Lagrangian fillings". In: Ann. of Math. (2) 195.1 (2022), pp. 207–249. ISSN: 0003-486X. DOI: 10.4007/annals.2022.
 195.1.3. URL: https://doi.org/10.4007/annals.2022.195.1.3.
- [CG24] Roger Casals and Honghao Gao. "A Lagrangian filling for every cluster seed". In: Invent. Math. 237.2 (2024), pp. 809–868. ISSN: 0020-9910,1432-1297. DOI: 10.1007/ s00222-024-01268-y. URL: https://doi.org/10.1007/s00222-024-01268-y.
- [Cha10] Baptiste Chantraine. "Lagrangian concordance of Legendrian knots". In: Algebr. Geom. Topol. 10.1 (2010), pp. 63–85. ISSN: 1472-2747,1472-2739. DOI: 10.2140/agt.2010.
 10.63. URL: https://doi.org/10.2140/agt.2010.10.63.
- [CL22] Roger Casals and Wenyuan Li. Conjugate Fillings and Legendrian Weaves. 2022. arXiv: 2210.02039 [math.SG].
- [CM18] Roger Casals and Emmy Murphy. "Differential algebra of cubic planar graphs". In: Advances in Mathematics 338 (2018), pp. 401-446. ISSN: 0001-8708. DOI: https: //doi.org/10.1016/j.aim.2018.09.002. URL: https://www.sciencedirect.com/ science/article/pii/S0001870818303335.
- [CN22] Roger Casals and Lenhard Ng. "Braid loops with infinite monodromy on the Legendrian contact DGA". In: J. Topol. 15.4 (2022), pp. 1927–2016. ISSN: 1753-8416,1753-8424. DOI: 10.1112/topo.12264. URL: https://doi.org/10.1112/topo.12264.

- [CW24] Roger Casals and Daping Weng. "Microlocal theory of Legendrian links and cluster algebras". In: *Geom. Topol.* 28.2 (2024), pp. 901–1000. ISSN: 1465-3060,1364-0380.
 DOI: 10.2140/gt.2024.28.901. URL: https://doi.org/10.2140/gt.2024.28.901.
- [CZ22] Roger Casals and Eric Zaslow. "Legendrian weaves: N-graph calculus, flag moduli and applications". In: Geom. Topol. 26.8 (2022), pp. 3589–3745. ISSN: 1465-3060,1364-0380.
 DOI: 10.2140/gt.2022.26.3589. URL: https://doi.org/10.2140/gt.2022.26.3589.
- [DMN20] Jacques Distler, Mario Martone, and Andrew Neitzke. "On the BPS spectrum of the rank-1 Minahan-Nemeschansky theories". In: J. High Energy Phys. 2 (2020), pp. 100, 22. ISSN: 1126-6708,1029-8479. DOI: 10.1007/jhep02(2020)100. URL: https://doi.org/10.1007/jhep02(2020)100.
- [Dum+21] Olivia Dumitrescu et al. "From the Hitchin section to opers through nonabelian Hodge". In: J. Differential Geom. 117.2 (2021), pp. 223–253. ISSN: 0022-040X,1945-743X. DOI: 10.4310/jdg/1612975016. URL: https://doi.org/10.4310/jdg/ 1612975016.
- [Duv16] Julien Duval. "On a result by Y. Groman and J. P. Solomon". In: *Math. Ann.* 364.3-4 (2016), pp. 1361–1363. ISSN: 0025-5831. DOI: 10.1007/s00208-015-1258-0. URL: https://doi.org/10.1007/s00208-015-1258-0.
- [EGH00] Y. Eliashberg, A. Givental, and H. Hofer. "Introduction to symplectic field theory". In: GAFA 2000 (Tel Aviv, 1999). 2000, pp. 560–673. DOI: 10.1007/978-3-0346-0425-3_4. URL: https://doi.org/10.1007/978-3-0346-0425-3_4.
- [EHK16] Tobias Ekholm, Ko Honda, and Tamás Kálmán. "Legendrian knots and exact Lagrangian cobordisms". In: J. Eur. Math. Soc. (JEMS) 18.11 (2016), pp. 2627–2689.
 ISSN: 1435-9855,1435-9863. DOI: 10.4171/JEMS/650. URL: https://doi.org/10.4171/JEMS/650.
- [Ekh07] Tobias Ekholm. "Morse flow trees and Legendrian contact homology in 1-jet spaces".
 In: Geometry Topology 11.2 (May 2007), 1083-1224. ISSN: 1465-3060. DOI: 10.2140/gt.2007.11.1083. URL: http://dx.doi.org/10.2140/gt.2007.11.1083.
- [Ekh+13] Tobias Ekholm et al. "Knot contact homology". In: Geom. Topol. 17.2 (2013), pp. 975–1112. ISSN: 1465-3060. DOI: 10.2140/gt.2013.17.975. URL: https://doi.org/10.2140/gt.2013.17.975.
- [EL23] Tobias Ekholm and Yankı Lekili. "Duality between Lagrangian and Legendrian invariants". In: *Geometry Topology* 27.6 (Aug. 2023), 2049–2179. ISSN: 1465-3060. DOI: 10.2140/gt.2023.27.2049. URL: http://dx.doi.org/10.2140/gt.2023.27.2049.
- [EN85] David Eisenbud and Walter Neumann. Three-dimensional link theory and invariants of plane curve singularities. Vol. 110. Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, 1985, pp. vii+173. ISBN: 0-691-08380-0; 0-691-08381-9.

- [ENS18] Tobias Ekholm, Lenhard Ng, and Vivek Shende. "A complete knot invariant from contact homology". In: *Invent. Math.* 211.3 (2018), pp. 1149–1200. ISSN: 0020-9910,1432-1297. DOI: 10.1007/s00222-017-0761-1. URL: https://doi.org/10.1007/s00222-017-0761-1.
- [Erg+21] Behzat Ergun et al. "Factorized class S theories and surface defects". In: J. High Energy Phys. 12 (2021), Paper No. 041, 21. ISSN: 1126-6708,1029-8479. DOI: 10.1007/ jhep12(2021)041. URL: https://doi.org/10.1007/jhep12(2021)041.
- [Flo88] Andreas Floer. "Morse theory for Lagrangian intersections". In: J. Differential Geom. 28.3 (1988), pp. 513–547. ISSN: 0022-040X,1945-743X. URL: http://projecteuclid. org/euclid.jdg/1214442477.
- [Flo89a] Andreas Floer. "Symplectic fixed points and holomorphic spheres". In: Comm. Math. Phys. 120.4 (1989), pp. 575–611. ISSN: 0010-3616,1432-0916. URL: http://projecteuclid. org/euclid.cmp/1104177909.
- [Flo89b] Andreas Floer. "Witten's complex and infinite-dimensional Morse theory". In: J. Differential Geom. 30.1 (1989), pp. 207–221. ISSN: 0022-040X. URL: http://projecteuclid. org/euclid.jdg/1214443291.
- [FN21] Laura Fredrickson and Andrew Neitzke. "Moduli of wild Higgs bundles on CP¹ with C[×]-actions". In: Math. Proc. Cambridge Philos. Soc. 171.3 (2021), pp. 623–656. ISSN: 0305-0041,1469-8064. DOI: 10.1017/S0305004121000074. URL: https://doi.org/ 10.1017/S0305004121000074.
- [FN24] Daniel S. Freed and Andrew Neitzke. "3d spectral networks and classical Chern-Simons theory". In: Surveys in differential geometry 2021. Chern: a great geometer of the 20th century. Vol. 26. Surv. Differ. Geom. Int. Press, Boston, MA, 2024, pp. 51–155. ISBN: 978-1-57146-464-4.
- [FO97] Kenji Fukaya and Yong-Geun Oh. "Zero-loop open strings in the cotangent bundle and Morse homotopy". In: Asian J. Math. 1.1 (1997), pp. 96–180. ISSN: 1093-6106,1945-0036. DOI: 10.4310/AJM.1997.v1.n1.a5. URL: https://doi.org/10.4310/AJM. 1997.v1.n1.a5.
- [Fuk02] Kenji Fukaya. "Floer homology for families—a progress report". In: Integrable systems, topology, and physics (Tokyo, 2000). Vol. 309. Contemp. Math. Amer. Math. Soc., Providence, RI, 2002, pp. 33–68. ISBN: 0-8218-2939-4. DOI: 10.1090/conm/309/05341. URL: https://doi.org/10.1090/conm/309/05341.
- [Fuk+09a] Kenji Fukaya et al. Lagrangian intersection Floer theory: anomaly and obstruction. Part I. Vol. 46.1. AMS/IP Studies in Advanced Mathematics. American Mathematical Society, Providence, RI, 2009, pp. xii+396. ISBN: 978-0-8218-4836-4. DOI: 10.1090/ amsip/046.1. URL: https://doi.org/10.1090/amsip/046.1.
- [Fuk+09b] Kenji Fukaya et al. Lagrangian intersection Floer theory: anomaly and obstruction. Part II. Vol. 46. AMS/IP Studies in Advanced Mathematics. American Mathematical Society, Providence, RI, 2009, i-xii and 397-805. ISBN: 978-0-8218-4837-1. DOI: 10. 1090/crmp/049/07. URL: https://doi.org/10.1090/crmp/049/07.

144	REFERENCES
[Fuk93]	Kenji Fukaya. "Morse homotopy, A^{∞} -category, and Floer homologies". In: <i>Proceedings</i> of GARC Workshop on Geometry and Topology '93 (Seoul, 1993). Vol. 18. Lecture Notes Ser. Seoul Nat. Univ., Seoul, 1993, pp. 1–102.
[Fuk97]	Kenji Fukaya. "Morse homotopy and its quantization". In: <i>Geometric topology (Athens, GA, 1993)</i> . Vol. 2.1. AMS/IP Stud. Adv. Math. Amer. Math. Soc., Providence, RI, 1997, pp. 409–440. ISBN: 0-8218-0654-8. DOI: 10.1090/amsip/002.1/23. URL: https://doi.org/10.1090/amsip/002.1/23.
[Gab17]	Maxime Gabella. "Quantum holonomies from spectral networks and framed BPS states". In: <i>Comm. Math. Phys.</i> 351.2 (2017), pp. 563–598. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-016-2729-1. URL: https://doi.org/10.1007/s00220-016-2729-1.
[Gab+17]	Maxime Gabella et al. "BPS graphs: from spectral networks to BPS quivers". In: <i>J. High Energy Phys.</i> 7 (2017), 032, front matter+47. ISSN: 1126-6708,1029-8479. DOI: 10.1007/JHEP07(2017)032. URL: https://doi.org/10.1007/JHEP07(2017)032.
[Gei08]	Hansjörg Geiges. An introduction to contact topology. Vol. 109. Cambridge Studies in Advanced Mathematics. Cambridge: Cambridge University Press, 2008, pp. xvi+440. ISBN: 978-0-521-86585-2.
[GHK15]	Mark Gross, Paul Hacking, and Sean Keel. "Mirror symmetry for log Calabi-Yau surfaces I". In: <i>Publ. Math. Inst. Hautes Études Sci.</i> 122 (2015), pp. 65–168. ISSN: 0073-8301,1618-1913. DOI: 10.1007/s10240-015-0073-1. URL: https://doi.org/10.1007/s10240-015-0073-1.
[GHN23]	Alba Grassi, Qianyu Hao, and Andrew Neitzke. "Exponential networks, WKB and topological string". In: <i>SIGMA Symmetry Integrability Geom. Methods Appl.</i> 19 (2023), Paper No. 064, 44. ISSN: 1815-0659. DOI: 10.3842/SIGMA.2023.064. URL: https://doi.org/10.3842/SIGMA.2023.064.
[Ghy17]	Étienne Ghys. A singular mathematical promenade. ENS Éditions, Lyon, 2017, pp. viii+302. ISBN: 978-2-84788-939-0.
[Gir20]	Emmanuel Giroux. "Ideal Liouville domains, a cool gadget". In: <i>J. Symplectic Geom.</i> 18.3 (2020), pp. 769–790. ISSN: 1527-5256,1540-2347. DOI: 10.4310/JSG.2020.v18.n3.a5. URL: https://doi.org/10.4310/JSG.2020.v18.n3.a5.
[GLM15]	Dmitry Galakhov, Pietro Longhi, and Gregory W. Moore. "Spectral networks with spin". In: <i>Comm. Math. Phys.</i> 340.1 (2015), pp. 171–232. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-015-2455-0. URL: https://doi.org/10.1007/s00220-015-2455-0.
[GMN12]	Davide Gaiotto, Gregory W. Moore, and Andrew Neitzke. "Wall-crossing in coupled 2d-4d systems". In: J. High Energy Phys. 12 (2012), 082, front matter + 166. ISSN: 1126-6708,1029-8479. DOI: 10.1007/JHEP12(2012)082. URL: https://doi.org/10.1007/JHEP12(2012)082.
- [GMN13a] Davide Gaiotto, Gregory W. Moore, and Andrew Neitzke. "Framed BPS states". In: Adv. Theor. Math. Phys. 17.2 (2013), pp. 241–397. ISSN: 1095-0761,1095-0753. DOI: 10.4310/atmp.2013.v17.n2.a1. URL: https://doi.org/10.4310/atmp.2013.v17. n2.a1.
- [GMN13b] Davide Gaiotto, Gregory W. Moore, and Andrew Neitzke. "Spectral networks". In: Ann. Henri Poincaré 14.7 (2013), pp. 1643–1731. ISSN: 1424-0637,1424-0661. DOI: 10.1007/s00023-013-0239-7. URL: https://doi.org/10.1007/s00023-013-0239-7.
- [GMN13c] Davide Gaiotto, Gregory W. Moore, and Andrew Neitzke. "Wall-crossing, Hitchin systems, and the WKB approximation". In: Adv. Math. 234 (2013), pp. 239–403. ISSN: 0001-8708,1090-2082. DOI: 10.1016/j.aim.2012.09.027. URL: https://doi. org/10.1016/j.aim.2012.09.027.
- [GMN14] Davide Gaiotto, Gregory W. Moore, and Andrew Neitzke. "Spectral networks and snakes". In: Ann. Henri Poincaré 15.1 (2014), pp. 61–141. ISSN: 1424-0637,1424-0661.
 DOI: 10.1007/s00023-013-0238-8. URL: https://doi.org/10.1007/s00023-013-0238-8.
- [GPS20] Sheel Ganatra, John Pardon, and Vivek Shende. "Covariantly functorial wrapped Floer theory on Liouville sectors". In: *Publ. Math. Inst. Hautes Études Sci.* 131 (2020), pp. 73–200. ISSN: 0073-8301,1618-1913. DOI: 10.1007/s10240-019-00112-x. URL: https://doi.org/10.1007/s10240-019-00112-x.
- [GPS24a] Sheel Ganatra, John Pardon, and Vivek Shende. "Microlocal Morse theory of wrapped Fukaya categories". In: Ann. of Math. (2) 199.3 (2024), pp. 943–1042. ISSN: 0003-486X,1939-8980. DOI: 10.4007/annals.2024.199.3.1. URL: https://doi.org/10.4007/annals.2024.199.3.1.
- [GPS24b] Sheel Ganatra, John Pardon, and Vivek Shende. "Sectorial descent for wrapped Fukaya categories". In: J. Amer. Math. Soc. 37.2 (2024), pp. 499–635. ISSN: 0894-0347,1088-6834. DOI: 10.1090/jams/1035. URL: https://doi.org/10.1090/jams/1035.
- [Gro85] M. Gromov. "Pseudo holomorphic curves in symplectic manifolds". In: *Invent. Math.* 82.2 (1985), pp. 307–347. ISSN: 0020-9910,1432-1297. DOI: 10.1007/BF01388806. URL: https://doi.org/10.1007/BF01388806.
- [GS14] Yoel Groman and Jake P. Solomon. "A reverse isoperimetric inequality for J-holomorphic curves". In: Geom. Funct. Anal. 24.5 (2014), pp. 1448–1515. ISSN: 1016-443X. DOI: 10.1007/s00039-014-0295-2. URL: https://doi.org/10.1007/s00039-014-0295-2.
- [GS16] Yoel Groman and Jake P. Solomon. "J-holomorphic curves with boundary in bounded geometry". In: Journal of Symplectic Geometry 14.3 (2016), 767–809. ISSN: 1540-2347. DOI: 10.4310/jsg.2016.v14.n3.a5. URL: http://dx.doi.org/10.4310/JSG.2016.v14.n3.a5.

- [HHN20] Qianyu Hao, Lotte Hollands, and Andrew Neitzke. "BPS states in the Minahan-Nemeschansky E₇ theory". In: J. High Energy Phys. 4 (2020), pp. 039, 19. ISSN: 1126-6708,1029-8479. DOI: 10.1007/jhep04(2020)039. URL: https://doi.org/10.1007/jhep04(2020)039.
- [HJ22] Andreas Hohl and Konstantin Jakob. "Stokes matrices for Airy equations". In: Tohoku Math. J. (2) 74.4 (2022), pp. 501–520. ISSN: 0040-8735,2186-585X. DOI: 10.2748/tmj. 20210506. URL: https://doi.org/10.2748/tmj.20210506.
- [HKS24] F. Haiden, L. Katzarkov, and C. Simpson. "Spectral networks and stability conditions for Fukaya categories with coefficients". In: Comm. Math. Phys. 405.11 (2024), Paper No. 264, 68. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-024-05138-9. URL: https://doi.org/10.1007/s00220-024-05138-9.
- [HLOD04] C. J. Howls, P. J. Langman, and A. B. Olde Daalhuis. "On the higher-order Stokes phenomenon". In: Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 460.2048 (2004), pp. 2285–2303. ISSN: 1364-5021,1471-2946. DOI: 10.1098/rspa.2004.1299. URL: https://doi.org/10.1098/rspa.2004.1299.
- [HM79] John Hubbard and Howard Masur. "Quadratic differentials and foliations". In: Acta Math. 142.3-4 (1979), pp. 221–274. ISSN: 0001-5962,1871-2509. DOI: 10.1007/BF02395062.
 URL: https://doi.org/10.1007/BF02395062.
- [HN16] Lotte Hollands and Andrew Neitzke. "Spectral networks and Fenchel-Nielsen coordinates". In: Lett. Math. Phys. 106.6 (2016), pp. 811–877. ISSN: 0377-9017,1573-0530.
 DOI: 10.1007/s11005-016-0842-x. URL: https://doi.org/10.1007/s11005-016-0842-x.
- [HN17] Lotte Hollands and Andrew Neitzke. "BPS states in the Minahan-Nemeschansky E₆ theory". In: Comm. Math. Phys. 353.1 (2017), pp. 317–351. ISSN: 0010-3616,1432-0916.
 DOI: 10.1007/s00220-016-2798-1. URL: https://doi.org/10.1007/s00220-016-2798-1.
- [HN20] Lotte Hollands and Andrew Neitzke. "Exact WKB and abelianization for the T_3 equation". In: Comm. Math. Phys. 380.1 (2020), pp. 131–186. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-020-03875-1. URL: https://doi.org/10.1007/s00220-020-03875-1.
- [Hug23] James Hughes. "Lagrangian fillings in A-type and their Kálmán loop orbits". In: Rev. Mat. Iberoam. 39.5 (2023), pp. 1681–1723. ISSN: 0213-2230,2235-0616. DOI: 10.4171/ rmi/1436. URL: https://doi.org/10.4171/rmi/1436.
- [IK22] Kohei Iwaki and Omar Kidwai. "Topological recursion and uncoupled BPS structures
 I: BPS spectrum and free energies". In: Adv. Math. 398 (2022), Paper No. 108191,
 54. ISSN: 0001-8708,1090-2082. DOI: 10.1016/j.aim.2022.108191. URL: https://doi.org/10.1016/j.aim.2022.108191.
- [IK23] Kohei Iwaki and Omar Kidwai. "Topological recursion and uncoupled BPS structures II: Voros symbols and the τ -function". In: Comm. Math. Phys. 399.1 (2023), pp. 519–

572. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-022-04563-y. URL: https://doi.org/10.1007/s00220-022-04563-y.

- [IM21] Matei Ionita and Benedict Morrissey. Spectral Networks and Non-abelianization. 2021. arXiv: 2103.12285 [math.AG]. URL: https://arxiv.org/abs/2103.12285.
- [Kat+15] Ludmil Katzarkov et al. "Harmonic maps to buildings and singular perturbation theory". In: Comm. Math. Phys. 336.2 (2015), pp. 853–903. ISSN: 0010-3616,1432-0916.
 DOI: 10.1007/s00220-014-2276-6. URL: https://doi.org/10.1007/s00220-014-2276-6.
- [Kat87] Nicholas M. Katz. "On the calculation of some differential Galois groups". In: Invent. Math. 87.1 (1987), pp. 13–61. ISSN: 0020-9910,1432-1297. DOI: 10.1007/BF01389152. URL: https://doi.org/10.1007/BF01389152.
- [KKP08] L. Katzarkov, M. Kontsevich, and T. Pantev. "Hodge theoretic aspects of mirror symmetry". In: From Hodge theory to integrability and TQFT tt*-geometry. Vol. 78. Proc. Sympos. Pure Math. Amer. Math. Soc., Providence, RI, 2008, pp. 87–174. ISBN: 978-0-8218-4430-4. DOI: 10.1090/pspum/078/2483750. URL: https://doi.org/10. 1090/pspum/078/2483750.
- [KLS23] Hyun Kyu Kim, Thang T. Q. Lê, and Miri Son. "SL₂ quantum trace in quantum Teichmüller theory via writhe". In: *Algebr. Geom. Topol.* 23.1 (2023), pp. 339–418.
 ISSN: 1472-2747,1472-2739. DOI: 10.2140/agt.2023.23.339. URL: https://doi.org/10.2140/agt.2023.23.339.
- [Kon95] Maxim Kontsevich. "Homological algebra of mirror symmetry". In: Proceedings of the International Congress of Mathematicians, Vol. 1, 2 (Zürich, 1994). Birkhäuser, Basel, 1995, pp. 120–139. ISBN: 3-7643-5153-5.
- [Kuw24] Tatsuki Kuwagaki. On the generic existence of WKB spectral networks/Stokes graphs. 2024. arXiv: 2408.05399 [math.SG]. URL: https://arxiv.org/abs/2408.05399.
- [Lag73] Joseph Louis Lagrange. Oeuvres. Tome 2. Publiées par les soins de J.-A. Serret, Nachdruck der Ausgabe Paris 1868. Georg Olms Verlag, Hildesheim-New York, 1973, pp. v+727.
- [Lon18] Pietro Longhi. "Wall crossing invariants from spectral networks". In: Ann. Henri Poincaré 19.3 (2018), pp. 775–842. ISSN: 1424-0637,1424-0661. DOI: 10.1007/s00023-017-0635-5. URL: https://doi.org/10.1007/s00023-017-0635-5.
- [LP16] Pietro Longhi and Chan Y. Park. "ADE spectral networks". In: J. High Energy Phys. 8 (2016), 087, front matter+90. ISSN: 1126-6708,1029-8479. DOI: 10.1007/ JHEP08(2016)087. URL: https://doi.org/10.1007/JHEP08(2016)087.
- [LP17] Pietro Longhi and Chan Y. Park. "ADE spectral networks and decoupling limits of surface defects". In: J. High Energy Phys. 2 (2017), 011, front matter+48. ISSN: 1126-6708,1029-8479. DOI: 10.1007/JHEP02(2017)011. URL: https://doi.org/10.1007/JHEP02(2017)011.
- [Lur17] Jacob Lurie. "Higher algebra". In: *Preprint, available at https://www.math.ias.edu/ lurie/* (2017).

148	REFERENCES
[Mil63]	J. Milnor. <i>Morse theory</i> . Vol. No. 51. Annals of Mathematics Studies. Based on lecture notes by M. Spivak and R. Wells. Princeton University Press, Princeton, NJ, 1963,
[MS04]	pp. v1+153. Dusa McDuff and Dietmar Salamon. <i>J-holomorphic curves and symplectic topology</i> . Vol. 52. American Mathematical Society Colloquium Publications. American Math- ematical Society, Providence, RI, 2004, pp. xii+669. ISBN: 0-8218-3485-1. DOI: 10.
[MS98]	1090/col1/052. URL: https://doi.org/10.1090/col1/052. Dusa McDuff and Dietmar Salamon. <i>Introduction to symplectic topology</i> . Second. Oxford Mathematical Monographs. The Clarendon Press, Oxford University Press, New York, 1998, pp. x+486, ISBN: 0-19-850451-9.
[Nei14]	Andrew Neitzke. "Cluster-like coordinates in supersymmetric quantum field theory". In: <i>Proc. Natl. Acad. Sci. USA</i> 111.27 (2014), pp. 9717–9724. ISSN: 0027-8424,1091- 6490. DOI: 10.1073/pnas.1313073111. URL: https://doi.org/10.1073/pnas. 1313073111.
[Nei21]	Andrew Neitzke. "Lectures on BPS states and spectral networks". In: <i>Quantum field theory and manifold invariants</i> . Vol. 28. IAS/Park City Math. Ser. Amer. Math. Soc., Providence, RI, [2021] ©2021, pp. 439–476. ISBN: 978-1-4704-6123-2.
[Nho24]	Yoon Jae Nho. Family Floer theory, non-abelianization, and Spectral Networks. 2024. arXiv: 2307.04213 [math.SG] UBL: https://arxiv.org/abs/2307.04213
[NY20]	Andrew Neitzke and Fei Yan. "q-nonabelianization for line defects". In: J. High Energy Phys. 9 (2020), pp. 153, 65. ISSN: 1126-6708,1029-8479. DOI: 10.1007/jhep09(2020) 153. URL: https://doi.org/10.1007/jhep09(2020)153.
[Oh15]	Yong-Geun Oh. Symplectic topology and Floer homology. Vol. 2. Vol. 29. New Mathematical Monographs. Floer homology and its applications. Cambridge University Press, Cambridge, 2015, pp. xxiii+446, ISBN: 978-1-107-10967-4.
[Oh93]	Yong-Geun Oh. "Floer cohomology of Lagrangian intersections and pseudo-holomorphic disks. II. $(\mathbb{CP}^n, \mathbb{RP}^n)$ ". In: Comm. Pure Appl. Math. 46.7 (1993), pp. 995–1012. ISSN: 0010-3640,1097-0312. DOI: 10.1002/cpa.3160460703. URL: https://doi.org/10.1002/cpa.3160460703.
[PSS96]	S. Piunikhin, D. Salamon, and M. Schwarz. "Symplectic Floer-Donaldson theory and quantum cohomology". In: <i>Contact and symplectic geometry (Cambridge, 1994)</i> . Vol. 8. Publ. Newton Inst. Cambridge Univ. Press, Cambridge, 1996, pp. 171–200. ISBN: 0-521-57086-7.
[RS20]	Dan Rutherford and Michael Sullivan. "Cellular Legendrian contact homology for surfaces, part I". In: <i>Adv. Math.</i> 374 (2020), pp. 107348, 71. ISSN: 0001-8708,1090-2082. DOI: 10.1016/j.aim.2020.107348. URL: https://doi.org/10.1016/j.aim. 2020.107348.
[Rua06]	Wei-Dong Ruan. The Fukaya category of symplectic neighborhood of a non-Hausdorff manifold. 2006. arXiv: math/0602119 [math.SG].

- [Sab13] Claude Sabbah. Introduction to Stokes structures. Vol. 2060. Lecture Notes in Mathematics. Springer, Heidelberg, 2013, pp. xiv+249. ISBN: 978-3-642-31694-4; 978-3-642-31695-1. DOI: 10.1007/978-3-642-31695-1. URL: https://doi.org/10.1007/978-3-642-31695-1.
- [Sei08a] Paul Seidel. "A biased view of symplectic cohomology". In: Current developments in mathematics, 2006. Int. Press, Somerville, MA, 2008, pp. 211–253. ISBN: 978-1-57146-167-4.
- [Sei08b] Paul Seidel. Fukaya categories and Picard-Lefschetz theory. Zurich Lectures in Advanced Mathematics. European Mathematical Society (EMS), Zürich, 2008, pp. viii+326.
 ISBN: 978-3-03719-063-0. DOI: 10.4171/063. URL: https://doi.org/10.4171/063.
- [Sei15] Paul Seidel. "Homological mirror symmetry for the quartic surface". In: Mem. Amer. Math. Soc. 236.1116 (2015), pp. vi+129. ISSN: 0065-9266,1947-6221. DOI: 10.1090/ memo/1116. URL: https://doi.org/10.1090/memo/1116.
- [She+19] Vivek Shende et al. "Cluster varieties from Legendrian knots". In: Duke Math. J. 168.15 (2019), pp. 2801–2871. ISSN: 0012-7094,1547-7398. DOI: 10.1215/00127094–2019–0027. URL: https://doi.org/10.1215/00127094–2019–0027.
- [Sma61] Stephen Smale. "On gradient dynamical systems". In: Ann. of Math. (2) 74 (1961), pp. 199–206. ISSN: 0003-486X. DOI: 10.2307/1970311. URL: https://doi.org/10.2307/1970311.
- [Smi15] Ivan Smith. "Quiver algebras as Fukaya categories". In: Geom. Topol. 19.5 (2015), pp. 2557–2617. ISSN: 1465-3060,1364-0380. DOI: 10.2140/gt.2015.19.2557. URL: https://doi.org/10.2140/gt.2015.19.2557.
- [Smi21] Ivan Smith. "Floer theory of higher rank quiver 3-folds". In: Comm. Math. Phys. 388.3 (2021), pp. 1181–1203. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-021-04252-2. URL: https://doi.org/10.1007/s00220-021-04252-2.
- [Sto09] George Gabriel Stokes. "1857. On the Discontinuity of Arbitrary Constants which appear in Divergent Developments". In: *Mathematical and Physical Papers*. Cambridge Library Collection - Mathematics. Cambridge University Press, 2009, 77–109.
- [Str84] Kurt Strebel. Quadratic differentials. Vol. 5. Ergebnisse der Mathematik und ihrer Grenzgebiete (3) [Results in Mathematics and Related Areas (3)]. Springer-Verlag, Berlin, 1984, pp. xii+184. ISBN: 3-540-13035-7. DOI: 10.1007/978-3-662-02414-0. URL: https://doi.org/10.1007/978-3-662-02414-0.
- [Su25] Tao Su. "Dual boundary complexes of Betti moduli spaces over the two-sphere with one irregular singularity". In: Adv. Math. 462 (2025), Paper No. 110101. ISSN: 0001-8708,1090-2082. DOI: 10.1016/j.aim.2024.110101. URL: https://doi.org/10. 1016/j.aim.2024.110101.
- [Sue24] Yat-Hin Suen. "Toric Vector Bundles, Non-abelianization, and Spectral Networks". In: Int. Math. Res. Not. IMRN 24 (2024), pp. 14576–14599. ISSN: 1073-7928,1687-0247.
 DOI: 10.1093/imrn/rnae250. URL: https://doi.org/10.1093/imrn/rnae250.

150	REFERENCES
[SW06]	D. A. Salamon and J. Weber. "Floer homology and the heat flow". In: <i>Geom. Funct.</i> <i>Anal.</i> 16.5 (2006), pp. 1050–1138. ISSN: 1016-443X,1420-8970. DOI: 10.1007/s00039- 006-0577-4. UBL: https://doi.org/10.1007/s00039-006-0577-4.
[Syl19]	Zachary Sylvan. "On partially wrapped Fukaya categories". In: <i>J. Topol.</i> 12.2 (2019), pp. 372–441. ISSN: 1753-8416,1753-8424. DOI: 10.1112/topo.12088. URL: https://doi.org/10.1112/topo.12088
[Vit99]	C. Viterbo. "Functors and computations in Floer homology with applications. I". In: <i>Geom. Funct. Anal.</i> 9.5 (1999), pp. 985–1033. ISSN: 1016-443X,1420-8970. DOI: 10.1007/s000390050106. URL: https://doi.org/10.1007/s000390050106.
[Wal75]	C. T. C. Wall. "Regular stratifications". In: Dynamical systems—Warwick 1974 (Proc. Sympos. Appl. Topology and Dynamical Systems, Univ. Warwick, Coventry, 1973/1974; presented to E. C. Zeeman on his fiftieth birthday). Vol. Vol. 468. Lecture Notes in Math. Springer, Berlin-New York, 1975, pp. 332–344.
[Wil16]	Harold Williams. "Toda systems, cluster characters, and spectral networks". In: <i>Comm. Math. Phys.</i> 348.1 (2016), pp. 145–184. ISSN: 0010-3616,1432-0916. DOI: 10.1007/s00220-016-2692-x. URL: https://doi.org/10.1007/s00220-016-2692-x.
Universit Email addr	Y OF CALIFORNIA DAVIS, DEPT. OF MATHEMATICS, USA ress: casals@ucdavis.edu

University of California Davis, Dept. of Mathematics, USA $\mathit{Email}\ address: \texttt{yjnho@ucdavis.edu}$